AMAP Assessment 2013: Arctic Ocean Acidification

AMAP Arctic Monitoring and Assessment Programme (AMAP)

Educational use: This report (in part or in its entirety) and other AMAP products available from www.amap.no for which AMAP holds the copyright can be used freely as teaching materials and for other educational purposes.

The only condition of such use is acknowledgement of AMAP as the source of the material according to the recommended citation.

In case of questions regarding educational use, please contact the AMAP Secretariat (amap@amap.no).

Note: This report may contain material (e.g. photographs) for which permission for use will need to be obtained from original copyright holders.

Disclaimer: The views expressed in this peer-reviewed report are the responsibility of the authors of the report and do not necessarily reflect the views of the Arctic Council, its members or its observers.

AMAP Assessment 2013: Arctic Ocean Acidification

AMAP Arctic Monitoring and Assessment Programme (AMAP)

AMAP Assessment 2013: Arctic Ocean Acidification

Citation

AMAP, 2013. AMAP Assessment 2013: Arctic Ocean Acidification. Arctic Monitoring and Assessment Programme (AMAP), Oslo, Norway. viii + 99 pp.

ISBN - 978-82-7971-082-0

© Arctic Monitoring and Assessment Programme, 2013

Published by

Arctic Monitoring and Assessment Programme (AMAP), Oslo, Norway (www.amap.no)

Ordering

This report can be ordered from the AMAP Secretariat, Gaustadalléen 21, N-0349 Oslo, Norway (amap@amap.no) This report is also published as electronic documents, available from the AMAP website at www.amap.no

Production

Production management Simon Wilson (AMAP Secretariat)

Scientific, technical and linguistic editing Carolyn Symon (carolyn.symon@btinternet.com), Simon Wilson (AMAP Secretariat)

Lay-out and technical production Burnthebook, United Kingdom (www.burnthebook.co.uk)

Design and production of computer graphics

Simon Duckworth (simon@burnthebook.co.uk), Hugo Ahlenius (hugo.ahlenius@nordpil.com), Simon Wilson (s.wilson@inter.nl.net), John Bellamy (johnbellamy@swipnet.se)

Cover photograph

Barnacles (Semibalanus balanoides); one of the Arctic species being investigated for ocean acidification effects. Photo: Fredrik Pleijel.

Printing

Narayana Press, Gylling, DK-8300 Odder, Denmark (www.narayanapress.dk).



AMAP Working Group

Russel Shearer (Chair, Canada), Fred Wrona (Canada), Mikala Klint (Denmark), Morten Olsen (Vice-Chair, Denmark), Outi Mähönen (Vice-Chair, Finland), Helgi Jensson (Iceland), Per Døvle (Norway), Tove Lundberg (Sweden), Yuri Tsaturov (Russia), Tom Armstrong (USA).

AMAP Secretariat

Lars-Otto Reiersen, Simon Wilson, Jon Fuglestad, Jan-Rene Larsen, Janet Pawlak, Inger Utne.

Arctic Council Member States and Permanent Participants of the Council:

Canada, Denmark/Greenland/Faroe Islands, Finland, Iceland, Norway, Russia, Sweden, United States, Aleut International Association (AIA), Arctic Athabaskan Council (AAC), Gwitch'in Council International (GCI), Inuit Circumpolar Council (ICC), Russian Association of Indigenous Peoples of the North (RAIPON), Saami Council.

Acknowledgements

Helene Amundsen, Leif Anderson, Andreas Andersson, Kumiko Azetsu-Scott, **Richard Bellerby**, Michael Beman, **Howard I. Browman**, Craig Carlson, William W.L. Cheung, Melissa Chierici, Tonya Clayton, Sarah Cooley, Peter Croot, Nils Daan, Carlos Duarte, Sam Dupont, Maoz Fine, Ola Flaaten, Jan Helge Fosså, Agneta Fransson, Arild Gjertsen, Jason Hall-Spencer, Pamela Hallock-Muller, Jon Havenhand, Nathalie Hilmi, Grete K. Hovelsrud, Thomas P. Hurst, Debora Iglesias-Rodriguez, Emil Jeansson, Paul Knorr, Haruko Kurihara, Vicky W.Y. Lam, John Lisle, Robie Macdonald, Fred Mackenzie, Clara Manno, Jeremy Mathis, Sophie McCoy, Frank Melzner, Lisa Miller, Philip Munday, Jon Olafsson, Are Olsen, Ute Passow, Hans-Otto Pörtner, Lars-Otto Reiersen, Justin Ries, Lisa Robbins, Dominique Robert, Jeffrey Runge, Alain Safa, David Scott, Hein Rune Skjoldal, Nadja Steiner, **U. Rashid Sumaila,** Keita Suzuki, Frede Thingstad, Simon Wilson, Tim Wootton, Michiyo Yamamoto-Kauai

Bold: coordinating authors

Contents

Acknowledgements	iii
Preface	vii
1. Introduction	1
1.1 What is ocean acidification?	1
1.2 Ocean acidification in the global system	2
1.3 Why assess ocean acidification in the Arctic?	3
1.3.1 The sensitivity of the Arctic Ocean to acidification	3
1.3.2 The sensitivity of Arctic marine ecosystems to acidification	4
1.3.3 The socio-economic value of the Arctic – and the potential cost of Arctic Ocean acidification	4
1.4 The geographical context of the Arctic Ocean acidification assessment	5
1.4.1 What are the Arctic Ocean and its marginal seas?	5
1.4.2 The functioning of the Arctic Ocean in the context of acidification.	6
1.5 Aim of the present assessment	7
1.6 Reader's guide to this assessment	7
2. Acidification in the Arctic Ocean	9
2.1 Introduction	9
2.2 Carbonate system in seawater.	9
2.3 Influences on the marine carbonate system in the Arctic Ocean	11
2.3.1 Sea ice	11
2.3.2 Freshwater	12
2.3.3 Solubility pump	13
2.3.4 Biological pump	14
2.3.5 Carbonate pump	15
2.3.6 Air-sea exchange	16
2.3.7 Geological methane and carbon dioxide seeps	17
2.3.8 Other 'acidifying' compounds	17
2.3.9 Terrestrial organic carbon	17
2.4 Ocean processes sensitive to acidification	18
2.4.1 Macronutrients	. 18
2.4.2 Trace metals	19
2.4.3 Climate-relevant gases	2.0
2.5 Current Arctic Ocean acidification	21
2.5.1 Inflow seas	21
2.5.2 Shelf seas	23
2.5.3 Central Arctic Ocean	26
2.6 Eutrue scenarios	26
2.6.1 Introduction	20
2.6.2 Ocean acidification changes projected over the aut contury	20
2.6.3 Persional modeling studies	27
2.6.1 Limitations of future projections	2)
2.6.5 Lifetime of ocean acidification	51
2.0.9 Enclusions	51
Chapter a Appendix: Compilation of publiched data (1960 2012) on trace elements and their instance in the Artic	52
Chapter 2 Appendix, Compliation of published data (1900-2012) on trace elements and their isotopes in the Arctic	34
3. Biological responses to ocean acidification	37
3.1 Introduction	37

3.2 Biogeochemical, biological, and physiological context for assessing the impact of ocean acidific	ation 3
3.2.1 Natural variability in pH and carbon dioxide in time and space	
3.2.2 General biological responses to lowered pH / elevated CO	
3.3 Arctic marine food webs	
3.3.1 Arctic habitats and biogeography	
3.3.2 Arctic ecosystems and food webs	
3.4 Taxon-specific responses to ocean acidification	
3.4.1 Effects of ocean acidification on viruses	
3.4.2 Effects of ocean acidification on bacteria	
3.4.3 Effects of ocean acidification on phytoplankton	
3.4.4 Effects of ocean acidification on foraminifera	
3.4.5 Effects of ocean acidification on macroalgae	
3.4.6 Effects of ocean acidification on corals	
3.4.7 Effects of ocean acidification on mollusks	
3 4 8 Effects of ocean acidification on echinoderms	4
3 4 9 Effects of ocean acidification on crustaceans	4
3.4.10 Effects of ocean acidification on other invertebrates	і Д
3.4.11 Effects of ocean acidification on fiches	н Д
3.4.12 Effects of ocean acidification on coshids and maxing mammals	
3.5 Ecosystem-level responses to ocean acidification	
2.6 Observations in naturally occurring low pH and/or high CO, regions	
2.7 Generate	
3.7.1 Time-scales of experiments in the context of acclimation and adaptation	
3.7.2 Multiple stressors	
2.9 Deceases priorities	5
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fishe 	ries
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fishe 4.1 Introduction. 	
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fishe 4.1 Introduction. 4.2 People in the Arctic. 	
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fishe 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 	ries
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fishe 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 	
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fishe 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 5
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fishe 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 	
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fishe 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 27 (Northeast Atlantic). 	ries
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fishe 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 27 (Northeast Atlantic) 4.4.4 FAO Fishing Area 61 (Northwest Pacific) and 67 (Northeast Pacific). 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 6 6
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fishe 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 21 (Northwest Atlantic). 4.4.4 FAO Fishing Areas 61 (Northwest Pacific) and 67 (Northeast Pacific). 4.5 Impacts of ocean acidification on Arctic fisheries. 	ries 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fisher 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 27 (Northeast Atlantic). 4.4.4 FAO Fishing Areas 61 (Northwest Pacific) and 67 (Northeast Pacific). 4.5 Impacts of ocean acidification on Arctic fisheries. 4.5.1 Economic impacts of ocean acidification on Arctic fisheries. 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 6 6 6 6
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fishe 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 27 (Northeast Atlantic) 4.4.4 FAO Fishing Area 61 (Northwest Pacific) and 67 (Northeast Pacific). 4.5 Impacts of ocean acidification on Arctic fisheries 4.6 Impacts of ocean acidification on Arctic indigenous peoples and local communities. 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 6 6 6 6
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fisher 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 27 (Northeast Atlantic). 4.4.4 FAO Fishing Areas 61 (Northwest Pacific) and 67 (Northeast Pacific). 4.5 Impacts of ocean acidification on Arctic fisheries. 4.5.1 Economic impacts of ocean acidification on Arctic fisheries. 4.6 Impacts of ocean acidification on the recreational value of Arctic ecosystems 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 6 6 6 6 6 6 6 6
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fisher 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 27 (Northeast Atlantic) 4.4.4 FAO Fishing Areas 61 (Northwest Pacific) and 67 (Northeast Pacific). 4.5 Impacts of ocean acidification on Arctic fisheries. 4.5.1 Economic impacts of ocean acidification on Arctic fisheries. 4.6 Impacts of ocean acidification on the recreational value of Arctic ecosystems 4.8 Marine management in the face of ocean acidification 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 6 6 6 6 6 6 6 6 6 6
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fishe 4.1 Introduction 4.2 People in the Arctic. 4.3 Socio-economic factors 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 27 (Northeast Atlantic). 4.4.4 FAO Fishing Area 361 (Northwest Pacific) and 67 (Northeast Pacific). 4.5 Impacts of ocean acidification on Arctic fisheries 4.5.1 Economic impacts of ocean acidification on Arctic fisheries 4.6 Impacts of ocean acidification on the recreational value of Arctic ecosystems 4.8 Marine management in the face of ocean acidification 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 6
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fisher 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 27 (Northeast Atlantic). 4.4.4 FAO Fishing Areas 61 (Northwest Pacific) and 67 (Northeast Pacific). 4.5 Impacts of ocean acidification on Arctic fisheries. 4.6 Impacts of ocean acidification on Arctic indigenous peoples and local communities. 4.7 Impact of ocean acidification on the recreational value of Arctic ecosystems 4.8 Marine management in the face of ocean acidification 4.9 Conclusions and further work 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 6 6
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fisher 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 18 (Arctic Sea). 4.4.3 FAO Fishing Area 27 (Northeast Atlantic). 4.4.4 FAO Fishing Area 50 (Northwest Atlantic). 4.5 Impacts of ocean acidification on Arctic fisheries. 4.5.1 Economic impacts of ocean acidification on Arctic fisheries. 4.6 Impacts of ocean acidification on Arctic indigenous peoples and local communities. 4.7 Impact of ocean acidification on the recreational value of Arctic ecosystems 4.8 Marine management in the face of ocean acidification 4.9 Concluding remarks. 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 6 6 6 <t< td=""></t<>
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fisher 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 27 (Northeast Atlantic). 4.4.4 FAO Fishing Areas 61 (Northwest Pacific) and 67 (Northeast Pacific). 4.5 Impacts of ocean acidification on Arctic fisheries. 4.6 Impacts of ocean acidification on Arctic indigenous peoples and local communities. 4.7 Impact of ocean acidification on the recreational value of Arctic ecosystems 4.8 Marine management in the face of ocean acidification. 4.9 Conclusions and further work 5.1 The changing Arctic 5.2 Experimental direction. 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 6 6 6 <t< td=""></t<>
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fisher 4.1 Introduction 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 27 (Northeast Atlantic). 4.4.4 FAO Fishing Area 61 (Northwest Pacific) and 67 (Northeast Pacific). 4.5 Impacts of ocean acidification on Arctic fisheries. 4.5.1 Economic impacts of ocean acidification on Arctic fisheries. 4.6 Impacts of ocean acidification on the recreational value of Arctic ecosystems 4.8 Marine management in the face of ocean acidification 4.9 Concluding remarks. Conclusions and further work 5.1 The changing Arctic 5.2 Experimental direction. 5.3 Socio-economic advances.	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 6 6 6 <t< td=""></t<>
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fisher 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 21 (Northwest Atlantic). 4.4.4 FAO Fishing Area 21 (Northwest Atlantic). 4.4.4 FAO Fishing Area 27 (Northeast Atlantic). 4.5 Impacts of ocean acidification on Arctic fisheries. 4.5.1 Economic impacts of ocean acidification on Arctic fisheries. 4.6 Impacts of ocean acidification on Arctic indigenous peoples and local communities. 4.7 Impact of ocean acidification on the recreational value of Arctic ecosystems 4.8 Marine management in the face of ocean acidification. 4.9 Conclusions and further work. 5.1 The changing Arctic 5.2 Experimental direction. 5.3 Socio-economic advances. 5.4 An interdisciplinary approach. 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 7 6 7 7
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fisher 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 27 (Northeast Atlantic). 4.4.4 FAO Fishing Area 50 (Northwest Pacific) and 67 (Northeast Pacific). 4.5 Impacts of ocean acidification on Arctic fisheries. 4.5.1 Economic impacts of ocean acidification on Arctic fisheries. 4.6 Impacts of ocean acidification on Arctic indigenous peoples and local communities. 4.7 Impact of ocean acidification on the recreational value of Arctic ecosystems 4.8 Marine management in the face of ocean acidification. 4.9 Conclusions and further work 5.1 The changing Arctic 5.2 Experimental direction. 5.3 Socio-economic advances. 5.4 An interdisciplinary approach. 5.5 Monitoring networks 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 6 6 6 6 6 6 6 6 6 6 6 6 6 6 7 7
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fisher 4.1 Introduction. 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 27 (Northwest Atlantic). 4.4.4 FAO Fishing Areas 61 (Northwest Pacific) and 67 (Northeast Pacific). 4.5 Impacts of ocean acidification on Arctic fisheries. 4.5.1 Economic impacts of ocean acidification on Arctic fisheries. 4.6 Impacts of ocean acidification on the recreational value of Arctic ecosystems 4.8 Marine management in the face of ocean acidification. 4.9 Conclusions and further work. 5.1 The changing Arctic. 5.2 Experimental direction. 5.3 Socio-economic advances. 5.4 An interdisciplinary approach. 5.5 Monitoring networks. 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 6 6 6 6 6 6 6 6 6 6 7 7 7 7
 3.8 Research priorities. 3.9 Conclusions and recommendations Potential economic and social impacts of ocean acidification on Arctic fisher 4.1 Introduction 4.2 People in the Arctic. 4.3 Socio-economic factors. 4.4 Marine fisheries in the Arctic region. 4.4.1 FAO Fishing Area 18 (Arctic Sea). 4.4.2 FAO Fishing Area 21 (Northwest Atlantic). 4.4.3 FAO Fishing Area 27 (Northeast Atlantic). 4.4.4 FAO Fishing Areas 61 (Northwest Pacific) and 67 (Northeast Pacific). 4.5 Impacts of ocean acidification on Arctic fisheries. 4.5.1 Economic impacts of ocean acidification on Arctic fisheries. 4.6 Impacts of ocean acidification on Arctic indigenous peoples and local communities. 4.7 Impact of ocean acidification on the recreational value of Arctic ecosystems 4.8 Marine management in the face of ocean acidification. 4.9 Conclusions and further work 5.1 The changing Arctic 5.2 Experimental direction. 5.3 Socio-economic advances. 5.4 An interdisciplinary approach. 5.5 Monitoring networks 	5 ries 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 6 6 6 6 6 6 6 6 6 6 7 7 7 7

This assessment report presents the results of the 2013 AMAP Assessment of Arctic Ocean Acidification (AOA). This is the first such assessment dealing with AOA from an Arctic-wide perspective, and complements several assessments that AMAP has delivered over the past ten years concerning the effects of climate change on Arctic ecosystems and people.

The Arctic Monitoring and Assessment Programme (AMAP) is a group working under the Arctic Council. The Arctic Council Ministers have requested AMAP to:

- produce integrated assessment reports on the status and trends of the conditions of the Arctic ecosystems;
- identify possible causes for the changing conditions;
- detect emerging problems, their possible causes, and the potential risk to Arctic ecosystems including indigenous peoples and other Arctic residents; and to
- recommend actions required to reduce risks to Arctic ecosystems.

This report provides the accessible scientific basis and validation for the statements and recommendations made in the Arctic Ocean Acidification Assessment Summary for Policy-makers¹ that was delivered to Arctic Council Ministers at their meeting in Kiruna, Sweden in May 2011 and the related AMAP State of the Arctic Environment report Arctic Ocean Acidification 2013: An Overview². It includes extensive background data and references to the scientific literature, and details the sources for figures reproduced in the overview report. Whereas the Summary for Policy-makers report contains recommendations that focus mainly on policy-relevant actions concerned with addressing the consequences of AOA, the conclusions and recommendations presented in this report also cover issues of a more scientific nature, such as proposals for filling gaps in knowledge, and recommendations relevant to future monitoring and research work.

The AOA assessment was conducted between 2010 and 2013 by an international group of over 60 experts. Lead authors were selected based on an open nomination process coordinated by AMAP. A similar process was used to select international experts who independently reviewed this report.

Information contained in this report is fully referenced and based first and foremost on peer-reviewed and published results of research and monitoring undertaken since 2006. It also incorporates some new (unpublished) information from monitoring and research conducted according to wellestablished and documented national and international standards and quality assurance/ quality control protocols. Care has been taken to ensure that no critical probability statements are based on non-peer-reviewed materials. Access to reliable and up-to-date information is essential for the development of science-based decision-making regarding ongoing changes in the Arctic and their global implications. The AOA assessment summary reports^{1,2} and films have therefore been developed specifically for policy-makers, summarizing the main findings of the AOA assessment. The AOA lead authors have confirmed that both this report and its derivative products accurately and fully reflect their scientific assessment. The AOA reports and the films are freely available from the AMAP Secretariat and on the AMAP website: www.amap.no, and their use for educational purposes is encouraged.

AMAP would like to express its appreciation to all experts who have contributed their time, efforts and data, in particular the lead authors who coordinated the production of this report. Thanks are also due to the reviewers who contributed to the AOA peer-review process and provided valuable comments that helped to ensure the quality of the report. A list of the main contributors is included at the start of each chapter. The list is not comprehensive. Specifically, it does not include the many national institutes, laboratories and organizations, and their staff, which have been involved in various countries in AOA-related monitoring and research. Apologies, and no lesser thanks are given to any individuals unintentionally omitted from the list.

The support from the Arctic countries and non-Arctic countries implementing research and monitoring in the Arctic is vital to the success of AMAP. The AMAP work is essentially based on ongoing activities within these countries, and the countries that provide the necessary support for most of the experts involved in the preparation of the AMAP assessments. In particular, AMAP would like to acknowledge Norway for taking the leadcountry role in this assessment and thank Canada, Norway, Sweden, USA and the Nordic Council of Ministers for their financial support to the AOA work.

The AMAP Working Group is pleased to present its assessment to the Arctic Council and the international science community.

Richard Bellerby (AOA assessment Chair)

Russel Shearer (AMAP Chair)

Lars-Otto Reiersen (AMAP Executive Secretary)

Oslo, May 2013

¹ AMAP, 2013. AMAP Arctic Ocean Acidification Assessment: Summary for Policy-makers. (www.amap.no/documents/doc/amap-arctic-ocean-acidificationassessment-summary-for-policy-makers/808).

² AMAP, 2013. Arctic Ocean Acidification 2013: An Overview. Arctic Monitoring and Assessment Programme (AMAP), Oslo, Norway.

1. Introduction

Authors: Lars-Otto Reiersen, Robie Macdonald, Leif Anderson, Richard Bellerby, Simon Wilson

1.1 What is ocean acidification?

Ocean acidification refers to a reduction in the pH of the ocean over an extended period, typically decades or longer, which is caused primarily by uptake of carbon dioxide from the atmosphere, but can also be caused by other chemical additions or subtractions from the ocean. Anthropogenic ocean acidification refers to the component of pH reduction that is caused by human activity. (IPCC, 2011)

As a consequence of the higher amounts of carbon dioxide (CO₂) in the atmosphere, the upper ocean has been forced to take up more CO₂ through gas exchange (Figure 1.1) such that it now holds approximately one quarter of the human-generated CO₂ load. This uptake of CO₂ occurs because the surface ocean

Box 1.1 Seawater pH

Seawater pH, at the heart of discussions concerning ocean acidification, is a confusing and poorly understood concept. The hydrogen ion concentration ([H⁺]) cannot be directly measured and can only be estimated using operationally defined scales, which are based on the nature of the waters being analyzed and the chemistry of the buffers used to calibrate them. Therefore, different disciplines use different pH scales, and there is no unifying agreement on a single, appropriate scale to use.

Nevertheless, independent of the chosen scale, pH represents the *acidity* or concentration of hydrogen ions in a solution. Expressed on a log scale, the pH range between 0 and 14 approximately represents hydrogen ion concentrations between 1 and 10⁻¹⁴ moles per liter. Thus, a ten-fold change in the hydrogen ion concentration results in a pH change of 1. The solution is said to be acidic if the pH is below 7, to be alkaline if above 7 and to have a neutral pH at pH=7 (see graphic for examples of common solutions and their pH values). Importantly, a solution can increase in acidity (i.e.,

attempts to match its dissolved CO_2 content to the partial pressure of atmospheric CO_2 , which has risen from about 280 parts per million, prior to the Industrial Revolution (ca. 1850), to the present 395 parts per million. As CO_2 in the atmosphere continues to increase with further burning of fossil fuels, so too will the amount of CO_2 dissolved in the ocean. Combined with water, dissolved CO_2 forms carbonic acid (H₂CO₃), a weak acid, which then partially dissociates to release H⁺ ions (Figure 1.1) thereby increasing the acidity of the ocean. Because the concentration of H⁺ ions ([H⁺]) in aqueous solutions can vary through many orders of magnitude, chemists use the term pH to describe the acidity of water (see also Box 1.1), where pH is defined as minus the logarithm of [H⁺] (i.e., pH = $-log_{10}$ [H⁺]).

the pH goes down) without becoming acidic, and an alkaline solution does not have to have a high alkalinity.

Ocean acidification is the ongoing reduction in seawater pH predominantly due to the uptake of CO_2 from the atmosphere. As CO_2 increases, the acidity increases and pH decreases. From a mean global pH of about 8.2 at the beginning of the Industrial Revolution pH has decreased by about 0.1 and with continuing combustion of fossil fuels may fall another 0.25 units by the end of the century (note that these are global averages and many regions will experience greater and lesser pH reductions). These changes represent about a 30% and 125% increase, respectively, in hydrogen ion concentration since 1860 (see right-hand part of graphic for the percentage change in ocean acidity relative to a starting point of pH 8.2 – the value at the start of the industrial era).

Ocean acidification will not, however, lead to an acidic ocean – there is not enough fossil fuel carbon to burn to result in a seawater pH below 7.



Source: Modified from National Oceanic and Atmospheric Administration (NOAA) Pacific Marine Environmental Laboratory (PMEL).



Figure 1.1. A schematic illustration of the inorganic carbon system in the Arctic Ocean. Black lettering shows the equilibria between inorganic carbon components in water; brown lettering shows the potential for terrestrial organic carbon to contribute carbon dioxide (CO_2) through oxidation; green lettering shows the potential for the marine organic carbon cycle to remove CO_2 through primary production or add CO_2 through oxidation of organic carbon, and to transfer CO_2 to the deep ocean through the rain of particulate organic carbon; red lettering shows the potential to remove carbonate ions (CO_3^{-2}) by precipitation of solid carbonate or to add CO_3^{-2} by dissolving solid carbonate. A non-intuitive consequence of this equilibrium system is that the addition of CO_2 at the left side of the equation, either through uptake from the atmosphere or oxidation of organic carbon, promotes the dissolution of solid carbonate at the right side of the equation.

Processes other than atmospheric gas exchange can produce CO₂ or remove it from the ocean and thus affect ocean pH. As is evident in Figure 1.1, carbon dioxide concentration in the ocean $([CO_2]_{aq})$ may also be strongly affected by the organic carbon cycle. Terrestrial organic carbon from rivers and coastal erosion enters the ocean where a proportion of it oxidizes through photolytic or microbial oxidation, producing CO_2 (left-hand side, Figure 1.1). The addition of $[CO_2]_{aa}$ by this process will lead to higher [H+] and a decline in pH in exactly the same way that CO₂ invasion from the atmosphere does. Within the ocean itself, CO₂ may be removed through primary production during which phytoplankton convert CO₂ to organic carbon. This removal process then leads to an increase in pH. Much of the organic carbon produced by phytoplankton becomes oxidized back into CO₂ through grazing by zooplankton, metabolism and respiration and by heterotrophic microorganisms (Figure 1.1, green lettering). However, a small proportion of the organic carbon produced by phytoplankton ultimately enters the deep ocean or becomes buried in sediments and therefore contributes a net loss of CO₂ from the upper ocean.

1.2 Ocean acidification in the global system

During the past decade, ocean acidification has emerged as a new challenge to oceans around the world. Although it has long been understood that the addition of CO_2 to water leads to acidification, it was only in 2004, following an international symposium entitled *The Ocean in a High-CO_2 World* and a report by the Royal Society (The Royal Society, 2005) that acidification became an urgent issue in ocean research. This 'other CO_2 problem' (Turley and Blackford, 2005; Doney et al., 2009) is a direct consequence of the increase in atmospheric CO_2 during the past two centuries due primarily to human use of fossil fuels like coal, oil and gas (Sabine et al., 2004; Forster et al., 2007; IPCC, 2007).

Given the number of processes in the oceans that impact $[CO_2]_{aq}$, and thus also pH, there is large variability with time (diurnally, annually) and space, especially near the sea surface.

Within the large seasonal and spatial variability in pH, the consequence of fossil fuel use can be seen in the Pacific Ocean, where a time series collected since 1989 in surface waters near Hawaii, shows a slow, relentless decline in average pH that mirrors the increase in atmospheric CO_2 (Figure 1.2). Since the beginning of the Industrial Revolution, the CO_2 accumulation in ocean surface waters has led to an increase in [H⁺] of ~30%, which corresponds to an average decrease in pH by 0.1 (Caldeira and Wickett, 2003; Doney et al., 2009).

A pH decline in the ocean is of concern because it has the potential to exert far-reaching effects on biological and geochemical systems (e.g., Orr et al., 2005; Guinotte and Fabry, 2008; Hester et al., 2008; Cooley and Doney, 2009; Dore et al., 2009).

Acidification may affect biological systems in a number of ways, these include:

• Directly, through for example interference in the calcification process required by many organisms to build hard-body structure (see also Box 1.2). The list of marine taxa relying on structural carbonate is large and varied, including for example, plankton, mollusks, echinoderms and corals.



Figure 1.2. Time series of atmospheric CO_2 at station Mauna Loa, and surface water pCO_2 and pH at station ALOHA in the subtropical North Pacific Ocean. Source: Modified after Doney et al. (2009).

- Directly, through physiological impacts at various life-stages, impacts on health through, for example, compromised immune systems, changes to behavior, competition and resource partitioning.
- Indirectly, through alteration of food webs, which could result from loss of carbonate species, and interference with olfaction, respiration and metabolism, and impairment of reproductive success (Gattuso and Hannson, 2009; Williamson and Turley, 2012). On the other hand, some species may thrive in a higher CO₂ world, for example sea-grasses (Hall-Spencer et al., 2008), while others may adapt (Langer et al., 2006).

Box 1.2 Interference in calcification

Interference with the calcification process as a consequence of ocean acidification is one of the more well-known biological responses to ocean acidification. However, this response is complex and cannot be understood by reference to pH by itself. The precipitation of calcium carbonate (CaCO₃), or its dissolution, depends on the product of the concentrations of carbonate and calcium ions (i.e., $[CO_3^{2-}] \times [Ca^{2+}])$ (Figure 1.1, right side). Although $[CO_3^{2-}]$ is related directly to pH through the ocean's carbonate system as depicted in Figure 1.1, [Ca²⁺] may be affected by other processes, for example by dilution with river water or sea-ice melt containing lower concentrations of dissolved calcium, and these processes must be factored into the calculations before any inference can be made about the stability of solid CaCO₃ Finally, CaCO₃ manifests a number of different crystalline structures: in the ocean, the two predominant biological forms are aragonite and calcite with the former tending to dissolve first as pH is lowered. These issues are discussed in more detail in Chapter 2, section 2.2.

1.3 Why assess ocean acidification in the Arctic?

1.3.1 The sensitivity of the Arctic Ocean to acidification

With respect to ocean acidification, conditions in the Arctic Ocean are different from temperate oceans in several important respects. For example, primary production has a smaller diurnal signal at high latitudes, also reflected in pH, because the light conditions do not vary as much throughout the day. On the other hand, the surface water annual pH cycle may be substantial. For example, a $pH^{tot}_{in situ}$ increase in the surface waters of the Barents Sea of between 0.1 in the south to 0.25 in the north was observed from March to May 1998 (Kaltin, et al., 2002). The spatial variability can be even greater, with $pH^{tot}_{in situ}$ values as low as 7.5 in the top 10 m close to the coast and up to about 8.3 in the surface waters of the deep basin. This regional variation may be attributed to oxidation of terrestrial organic matter near the coast.

Commencing in 1998, a sequence of AMAP assessments has shown, without exception, that the Arctic is especially sensitive to human activities at the global scale (AMAP, 1998, 2011a,b; ACIA, 2005; McGuire et al., 2010). Indeed, in a prescient note, Walsh (1991) suggested that ice cover in the Arctic might provide a bellwether for global change. All oceans face acidification due to higher amounts of CO₂ in the atmosphere, but based on experience, and knowing that the organic carbon cycle in the Arctic contains a number of feedbacks that provide opportunities for surprise (Bates and Mathis, 2009; McGuire et al., 2010) it seems prudent to anticipate that the Arctic Ocean and its marginal seas will in some way be especially sensitive to CO₂ - that is, the Arctic Ocean and its marginal seas are also likely to provide a bellwether for ocean acidification (Bellerby et al., 2005; Fabry et al., 2009; Steinacher et al., 2009; Turley et al., 2010).

A number of points of Arctic Ocean sensitivity to acidification can be anticipated. Seasonal under-saturation for aragonite in surface and shallow subsurface waters of the northern polar seas has already been observed (Yamamoto-Kawai et al., 2009, 2011; Bates et al., 2009), and this ocean is projected through models to be persistently under-saturated by the mid-21st century (Steinacher et al., 2009; Denman et al., 2011). Even if acidification does not lead to local or regional extinction of sensitive species that rely on aragonite for structure, it may set the stage for invasive species to displace such species (e.g., Vermeij and Roopnarine, 2008), which can no longer compete due to stress from acidification.

The recent extensive loss of summer sea ice in the Arctic Ocean has accelerated the invasion of anthropogenic CO_2 into the ocean while at the same time providing a widespread source of brackish water that is impoverished in Ca^{2+} . Sea-ice melt is augmented by a very large cumulative inflow of river water, also low in $[Ca^{2+}]$, further leading toward under-saturation of aragonite. It may be speculated as to whether or not the northern polar ocean has crossed a sea-ice tipping point (Serreze, 2011; Wadhams, 2012), but it seems clear that we are heading for the aragonite tipping point.

The Arctic shelves have several points of vulnerability. They are well supplied with dissolved and particulate organic carbon (DOC and POC, respectively) from rivers and coastal erosion (estimated at ~11 Mt/y for POC alone; Stein and Macdonald, 2004a). Recently, Vonk et al. (2012) suggested that this terrigenous organic carbon supply may have been seriously underestimated and that as much as 44 ± 10 Mt/y is released to the East Siberian shelf seas alone by the Siberian ice complex, with two-thirds becoming oxidized to CO₂ and the remainder buried in shelf sediments. Whether or not this enormous organic carbon release is a consequence of recent permafrost thawing, if it becomes metabolized in water or sediments beneath the ice, this terrigenous organic carbon will provide a large source of CO₂ (Figure 1.1) that may not be easily evaded to the atmosphere. A particularly sensitive location, in this regard, is likely to be the shallow, nearshore waters that are seasonally covered by continuous landfast ice, thus shutting down CO₂ evasion while still permitting metabolism of terrigenous and marine organic carbon (Garneau et al., 2006). The Arctic shelves tend to have higher primary productivity than the interior ocean, especially the Chukchi and eastern East Siberian Seas, which exhibit some of the highest productivities observed in the world ocean (Grebmeier et al., 2006; Anderson et al., 2011a). The downward flux of organic-rich particulate matter associated with this production then supplies a powerful metabolic engine that releases CO₂ and thus contributes low pH (Bates et al., 2009). Upwelling of CO2-rich water onto shelves is an effective pathway to place CaCO₃-corrosive water near the surface (Feely et al., 2008). With the loss of sea ice, upwelling has been projected to increase in the Arctic (Carmack and Chapman, 2003), and in the western Arctic Ocean upwelling is likely to access CO₂-rich halocline water, whose origin is the Pacific Ocean via the Chukchi Shelf.

Global warming has the potential to destabilize the large amounts of methane hydrates stored in sediments along the continental margins, releasing methane (CH₄) into the water column and possibly the atmosphere, with the Arctic particularly vulnerable in this respect. Release of CH₄ from melting hydrates in these areas could enhance ocean acidification and oxygen depletion in the water column (Biastoch et al., 2011). Recently, large amounts of CH₄ have been observed to seep out of bottom sediments in the Siberian seas (Shakhova et al., 2010). This phenomenon may be widespread (Damm et al., 2007; McGuire et al., 2010) over Arctic shelves and offer the possibility of another sort of tipping point (Archer et al., 2009b). In the context of ocean acidification, the metabolism of CH_4 would provide another pathway to contribute CO, to bottom waters.

1.3.2 The sensitivity of Arctic marine ecosystems to acidification

Arctic Ocean acidification has a number of potential biological and ecological consequences, including biogeochemical feedbacks. Arctic ecosystems are characterized by low biodiversity and simple food webs. The large dataset of global responses to ocean acidification contrasts with very limited and scarce data concerning responses of Arctic species and ecosystems, however parallels imply that Arctic ecosystems are likely to be at risk from the effects of ocean acidification.

1.3.3 The socio-economic value of the Arctic – and the potential cost of Arctic Ocean acidification

Arctic fisheries are important both regionally and globally. In 2002, the total catch of wild fish in the Arctic amounted to 7.26 million tonnes, around 10% of the world catch of fish. Shrimp and crabs, and farmed fish contributed a further 0.36 and 0.1 million tonnes, respectively (Statistics Norway, 2006). Commercial fisheries also provide an important component in the economy of the Arctic region. Fishing and fish processing accounted for almost 2% of the Arctic regional economies of Alaska, Iceland, Greenland and the Faroe Islands and the Arctic regions of Canada, Finland, Norway, Russia and Sweden in 2005 (Statistics Norway, 2009). The industry is particularly important in the Faroe Islands, Greenland, Iceland, and Arctic Norway, contributing approximately 20%, 12%, 7% and 7% to the regional economies of these four Arctic countries/regions.

Climate change, by itself, is expected to alter northern fisheries, both through changes in productivity (which is generally expected to increase with warming and reduction of sea ice – although this is not certain) and changes in species composition and food-web structure (e.g., as new species move into the Arctic). Arctic Ocean acidification adds another dimension of change, targeting species in a very different way. Marine resource utilization is central to the cultures and lifestyle of northern communities, in particular coastal communities.

People have been living in and adapting to changing conditions in the Arctic for thousands of years. The Arctic is home to around four million residents of which about 10% are indigenous peoples, spread over many communities in the Arctic (AMAP, 1998). The proportion of the population that is indigenous varies widely. In Nunavut, Canada about 90% of the population is Inuit; Inuit also make up most of the population of Greenland, whereas the Saami account for only about 2.5% of the population in northern Scandinavia. Arctic indigenous peoples have survived by using the natural resources available to them. Traditional food use is culturally specific and varies widely between the different communities and regions of the Arctic. People living along the coast mainly rely on marine foods, including both marine mammals and fish, while people living inland eat mainly reindeer/caribou/muskox and freshwater fish. These traditional foods are rich in nutrients and energy, and thus permitted the first migrations into the region and have continued to support life there ever since.

Over the past three decades AMAP has documented the contamination of important traditional food items by persistent organic pollutants, mercury and radionuclides as a result of long-range transported and local contamination, (AMAP, 1997, 2002, 2009, 2011b). This aspect of food security is a major concern for Arctic indigenous peoples. In cooperation with national governments and responsible agencies, AMAP has promoted actions at both the international level (for example through the Stockholm Convention on Persistent Organic Pollutants and the negotiation of a global agreement on mercury under UNEP) and local level (such as through food consumption advice) to reduce human exposure to these contaminants.

Some parts of the Arctic are experiencing increasing immigration of people from outside the area due to the employment opportunities provided by oil and gas resource development, mining and other industries. One consequence of this, and of the introduction in the Arctic of modern communications and associated media and advertising, has been an increasing introduction of western foods resulting in dietary shifts that are both more expensive and less healthy.

The identification of ocean acidification as a new factor in the Arctic food security debate has led some indigenous peoples (reliant on marine foods) to express concern about what may happen to marine life if the Arctic Ocean becomes more acidic, and in this connection its potential effect on their marine foods and their traditional culture. This assessment is an attempt to clarify the situation regarding the future consequences of increasing Arctic Ocean acidification also in this regard.

Ocean acidification does not act alone; it is occurring along with other global stressors that are now defining the anthropocene (Steffen et al., 2011) including climate change (e.g., warming, altered hydrology), change in the carbon and nitrogen cycles, alteration of marine foodwebs, and contamination by industrial chemicals. The present assessment rests on a foundation comprising a sequence of assessments dealing comprehensively with these sorts of stressors as they relate to the Arctic (AMAP, 1998, 2011a; ACIA, 2005; McGuire et al., 2010).

1.4 The geographical context of the Arctic Ocean acidification assessment

1.4.1 What are the Arctic Ocean and its marginal seas?

The geographical delineation of the Arctic Ocean and its marginal seas used in this assessment follows previous boundaries defined by AMAP (AMAP, 1998, 2011a). Specifically, the ocean regions considered include the Arctic Ocean, the Bering Sea, the Greenland-Iceland-Norwegian Seas, the Labrador Sea, the Canadian Arctic Archipelago and Hudson Bay (Figure 1.3). This northern Arctic region includes deep basins (Eurasian and Canada Basins, Baffin Bay, Nordic Seas), broad shelf seas (Barents, Kara, Laptev, East Siberian, Chukchi, Bering, Beaufort) and Canadian inland seas (Canadian Arctic Archipelago, Hudson Bay).



Figure 1.3. The Arctic Ocean and its marginal seas.

1.4.2 The functioning of the Arctic Ocean in the context of acidification

The oceanography of the Arctic Ocean has been well reviewed in previous assessments (ACIA, 2005), and is discussed further in Chapter 2 mainly in the context of features that are crucial for understanding the Arctic's response to ocean acidification. The Arctic Ocean receives its salt water from the Atlantic and Pacific Oceans (Figure 1.4). The Atlantic water enters partly over the Barents Shelf, where it is strongly modified by cooling and ice formation, and partly through Fram Strait. This latter passage is deep, and provides access for the exchange of all the deep water in the Arctic Ocean.

Water from the Pacific Ocean enters the Arctic Ocean via Bering Strait, where it must pass over the broad Chukchi Shelf. Because the Bering Strait sill is shallow (~50 m) and the Pacific water is fresher than the Atlantic water, the Pacific inflow rides above the Atlantic layer within the Arctic, supplying most of the surface water throughout the Canada Basin (e.g., see Ekwurzel et al., 2001). The disposition of Pacific water in the Arctic Ocean is subject to large decadal-scale variation subject to atmospheric pressure fields which vary with the Arctic Oscillation (McLaughlin et al., 1996). The Pacific water entering the Arctic supplies most of the water exiting through the Canadian Arctic Archipelago (McLaughlin et al., 2006) and thence through Hudson Bay and into Baffin Bay (Jones et al., 2003) (Figure 1.4). Because the North Pacific Ocean lies at the end of the global thermohaline circulation system (Schmitz, 1995), its deep water contains high accumulated CO₂ concentrations due to regeneration of organic matter without ventilation to release the CO₂ during the thousand or more years taken to arrive in the North Pacific Ocean. Accordingly, this water is among the most corrosive in the world oceans to CaCO₃ (Feely et al., 2004, 2008; Sabine et al., 2004).

More than 3300 km³/y of runoff (Serreze et al., 2006) together with net precipitation and the relatively fresh Pacific water ensures strong, pervasive stratification near the surface of the Arctic Ocean (Figure 1.5). This strong layering, which has long been recognized as the defining feature of the Arctic Ocean's water column (Treshnikov, 1959), led McClelland et al. (2012) to liken this ocean to a large, complex estuary. The stratification of the Arctic Ocean inhibits exchange between shallow (~50-100 m) and deep ocean waters. Stratification thus has the important role of limiting deep ventilation and atmospheric CO₂ exchange in the interior ocean, and predominantly, these processes must occur at or near the Arctic Ocean's margins, with recently ventilated water then transporting into the interior Arctic Ocean horizontally (Figure 1.5). An important feature of riverine water and precipitation with respect to ocean acidification is that these freshwaters have a lower capacity to neutralize acid than seawater (Yamamoto-Kawai et al., 2005). The accompanying lower Ca2+ concentrations, therefore, lead to under-saturation with respect to aragonite more easily than found in seawater (see Figure 1.1).

Melting during spring and summer converts the 1-2 m layer of floating sea ice into brackish water, which mixes into the surface ocean, further enhancing stratification. Even more important insofar as acidification is concerned, ice melt contains very low concentrations of ions like Ca2+, which therefore means that these waters have even greater sensitivity than river water toward CaCO₃ dissolution when acidified by CO₂ (Yamamoto-Kawai et al., 2011). When sea ice forms in winter it rejects brine into the water column, thus promoting mixing and convection, and under favorable circumstances, deep convection, which can



Figure 1.4. Major surface ocean currents within the High Arctic, Low Arctic and sub-Arctic areas of the Arctic Ocean and its



Figure 1.5. Stratification of the Arctic Ocean due to freshwater inputs. Low salinity waters are added to the surface and halocline layers via precipitation and runoff, Pacific inflow via the Bering Strait, and the sea-ice distillation process. Low salinity waters and sea ice are subsequently advected through Fram Strait and the Canadian Archipelago into the convective regions of the North Atlantic. Stratification, and the addition of poorly-buffered freshwater are two important factors that enhance the Arctic Ocean's sensitivity to ocean acidification.

sequester CO_2 to depth in the ocean (Rysgaard et al., 2007). Sea ice cover may also present a partial barrier to exchange of gases between the ocean and atmosphere, but with respect to CO_2 it plays an active role in this exchange through the formation of brine and its subsequent movement within brine channels (Miller et al., 2011). Finally, in regions of the Arctic where sea ice is lost seasonally, the cadence of primary production during open water versus regeneration during ice cover has been proposed as a rectifying mechanism that enhances CO_2 uptake by the surface ocean (Yager et al., 1995).

A special feature of the Arctic Ocean is a pervasive, cold stratified layer occupying the water column from about 150 to 250 m depth. The water in this layer, termed the Arctic halocline (Aagaard et al., 1981), is produced predominantly over the Barents and Chukchi Shelves through cooling and sea-ice formation, which increase the water's density (Jones and Anderson, 1990; Rudels et al., 2004). This water, especially the component produced over the Chukchi shelf, contains elevated concentrations of regeneration products including CO_2 . Like other nutrient-rich waters, when this water is upwelled it delivers to the surface water that is relatively corrosive to $CaCO_3$ (e.g., see Feely et al., 2008).

The Arctic Ocean contains enormous shelf areas, fully half of the ocean's area. Accordingly, processes over the shelves are extremely important to the functioning of this ocean, and these regions are also most important to the inhabitants of the Arctic in terms of travel and access to food. And yet, these regions are also among the most sensitive to climate change (ACIA, 2005).

A last consideration for the Arctic Ocean is the question of how quickly water masses can respond to change in surface-water CO_2 content due to the ongoing loading by anthropogenic CO_2 . Anderson et al. (1998) pointed out that, in terms of inorganic carbon budgets, the Arctic Ocean is playing catch-up with the atmosphere in a process where the outflow at Fram Strait exports old water equilibrated with a former atmosphere, containing less anthropogenic CO_2 , to be replaced by recently ventilated inflowing waters reflecting the present atmospheric CO_2 concentration. For the surface layer (top ~50 to 100 m) in the Arctic Ocean, the residence times are relatively short (<10 y) suggesting that these waters are influenced more directly by the atmosphere and they can respond faster to changing atmospheric CO₂ increases. Although sea ice may impede exchange with atmospheric CO₂ in this layer, the recent demise of permanent ice cover over an extensive area of the Arctic Ocean (Stroeve et al., 2007; Schiermeier, 2012) suggests that exchange now occurs with much less impediment from the ice (Bates et al., 2006). Deeper in the water column, the Atlantic layer pervades the Arctic Ocean. Circulation occurs more rapidly along the boundaries, with exchange into the interior ocean occurring more slowly such that the ventilation age in this water is approximately 30 years, or more and then more slowly into the interior regions of the ocean, taking approximately 30 years to transit the interior ocean. This time frame suggests that CO₂ concentration in the Atlantic layer waters will lag that in the atmosphere by up to three decades. Finally, the basin waters have been out of contact with the surface for a century or more, suggesting that these will reflect periods well before the recent rise in atmospheric CO₂.

1.5 Aim of the present assessment

The objectives of the present assessment are to provide the Arctic Council with timely, up-to-date, and synthesized scientific knowledge about the present status, processes, trends, and future consequences of changes in Arctic Ocean acidification.

This assessment complements previous AMAP assessments of climate change in the Arctic, including the Arctic Climate Impact Assessment (ACIA, 2005) and the Snow, Water, Ice and Permafrost in the Arctic (SWIPA) assessment (AMAP, 2011a).

1.6 Reader's guide to this assessment

This scientific report presents the findings of the Arctic Ocean acidification assessment as developed by the report authors, produced under their responsibility. A separate overview report and summary for policymakers were produced under the responsibility of the AMAP Working Group. The summary for policy-makers was developed for presentation to the Arctic Council Ministers at their meeting 14-15 May, 2013 in Kiruna, Sweden, and includes policyrelevant scientific recommendations. The scientific report (the present report) provides the validated scientific basis for all statements made in the overview report and the summary for policymakers, as confirmed by the lead authors of the scientific assessment.

This report contains five chapters.

Chapter 1 sets the stage for the assessment and describes its scope.

Chapter 2 presents an introduction to the carbon biogeochemical system in the Arctic Ocean: processes that influence the carbon system; processes sensitive to ocean acidification; present sources and sinks of carbon in the Arctic Ocean; current features of ocean acidification in the different Arctic seas; and a discussion of future scenarios.

Chapter 3 provides a description of the biological responses to ocean acidification: impacts on calcification rate, on polymorph mineralogy and elemental partitioning in hard parts like skeletons and shells; viral effects on marine organisms including fish and mammals; effects observed within the Arctic area and in other relevant areas including sub-Arctic waters and Antarctic waters; and interactions between multiple stressors.

Chapter 4 presents analyses of how changes in ocean acidification may affect the economics of marine fisheries in regions of the Arctic and on food security and cultural issues for coastal Arctic indigenous communities.

Chapter 5 presents an overall summary of the major findings and gaps in knowledge on Arctic Ocean acidification. The summary is based on the logical consequences of and conclusions stemming from the scientific findings presented in the preceding chapters. Lead authors: Richard Bellerby, Leif Anderson, Kumiko Azetsu-Scott, Peter Croot, Robie Macdonald, Lisa Miller, Jon Olafsson, Nadja Steiner

Contributing authors: Andreas Andersson, Craig Carlson, Melissa Chierici, Agneta Fransson, Emil Jeansson, Fred Mackenzie, Jeremy Mathis, Are Olsen, Ute Passow, Michiyo Yamamoto-Kauai

2.1 Introduction

A consequence of the persistent release of carbon dioxide (CO₂) to the atmosphere following fossil fuel combustion and changes in land use is that there is an increasing net air-to-sea transport of CO₂. Although this oceanic uptake will reduce the potential for greenhouse warming that would have arisen had the gas remained in the atmosphere, it will also result in major changes in ocean chemistry. The most obvious signal in this respect is the fall in ocean pH and the change in the speciation of the marine carbonate system. The Arctic Ocean is one of the regions where ocean acidification is occurring fastest. From a baseline where the seawater is already poorly buffered and thus small changes in CO₂ content have large changes in pH, there are a multitude of stressors that act on the Arctic Ocean amplifying the acidification.

This chapter summarizes carbonate chemistry in seawater (Section 2.2) and reviews the major processes influencing the Arctic Ocean carbonate system (Section 2.3). The chapter also describes some of the biogeochemical processes sensitive to ocean acidification (Section 2.4). Section 2.5 addresses the major sources and sinks of carbon to the Arctic Ocean, and presents a regional breakdown of contemporary rates of ocean acidification. Finally, simulations from earth system models and regional models are analyzed to project potential changes to the Arctic Ocean carbonate system (Section 2.6).

2.2 Carbonate system in seawater

Carbon dioxide acidifies seawater, because CO_2 is one of a number of compounds (including sulfur and nitrogen dioxides) that hydrolyze water in solution. That is, when CO_2 is added to water, it splits the water molecule, releasing protons (acid):

$$CO_2 + H_2O \Leftrightarrow H_2CO_3 \Leftrightarrow H^+ + HCO_3^- \Leftrightarrow 2H^+ + CO_3^{2-}$$
 Eqn. 1

In seawater and other solutions with high concentrations of carbonate (as well as other compounds participating in acid-base



The precise relative concentrations of the carbonate system species, as well as the pH, in seawater are related through the conditional equilibrium constants:

$$K'_{1} = \frac{[H^{+}][HCO_{3}]}{[H_{2}CO_{3}]}$$
 Eqn. 2

$$K'_{2} = \frac{[H^{+}][CO_{3}^{2}]}{[HCO_{3}^{-}]},$$
 Eqn. 3

where, K'_1 and K'_2 are defined empirically, based on experiments in sodium chloride and real and artificial seawater solutions (e.g., Hansson, 1973; Mehrbach et al., 1973; Dickson and Millero, 1987; Roy et al., 1993; Lueker et al., 2000). When K'_1 and K'_2 are coupled with the definitions of the directly measurable variables pH, total alkalinity (A_T), total inorganic carbon concentration (C_T), and CO₂ fugacity (f_{CO2} ; closely related to partial pressure),

$$pH = -log [H^+]$$
Eqn. 4

$$A_{T} \approx [HCO_{3}^{-}] + 2[CO_{3}^{2-}] + [B(OH)_{4}^{-}] + [OH^{-}] - [H^{+}] Eqn. 5$$

$$C_{T} = [H_{2}CO_{3}] + [HCO_{3}] + [CO_{3}]$$
 Eqn. 6

$$fCO_2 = \frac{[H_2CO_3]}{K_0}$$
 Eqn. 7

(where K'_0 represents the CO₂ solubility in seawater; Weiss, 1974), it becomes possible to fully define the carbonate system with measurements of only two variables (e.g., pH and alkalinity can be calculated from measurements of C_T and *f*CO₂).



Figure 2.1. Changes in carbonate species concentration with changes in seawater pH. The light grey band indicates the decline in surface seawater pH since the Industrial Revolution and the dark grey band represents the projected change to the end of the century. Dickson et al. (2007) presented full details of these calculations, including pitfalls to avoid. Certain pairs of variables are preferable for calculating the others, due to their superior analytical precision (e.g., Dickson and Riley, 1978). As analytical methodologies and instrumentation have evolved and as reference materials have become available, the two best variables to measure have not only changed over the years, but also differ between laboratories, depending on resources available. This complicates efforts to compare results from different studies, particularly those from more than 15 years ago.

Although the carbonate system in seawater is more complicated than implied by Equation 1, and therefore less intuitive, adding C_T to a seawater solution of otherwise fixed composition, does simply decrease the pH (Figure 2.2). An important effect of the high total alkalinity and the accompanying complicated chemistry in seawater is to buffer the solution; not only is the pH of seawater higher than in freshwaters, despite the higher C_T concentrations in seawater, but the ocean also resists more strongly further decreases in pH as more C_T is added (Table 2.1). As the table shows, the freshwater influences in the Arctic (i.e., river inflows and sea-ice melt) serve to decrease the pH and reduce the buffering capacity, at least in surface waters.

One of the primary geochemical impacts of seawater acidification is to increase the solubility of calcium carbonate $(CaCO_3)$ minerals. Thus, somewhat counter-intuitively, adding CO₂ to seawater causes CaCO₃ to dissolve:

$$CaCO_{3(s)} + CO_2 + H_2O \leftrightarrow Ca^{2+} + 2HCO_3^{-}$$
 Eqn. 8

The saturation state of a CaCO₃ mineral (i.e., its tendency to resist dissolution) in a specific solution is represented by the saturation state:

$$\Omega = \frac{[Ca^{2+}] [CO_3^{2-}]}{K'_{sp}}$$
 Eqn. 9

where K'_{sp} is the equilibrium solubility product (which depends on salinity, temperature, pressure, etc.):

$$K'_{sp} = [Ca^{2+}]_{sat} [CO_3^{2-}]_{sat}$$
 Eqn. 10

The saturation state for a specific mineral is generally indicated with a subscript: Ω_{Ar} represents the saturation state for aragonite and Ω_{Ca} that for calcite.





Oceanic carbon uptake

Figure 2.2. Changes in total inorganic carbon (C_T) , pH and the partial pressure of carbon dioxide in seawater with increasing ocean carbon uptake and constant total alkalinity (A_T) .

In general, if Ω is greater than 1, the mineral is stable, but if Ω is less than 1, the mineral is vulnerable to dissolution. Each specific mineral has a different solubility product; for example, of the two most common biogenic CaCO₃ minerals in the oceans - aragonite and calcite - aragonite has a higher solubility product than calcite (Mucci, 1983). This means that aragonite dissolves more readily than calcite, becoming unstable even in waters at relatively high pH values (see Riebesell et al., 2010 for a more detailed discussion). However, calcite minerals in which some Ca2+ ions are replaced by Mg2+ ions, called hereafter Mg-calcite, are even more soluble than aragonite. With lower initial surface water Ω and strong freshwater inputs (thus lower Ca²⁺ concentrations), CaCO₃ minerals in the Arctic Ocean are particularly susceptible to dissolution. Depths at which $\Omega = 1$ are often referred to as the 'saturation horizon'. In the Arctic Ocean, not only is the pH decrease from anthropogenic CO₂ causing the dissolution of CaCO₃, but increasing river run-off and the pulse of sea-ice melt are further diluting the seawater, causing additional dissolution. (See Section 2.3.5 for further details on this topic.)

Surface water type	C _r , μmol/kg	Α _T , μmol/kg	pH_{T}	pH change with +1 μ mol/kg C _T
Arctic seawater	2140ª	2290ª	8.12	-0.003
Tropical seawater	2040ª	2390ª	8.18	-0.002
High-alkalinity river (e.g., Yukon)	1700ª	1710ª	7.84	-0.005
Low-alkalinity river (e.g., Kolyma)	600ª	500ª	7.28	-0.004
Sea-ice melt	400ª	400ª	7.97	-0.026
Pure water	0	0	7	-0.992

^a Typical values used solely for comparison purposes.

2.3 Influences on the marine carbonate system in the Arctic Ocean

2.3.1 Sea ice

Characteristic of polar oceans, sea ice can substantially affect airsea CO_2 exchange and, thereby, the acidification state of Arctic waters. In the simplistic view, the presence of a solid sea-ice cover has been thought to inhibit both primary production and direct air-sea CO_2 transfer, however, a number of sea-ice associated processes also act to enhance gas exchange, counteracting and possibly even superseding this inhibition effect.

Although sea ice reduces light transmission to the underlying phytoplankton community, these communities appear to be well-adapted to low-light conditions, and observations of extensive under-ice phytoplankton blooms (e.g., Fortier et al., 2002; Mundy et al., 2009; Arrigo et al., 2012) indicate that at least first-year sea ice may not limit surface water primary production as much as has been assumed. In fact, sea-ice algal communities are important contributors to Arctic primary production, both within the sea ice and as possible seed populations for under-ice blooms (e.g., Arrigo et al., 2010). In addition, the timing of sea-ice retreat and formation can apparently have dramatic impacts on water-column primary production, depending on geographical and meteorological conditions, because in the absence of ice, upwelling events and wind-induced mixing can break down surface stratification and thus foster blooms (Brugel et al., 2009; Tremblay et al., 2011).

Contrary to the long-standing assumption that sea ice is a barrier to gas transport, it is now clear that gases, including CO₂, are able to move within and through sea ice (Gosink et al., 1976; Semiletov et al., 2004; Zemmelink et al., 2006; Nedashkovsky and Makshtas, 2010; Nomura et al., 2010a,b; Miller et al., 2011; Papakyriakou and Miller, 2011). In the presence of a solid, cold ice cover, the fluxes appear to be small, with upward fluxes from the ice to the atmosphere in winter switching to downward fluxes from the atmosphere to the ice as spring progresses (Nomura et al., 2010a; Miller et al., 2011; Papakyriakou and Miller, 2011). These fluxes are clearly between the atmosphere and the ice, not the underlying water; that is, the ice is not a passive conduit between the surface waters and the atmosphere but constitutes a distinct carbon pool which exchanges with both the atmosphere and the ocean. While sea ice contains less inorganic carbon than an equal volume of the underlying seawater (Rysgaard et al., 2009; Miller et al., 2011), the opposite may be the case for organic carbon for which concentrations can be much higher in sea ice than seawater (e.g., Thomas et al., 2010), and pCO₂ can be very high in sea-ice brines, particularly under very cold conditions (e.g., Loose et al., 2011).

Direct flux measurements (Else et al., 2012) have confirmed the hypothesis developed from water-column data (Anderson et al., 2004; Omar et al., 2005; Loose and Schlosser, 2011) that in the presence of an incomplete or broken ice cover, CO_2 fluxes can be much higher than in open, ice-free waters. Two possible mechanisms may be responsible: boundary-layer turbulence enhancement in both the air and water caused by the presence of sea ice and its formation (e.g., McPhee and Stanton, 1996), and small-scale disruption of the sea-surface microlayer by

frazil ice during ice formation (as suggested by Anderson et al., 2004). High CO₂ fluxes in the presence of a broken and mobile ice cover have profound implications for the net annual air-sea CO₂ budget in an Arctic Ocean covered by first-year, instead of multi-year, sea ice. Winter data are still sparse, but if Arctic surface waters remain undersaturated in CO, throughout most of the winter (Miller et al., 2011; Else et al., 2012), high air-sea fluxes in the presence of a broken ice cover would increase the net Arctic CO₂ sink substantially. On the other hand, in areas where surface waters become supersaturated during winter (e.g., Kelley, 1968; Miller et al., 2002), new understanding of the potential for winter gas exchange indicates that outgassing could completely cancel out all summer CO₂ drawdown in those regions. Steiner et al. (2013) tested the implications of enhanced flux in ice-covered areas in an earth system model application. They found that enhancing the flux in ice-covered areas of the Arctic alters the seasonal cycle by extending the maximum CO₂ uptake in autumn and reducing uptake in summer. The reduced uptake in summer is likely to be due to earlier ocean pCO, equilibration with enhanced flux during autumn and winter, but also hints at a limited uptake capacity of Arctic surface waters as discussed by Cai et al. (2010). Results by Steiner et al. (2013) also show the relative change due to the enhanced flux parameterization in the annual mean to be small if integrated over the whole Arctic region (below 2-3% for the area north of 68° N) but to be significantly higher for the predominantly ice-covered central Arctic (up to 21% north of 80° N). With respect to acidification, they confirmed accelerated ocean acidification with enhanced fluxes. They found that enhancing the CO₂ flux by about 20% increases the surface pCO_2 and decreases the aragonite saturation by over 50% relative to what would be expected if sea ice did not inhibit the flux at all (a 10% enhancement allows for more than a third of the pCO_2 increase / aragonite saturation decrease). The authors concluded that the enhanced uptake could account for a one to two decade difference in the projection of when aragonite undersaturation is reached in certain areas.

A 'sea ice pump' may also help export CO₂ to depth (Loose et al., 2011; Rysgaard et al., 2011). When sea ice forms, the salts in the seawater, including carbonates, are not included in the ice lattice but are concentrated in brines. When the brines drain into the underlying seawater, they carry substantial quantities of inorganic carbon. Data indicating that sea-ice brine drainage can indeed inject dissolved inorganic carbon into the surface waters below the ice have been reported by Miller et al. (2011) and Chierici et al. (2011). Because brines rejected from sea ice are important contributors to deep-water formation, the C_{T} in ice brines may be a significant component of the solubility pump (see Section 2.3.3). In addition, CaCO₃ can precipitate from the concentrated sea-ice brines within the ice as it freezes, and while this only consumes a small fraction of the C_{T} in the brines, it is responsible for a proportionally greater decrease in A_T (when 1 unit of CaCO₃ precipitates, C_T in the remaining solution is decreased by 1 unit, while A_{T} is decreased by 2 units). If the solid CaCO₃ remains behind in the ice when the brine drains away, enriching the ice in alkalinity over C_T, and if those CaCO₃ salts dissolve when the ice melts in the following spring and summer, the resulting excess A_{T} in the surface waters would encourage atmospheric CO₂ absorption, which could then feed

C_T Pump Scenario



Brine detrains in ML

Figure 2.3. Potential mechanism by which sea ice growth and retreat may enhance atmospheric CO_2 drawdown. Source: modified after Loose et al. (2011).

Schematic of the total inorganic carbon (C_{T}) pump. In the upper panel, the scenario depicts abyssal water formation in polynyas and the offshore transport of sea ice so that freeze and melt occur in different regions of the ocean. Under such a scenario, if carbonate alkalinity (A_T) remains trapped in the ice as salt crystals and dissolved inorganic carbon (C_T) drains with the brine, then a 'C_T pump' might result in a net transport of inorganic carbon to the deep ocean. The lower panel depicts a scenario where ice forms and melts over a water column with a homogeneous mixed layer (ML). In this case, brine from sea ice is likely to detrain in the mixed layer and subsequently recombine with carbonate alkalinity in the ice. This scenario depicts a zero-sum change in the mixed layer inorganic carbon budget.

further carbon export with sinking brines when the surface waters refreeze the following winter (Figure 2.3). However, this proposed pumping mechanism is still speculative, and the few data available on the $A_T:C_T$ ratio in sea ice are variable and ambiguous.

Estimates of the impact of these various sea-ice processes on the net atmospheric CO_2 drawdown for the Arctic range from 3 to 36 TgC/y (Omar et al., 2005; Nedashkovsky and Shvetsova, 2010; Rysgaard et al., 2011). Bates et al. (2009) also estimated that if summer sea-ice loss increases the exposure of cold, CO_2 -undersaturated surface water to the atmosphere, the meltseason CO_2 drawdown could increase by 2 TgC/y, although such an increase in the summer drawdown would also require a substantial increase in wind-mixing to break down meltinduced stratification. Nevertheless, in terms of the physical processes through which sea-ice directly impacts air-sea CO_2 exchange over the annual cycle, the increase in seasonal sea-ice extent at the expense of the multi-year pack currently taking place in the Arctic (AMAP, 2012) is expected to increase the seaward flux of CO_2 over the coming decades.

2.3.2 Freshwater

In the Arctic Ocean, the input of freshwater is one of the most important factors controlling biological and geochemical processes. In particular, the Arctic Ocean receives ~11% of the global river discharge, but contains only 1% of the global ocean volume, and 2.8% of the area. The resultant widespread stratification of the top few hundred meters of the water column due to this freshwater input strongly affects other processes, such as the penetration of atmospheric CO_2 to the depths, and the return to the surface of dissolved inorganic carbon (C_T) and nutrient-rich water from below the surface mixed layer. Approximately 3300 km³/y of freshwater enters the Arctic Ocean directly via rivers (Serreze et al., 2006), four of which (Yenisey, Lena, Ob, Mackenzie) are among world's largest. This inflow, which exhibits a strong Arctic-wide freshet in May-June, is dominated by snow melt, although significant flow is maintained throughout the year in the larger rivers. Runoff also enters the Arctic Ocean indirectly, carried by the Pacific Inflow through Bering Strait (2500 km³y) and by the Atlantic Inflow in the Norwegian Coastal Current and Fram Strait (750 km³y) (Serreze et al., 2006). Runoff, which is distributed unevenly over shelves and basins because it enters from the sides of the ocean, is augmented by a more evenly distributed net input of ~1300 km³/y directly onto the ocean through precipitation (Serreze et al., 2006). The addition of all this freshwater has led to the view that the Arctic Ocean should be considered somewhat like a grand estuary (McClelland et al., 2012).

mix in ML

A residence time of about ten years for surface water in the Arctic Ocean means that there is a considerable storage of the runoff after it enters the ocean (Aagaard and Carmack, 1989), which magnifies the effect on stratification. Climate change and variability are known to affect the runoff component of the freshwater balance in at least two ways. First, there has been an increase in runoff (1964-2000) of ~120 km³/y, unevenly distributed among rivers (McClelland et al., 2006); part of this increase is likely to have been contributed by permanent snows and glaciers, now in retreat due to warming (Olsen et al., 2011; Sharp et al., 2011). One may anticipate that, likewise, with the projected intensification of the hydrological cycle, net direct precipitation onto the ocean is also increasing (e.g., see White et al., 2007). Second, the storage of freshwater can be strongly affected by decadal variation in circulation, for example forced by the Arctic Oscillation (McPhee et al., 2009; Giles et al., 2013; McClelland et al., 2012).

The formation and melting of sea ice provides another, very different, source of freshwater to the Arctic Ocean. In winter,

~2 m of sea ice forms from seawater, rejecting most of the brine to the surface ocean (first-year sea ice has a salinity of ~5–8 compared to the surface ocean at ~33–35). This rejected brine facilitates water-column mixing, which leads to the formation of the polar-mixed layer (PML: 40–50 m), and therefore also helps to redistribute freshwater from runoff into this same layer. During spring and summer, the process reverses; the sea ice melts and 1–2 m of brackish water is added to the ocean surface. The sea-ice component of freshwater is also subject to climate change. The recent replacement of large regions of thick, multi-year ice by first-year ice means that these regions now completely melt their ice cover by the end of summer (e.g., Stroeve et al., 2012a). Furthermore, the saltiness of first-year ice produces an ice phase that is physically and chemically very different from multi-year ice (which has lost almost all of its salt), or freshwater ice.

Aside from controlling the physics and chemistry of the upper ocean, the addition of freshwater from runoff, precipitation and sea-ice melt have a direct effect on the sensitivity of the surface water to added CO_2 . Specifically, these waters are more poorly buffered and contain a lower [Ca^{2+}], with the result that the mixed water exhibits a greater decline in pH with added CO_2 (Table 2.1), and crosses mineral solubility thresholds sooner. Indeed, it appears that brackish surface water impacted by both runoff and seaice melt is already sufficiently corrosive to dissolve aragonite (Bates et al., 2009; Yamamoto-Kawai et al., 2011). In summary, all freshwater additions (precipitation, sea-ice melt, runoff) lead to greater decreases in pH and Ω at the surface of the ocean to an extent depending on the source of freshwater (Table 2.1) and the mixing ratio with seawater (Figure 2.4; Azetsu-Scott et al., 2010).

2.3.3 Solubility pump

Atmospheric gases, including CO₂, all dissolve in seawater. The same process occurs in the Labrador Sea, where deep convections produce the Labrador Sea Water with high anthropogenic CO₂ content. Their solubilities are different but all increase as water temperature decreases. The air-sea gas flux is governed by Henry's Law by which equilibrium is reached when the partial pressure of the gas is the same in both the air and surface seawater. This is the foundation of the solubility pump. Envisage the North Atlantic with 25 °C Gulf Stream water and the North Atlantic Drift flowing toward high latitudes. In this progress the seawater gives off heat to the atmosphere and, as the temperature decreases, the CO₂ solubility of the surface water increases by as much as 100% (Figure 2.5). The surface seawater becomes undersaturated and by Henry's Law the CO₂ gas transfers from the atmosphere into the sea. As the North Atlantic Drift reaches the Nordic Seas, the seawater has dropped in temperature by 20 °C but still retains a relatively high salinity. In winter, further cooling leads to a density increase with consequent deep vertical convection and formation of Arctic Intermediate- and Arctic Bottom-Water. This process is termed Deep Water Formation. The dense water masses fill the deeps of the Nordic Seas north of the Greenland-Iceland-Faroe-Scotland Ridge and only escape by flowing southward over the submarine ridges in the Denmark Strait, the Faroe Channel and between Iceland and the Faroe Islands.

These deep currents initiate the Thermohaline Circulation which extends through the depths of the Atlantic, Indian and Pacific



Figure 2.4. Effect of dilution by different freshwater sources on the saturation state for aragonite (Ω_{Ar}). Solid symbols are results that include a [Ca²⁺] contribution from the rivers assuming a conservative mixing between freshwater and seawater end members. Open symbols are results using [Ca²⁺] calculated from the relationship [Ca²⁺] = 10.28 × (salinity/35). Source: after Azetsu-Scott et al. (2010).





Figure 2.5. The solubility of carbon dioxide in seawater increases as the seawater cools. Shown for a salinity of 35.

Oceans. The solubility pump in this way transfers CO_2 from the atmosphere to the ocean depths to be stored there for centuries. The solubility pump is most intense in the sub-Arctic region of the North Atlantic (Takahashi et al., 2002) where the ocean heat loss to the atmosphere is high, but it also operates to a smaller extent within the Arctic basin, where dense waters formed on the shelves can sink into the halocline and even deeper, isolating CO_2 from the atmosphere for up to hundreds of years (e.g., Jones et al., 1995; Anderson et al., 2010). The solubility pump is sensitive to climatic variations: it is intimately linked to deepwater formation, which is sensitive to freshening, and will become less effective with increasing sea-water temperature.

The solubility pump, and its transfer of anthropogenic carbon into deep waters, is one of the primary drivers of increasing acidification in the Arctic Ocean.



Figure 2.6. A schematic illustration of the biological pump in the Arctic Ocean. Number units are gC/m²/y, pro-rated to the area of the receiving environment (total shelf or total basin). For productivity, the rates are given as total production (export production). The flux estimates are derived from: primary production and shelf export to halocline (Sakshaug, 2004), vertical flux (Honjo et al., 2010; O'Brien et al., 2013), and sediment burial rates. Source: Stein and Macdonald, (2004b). Abbreviations: POC, particulate organic carbon; POM, particulate organic matter; DOC, dissolved organic carbon; A-MR, Alpha-Mendeleev Ridge; MB, Makarov Basin; LR, Lomonosov Ridge; AB, Amundsen Basin; GR, Gakkel Ridge; NB, Nansen Basin.

2.3.4 Biological pump

The term biological carbon pump describes the suite of biologically driven processes that move carbon from surface waters to depth in the ocean (de la Rocha, 2006). C_{T} is removed near the surface by conversion to organic matter (OM) during primary production (Figure 2.6). A proportion of that carbon is sequestered at depth in the ocean by the sinking of particulate organic matter (POM) due to gravitational settling, by vertical migration of zooplankton (copepods), or by the advection or mixing of dissolved organic matter (DOM) to depth. The organic particles include phytoplankton, aggregates comprising organic and inorganic detritus, fecal pellets produced by zooplankton, and carcasses of zooplankton or, more rarely, larger animals up to and including whales (e.g., Stockton and DeLaca, 1982; Turner, 2002; Honjo et al., 2008). Because much of this POM is remineralized while sinking (Bishop, 1989), the net effect of the biological pump is to decrease C_{T} (increase pH) in the upper ocean and increase C_{r} (lower pH) at depth in the ocean. The biological pump acts on many other elements involved in biological cycles including silicon, phosphorus, nitrogen, metals and metaloids, but from the perspective of ocean acidification the most important component other than POM/DOM is CaCO₃, which is used by a number of planktonic species (e.g., coccolithophores, pteropods) to form structures. This is discussed in Section 2.3.5.

In the textbook view of ocean productivity, the large areas of the interior ocean basins support more total primary production than the smaller total shelf area, despite higher per-area production rates on the shelves. However, this description does not properly represent the Arctic Ocean. Enormous shelves, high stratification, and sea-ice cover in the Arctic Ocean combine to create exceptional circumstances where primary production over shelves plays a much larger role than in the global ocean (Figure 2.6). In contrast to the shelves, the oligotrophic interior parts of the Arctic Ocean produce little sinking particulate matter, limiting the efficiency of the biological pump in the central basins (Honjo et al., 2010).

The Arctic Ocean exhibits large spatial heterogeneity: shelf regions supplied externally by nutrients from the Pacific (Chukchi Sea; Lepore et al., 2007) or the Atlantic (Barents Sea; Reigstad et al., 2002) and persistent regions of open water like the Cape Bathurst Polynya (Arrigo and van Dijken, 2004) and the Northwater Polynya (Tremblay et al., 2006) are among the most productive locations in the world ocean, whereas the interior of the Canada Basin has an almost non-existent export production (Hargrave et al., 1994; Honjo et al., 2010; O'Brien et al., 2013). Because of this disparity, the Arctic's biological pump can only be understood by considering the shelf-to-basin OM export. That is, the Arctic Ocean has a strong continental shelf pump (Liu et al., 2000; Yool and Fasham, 2001; Anderson et al., 2010), partly due to the contrast in productivity between shelf and interior, and partly due to dense water production over the shelves during winter. This dense water can accumulate OM and metabolic products, including C_T, as the water flows along the shelf bottom thence to be exported to the interior ocean. Aagaard et al. (1981) estimated that as much as 79×10^3 km³/y of brine-enriched water was supplied to the halocline from shelves by this process alone. In the western Arctic Ocean, the upper halocline is a dominant feature that has been enriched with nutrients and C_T and impoverished in dissolved oxygen

during its formation in the Chukchi and East Siberian Seas (e.g., Bates et al., 2005; Mathis et al., 2007; Nishino et al., 2009; Anderson et al., 2010; Griffith et al., 2012). The sequence of high primary production, sinking of POM, partial regeneration to C_T or conversion to DOM near the bottom and then transport of labile OM and regeneration products to the interior ocean produces a widespread nutrient maximum / oxygen minimum within the Arctic Ocean halocline (Figure 2.6). The Barents Sea, likewise, supports the C_T and nutrient enrichment in the Arctic Ocean halocline, albeit to a lesser extent (e.g., see Jones and Anderson, 1986).

A second set of processes, whose importance has only recently been recognized, occurs over the outer shelf in the Arctic. Boundary currents, turbulence and eddies provide the circumstances to re-suspend surface sediment containing OM that has settled there temporarily (e.g., algal mats and other organic detritus from annual production), and then transport this material and accompanying dissolved components along isopycnals into the interior ocean (Schauer et al., 2002; O'Brien et al., 2006, 2011, 2012; Forest et al., 2007; Mathis et al., 2007; Kadko and Aagaard, 2009). During transport to the interior, the POM can sink and become part of the vertical flux, or become metabolized to CO₂ or converted to DOM, both of which remain in the water column (Figure 2.6). Thus, for the Arctic Ocean much of the biological pump operates from the side of the basin, driven by shelf productivity, rather than vertical export of POM produced in the interior. According to this scheme (Figure 2.6) shelf export production, estimated to average ~ 8 g C/m²/y (Sakshaug, 2004), is delivered to shelf bottom waters with as much as 35% of the export then distributed out into the haloclines or deeper into the Arctic Ocean leading to net CO, drawdown in shelf surface water and net increase in CO₂ in shelf bottom water and haloclines.

How might climate change affect the biological pump in the Arctic Ocean? It seems clear that the major factor driving change will be the continued conversion of multi-year sea ice to first-year ice until the Arctic Ocean becomes seasonally clear of sea ice, perhaps within two decades (Schiermeier, 2012). This, by itself, will lead to greater light penetration into the water and a significant warming of surface waters above freezing. Open water will also promote upwelling at the shelf edge, greater mixing by winds, and greater light penetration into the water column, all of which would enhance primary production and lead to an increase in the strength of the biological pump. However, upwelling of nutrients is also accompanied by $\mathrm{C}_{_{\mathrm{T}}}$ thereby limiting carbon sequestration. On the other hand, there is likely also to be an increase in river inflow and greater amounts of sea-ice melt in summer, which would lead to greater stratification, which would generally inhibit nutrient supply to the photic zone. How these various factors might play out in the future is very uncertain, but the response to change is likely to vary widely between shelves, polynyas, and the basins, considering how differently each of these environments function (Carmack et al., 2006; Nishino et al., 2011). Using a sea ice-ocean-biologically coupled 1-D model, Lavoie et al. (2010) simulated the effect of climate change on primary production at the shelf edge of the Beaufort Sea. Their model projected that by the end of the 21st Century, ice algal production would have declined in relative importance due to the rapid loss of ice early in the year, but export production would have increased by ~16%. Slagstad et al. (2011), using a 3-D coupled physical-biological model, found that gross primary production over the Arctic Basin and Eurasian shelves would increase substantially with an 8 °C temperature increase (10 to 42 g C/m²/y and 29 to 61 g C/m²/y, respectively), but hardly change in the Barents Sea. These disparities suggest that the strength of the biological pump might increase on average, but that the scenario of a relatively strong biological pump over the shelves, a weak pump in the ocean interior, and a large shelf-basin export of particulate organic carbon (POC), DOC and C_T is likely to remain a dominant feature in the future Arctic Ocean.

2.3.5 Carbonate pump

Another biologically mediated carbon pump is associated with the precipitation of CaCO₃ from seawater by marine organisms to form, mainly, exoskeletons and shells. This *carbonate pump* is dominated by surface calcification in marine plankton and the subsequent transfer of carbon in the form of CaCO₃ from the sun-lit surface into the interior and deep-ocean (Volk and Hoffert, 1985). However, benthic and mesopelagic CaCO₃ production also play a significant role.

The carbonate pump has what may be considered a counterintuitive effect on the marine carbonate system (see also Section 2.2). As CaCO₃ production partitions carbon from seawater through the reversal of Equation 8 it has the net effect of removing one CO_3^{2-} ion and one Ca^{2+} ion thus reducing A_{T} and C_{T} in the proportion 2:1 and increasing pCO₂ and reducing pH (Equation 1). The stability of the CaCO, will depend on the organic matrix of the skeleton or shell and how effective the organism is at protecting the shell or skeleton. However, in geochemical terms, exposed or unprotected CaCO₃ will become thermodynamically unstable if Ω goes below 1 (Equation 9) and will dissolve. Each specific mineral has a different solubility product; for example, of the two most common biogenic CaCO₃ minerals in the oceans, aragonite has a higher solubility product than calcite (Mucci, 1983), and thus aragonite dissolves more readily than calcite. However, calcite minerals in which some Ca²⁺ ions are replaced with Mg²⁺, called hereafter Mg-calcite, are even more soluble than aragonite.

As the organisms die they often sink through the water column into waters that have a lower Ω following respiration of OM (the biological pump, see Section 2.3.4). There are three depths that are commonly used for understanding the distribution of CaCO₃ in the water column and on the ocean sediment surface. The first is the saturation horizon (ASH or CSH for aragonite and calcite, respectively) where Ω is unity. As sedimenting calcareous material does not dissolve immediately on passing the saturation horizon, CaCO₃ often continues to a depth where a strong gradient in CaCO₃ is seen on the ocean floor, termed the lysocline. This depth is rather ambiguous and is termed the lysocline. Below here some CaCO₃ will continue its downward journey until a depth where all is dissolved. This demarcation is termed the compensation depth. Dissolution results in an increase in C_T and A_T of the subsurface waters in



Figure 2.7. Percentage of calcium carbonate in surface sediments in the Arctic Ocean. Core data were extracted from the Pangaea database (www.pangaea.de) and peer-reviewed publications. Left panel shows sampling locations.

tandem with respiration in the biological pump. Dissolution, however, counteracts the pH reduction and pCO_2 increases due to the biological pump.

In most of the surface Arctic Ocean, $CaCO_3$ is stable as Ω is greater than 1 for both aragonite and calcite. However, with lower initial surface water Ω and strong freshwater inputs (thus lower Ca²⁺ concentrations), CaCO₃ minerals in the Arctic Ocean are particularly susceptible to dissolution following only small additions of CO₂.

The present-day distribution of CaCO₃ forming planktonic organisms in the Arctic Ocean can be partially inferred from CaCO₃ core top data (Figure 2.7). Highest percentages of CaCO₃ in the sediments are found in the Nordic Seas and this has been linked to the inflow of warm Atlantic Waters (Taylor et al., 2002; Giraudeau et al., 2010). The typical planktonic CaCO₃ forming organisms found in the ice-free regions of the Arctic is the coccolithophorid Emiliania huxleyi which synthesizes external CaCO₃ platelets known as coccoliths. In the latter stages of a bloom these coccoliths are released in large numbers giving the water a turquoise-white appearance which is easily detectable both at the surface and from space. Satellite data are now routinely used to map blooms of these organisms using the distinctive brightness and reflectance of the coccoliths (Groom and Holligan, 1987; Moore et al., 2012). Using this approach the distribution of these organisms in the global ocean has been assessed (Brown and Yoder, 1994; Moore et al., 2012). A number of studies have shown large annual blooms of coccolithophorids in the Arctic, most noticeably along the Arctic Front between the Greenland and Iceland seas (Trees et al., 1992), in the vicinity of Iceland (Holligan et al.,

1993), along the Norwegian coast (Findlay et al., 2008; Moore et al., 2012) and in the Barents (Smyth et al., 2004) and Bering Seas (Merico et al., 2003). A comparison between the satellite data showing the occurrence of coccolithophorid blooms and the percentage of $CaCO_3$ in the sediments is generally best in regions with deeper waters, as in shallower waters the lower carbonate saturation state (see Section 2.2) will lead to greater dissolution of the coccoliths (e.g., Barents Sea). Recent analysis of longer term satellite data suggests that the increased occurrence of coccolithophorid blooms in the Barents Sea (Smyth et al., 2004) may be related to climate change, similarly satellite investigations in the Bering Sea since the 1970s suggest an ecosystem shift toward coccolithophorids driven by climate change (Harada et al., 2012).

2.3.6 Air-sea exchange

The net result of the combined impacts of freshwater and sea-ice distributions in the Arctic Ocean on CO_2 solubility and primary production is that most surface waters within the Arctic Ocean are undersaturated in CO_2 relative to atmospheric values (e.g., Jutterström and Anderson, 2010). To date, in autumnperiod studies, sea ice has formed before air-sea exchange has balanced the partial pressures (Miller et al., 2002; Mucci et al., 2010; Else et al., 2012). Consequently, the gas exchange rate largely determines the net air-sea flux, and that rate is heavily dependent on sea-ice conditions, as well as winds. The air-sea flux has been estimated for a number of Arctic Ocean areas (Table 2.2). These estimates are both uncertain and variable and consequently, must be confirmed carefully in this time of changing sea-ice coverage.

Region	Sea-air exc	Source	
_	(mmol/m²/d)ª	10 ¹² mol/y	_
Central Basins	-0.3 to 1		Bates et al., 2006
Barents Sea	-3.7±0.9 ^b	$\begin{array}{r} -0.8 \\ -2.4 \pm 1 \\ -3.8 \pm 1.5 \\ -4.3 \pm 0.7 \\ -1.8 \pm 1 \end{array}$	Fransson et al., 2001 Kaltin et al., 2002 Nakaoka et al., 2006 Omar et al., 2007 Kivimäe et al., 2010
Nordic Seas		-5.8	Jeansson et al., 2011
Kara Sea	-0.1 ^b		Fransson et al., 2001
Laptev Sea	-1.2 to 1.7	-2.1	Nitishinsky et al., 2007 Semiletov et al., 2007
East Siberian Sea	-2 to 10	0.04 to 0.28	Pipko et al., 2011
Chukchi Sea	~ -30 to -90 -15±13 -18 to 3	-2.6 to -3.8 -1.8	Bates, 2006 Pipko et al., 2002 Murata and Takizawa, 2003 Kaltin and Anderson, 2005
Beaufort Sea	-2 -17 to 0 -17 to 3 -25 to -9		Mucci et al., 2010 Murata and Takizawa, 2003 Shadwick et al., 2011 Else et al., 2011
Canadian Archipelago	-7 to 0 -140 to 18		Miller et al., 2002 Fransson et al., 2009
All Arctic Ocean		-2 to ~ -8	McGuire et al., 2009

Table 2.2. Sea-air exchange in different regions of the Arctic Ocean as reported in the literature. Negative fluxes are from the atmosphere into the ocean.

^a Data are from the summer period with limited ice coverage; ^b annual mean estimate.

2.3.7 Geological methane and carbon dioxide seeps

Carbon is added to Arctic waters from the sediment both as C_T and methane, which can be oxidized to CO₂, thereby contributing to ocean acidification (Elliott et al., 2010). Sedimentary processes in the Arctic Ocean have much in common with those in other oceanic regions, with the main difference being the low seawater temperature. Furthermore there are large areas on the Siberian shelves with subsea permafrost conditions that contain large deposits of methane hydrates. Since the Holocene transgression, these deposits have been slowly seeping out of the sediment as a result of increasing temperatures. Seeps of methane from hydrate have been observed as gas plumes near West Spitsbergen (Westbrook et al., 2009), on the East Siberian Shelf (Shakhova et al., 2010) and in the Beaufort Sea (Paull et al., 2007). The magnitudes as well as the temporal and spatial variability of the methane fluxes are not well known (Shakhova et al., 2010).

However since the methane concentration within the water column in these seep areas typically reaches a few hundred nM and the oxidation rate has been reported to be very low in cold oxic waters (e.g., Tsurushima et al., 1996), the total CO_2 addition from methane oxidation must be substantially lower than that from decaying organic matter. Although the reported water column methane concentrations in these studies have typically been low, it is possible that destabilization of shelf methane hydrate due to rising seawater temperatures may increase methane release to the water with consequences for ocean pH (Biastoch et al., 2011; AMAP, 2012).

2.3.8 Other 'acidifying' compounds

Sulfide oxidation by heterotrophic bacteria (HS⁻ + 2O₂ \rightarrow SO₄²⁻ + H⁺) and by autotrophic bacteria (2H₂S + CO₂ \rightarrow CH₂O + 2S + H₂O), and also sulfate reduction (SO₄²⁻ + 2CH₂O + H⁺ \rightarrow 2CO₂ + HS⁻ + 2H₂O) can change carbon chemistry, including pH. However, no study has yet been done to quantify the effect of these processes on ocean acidification in the Arctic.

Other anthropogenic gases such as sulfur oxides (SO_x), nitrogen oxides (NO_x) and ammonia (NH₃) have the potential to alter ocean pH and carbon chemistry. Doney et al. (2007) assessed the impact of anthropogenic nitrogen and sulfur deposition on ocean acidification using atmospheric and ocean models. Their study found that the global effect of these species on surface water chemistry is only a few percent, at most, of that caused by anthropogenic CO2. They also concluded that these substances may cause substantial impacts in coastal waters. Hunter et al. (2011) investigated impacts of SO_x, NO_x and NH₃ on ocean acidification in coastal waters and shipping lanes. Their analysis showed that the pH reduction caused by these pollutants is almost completely cancelled out by buffering reactions, but at the expense of a significant decrease in the uptake of atmospheric CO₂ in the same region. More shipping traffic is expected in the Arctic coastal area due to the possibility of an ice-free Northwest Passage and Northern Sea Route and increased resource exploration (AMAP, 2012), so the input of these anthropogenic gases needs to be evaluated.

2.3.9 Terrestrial organic carbon

The Arctic Ocean receives exceptional quantities of organic carbon from land via rivers or coastal erosion (McGuire 18

et al., 2009). Stein and Macdonald (2004a) estimated that 11 Mt/y of POC was supplied to the Arctic's shelves in this manner, of which ~ 4 Mt/y was metabolized, forming CO₂. Decay of terrestrial OM, most likely occurring at the sediment surface, has been shown to greatly decrease pH, producing large regions of both aragonite and calcite undersaturation (Anderson et al., 2011b). Bottom waters of the shelf seas are clearly most vulnerable to this process, especially for the large Siberian seas (e.g., Guo et al., 2004; Vonk et al., 2010; Alling et al., 2012). In this latter context, recent measurements for the East Siberian Sea by Vonk et al. (2012) suggest that the release of old particulate carbon by coastal erosion and subsea permafrost may have been grossly underestimated, or has recently been undergoing a large increase. In that paper, the authors suggested that 44 ± 10 Mt/y of terrigenous POC is presently released, ~66% of which is metabolized. Clearly, warming and thawing of permafrost leading to accelerated release of stored POC (e.g., deConto et al., 2012), could be of immense importance to future acidification of shelf bottom waters in the Arctic.

2.4 Ocean processes sensitive to acidification

2.4.1 Macronutrients

2.4.1.1 Nitrogen species

Ocean acidification should not alter the speciation of the majority of inorganic nitrogen species (N₂, NO₃⁻ and NO₂⁻). The exception is NH₃ which is a weak base (pK_a ~ 9.2) in equilibrium with its conjugate acid ammonium (NH₄⁺) (Clegg and Whitfield, 1995). Ocean acidification will push the equilibrium toward an increase in the relative abundance of NH₄⁺ and lead to a possible reduction in the air-sea gas exchange of NH₃.

Nitrification

Recent work from the tropics indicates that a decrease in the pH of seawater leads to a reduction in oceanic nitrification rates (Beman et al., 2011). This arises as both ammoniaoxidizing bacteria and ammonia-oxidizing archaea utilize NH, as a substrate (Martens-Habbena et al., 2009). The abundance of ammonia-oxidizing bacteria and ammonia-oxidizing archaea in the Beaufort and Chukchi Seas has recently been reported over a seasonal cycle and is strongly related to NH₄⁺ concentrations (Christman et al., 2011). Decreasing surface water pH in the Arctic could lead to slower nitrification rates in the water column resulting in increased NH₄⁺ concentrations and decreased nitrate (NO3-) levels which may cause shifts in the phytoplankton community toward smaller organisms that are more competitive for NH⁺. Organisms that are specialists for NO₃⁻ uptake, such as large diatoms, may be disadvantaged under these conditions, leading to potential changes in the oceanic food web (Beman et al., 2011).

Some archaea in the water column in the Arctic also possess a urease gene providing them with the potential use of urea to fuel nitrification (Alonso-Saez et al., 2012). Earlier studies have shown that urea is frequently found throughout the water column in

parts of the Arctic at concentrations above that of NH_3 (Harrison et al., 1985). This suggests that the biological decomposition of urea may be a significant nitrification pathway in the Arctic.

Denitrification

Sedimentary denitrification occurs predominantly on continental margins (Christensen et al., 1987) and thus the Arctic may be an important contributor to global marine sedimentary denitrification. Sedimentary denitrification rates in the Arctic generally decrease with increasing water depth and correspond to differences in the overlying primary productivity and organic matter flux (Koike and Hattori, 1979; Devol et al., 1997; Chang and Devol, 2009). Denitrification has also been found to occur in Arctic sea ice (Rysgaard et al., 2008), yet water column denitrification overall appears to be a minor process compared to sedimentary denitrification (Kaltin and Anderson, 2005; Lehmann et al., 2005). Denitrification, but modeling studies predict increases in denitrification (Schmittner et al., 2008) due to global increases in the volume of suboxic waters.

Nitrogen fixation

Nitrogen-fixing cyanobacteria from the tropical ocean have been found to increase their fixation rates under high CO_2 conditions (Hutchins et al., 2007). Nitrogen fixation is currently not thought to be an important part of the nitrogen cycle in the Arctic although the distribution of diazotrophs is more widespread in the Arctic than previously thought (Blais et al., 2012).

Апаттох

Anammox has recently been detected in the melt waters of Arctic sea ice (Rysgaard et al., 2008) and also appears to be a small contributor to nitrogen cycling in the sediments of an Arctic fjord (Gihring et al., 2010). As the anammox process, which converts NH_4^+ and NO_2^- to N_2 and H_2O , may also be pH dependent, as organisms utilize NH_3 as the substrate for hydrazine hydrolase (HH), the first enzyme involved in anammox (Kuenen, 2008; Karlsson et al., 2009). The final enzymatic step in anammox, converting hydroxylamine to N_2 via hydroxylamine oxidoreductase (HAO), has been reported to be strongly pH dependent (Zou et al., 2011). Currently there are, however, no data on the overall pH sensitivity of anammox in the marine environment, or on whether under climate change anammox will become more or less important in the Arctic.

2.4.1.2 Phosphate speciation

There will be minimal changes in the distribution of phosphate species with ocean acidification as the HPO₄²⁻ species strongly predominates throughout the pH range 7.5–8.1. There are no reports of ocean acidification affecting the uptake of inorganic phosphate by organisms. The hydrolysis of dissolved organic phosphorus (DOP) compounds is, however, pH dependent (Price and Morel, 1990), as are many other enzymatic processes (Yamada and Suzumura, 2010). Indeed the activity of the zinccentered enzyme alkaline phosphatase decreases with decreasing pH (Kuenzler and Perras, 1965). While DOP can exist in the Arctic in significant concentrations (Simpson et al., 2008) there is currently no evidence to suggest that ocean acidification will

significantly influence the uptake of either dissolved inorganic phosphorus (DIP) or DOP by phytoplankton or bacteria.

The Arctic has been identified as an important mediator between regions of denitrification (the Bering and Chukchi Seas and the North Pacific) and of nitrogen fixation (the Atlantic) through excess phosphate, relative to nitrogen, transported east through the Arctic Ocean (Yamamoto-Kawai et al., 2006). Thus ocean acidification impacts on denitrification (see Section 2.4.1.1) may impact the excess P:N in the Arctic throughflow, with implications for N, fixation in the North Atlantic.

2.4.1.3 Silicate species

Under ocean acidification the speciation of silicate will change only slightly as it is dominated by the Si(OH), species. As expected, studies investigating the uptake rate of silicate by diatoms find that it is not influenced by pH (Milligan et al., 2004). This has also been seen in identical silicate utilization rates in mesocosms under different pCO₂ regimes (Bellerby et al., 2008). However lower Si:C ratios have been found in culture experiments at low pH due to higher Si efflux from the cell (Milligan et al., 2004). Ocean acidification has also been found to enhance the silica dissolution rate of empty diatom frustules (Milligan et al., 2004). This may have an influence on the ballasting of particulate material, leading to faster rates of remineralization and the nutricline shifting toward the surface. This could lead to a decrease in biological pump efficiency, and a reduction in material reaching benthic communities, as has been suggested for the CaCO₂ component of particulate matter under ocean acidification conditions (de Jesus Mendes and Thomsen, 2012). However floating sediment trap studies in the Arctic (Honjo et al., 2010) indicate that there is already an absence of ballast particles in this environment for sinking fine, lightweight POC to depth (see also Section 2.3.4), so the existing condition may be reinforced under ocean acidification with implications for the cycling of some trace metals and organic pollutants.

2.4.2 Trace metals

2.4.2.1 Inorganic speciation and reactivity

On first principles, ocean acidification will directly influence any chemical reaction that depends on either H^+ or OH^- . Thus the chemical speciation of elements that undergo hydrolysis will be altered by changes in pH. Similarly, enzymatic reactions that depend on substrates that are influenced by pH will also be affected.

There have been a number of studies examining the changes in trace metal distributions in the Arctic (see Appendix, Table A1), particularly in the estuarine mixing zone on the Arctic shelf (Martin et al., 1993; Garnier et al., 1996; Guieu et al., 1996; Hölemann et al., 2005). More recently, basin-scale elemental budgets have been constructed for manganese and aluminum in the Arctic Ocean through analysis of box core data (Macdonald and Gobeil, 2012). However, there is a distinct lack of trace metal speciation studies from the Arctic (see Appendix, Table A2), with most of the work in the Bering Sea. Substantial information is available on the distributions of radioactive isotopes in the Arctic (see Appendix, Table A3), The inorganic speciation of trace metals in seawater is reasonably well described due to the extensive work performed by physical chemists in simple salt solutions. In more complex media such as seawater, the use of Pitzer equations (Pitzer, 1973) is required, but for many species in seawater the relevant data are still missing. A number of detailed reviews have covered the inorganic speciation of trace metals in natural waters (Turner et al., 1981; Byrne et al., 1988), and recent papers by Byrne (2010) and Millero et al. (2009) have focused on those elements whose inorganic speciation is dominated by hydroxide and/ or carbonate species and are therefore particularly sensitive to decreases in pH and increasing CO₂ concentrations.

2.4.2.2 Organic speciation of trace metals

Many of the biologically relevant trace metals are predominantly complexed by organic ligands in seawater; most notably iron (Fe; Gledhill and van den Berg, 1994; Rue and Bruland, 1995), copper (Cu; Coale and Bruland, 1988; Moffett et al., 1990), cobalt (Co; Ellwood and van den Berg, 2001; Saito and Moffett, 2001), nickel (Ni; Nimmo et al., 1989; Saito et al., 2004), cadmium (Cd; Capodaglio et al., 1991; Bruland, 1992) and zinc (Zn; van den Berg, 1985; Bruland, 1989). However very little is known about these metal-organic complexes, although they do appear to be produced by organisms in response to limitation (Reid and Butler, 1991; Wilhelm and Trick, 1994) or metal-induced stress (Moffett and Brand, 1996; Croot et al., 2000). Only a few of these ligands have been isolated and their chemical structures determined (Reid and Butler, 1991; Ahner et al., 1994; Martinez et al., 2001; Vraspir and Butler, 2009).

The relationship between trace metal speciation and bioavailability is an ongoing and evolving field of research, but the impact of ocean acidification on the organic speciation, and hence bioavailability, of trace metals is not straightforward to assess. Little is known regarding the pH dependence of the conditional stability constants of these complexes in seawater. From first principles, the free metal concentration could be expected to increase due to the competition between H⁺ and metal for the same ligand (L) complexation sites (see Equation 11).

$$M^++HL \leftrightarrow ML+H^+$$
 Eqn. 11

However, in reality this depends on a number of factors; chiefly the acid dissociation constant for the ligand and the conditional stability constant for the metal-ligand complex (which includes other reactions between seawater components and the ligand).

A recent study by Shi et al. (2010) suggested that iron bioavailability decreases with decreasing pH for diatoms grown using media containing the artificial complexing agent EDTA. However these authors also noted that for the natural iron chelator azotochelin there was little change in inorganic iron as a function of pH within the range anticipated from ocean acidification. Other recent experimental data suggest that the redox potential of the iron complex controls iron uptake by phytoplankton (Maldonado and Price, 2001; Shaked et al., 2004, 2005; Maldonado et al., 2005; Morel et al., 2008). The redox potential for the iron organic complex is also related to its thermodynamic stability and may be subject to pH-induced shifts (Spasojevic et al., 1999; Crumbliss and Harrington, 2009; Harrington and Crumbliss, 2009). However, presently there is little information on the nature of the ligands present in seawater that are responsible for iron binding, and so it is difficult to *a priori* assess the impact of ocean acidification on organic iron complexation or bioavailability.

Similarly, little is known about the pH effects on organic complexation of other elements. One published study directly examined Cu complexation as a function of pH, using natural organic matter isolated from coastal seawater (Louis et al., 2009) and found that over the pH range 7–8 there was little change in the Cu speciation. Recent fieldwork by Xu et al. (2012) indicated that the uptake rates of Cd and Zn are reduced under ocean acidification conditions. These authors suggested that their observations were related to the presence of weak ligands that act to maintain some bioavailable metal at higher pH values.

As the Arctic Ocean is supplied with trace metals predominantly from riverine sources, it is also of interest to examine the influence of high CO_2 , low pH on the river systems. Studies in the Baltic examined the role that acidic leaching of trace metals from rivers plays in forming harmful algal blooms (Granéli and Haraldsson, 1993). Changes in the humic content of the rivers have also been suggested to play a role in shifting the species composition from diatoms to dinoflagellates (Granéli and Moreira, 1990).

Ocean acidification and increasing pCO_2 may also impact on the metal requirements of phytoplankton. Recent studies have shown high pCO_2 conditions influence cellular metal quotas; with data for Fe (Milligan et al., 2009), Cd (Cullen et al., 1999; Cullen and Sherrell, 2005) and Zn (Sunda and Huntsman, 2005).

2.4.2.3 Redox speciation – importance of kinetics

For many of the bio-essential trace metals there are major differences in the reactivity, bioavailability and toxicity between redox species. In many cases pH is a critical parameter in the overall kinetics for key redox processes: oxidation of Fe(II) by O₂ (Millero et al., 1987; King, 1998; Santana-Casiano et al., 2005) and H₂O₂ (Millero and Sotolongo, 1989; King and Farlow, 2000; Gonzalez-Davila et al., 2005), and oxidation of Cu(I) by O₂ (Moffett and Zika, 1983; Millero, 1989) and H₂O₂ (Sharma and Millero, 1989; González-Dávila et al., 2009). The overall impact of ocean acidification on the redox processes affecting Fe and Cu will be a shift toward the slower oxidation of the reduced species; Fe(II) and Cu(I). This may result in an increased bioavailability of both elements, particularly in the case of iron, as Fe(II) is significantly more soluble than the oxidized form, Fe(III). Results from a mesocosm experiment have confirmed that a lower pH can result in higher Fe(II) concentrations (Breitbarth et al., 2010).

An important further consideration is that reductions in oxygen concentrations will also be important in maintaining significant concentrations of thermodynamically unstable redox species in solution. The combined effect of low pH and low O_2 leads to higher fluxes of redox-sensitive metals (e.g., Fe, Mn and Co) and nutrients (e.g., phosphate released from metal oxides) from the sediment to the water column (Ardelan and Steinnes, 2010). Given that the Arctic Ocean contains an extensive area of continental shelf, the potential for increased fluxes from the sediment could be a positive feedback on increased primary productivity in surface waters and reduced O_2 in the bottom waters. This could have implications for regions like the Bering Sea Green Belt – where shelf edge processes are already important to primary productivity (Springer et al., 1996).

2.4.3 Climate-relevant gases

There are few data on the impacts of ocean acidification on the production of other climate relevant gases in general for marine systems and only a few studies of particular relevance to the Arctic.

2.4.3.1 Halocarbons (natural production)

Halocarbons (e.g., iodo-organics) are chemical compounds formed between carbon and halogen atoms (fluorine, chlorine, bromine or iodine). In the marine environment the low molecular weight compounds are typically gases and, along with sea spray, are an important source of halogens to the marine boundary layer of the atmosphere (Carpenter et al., 2012). Sea-air exchange of gaseous organic iodine is the major source of iodine to the atmosphere and is believed to be a major source of new particles in the atmosphere as the iodocarbons are rapidly oxidized to iodine oxide (IO) (O'Dowd et al., 2004; Seitz et al., 2009). Iodine- and bromine-containing halogen species are also believed to have a significant global effect on the concentration and lifetimes of other climatically active gases. This includes catalytic reactions with ozone and modifying the cycling of NO_x and hydrogen oxides (HO_x) in the atmosphere (Carpenter et al., 2012).

Two CO₂ treatments were used during a mesocosm CO₂ perturbation experiment in a Norwegian fjord (present day: 380 ppmv and year 2100: 750 ppmv), and phytoplankton blooms were stimulated by the addition of nutrients (Hopkins et al., 2010). Seawater trace gas concentrations were monitored over the growth and decline of the blooms, revealing that concentrations of methyl iodide were significantly reduced under high CO₂ conditions. Additionally, large reductions in concentrations of other iodocarbons were observed. The response of bromocarbons to high CO₂ was less clear cut. More recent mesocosm experiments in Svalbard performed over a wider range of pCO_2 concentrations showed that in general the response of halocarbons was either very subtle or undetectable (Hopkins et al., 2012). Only the air-sea flux of diiodomethane (CH₂I₂) showed a significant positive response to pCO₂. Iodoform (CH₃I) was strongly associated with biological parameters suggesting a biological source and active turnover. Losses of bromoform (CH, Br) correlated with total bacteria, suggesting biological consumption of this compound. Thus, based on the studies conducted so far, the direct effects of ocean acidification on halocarbons appear to be marginal, although no long-term studies have been made as yet.

2.4.3.2 Dimethylsufide

The biogenic gas dimethylsulfide (DMS; CH₃SCH₃) has been implicated in atmospheric aerosol formation (Charlson et al., 1987), a role that is still vigorously debated (Levasseur, 2011; Quinn and Bates, 2011), and its precursor compound DMSP (dimethylsulfoniopropionate) which is used by phytoplankton as an osmolyte, for regulating the internal pressure of the cell, and potentially as a cryoprotectant in ice algae (Kasamatsu et al., 2004) and as an anti-oxidant (Carpenter et al., 2012). The flux of DMS from the ocean to the atmosphere is important because DMS is oxidized in the atmosphere to sulfate aerosols, which in turn act as cloud condensation nuclei resulting in an increased cloud albedo. The potential negative feedback between DMS, algal production and irradiance is the central thesis of the CLAW hypothesis (Charlson et al., 1987).

Ocean acidification should not alter DMS speciation in seawater directly. However, there may be changes in the production and decomposition pathways of DMS from DMSP as, for example, the lyases for DMSP in many organisms are strongly pH sensitive (de Souza and Yoch, 1995; Stefels and Dijkhuizen, 1996). However, the abiotic oxidation of DMS to dimethylsulfoxide (DMSO) by H_2O_2 is not pH dependent under ocean acidification conditions (Amels et al., 1997).

Mesocosm studies simulating ocean acidification have given conflicting results, with claims of increased DMS under conditions of higher CO₂ (Wingenter et al., 2007), while another group reported that when integrated over the entire experiment, neither DMS levels nor DMSP-lyase activity were different between the CO₂ treatments (Vogt et al., 2008). Other mesocosm studies (Hopkins et al., 2010; Avgoustidi et al., 2012) found that DMS and DMSP concentrations were reduced under high CO₂ conditions. Part of the difference observed in the mesocosms may be related to grazing activity, as a further study suggested that DMS will increase due to CO₂induced grazing (Kim et al., 2010). Global modeling results suggest that DMS production will increase at high latitudes (Cameron-Smith et al., 2011), although the model was not tuned correctly and exhibited a cold bias in the Arctic with unrealistic sea-ice extent and persistence. More recent work from a pelagic mesocosm experiment off Spitsbergen (Archer et al., 2012) found that DMS concentrations were inversely related to acidity. Contrastingly, the precursor compound DMSP was elevated with increasing acidity due to elevated rates of DMSP synthesis in parallel with increased rates of inorganic carbon fixation. The elevated DMSP production under ocean acidification conditions was suggested to arise from the increased dinoflagellate abundance under these conditions, again indicating that changes in the phytoplankton community speciation and grazing pressure induced by ocean acidification are more important than any direct ocean acidification effects on DMS production and release.

2.5 Current Arctic Ocean acidification

The complex variety of processes occurring in the Arctic and their interactions (Section 2.3) result in wide ranges of pH and $CaCO_3$ saturation state, with dramatic variations observed both geographically (Table 2.3) and seasonally. Most importantly, the Arctic does not behave in a single, consistent manner, but different

Table 2.3. Observed pH and aragonite saturation states in Arctic seas.

Region	pН	$\Omega_{_{ m Ar}}$	Source
Nordic seas			
Surface	8.1-8.4	1.5-3.5	Olsen et al., 2009
Bottom	7.9-8.3	0.7-2.2	Olsen et al., 2009
Bering Sea			
Surface	7.9-8.3	0.7-2.9	Winn and Millero, 1993; Mathis et al., 2011a
Bottom	7.0-7.7	0.1-2.0	Winn and Millero, 1993; Mathis et al., 2011a
Siberian shelves			
Surface ^a	7.5-8.1	0.2-2.5	Anderson et al., 2011b
Bottom	7.4-7.9	0.2-1.4	Anderson et al., 2011b
Chukchi & Beaufort shelves			
Surface	7.9-8.4	0.8-2.0	Mucci et al., 2010; Shadwick et al., 2011; Bates et al., 2009
Bottom	7.8-8.1	0.8-2.0	Shadwick et al., 2011; Bates et al., 2009
Canadian Archipelago			
Surface	8.0-8.3	0.8-2.2	Chierici and Fransson, 2009; Azetsu-Scott et al., 2010
Bottom	7.6-8.1	0.6-1.4	Azetsu-Scott et al., 2010
Central Arctic			
Surface	8.0-8.2	1.3-1.8	Jutterström and Anderson, 2005
Deep ^b	8.1±0.02	0.6-1.0	Jutterström and Anderson, 2005

^a Includes data from close to river mouths; ^b >2000 m.

processes dominate in different regions, resulting in a complex mosaic of acidification states and vulnerabilities. The following sections summarize current understanding of regional acidification processes in the Arctic, emphasizing the most important processes in each region and the interactions between them.

2.5.1 Inflow seas

2.5.1.1 Nordic Seas

The Nordic Seas (i.e., the Greenland, Iceland, and Norwegian Seas) are situated north of the Greenland-Scotland Ridge and export dense, cold overflow water into the North Atlantic (e.g., Hansen and Østerhus, 2000). Deep waters formed in the Nordic Seas are an important source for the global thermohaline circulation, and the surface waters entering the Nordic Seas from the North Atlantic Ocean are a vital source of dissolved carbon entering the Arctic Ocean. Approximately one third of that Atlantic carbon in the Nordic Seas enters the Arctic Ocean with the Atlantic Water inflow via the Barents Sea and through Fram Strait (Jeansson et al., 2011).

Vertical mixing, both of the Atlantic Water in the Norwegian Sea and within the convective gyres of the Greenland and



Figure 2.8. Recent changes in surface pH in the Iceland Sea and the Norwegian Sea. Source: after Olafsson et al. (2009).

Iceland Seas, transports dissolved carbon from the surface to deeper water masses (see Section 2.3.3, Olsen et al., 2006, 2010). This deep convection also serves as a conduit transporting anthropogenic CO_2 (C_{ant}) to the intermediate and deep waters of the Nordic Seas, decreasing the pH and shoaling the aragonite saturation horizon.

In the Iceland Sea, water column measurements between 1985 and 2011 show that pH and the carbonate saturation states are indeed decreasing rapidly (Olafsson et al., 2009). The cold waters, with a mean winter temperature of -0.3 °C, show clear trends in both pH (-0.0024/y) and $\Omega_{\rm Ar}$ and $\Omega_{\rm Ca}$, which have decreased over the observation period by -0.0117/y and -0.00721/y, respectively (Figure 2.8). In the deep waters (>1500 m), pH has fallen 0.0006/y, about four times slower than at the surface, while at the same time, the aragonite saturation horizon ($\Omega_{\rm Ar} = 1$) has shoaled slightly, to about 1700 m, and the saturation state has decreased by 0.0009/y. As a result, substantial sea-floor areas are in transition from supersaturated to undersaturated conditions.

In the Norwegian Sea, surface ocean pCO_2 increased between 2002 and 2006 at a rate of $2.6 \pm 1.2 \mu \text{atm/y}$ (Skjelvan et al., 2008). This has resulted in an even stronger decrease in pH (-0.0038/y; Figure 2.8) than seen in the Iceland Sea. The pH evolution in the Greenland Sea surface water has not been evaluated thoroughly, but the CARINA data product (e.g., Key et al., 2009) reveals a large variability with no obvious trend.

There are two major reasons for the rapid acidification of the Nordic Seas. First, the buffer capacity of cold Arctic waters is less than that of subtropical waters. Thus, for the same amount of CO_2 added to seawater, the reduction in pH is greater in the Nordic Seas than at lower latitudes. Second, the Nordic Seas surface waters are undersaturated with respect to atmospheric CO_2 throughout the year due to low temperatures and intense biological utilization (e.g., Skjelvan et al., 1999). This results in an air-to-sea CO_2 flux throughout the year.

Figure 2.9 shows estimates of the total changes in surface Ω_{Ar} in the Nordic Seas since the pre-industrial period. This is based on the estimates of C_{ant} in the Nordic Seas prepared by Olsen et al. (2010). The greatest changes have occurred in

the Atlantic water; a consequence of the fact that this water mass carries the largest concentrations of C_{ant} . Nonetheless, the most vulnerable surface waters appear to be in the Iceland Sea, where Ω_{Ar} at 200 m is already 1.2–1.4. This vulnerability is largely because aragonite saturation was already quite low in the pre-industrial period.

The Barents Sea serves as a gateway between the Nordic Seas and the Arctic Basin, including the Laptev and East Siberian Seas (Section 2.5.2.1). Although apparently no or very little C_{aut} uptake takes place within the Barents Sea (Anderson and Olsen, 2002; Omar et al., 2003; Jeansson et al., 2011), the anthropogenic carbon signal is likely to be very strong in this area. Autumn data for 1967 indicate that over the subsequent three decades, pH may have dropped by 0.0019/y and $\Omega_{_{\mathrm{Ar}}}$ by 0.0081/y. Calcium carbonate undersaturation has not yet been observed in the Barents Sea (Jutterström et al., 2010), but Ω_{Ar} approaches 1 in some of the bottom waters. With further anthropogenic acidification and possible increases in primary production, organic matter export, and remineralization in the bottom waters and sediments, parts of this biologically and economically important area could be tipped into aragonite undersaturation in the near future.



Figure 2.9. Estimated changes in surface aragonite saturation in the Nordic Seas between 1860 and 2004, based on anthropogenic CO_2 concentrations derived by Olsen et al. (2010).

2.5.1.2 Bering Sea

Despite its shallow sill depth, the Bering Sea is an important source of corrosive waters to the Arctic Ocean (Figure 2.10), because of the generally high CO_2 content of aged Pacific waters. However, to date, few carbonate system data are available for the Bering Sea, and most are from the eastern side, off Alaska, although those data span nearly 40 years, back to 1970.

While extremely low pH and CaCO₃ saturation states occur in the deep bottom waters of the central Bering Sea (pH = 7.0 and Ω_{Ar} = 0.1), even at the depth of the Bering Strait overflow (about 50 m), pH values as low as 7.6 and Ω_{Ar} down to 0.7 have been observed (Winn and Millero, 1993). Seasonal physical and biological processes influencing the carbonate chemistry of the Pacific inflow as it crosses the Bering Sea (Figure 2.10), result in wide spatial, seasonal and interannual variability in pH and CaCO₃ saturation states (Table 2.3).

A few studies have recognized that upwelling frequently brings high-CO, waters to the surface, particularly in the southern Bering Sea, along the Aleutian Islands (Kelley et al., 1971; Fransson et al., 2006), but also south of the Chukchi Peninsula (Chierici and Fransson, 2009). However, most studies have focused on the dramatic spring and summer biological CO₂ consumption observed on the eastern Bering Sea Shelf (Codispoti et al., 1986; Kaltin and Anderson, 2005; Mathis et al., 2011a), which mitigates acidification in the surface waters while exacerbating acidification in bottom waters. Across the Bering Sea shelf, primary production rates average 330 mg C/m²/d, making it one of the most productive regions in the global ocean (Sambrotto et al., 2008; Mathis et al., 2011a) and generating strong CaCO₃ supersaturations in the surface waters, at least during spring and summer (Mathis et al., 2011a). Much of the organic matter produced during the spring phytoplankton bloom is exported vertically out of the surface mixed layer, introducing a mid-summer pulse of low CaCO₃ saturation into subsurface waters as shallow as 50 m (Mathis et al., 2011a).

Large interannual variability of the marine carbonate system in the Bering Sea coupled with the intermittent frequency of the research to date precludes any concrete, quantitative conclusions about acidification trends in this area. Nonetheless, addition of anthropogenic CO₂ to this system certainly augments the large natural seasonal interactions between ocean biology and seawater carbonate chemistry, additionally lowering both pH and CaCO₃ saturation states.

2.5.2 Shelf seas

2.5.2.1 Siberian Shelf

The Siberian shelf seas encompass the Kara, Laptev, and East Siberian Seas, although to date, carbonate system data are only available for the Laptev and East Siberian Seas. These seas receive oceanic waters from the Atlantic in the west and the Pacific in the east and are highly biogeochemically active (e.g., Semiletov et al., 2005). In these shallow shelf seas, river runoff decreases the salinity and calcium ion concentrations to extremely low values, particularly close to the river mouths. In addition, the substantial quantities of organic matter (both dissolved and particulate; e.g., Semiletov et al., 2011, 2012) brought into the coastal waters by river runoff and coastal erosion are subject to microbial decay that produces CO, (Pipko et al., 2008, 2009) and thus, lowers pH. In combination, dilution by freshwater and acidification by respiratory CO₂ increase CaCO₃ solubility (Anderson et al., 2011b). While these are natural processes, to at least some extent, climate change is increasing both river flow and organic matter release, as permafrost thaws.

On the other hand, substantial primary production takes place within the Siberian shelf seas, consuming CO_2 and thus, increasing pH. One effect of these two counteracting processes is that pH spans a very large range in surface waters (Table 2.3). For instance, values from 7.5 to 8.1 were observed in summer 2008 in the East Siberian and Laptev Seas (Anderson et al., 2011b).



Figure 2.10. Arctic inflow into the Bering Sea (left; Kaltin and Anderson, 2005) and a generalized description of the processes affecting the carbonate chemistry of the eastern Bering Sea shelf (right; Mathis et al., 2011b). River runoff and upwelling create seasonal sources of CO_2 to the atmosphere while reducing carbonate mineral saturation states. Seasonally high rates of primary production lead to a rapid CO_2 drawdown at the surface, creating a strong seasonal atmospheric CO_2 sink and raising surface $CaCO_3$ saturation states. Organic matter export contributes to remineralization at depth, which lowers saturation states.



Figure 2.11. Saturation states for aragonite and calcite and pH in the Laptev, East Siberian and western Chukchi Seas in August-September 2008 for surface water (left) and bottom water (right). Source: after Anderson et al. (2011b).

The marine primary production cycle dominates the carbonate system in the eastern East Siberian Sea, where the Pacific is the dominant seawater source. This is reflected in CaCO₃ supersaturations ($\Omega_{Ar} > 1$) in the surface waters (Figure 2.11). In contrast, the western East Siberian and Laptev Seas are heavily affected by river runoff, and the saturation states in surface waters are less than 1. The bottom waters are undersaturated with respect to both aragonite and calcite in all regions, except for some stations in the Herald Valley and at the continental slope; indeed, the lowest saturation states observed anywhere in the Arctic seas are seen in the bottom waters on the Siberian shelf.

The marine climate in this large shallow region is rapidly changing, with summer sea-ice extent in the East Siberian Sea declining from more than 50% some ten years ago to less than 10% today (Cavalieri et al., 1996 and updated annually). This has many effects: light conditions improve for primary production in areas where terrestrial runoff impact is low; more terrestrial particulate and dissolved organic matter is added to the sea and degraded to CO_2 ; and cyclone frequency and strength will increase, adding to more vertical mixing of the water column. The dominating effect of these processes is to amplify ocean acidification in surface waters.

2.5.2.2 Western Arctic shelves: Chukchi and Beaufort Seas

The shelf waters of both the Chukchi and Beaufort seas are strongly influenced by the Pacific inflow, which introduces naturally low-pH waters into the Arctic (Section 2.5.1.2). Local, seasonal processes act on the Pacific inflow to further reduce pH and $CaCO_3$ saturation states, and thus, this area is vulnerable to acidification, even without anthropogenic perturbation. Although different definitions are used for the boundaries of these seas, this report considers the shelf areas, out to depths of about 250 m (Figure 2.12).



Figure 2.12. Surface currents and sampling points on the Chukchi and Beaufort shelves (upper) and aragonite saturation state (middle) and pH (lower) along the Beaufort Shelf between 1975 and 2010. Source: after Giesbrecht et al. (2013).

Potentially corrosive waters are found in subsurface waters on both the Chukchi (Bates et al., 2009) and Beaufort (Shadwick et al., 2011) shelves, and can readily come to the surface with upwelling (Chierici and Fransson, 2009; Mucci et al., 2010; Mathis et al., 2012). In both areas, summertime phytoplankton primary production increases surface pH and CaCO₃ saturation states (Murata and Takizawa, 2003; Bates et al., 2009; Mucci et al., 2010). Sufficient quantities of the resulting organic matter are apparently exported into deeper waters and remineralized back to CO₂ to generate seasonal decreases in pH and saturation state in subsurface waters (Bates et al., 2009). The converse wintertime process, with decreasing pH and saturation states at the surface due to excess respiration and mixed-layer deepening with sea-ice brine rejection has not been observed consistently (Shadwick et al., 2011; Lansard et al., 2012), and is likely to be dependent on inter-annually variable processes such as ice formation rates and autumn water-column preconditioning (e.g., Melling and Moore, 1995). Thus, interannual variability in the carbonate system of this area is large (Figure 2.12; Murata and Takizawa, 2003; Shadwick et al., 2011; Lansard et al., 2012), although seasonal processes are more likely to tip the system into dangerously low pH and saturation state values as inventories of anthropogenic carbon increase.

2.5.2.3 Canadian Arctic Archipelago to the Labrador Sea

Observed changes in CaCO₃ saturation states in the Arctic surface waters propagate downstream from the Beaufort Sea through the Canadian Arctic Archipelago southward through the Labrador Sea to the Middle Atlantic Bight (Figure 2.13). Along this transect, the CaCO₃ saturation state of Pacific water that was already relatively low in pH decreases further through the addition of sea-ice meltwater and fluvial input, as well as organic matter respiration. Waters undersaturated in aragonite have already been observed at the surface in Coronation Gulf (Chierici and Fransson, 2009), and at only 50 m depth in Hudson Bay (Azetsu-Scott et al., 2008).

A core of undersaturated water at about 150 m observed in profiles in the middle of the Archipelago indicates that the Beaufort Sea and Canada Basin are the source of these waters (Azetsu-Scott et al., 2010), and that a subsurface core of lowsaturation state waters can be traced through the archipelago and Baffin Bay to Davis Strait (Figure 2.13). To date, the Amundsen Gulf is the only entrance area to the archipelago in which the carbonate system has been studied in detail (Chierici et al., 2011), although because of a shallow sill depth in Victoria Strait, other entrances appear to carry larger proportions of the total through-flow. While the outflow through Hudson Strait and local mixing in the northern Labrador Sea modifies the



Figure 2.13. Aragonite saturation states across critical gateways in the Canadian Arctic Archipelago. Source: after Azetsu-Scott et al. (2008, 2010), Lansard et al. (2012), Giesbrecht et al. (2013).

low- Ω Arctic outflow, low saturation states can still be identified over the Labrador Shelf, at least as far south as Hamilton Bank. Within the Labrador Sea, the position of the saturation horizon is depressed by local deep convection and mixing with the intermediate and deep water masses of the North Atlantic.

2.5.3 Central Arctic Ocean

The source of the surface waters of the central Arctic Ocean is mainly the surrounding shelf areas, with a small direct contribution from the inflow of Atlantic and Pacific water through the Fram and Bering Straits, respectively. These source waters have high CO_2 solubility, because of their low temperature and relatively high alkalinity, resulting in pH_T values in the range 8.0–8.2. The deep waters of the central Arctic have a smaller range, with the *in situ* pH_T averaging 8.1 below 2000 m (Table 2.3, Figure 2.14). These deep waters are also much less impacted by anthropogenic CO_2 than the surface waters. Aragonite saturation states below 1 occur deeper than about 2500 m, although Ω_{Ca} is still above 1 throughout the water column (Jutterström and Anderson, 2005).

In the Beaufort Gyre freshwater is trapped, and surface salinity can be as low as 30 or even lower, which causes a substantial lowering of the CaCO₂ solubility. Furthermore, the recent unprecedented summer melting of sea ice has substantially lowered Ca²⁺ concentrations and thus also CaCO₃ saturation state, and this area is one of the first where aragonite undersaturated waters have been observed at the surface (Yamamoto-Kawai et al., 2009). Summer sea-ice melting also exposes the surface waters to the atmosphere, and as the surface waters have to date been largely undersaturated in CO₂ (Jutterström and Anderson, 2010), the surface water could be expected to absorb more atmospheric CO₂, which would further decrease pH and CaCO₂ saturation state. However, increasing stratification, resulting in shallower surface mixed layers, may limit the depth to which CO₂ from air-sea gas exchange can penetrate to subsurface waters.

The halocline water of the Canada Basin has a strong signal of organic matter oxidization, a result of bottom water outflow from the highly productive Chukchi and East Siberian shelves. This water has high nutrient concentrations, high pCO_2 values, and low O_2 and pH. Consequently, the halocline is also undersaturated with respect to aragonite (e.g., Jutterström and Anderson, 2005; Bates and Mathis, 2009). In the past, when the central Arctic Ocean was mostly ice covered in summer as well as winter, this undersaturated water was isolated from the photic zone. However, with less sea ice, the energy for vertical mixing has increased and this water has started to penetrate the lower levels of the photic zone in recent years (Shimada et al., 2006).

2.6 Future scenarios

2.6.1 Introduction

Climate models project a significant retreat in sea-ice cover (e.g., Holland et al., 2006; Solomon et al., 2007; Stroeve et al., 2012a) as well as a shift from multi-year ice to thinner seasonal ice, suggesting a continuation of the trends already being observed (e.g., Maslanik et al., 2007; Giles et al., 2008; Kwok et al., 2009; Stroeve et al., 2012b). Models also suggest a loosening of the ice pack (Zhang et al., 2012) associated with the observed changes in ice conditions. Rampal et al. (2009) found an increase in sea-ice mean speed and deformation rate suggesting an increase in the number of cracks and small leads. All of these changes affect the CO₂ exchange between the atmosphere and ocean either directly (via changes in open water areas) or indirectly (e.g., via changes in biological uptake or changes in sea ice). As discussed in earlier sections, ocean acidification is intensified by increased atmospheric CO₂ and consequent ocean uptake as well as increased freshwater inflow from river runoff, ice melt and low-pH Pacific water, which makes Arctic waters especially vulnerable to a declining saturation state of CaCO, minerals such as aragonite and calcite



Figure 2.14. Section of pH_T (15 °C) from Alaska to the Gakkel ridge (lower panel) collected during the 2005 *Beringia* cruise. The upper panel shows the detail in the top 600 m. The insert shows station locations.


(e.g., Bates and Mathis, 2009; Yamamoto-Kawai et al., 2009, 2011; Denman et al., 2011).

The observed trend over the past two decades suggests continued ocean acidification on a basin-wide scale with further increase in atmospheric CO₂. The pattern and timing of future ocean acidification, however, will be a result of the interplay between many factors including excess CO₂ uptake, changes in ocean circulation, land-ocean exchanges, biological productivity and respiration, freshwater contributions and seaice dynamics. To develop scenarios of future change requires approximations of these mechanisms through informed process understanding incorporated into mathematical models. So far, model simulations of future Arctic Ocean acidification are largely limited to global earth system models (ESMs) which have limited resolution as well as limited skill in the Arctic region (e.g., Schneider et al., 2008; Steinacher et al., 2010), although both resolution and biogeochemical parameterizations have been improved in the latest suite of ESMs. Published results of ESM simulations generally show enhanced ocean acidification in polar regions (e.g., Orr et al., 2005; Steinacher et al., 2009; Denman et al., 2011; Joos et al., 2011; Steiner et al., 2013) and by now it is widely accepted that the Arctic will exhibit the greatest global reductions in pH (see also McNeil and Matear, 2007; Gangstø et al., 2008; Christian et al., 2010) and the earliest undersaturation in Ω_{Ar} (Figure 2.15). Concurrent with these greatest reductions in saturation state are the greatest percentage losses in CaCO₃ production (Gangstø et al., 2008).

Although the direction of change is understood, the rate, cyclicity and regional distribution of anticipated Arctic Ocean acidification is complicated by the non-linear responses of a plethora of drivers and compensatory mechanisms. For example, increases in CO_2 concentration lower pH and warming of the seawater further enhances this decline. However, the decrease in saturation state (Ω) that follows CO_2 uptake may be, partially, compensated for by ocean warming. On a global basis, temperature increases



Figure 2.15. Global aragonite saturation state for the surface ocean under the Intergovernmental Panel on Climate Change SRES A2 greenhouse gas emissions scenario. Graphic shows latitudinal means. Source: after Steinacher et al. (2009).

will be the dominant climate change feedback on saturation state (Orr et al., 2005; McNeil and Matear, 2007) but will have little effect on pH (McNeil and Matear, 2007).

2.6.2 Ocean acidification changes projected over the 21st century

Steinacher et al. (2009), using the Climate System Model (CSM) 1.4-Carbon of the US National Center for Atmospheric Research (NCAR), and Denman et al. (2011), using the Canadian ESM CanESM1, report model projections of aragonite and calcite saturation, carbonate concentration, and pH, based on SRES scenarios; all project significant acidification in the Arctic Ocean over the course of the 21st century (see Figures 2.15 and 2.16). CMIP5 results are now publicly available, allowing for a more thorough analysis; however published results are limited



Figure 2.16. Projected change in pH and aragonite saturation through the 21st century using the CanESM-1.0 model. Source: after Denman et al. (2011).

so far. Deal et al. (2013) reported results for the Pacific Arctic (Bering and Chukchi Seas) with the CanESM1.5, a precursor of the Canadian CMIP5 contribution CanESM2, which has already increased resolution in the ocean. They project surface pH for RCP8.5 forcing to further decrease to about 7.9 by mid-century and below 7.7 by the end of the century, which corresponds with the results reported by Steinacher et al. (2009). Projections of Ω_{Ar} and Ω_{Ca} , which also depend on salinity and are significantly affected by increased freshwater contributions from river inflow, precipitation, and ice melt (Yamamoto-Kawai et al., 2009; Denman et al., 2011), are hence affected by how these processes are represented in models. For the end of the 21st century, CanESM1.5 projects Ω_{Ar} to change by about 0.8-1.0 across most of the Pacific Arctic region with somewhat smaller changes north and northeast of the Bering Strait, where surface waters have already become undersaturated in current times (Deal et al., 2013).

Increases in primary production, which act to increase $\Omega_{\Lambda r}$ (and Ω_{Ca}) in the surface layer, might counteract a reduction in saturation states in certain areas at certain times. However, projections of Arctic primary production are highly variable. Steinacher et al. (2010) discussed discrepancies between results from empirical approaches and process-based ESMs and pointed out the importance of a realistic representation of nutrient cycling and distribution in order to project changes in primary production. Vancoppenolle et al. (2013) in a recent assessment of projected primary production, nutrient and seaice concentrations in 11 CMIP5 ESMs found that the mean model simulates Arctic-integrated primary production for 1998–2005 well, but that models neither agree on what limits primary production today, nor on the sign of future change. A balance of a decrease in available nutrients due to increased stratification and increased light availability due to a reduced sea-ice cover operates in all models; however, it depends on the particular model as to whether the decrease in available nitrate is sufficient to overcome or not the benefits of the light increase. Vancoppenolle et al. (2013) suggested that the main cause for the large inter-model spread was poorly constrained Arctic nitrate concentrations. In view of their results it is not clear whether future changes in Arctic primary production will accelerate or decelerate ocean acidification.

A multi-model comparison including the CanESM2 (Canada, e.g., Arora et al., 2011), IPSL-CM5A-LR (France, e.g., Dufresne et al., 2013), MPI-ESM-LR (Germany, e.g., Ilyina et al., 2013), GFDL-ESM2M (U.S.A., e.g., Dunne et al., 2012, 2013) and HadGEM2-ES (U.K., e.g., Collins et al., 2011) models shows the regionally and 20-year averaged annual mean surface pH to decrease from 8.11 in 1986–2005 to 8.00 in 2026–2045 and further to 7.92 in 2066–2085 for RCP8.5 (Figures 2.17 and 2.18). However, locally, pH values are projected to become even lower. Figure 2.17 shows an example of bi-decadally averaged Arctic Ocean pH in September, with pH below 7.7 in some areas. The very low values in the Hudson Bay for some models are likely to be unrealistic.

CMIP5 model results for the above-mentioned time periods (also including the MIROC-ESM, Japan, e.g., Watanabe, 2011) project Ω_{Ar} and Ω_{Ca} in the Arctic to reduce by about



Figure 2.17. Bi-decadal mean pH for September as projected by the CanESM2, IPSL-CM5A-LR, MPI-ESM-LR, GFDL-ESM2M and HadGEM2-ES models. Time periods shown are 1986–2005 (top), 2006–2025 (middle) and 2066–2085 (bottom). Latitude circles indicate 60° and 80° N.

Figure 2.18. Multi-model comparison of regionally averaged bi-decadal mean of pH, aragonite saturation state (annual mean) and sea-ice concentration (March and September). Models included are the CanESM2, IPSL-CM5A-LR, MPI-ESM-LR, GFDL-ESM2M, HadGEM2-ES and MIROC-ESM as indicated in the legend. Solid lines show the multi-model mean and error bars indicate model standard deviations.



0.7–1.0 in correspondence with reductions in summer sea-ice concentrations (Figure 2.18).

Furthermore, as anthropogenic CO_2 penetrates deeper into the oceans, the seawater saturation horizons with respect to various carbonate mineral phases, the layer where Ω equals 1.0, will become shallower (Feely et al., 2004). Consequently, the maximum depth at which deep water corals and other calcifying organisms are found will shoal (Guinotte et al., 2006, Turley et al., 2007). The saturation horizons generally are already comparatively shallower in high latitude temperate and Arctic waters than at lower latitudes (Broecker and Peng, 1982).

Simulated global time series of the zonal mean saturation horizon show a continuous shoaling from pre-industrial times to 2100 for all latitude bands with the exception of the Arctic (not shown). In the Arctic, in addition to a continuous shoaling of the deep saturation horizon, a shallow saturation horizon starts to form close to the surface. This is caused by the combination of increased CO₂ uptake and freshwater contributions at the surface which in a large part of the Arctic Ocean basins leads to surface and possibly subsurface undersaturation. This formation of multiple layers of undersaturation is a special characteristic of the Arctic Ocean. The models vary within the range of about 1.0 in the $\Omega_{\lambda a}$ of deeper layers (below 200 m) and the deep saturation horizon varies between ~2000 m to ~4000 m for the individual models. Waters in the European Basin follow the pattern seen in the major oceans with a rising saturation horizon from bottom to top and surface waters becoming undersaturated much later than the rest of the Arctic.

In addition to long-term changes in Arctic Ocean acidification, changes in the physical environment, specifically sea-ice cover, might alter the seasonal carbon cycling, affecting seasonal minima in pH and saturation states. In a sensitivity study with CanESM2, Steiner et al. (2013) found that retreating sea ice in the future leads to a shift in the seasonal cycle with an increase in CO_2 uptake in autumn and winter, a more continuous equilibration and hence a reduced uptake in summer. The reduction in summer suggests a limited CO_2 -uptake capacity of Arctic surface waters, which has also been suggested by Cai et al. (2010) and Jutterström and Anderson (2010).

2.6.3 Regional modeling studies

While global model simulations provide consistent projections of reduced pH and first estimates for the timing of when aragonite and calcite undersaturation may be reached in the Arctic, coarse vertical and horizontal resolution and crude (or no) representations of passages and through-flows limit the applicability at local scales. To address the question of specific regional effects, regional models need to be developed and run for future scenarios. AMAP has co-funded two case studies under this reporting project.

2.6.3.1 A 1-D model study for the Laptev Sea

To investigate the combined effect of changes in several forcings on carbon transformation and fluxes, a physicalbiogeochemical 1-D model was developed specifically for the Laptev Sea (Wåhlström et al., 2012, 2013). The model was constructed for a location in the central Laptev Sea and driven by atmospheric forcing, seawater influx from the central Arctic Ocean, and river runoff. The model was further developed to follow the evolution of the carbon system during the 21st century, applying an increase of atmospheric *p*CO, to 900 µatm



in 2100 (SRES-B1) and a resulting atmospheric temperature increase of 4 °C until 2100. The latter directly affects the temperature of the surface water (Figure 2.19) and thus also the length of the sea-ice period.

One result of the increased atmospheric pCO_2 is elevated CO_2 uptake by the surface seawater, which results in a pH decrease (Figure 2.19). The modeled maximum pH around June–July is due to CO_2 fixation by primary production. Reduced seaice cover positively affects this primary production; however, limited nutrient supply ultimately limits the annual new production. The net result is that the drawdown of CO_2 by primary production cannot compensate during the rest of the year for the increased pCO_2 from the atmospheric increase (Figure 2.19). However, one somewhat less significant effect of the increased primary production is the increase in organic matter sedimentation with subsequent remineralization all through the water column.

The effect on pH is directly influencing the solubility of calcium carbonates as illustrated by Ω_{Ar} (Figure 2.19). This reduction in saturation state will be amplified by lowered salinities, either from increased sea-ice melt or increased river discharge. Potential release of alkalinity trapped in sea ice (Section 2.3.1) or increased alkalinity in river waters might counteract this reduction in saturation state to a certain extent.

2.6.3.2 Atlantic-Arctic Greenland-Svalbard-Norway gateway

The Barents Sea, Fram Strait and the continental shelf north of Svalbard are productive shelf regions (Slagstad et al., 2011) and are a large contemporary CO_2 sink (Årthun et al., 2011). A considerable shelf pump transfers natural and excess carbon into the surface and intermediate Arctic Ocean (Kivimäe et al., 2010). Located at the frontier between the Atlantic and the polar waters, this region is very sensitive to changes in water temperature, ice cover and thus air-sea CO_2 exchange. Simulations with a regional

coupled sea-ice ocean biogeochemical model of modifications to the CO₂ system project large reductions in pH over the course of the 21st century (Bellerby et al., 2005, 2012). The pH reductions (between the means of the years 2006 to 2015 and 2099-2090) are large for the whole region and range from 0.24 to 0.35 (Figure 2.20). The greatest changes are associated with the Arctic waters of the East Greenland Current and to the north of Svalbard. Similarly, reductions in Ω_{Ar} are highest in the Arctic outflow waters (Figure 2.21). The changes are, however, very regional, and the different carbon system variables may respond differently on local scales. Increases in carbon are mostly a consequence of excess carbon transport with the Norwegian Atlantic Current and increased gas exchange promoted by decreased ice cover. Warming is especially prominent north of the present-day maximum extent of the marginal ice zone, and this acts to enhance ocean pH reduction and increase Ω_{Ar} . Freshening due to ice melt and increased run-off will lower both pH and $\Omega_{\lambda r}$. Deep wintertime mixing acts to dampen ocean acidification by diluting the surface excess carbon signal with waters with a lower C_{ant}.



Figure 2.20. Change in surface ocean pH (mean 2006–2015 minus mean 2090–2099) for the Atlantic-Arctic Greenland-Svalbard-Norway gateway region.

Figure 2.19. Change in temperature (upper), pH_T (middle) and aragonite saturation state (lower) in the central Laptev Sea for 2010 and 2050. Note that the plots for 2010 and 2050 have the same scales.



Figure 2.21. Simulated aragonite saturation state through the 21st century (mean 2006–2016 and mean 2090–2100) for the northern Nordic Seas, western Barents Sea and Eurasian continental shelf.

2.6.4 Limitations of future projections

Many of the processes affecting the pace and pattern of Arctic Ocean acidification are inadequately known, such as the changing freshwater distributions within the Arctic Ocean (e.g., Long and Perrie, 2013), the changes in primary production, and the timing and pattern of sea-ice retreat. For example, a model study by Yamamoto et al. (2012) highlighted the importance of the sea-ice reduction rate on acidification, suggesting that future reductions in pH and Ω_{Ar} with an increased pace in sea-ice reduction could be significantly faster than previously projected. Stroeve et al. (2012a) found that while IPCC AR5 models better capture the observed decline in Arctic sea ice than an earlier comparison project (Stroeve et al., 2005), the models exhibit a stronger seasonal cycle in both sea-ice extent and volume, and the inter-model scatter remains, particularly for summer. It is also clear that the spatial variability of ice thickness is not simulated well by the majority of models (Maslowski et al., 2012). The simulated gas flux through small leads and cracks is likely to be underestimated (Steiner et al., 2013) and carbon transport through the sea-ice matrix (e.g., Loose et al., 2011) is not represented in models at all. Another



Figure 2.22. Simulated change in pH across the Arctic over a 100-year period due to a 50% methane release. Source: after Biastoch et al. (2011).

unknown is the potential release of methane and its subsequent oxidation to CO_2 . Biastoch et al. (2011) showed that the local pH values could be lowered by over 0.25 units if 50% of the methane from hydrates is released over a period of 100 years (Figure 2.22).

The complexity of the ecosystem and carbon modules varies between models (e.g., Vancoppenolle et al., 2013). In many cases, the ecosystem structure is not sufficient to resolve the foodweb structure in the Arctic or to adequately represent complex processes and interactions. Other models might include higher complexities, but applied parameterizations might be inadequate, either due to limited understanding in general or, especially in the case of global models, because global parameterizations might not be suitable for Arcticspecific processes.

The fairly coarse resolution in global models does not make it possible to properly resolve the physical processes important for biogeochemical carbon exchange, especially within the Arctic, where shallow shelves and narrow passages are common features. Higher resolution regional models simulating ecosystem and carbon cycle processes will need to be applied to project the impacts of future climates in more detail. However, as seen above, these models are extremely limited, as yet.

2.6.5 Lifetime of ocean acidification

As CO₂ continues to be produced from fossil fuel combustion and other sources, ocean acidification will continue roughly in line with the increasing atmospheric CO₂ concentration. When geological carbon reserves become uneconomic, or legislation is in place and other energy sources are preferred, atmospheric CO, will decline, as will surface ocean acidification, as the anthropogenic carbon continues to be mixed into the ocean abyss, increasing deepwater acidification (Caldeira and Wickett, 2003; Archer, 2005) (Figures 2.23 and 2.24). Mixing will be the main moderator of ocean acidification excesses for the next 1000 years. To reduce ocean acidification requires the dissolution of CaCO₃ on the sea floor that will increase the A_r of seawater and therefore increase pH and reduce ocean pCO₂ and then atmospheric CO₂ (Archer, 2005; Ridgwell and Hargreaves, 2007). There is not, unfortunately, enough accessible CaCO₃ to buffer ocean acidification and CaCO₃

32



Figure 2.23. Example of mean ocean response to a theoretical carbon emission rate and extent showing the initial surface ocean acidification extreme and then dilution of the surface as carbon penetrates the interior ocean increasing ocean acidification at depth. Source: after Caldeira and Wickett (2003).

compensation will be exhausted over the next centuries. This is when the true geological scaling of ocean acidification will become apparent: the weathering of silicates is required to further neutralize the fossil fuel perturbation. Ultimately, the increasing alkalinity of the ocean will promote calcification and increased sedimentary deposition returning the carbon back into the geological reservoir. However, with a timescale of tens to hundreds of thousands of years (Archer, 2005; Archer et al., 2009a), a return toward pre-perturbation atmospheric CO_2 and ocean pH is of little use in managing ecosystem and societal needs of today.

2.7 **Conclusions**

There is a building awareness of ocean acidification and the role it has in shaping the chemical matrix of seawater and the effects this may have on ecosystems and biogeochemical cycling in the ocean. There can be no doubt that, as the atmospheric CO₂ concentration rises, the surface ocean pH will drop and there will be a change in the speciation of the carbonate system. The Arctic Ocean is one of the most sensitive ocean regions regarding the ocean acidification response to CO, uptake and will be the first ocean to undergo basin-wide acidification, crossing important geochemical thresholds. The seawater in the Arctic is already poorly buffered with low carbonate ion concentrations and a high $C_{_{\!\rm T}}$ to $A_{_{\!\rm T}}$ ratio which results in large changes in pH with modifications to the CO₂ concentration. Although current knowledge is based on few direct measurements, it is clear that the Arctic Ocean is undergoing rapid acidification. Unique on such a grand scale, in addition to direct anthropogenic CO₂ effects, ocean acidification is also being driven by large-scale freshening and increases in terrestrial carbon fluxes. The Arctic has a rapidly changing freshwater budget through increasing fluvial inputs and in addition a net supply from changes in sea-ice distribution.

Arctic Ocean acidification is being measured throughout the water column with the largest deep-water changes taking place in the Nordic Seas. However, it is in the surface and upper intermediate waters that changes are happening fastest. The bulk of the excess CO_2 is being transported to the Arctic from other ocean basins and is being taken up directly from the atmosphere.

Ocean acidification is changing the geochemistry of the Arctic. Decreases in CO_3^{2-} through CO_2 invasion and reduction in Ca^{2+} from freshwater dilution are reducing the stability of $CaCO_3$. Ocean acidification has been shown to influence nutrient cycling and trace element availability. The consequences of these changes on marine organisms and ecosystems are discussed in Chapter 3.



Figure 2.24. Geological lifetime of atmospheric CO₂. Model results for pCO_2 for different CO₂ emission scenarios and model runs without (left) and with (right) ocean temperature feedback. Source: after Archer (2005).

Ocean acidification, to the first degree, closely follows atmospheric CO₂ concentrations. Ocean acidification is occurring fast in the Arctic such that within decades the entire surface Arctic will be undersaturated with respect to CaCO₂. Complete undersaturation from the surface to the sea floor in the Arctic Ocean will occur due to the expansion of a deep saturation horizon upward following transport of anthropogenic CO₂ from outside the Arctic and the expansion of a shallow saturation horizon downward due to freshwater addition, enhanced local uptake, mixing, and shelf-ocean interior exchange. Over thousands of years, even when fossil fuel combustion has been reduced, ocean acidification will still be prevalent, although ocean overturning will reduce the surface signal to some extent. Geochemically, removal to marine organisms (corals and other calcifiers) and sedimentary CaCO₃ due to dissolution in the corrosive ocean is the intermediate pathway to buffer ocean acidification. Even if fossil fuel emissions would cease today, the ocean would not return to pre-industrial pH values for tens of thousands of years. However, reductions in emissions will curtail the extent and duration of ocean acidification.

Key points:

- The carbon inventory of the Arctic Ocean is increasing, leading to ocean acidification. *The amount of inorganic carbon in the Arctic Ocean is going up and this is lowering the pH of the seawater and changing the relative amounts of inorganic carbon species.*
- The major driver of increasing ocean acidification is anthropogenic carbon uptake. The major increase in inorganic carbon is derived from the combustion of fossil fuels.
- The degree and distribution of Arctic Ocean acidification is further controlled by ice cover, freshening, warming and terrestrial inputs. In addition to fossil fuel combustion, the other, smaller but significant, contributors to ocean acidification in the Arctic include the changing cryosphere which allows for more air-sea gas exchange, salinity reductions due to ice melt and increasing river run-off, ocean warming and increasing inputs of carbon from land.
- The carbon dioxide content and pH of seawater has high natural temporal and spatial variability. *High seasonality of the solubility and biological pumps together with the low buffer capacity of Arctic seawater results in high seasonal variability of the marine carbonate system.*
- There will be modifications to some nutrient cycling and trace metal speciation. *Changes in seawater pH can modify the forms of trace elements and nutrients, and thus their bioavailability will be altered.*
- Regions of the Arctic Ocean are, globally, among the first to reach geochemical thresholds of ocean acidification. *Forms of calcium carbonate will become geochemically unstable over the next decades. The Arctic Ocean is the first ocean region where this will become widespread.*
- Ocean acidification will be tenacious. *Geological carbon from fossil fuel combustion will remain in the ocean-atmosphere system for tens of thousands of years.* And even after this time, *the system will have not recovered to its pre-industrial marine carbonate system state.*

Chapter 2 Appendix: Compilation of published data (1960-2012) on trace elements and their isotopes in the Arctic

Table A1. Trace metal distribution studies in the Arctic.

Element	Region	Source
Aluminum	Lomonosov Ridge Canada and Eurasian Basins Norwegian Sea Amundsen, Makarov and Nansen Basins and the Barents and Laptev Seas	Moore, 1981 Measures, 1999 Hall and Measures, 1998 Middag et al., 2009
Barium	Canada and Eurasian Basins White, Barents, Kara, Laptev and Beaufort Seas Fram and Denmark Straits Canada and Eurasian Basins Canada Basin Makarov Basin Canadian Archipelago	Guay and Falkner, 1997 Guay and Falkner, 1998 Taylor et al., 2003 Alkire et al., 2007 Guay et al., 2009 Abrahamsen et al., 2009 Thomas et al., 2011
Cadmium	Lomonosov Ridge Nansen Basin Barents Sea and Fram Strait Nansen Sound (Ice Island) Fram Strait and Greenland Sea	Moore, 1981 Danielsson and Westerlund, 1983 Mart et al., 1984 Yeats and Westerlund, 1991 Pohl et al., 1993
Cobalt	Nansen Sound (Ice Island)	Yeats and Westerlund, 1991
Copper	Lomonosov Ridge Bering Sea Nansen Basin Barents Sea and Fram Strait Nansen Sound (Ice Island) Fram Strait and Greenland Sea	Moore, 1981 Heggie, 1982 Danielsson and Westerlund, 1983 Mart et al., 1984 Yeats and Westerlund, 1991 Pohl et al., 1993
Iron	Nansen Basin Canada and Eurasian Basins Bering Sea Bering Sea Chukchi Sea Amundsen, Makarov and Nansen Basins and the Barents and Laptev Seas	Danielsson and Westerlund, 1983 Measures, 1999 Aguilar-Islas et al., 2007 Hurst and Bruland, 2007 Nakayama et al., 2011 Klunder et al., 2012a,b
Lead	Barents Sea and Fram Strait Fram Strait and Greenland Sea	Mart et al., 1984 Pohl et al., 1993
Manganese	Nansen Sound (Ice Island) Bering Sea Amundsen, Makarov and Nansen Basins and the Barents and Laptev Seas	Yeats and Westerlund, 1991 Aguilar-Islas et al., 2007 Middag et al., 2011
Nickel	Nansen Basin Nansen Sound (Ice Island)	Danielsson and Westerlund, 1983 Yeats and Westerlund, 1991
Zinc	Lomonosov Ridge Nansen Basin Barents Sea and Fram Strait Nansen Sound (Ice Island) Fram Strait and Greenland Sea	Moore, 1981 Danielsson and Westerlund, 1983 Mart et al., 1984 Yeats and Westerlund, 1991 Pohl et al., 1993

Table A2. Trace metal speciation studies in the Arctic and sub-Arctic.

Element	Region	Source
Copper	Bering Sea	Moffett and Dupont, 2007
Iron	Bering Sea (as tracer) Amundsen, Nansen and Makarov Basins and the Kara Sea	Hurst and Bruland, 2007 Thuróczy et al., 2011
Zinc	Bering Sea (as tracer)	Hurst and Bruland, 2007

Table A3. Natural abundance of trace metal isotopes in the Arctic	Table A3. Natural abur	idance of trace metal	l isotopes in	the Arctic.
---	------------------------	-----------------------	---------------	-------------

Element	Region	Source
Beryllium	Canada, Amundsen and Nansen Basins and Fram Strait	Frank et al., 2009
Cadmium	Canada and Amundsen Basins	Ripperger et al., 2007
Hafnium	Canada Basin and Central Arctic	Zimmermann et al., 2009
Neodymium	Barents Sea and Fram Strait Canada, Amundsen, and Makarov Basins	Andersson et al., 2008 Porcelli et al., 2009
Lead	Canada Basin Canada and Eurasian Basins	Moore and Smith, 1986 Smith et al., 2003
Radium	Canada Basin Barents Sea, Eurasian Basin Canada and Eurasian Basins Kara and Laptev Seas Chukchi and Beaufort Seas Eurasian and Canadian Basins	Moore and Smith, 1986 Rutgers van der Loeff et al., 1995 Smith et al., 2003 Rutgers van der Loeff et al., 2003 Kadko and Muench, 2005 Kadko and Aagaard, 2009
Thorium	Alpha ridge Nansen Basin Norwegian and Denmark Seas Makarov, Amundsen and Nansen basins. Eurasian and Canada Basins Canada Basin Canada Basin Canada, Amundsen, Nansen and Makarov Basins Canada Basin Beaufort Sea Beaufort Sea Chukchi Sea	Bacon et al., 1989 Cochran et al., 1995 Moran et al., 1995 Scholten et al., 1995 Moran et al., 1997 Edmonds et al., 1998 Amiel et al., 2002 Edmonds et al., 2004 Trimble et al., 2004 Lepore and Moran, 2007 Amiel and Cochran, 2008 Yu et al., 2010
Uranium	Canada and Makarov Basins Beaufort Sea	Andersen et al., 2007 Not et al., 2012

3. Biological responses to ocean acidification

Lead authors: Howard I. Browman, Sam Dupont, Jon Havenhand, Lisa Robbins

Contributing authors: Michael Beman, Carlos Duarte, Maoz Fine, Jan Helge Fosså, Jason Hall-Spencer, Pamela Hallock-Muller, Thomas P. Hurst, Debora Iglesias-Rodriguez, Paul Knorr, Haruko Kurihara, John Lisle, Clara Manno, Sophie McCoy, Frank Melzner, Philip Munday, Hans-Otto Pörtner, Justin Ries, Dominique Robert, Jeffrey Runge, David Scott, Hein Rune Skjoldal, Keita Suzuki, Frede Thingstad, Tim Wootton

3.1 Introduction

This chapter presents an overview of what is known about the effects of ocean acidification on marine organisms and ecosystems. Information on Arctic species and ecosystem processes is extremely limited owing to the logistical difficulties of working in these environments (remoteness, ice cover, high cost, etc.) and in conducting experiments at very low temperatures (long generation times, working at below freezing temperatures for prolonged periods, etc.). This is clearly illustrated in Figure 3.1 using echinoderms as an example. Where possible, this report presents available data from Arctic and/or polar cases. However, for the reasons just stated, most of the material presented here is from studies on non-Arctic species and ecosystems, accompanied by an attempt to assess what this means in an Arctic context.

This chapter summarizes briefly the natural variability in pH and carbon dioxide (CO_2) in time and space (see Chapter 2 for further details) and presents an overview of the biological and physiological context for assessing the impact of ocean acidification on marine organisms and ecosystems (Section 3.2). This is followed by a description of Arctic ecosystems and food webs (Section 3.3). The chapter then reviews what is known about the effects of ocean acidification on different taxonomic groups; with each sub-section distilling what is known from non-Arctic species and then discussing how that may be relevant to Arctic species and to the key linkages in the Arctic food web (Section 3.4). Ecosystem-wide responses to ocean acidification are addressed in Section 3.5. There is then a brief discussion of ocean acidification analogs (areas of naturally lower pH and/or high carbon dioxide regions) and their potential for research (Section 3.6). The chapter concludes with some general caveats to this work (Section 3.7), proposals for research priorities (Section 3.8) and a series of conclusions and recommendations (Section 3.9).

3.2 Biogeochemical, biological, and physiological context for assessing the impact of ocean acidification

3.2.1 Natural variability in pH and carbon dioxide in time and space

Uptake of atmospheric CO_2 by the oceans is driving changes in the carbonate chemistry of the Arctic Ocean – this is commonly referred to as 'ocean acidification' and is described in detail in Chapter 2. There is high spatio-temporal variability in these changes with, for example, rapid and large fluctuations on the continental shelves and more stable conditions in ancient deep waters that currently contain no anthropogenic carbon. Organisms living in shallow coastal waters may already experience transient changes in pH as great as 7.4–9.2 on a daily basis (e.g., Middelboe and Hansen, 2007). Pelagic organisms will be exposed to less pronounced diel fluctuations, but will encounter seasonal changes in pH of the order of 7.5–9.7



Figure 3.1. Number of published studies investigating the impact of ocean acidification on sea urchins. Source: Sam Dupont, University of Gothenburg.

(Hofmann et al., 2011). The range in pH to which organisms might be exposed is particularly large (7.1–9.7) in eutrophic coastal waters (e.g., Hansen, 2002). In the Arctic, large changes in pH may occur in sea ice, but in open waters pH will typically be in the 7.7–8.3 range. The highest pH values (9.7) have been recorded during algal blooms, whereas the lowest (7.1) occur where respiration lowers the pH, such as below the pycnocline or during winter after prolonged dark periods.

A comprehensive dataset of continuous pH observations spanning polar to tropical ecosystems, and open-ocean to kelp forests and coral reefs, shows that pH in the upper 15 m of the water column can vary by up to 1.43 units (Hofmann et al., 2011). Polar sites (Antarctica) showed low fluctuation (≤ 0.1 pH units), although measurements were only made for two weeks. The most extreme fluctuations were observed in natural CO₂ vent areas (Ischia, Italy), however, kelp-forest, estuarine, and upwelling ecosystems all showed fluctuations of 0.4-0.6 pH units over a period of four weeks (Hofmann et al., 2011). Similar magnitude pH variations have been observed diurnally in tide pools (Wootton et al., 2008; Moulin et al., 2011). In all cases, lower-bound recorded pH values can be below, or well below, those projected for the end of the 21st century for the average surface ocean (IPCC, 2007). This demonstrates that many organisms are already experiencing pH conditions that are lower than previously considered likely. The consequences of fluctuating pH conditions are only just beginning to be investigated. The variability in ocean acidification and Arctic Ocean carbonate chemistry will increase in both space and time as the process of acidification continues. This means that the biological and ecological effects of ocean acidification will vary regionally and with its increase over time. Consequently, as well as a need for more data detailing diurnal/seasonal/annual fluctuations in pH in Arctic habitats, experimental results are also required to consider local scenarios and take into account the effects of fluctuating pH on Arctic species (McElhany and Busch, 2012).

The interpretations and conclusions of the ocean acidification challenge experiments described in this chapter should be set within the context of this natural variability in pH.

3.2.2 General biological responses to lowered pH / elevated CO

Marine organisms with external shells made up of calcium carbonate obtain the raw material used to build them from seawater (e.g., Roleda et al., 2012). Therefore, the carbonate chemistry of seawater is an important factor for these organisms as it will, at least in part, affect the shell-building process and, for those organisms whose shells are not protected by a chemically resistant organic layer, also the rate of dissolution (e.g., Lischka et al., 2011). For heterotrophic marine metazoans, pH and the partial pressure of CO_2 (pCO_2) are the main ocean acidification-related variables that impact them: effective respiration requires a diffusion gradient of CO₂ from the extracellular body fluids (blood, haemolymph or coelomic fluid) to the seawater. Typically, pCO_2 in ectothermic marine animal body fluids is about 1000 to 4000 µatm higher than that of seawater (Melzner et al., 2009a). Increases in seawater pCO_2 , typically increase body fluid pCO_2 by an equal amount, as

diffusion gradients must be maintained to guarantee constant rates of CO_2 excretion (see Melzner et al., 2013: Fig. 6). These unavoidable changes in body fluid pCO_2 impact physiology primarily through two mechanisms: (1) accumulation of bicarbonate ions in the body fluids in order to stabilize pH at higher levels (e.g., cephalopods, teleost fish, many decapod crustaceans), (2) no bicarbonate accumulation resulting in much larger decreases in pH (Seibel and Walsh, 2003; Melzner et al., 2009a). Both types of reaction can result in pathologies in short- to medium-term experiments (e.g., Melzner et al., 2013).

Many other direct effects of ocean acidification on organisms are possible; for example, sublethal effects on metabolism and energetics, growth rate, condition, reproductive success or behavior (e.g., see Chapters 4–10 in Gattuso and Hansson, 2011). There is also the possibility of indirect impacts through the food web (e.g., Eklof et al., 2012; Rossoll et al., 2012). To complicate matters further, these effects can be very different depending upon the life-history stage being evaluated (although essential, studies that consider the complete life cycle of an organism are rare).

3.2.2.1 Impact on calcification by marine organisms

The calcification rates of a wide range of marine organisms decline when subjected to the seawater pH values projected for the coming centuries due to atmospheric increases in CO₂ Review articles by Kleypas et al. (1999, 2006), Hoegh-Guldberg et al. (2007), Fabry et al. (2008), Doney et al. (2009), Gattuso and Hansson (2011), and Wicks and Roberts (2012) provide excellent summaries of many of the studies conducted to date. Ries et al. (2009) also found highly variable calcification responses of 18 species of benthic marine organisms to pCO_2 treatments of 400, 600, 900, and 2850 ppm. They observed linear negative responses (soft clam, oyster, scallop, periwinkle, whelk, serpulid tubeworm), threshold-negative responses (tropical urchin, temperate coral, conch, hard clam), no response (mussel), parabolic responses (coralline red alga, calcareous green alga, temperate urchin, limpet), a threshold-positive response (lobster), and linear positive responses (crab, shrimp). These disparate responses are attributed to differences in the organisms' ability to regulate pH at their site of calcification, the extent to which their outer shell is covered by an organic layer, the solubility of their biomineral polymorph, and whether they utilize CO₂ directly via photosynthesis. Numerous experimental studies have investigated the effect of CO2-induced ocean acidification on rates of calcification within marine organisms (see reviews cited above). Far fewer, however, have investigated its effects on other properties of their skeletons (e.g., Cohen and Holcomb, 2009; Dissard et al., 2010), such as polymorph mineralogy, elemental composition, ultrastructure, and biomechanics.

3.2.2.2 Physiological effects

The effect of ocean acidification on phytoplankton and primary production has been one of the main foci of research. Inorganic carbon is one of the essential building blocks of photosynthesis and, therefore, increases in pCO_2 might be expected to increase primary production. This pattern is generally seen in macroalgae

(see Section 3.4.5); however, no consistent pattern of response to ocean acidification has emerged in phytoplankton. This may be related to highly efficient carbon capture systems in many phytoplankton species, however different groups, different species, and even different clones within species show divergent responses to ocean acidification, revealing a highly contextually dependent pattern of photosynthesis, growth, carbon fixation, nutrient usage, and calcification. Phytoplankton blooms are appearing earlier (Kahru et al., 2011) and increased dominance of blooms by picophytoplankton has been attributed to ocean acidification (Newbold et al., 2012) and freshening (Li et al., 2009) of the Arctic Ocean. The extent to which these changes will have flow-on effects in the Arctic food web is unknown but could be substantial. These patterns and their likely consequences are dealt with in more detail in Section 3.4.3. Other work suggests that the harmful algal diatom pseudonitzchia produces more toxin (domoic acid) under high CO, conditions (Sun et al., 2011).

Assessing the sensitivity of Arctic marine animals to ocean acidification must consider the special circumstances and functional characteristics of life in Arctic waters. The Arctic Ocean is an open system in which variability in ambient conditions is driven partly by the influence of waters from the North Atlantic and North Pacific. In these areas, as well as the sub-Arctic, temperatures are low but variable. This contrasts with the more constant temperatures in Antarctic oceans due to its near isolation by the circum-Antarctic current. Understanding the evolution and adaptations of marine polar fauna to changing climate and ocean acidification requires an understanding of the key mechanisms and driving forces involved. The key driving force presently is temperature, although enhanced solubility of CO, in cold waters and body fluids may also shape their future sensitivity. High solubility of CO₂ decreases aragonite and calcite saturation states and lowers the pH of the water (see Chapter 2, Section 2.2), which has been hypothesized to threaten calcifying organisms at the poles (Orr et al., 2005; but see Roleda et al., 2012, who argued that calcifying organisms that do not rely on carbonate may not be affected). Multi-year ice melt, increased river input and associated salinity decreases exacerbate the extent of acidification in surface layers of the Arctic Ocean (Denman et al., 2011). The combined effects of warming, acidification, and freshening, on marine ecosystems have not yet been estimated (Ainsworth et al., 2011).

Understanding key physiological and life-history characteristics is critical to estimating the sensitivity of species to climate change. In the Antarctic, such efforts are based on the concept of oxygen- and capacity-limited thermal tolerance (OCLTT; Pörtner, 2006). This concept integrates molecular, cellular, whole organism and ecological characteristics with responses to climate change (Pörtner, 2001, 2002, 2010; Pörtner and Knust, 2007). In general, polar marine ectotherms ('cold-blooded' animals) minimize baseline energy costs in order to maximize growth potential in the cold, especially in the Antarctic. Similar patterns can also be seen in some ectotherms from the High Arctic (e.g., crustaceans, Walther et al., 2010) and sub-Arctic. The temperature variability that characterizes large sections of the sub-Arctic Ocean causes elevated levels of energy turnover in the cold-eurythermal fauna (animals that can tolerate a wide range of temperatures). This elevation can be seen in species that extend from temperate to Arctic regions (e.g., Atlantic cod, Gadus morhua) compared to temperate zone conspecifics (Pörtner and Farrell, 2008). Elevated baseline metabolic costs in cold-adapted eurytherms have been related to their broad thermal tolerance to cold. True polar characteristics as seen in animals from the Antarctic or High Arctic only develop under highly stable, extremely low temperatures around and below 0 °C. Low-energy costs of maintenance at low temperatures are also mirrored in low turnover rates of transmembrane ion exchange and in a lower capacity for acid-base regulation than seen in temperate to warm-water species (Pörtner, 2010). Lifestage and lifestyle dependent capacity for acid-base regulation has been suggested as an over-arching principle that shapes sensitivity to ocean acidification (Pörtner, 2008; Melzner et al., 2009a). Pre-adaptation to climatic regimes will thus influence these patterns, with a decreasing capacity for ion and acid-base regulation in polar species.

Insight into the biochemistry of cold adaptation in Arctic versus Antarctic species confirms these generalizations. Low metabolic rates found in Antarctic marine stenotherms (organisms that only function well within a narrow range of temperatures) contrast with the more costly patterns of metabolic regulation at elevated rates in cold-adapted (sub-Arctic) eurytherms. In light of very limited data, these principles would support the extrapolation that sub-Arctic eurytherms are likely to be more resistant to ocean acidification while Antarctic and High Arctic stenotherms may be more sensitive. Whether calcification processes are also more resistant to ocean acidification in eurythermal Arctic than in stenothermal Antarctic and Arctic species requires further investigation.

Various other environmental factors, such as CO₂ or pollution, have been suggested to constrain aerobic performance and fitness and to thereby narrow the thermal window (Pörtner and Farrell, 2008). They thus act as additional stressors that interact with extreme temperatures. By narrowing the window of aerobic scope, CO₂ elicits passive rather than active tolerance to thermal extremes. The extent to which species are able to acclimate to such challenges over long time scales is as yet unexplored, however recent examples from laboratory and field studies confirm the hypothesis of a reduced thermal window under elevated CO₂ for crustaceans (Metzger et al., 2007; Walther et al., 2009; Findlay et al., 2010b), fishes (Munday et al., 2009a), and corals (via enhanced bleaching; Anthony et al., 2008). Pollutants can also enhance thermal sensitivity (Lannig et al., 2008). These effects, while most likely representing unifying principles, remain to be quantified in Arctic species.

3.3 Arctic marine food webs

The Arctic Ocean is characterized by marine ecosystems in which a small number of taxa channel a large percentage of the energy from zooplankton to vertebrate predators including seabirds, seals, whales, and polar bears (*Ursus maritimus*). As such, the effects of ocean acidification on these ecosystems will depend greatly – and far more so than for ecosystems with a more complex food web – upon the susceptibility of the keystone taxa to ocean acidification and other stressors.

3.3.1 Arctic habitats and biogeography

The Arctic can be broadly divided into High Arctic, Low Arctic, and sub-Arctic (Dunbar, 1968; see also Chapter 1, Figure 1.4). The High Arctic corresponds to areas of the central Arctic Ocean with more or less permanent (multi-year) ice cover and persistently cold water, while the Low Arctic corresponds broadly to the zone with seasonal ice cover on the shelves surrounding the Arctic Ocean basin. The sub-Arctic zone is comprised of the adjacent areas which have little sea ice but relatively low winter temperatures. The Arctic Ocean basins (Canada and Eurasian basins separated by the Lomonosov Ridge) are connected to the deep basins of the Nordic Seas through the deep Fram Strait. The basin of Baffin Bay is separated from the Arctic Ocean by passages through the Canadian Arctic Archipelago with sill depths of about 150 m. The Bering Sea basin is separated from the Arctic Ocean by roughly 1000 km of very shallow areas, mostly <50 m deep.

The Arctic Ocean is in some sense a warm ocean with its deeper waters filled with Atlantic water from about 200–1000 m depths and temperatures of 0–3 °C. However, at the surface, cooling and formation of sea ice lead to seawater at near freezing point under the ice. This cold water may extend to the bottom of shallow shelf areas (100–200 m depths). When the sea ice melts in summer, the lighter meltwater floats as a buoyant upper layer and shields the deeper layer from seasonal warming (also see Chapter 2, Section 2.3.1). Cold Arctic water therefore prevails over shelves as a seasonally persistent layer at depths from about 25 m to 100–150 m. Shallower than this the nearshore waters may be warmed to some degree in summer, while at greater depths warmer water from the intermediate layer in the Arctic Ocean may intrude onto the outer shelf.

Persistently cold waters are found over the deeper slopes and basins of the Arctic Ocean (below about 1000 m) and over relatively shallow shelves surrounding the central Arctic Ocean, which are extensive on the Eurasian side from the northern Barents Sea to the Chukchi Sea, and also in large portions within the Canadian Arctic Archipelago. The Chukchi Sea differs from other Arctic shelf regions by the strong throughflow of Pacific water (Bering slope water) that pushes away the cold water during the summer season.

This temperature regime determines the distribution and biogeography of species. Endemic Arctic species are found mainly in the permanently cold shelf areas and in the deep basins. The number of endemic Arctic species is limited, and the majority of species in the Arctic is comprised of Arctic-boreal species or boreal species with northern ranges that extend into the Arctic. For example, of 242 marine fish species that have been recorded from the ice-covered Arctic, 25% (62 species) were classified as Arctic (or predominantly Arctic), 16% (38 species) as Arctic-boreal, while more than half (53%, 128 species) were boreal or predominantly boreal (Mecklenburg et al., 2011).

Many Arctic species have wide distributions, often circumpolar or near-circumpolar. This is the case for invertebrates as well as for fish, birds, and mammals. For invertebrates and fish, the majority of species is of Atlantic origin. For example, among the 62 species of fish classified as Arctic, 34 species are found in the Atlantic sector, six in the Pacific sector, and 29 in both. Arctic-boreal species are distributed more evenly, with 15 species in the Atlantic, 13 in the Pacific, and 13 in both sectors. The situation is very different among boreal species of marine fish if the Bering Sea is included as part of the sub-Arctic zone. For this category, the greatest number of species occurs in the Pacific sector with 286 species (out of 375), compared to 85 species in the Atlantic sector. Only four species are identified as amphiboreal (occurring in both Atlantic and Pacific sectors). However, there are many examples of closely related species in the two sectors, for example, twin Atlantic and Pacific species of cod (*Gadus morhua* and *G. macrocephalus*), halibut (*Hippoglossus hippoglosus* and *H. stenolepis*), and herring (*Clupea harengus* and *C. pallasii*) as examples among commercially (and ecologically) important species.

3.3.2 Arctic ecosystems and food webs

Broad species distributions, many closely related species, and similar physical regimes create a large element of commonality in the structure and function of Arctic marine ecosystems. However, marked differences in sea ice, bathymetry, and oceanography also lead to pronounced zonal patterns between the High, Low and sub-Arctic.

Very strong seasonal changes in light and ice conditions between summer and winter are key features that are reflected in the patterns of primary production and its utilization in the food web. The extensive migrations of birds and mammals into the Arctic to breed and feed during the short and hectic Arctic summer benefit from this production: the Arctic provides major breeding - and staging - grounds for most global populations of shorebirds and geese and many populations of ducks (e.g., eiders and scoters) and seabirds (e.g., Arctic tern Sterna paradisaea, gulls, and jaegers or skuas). Large populations of many species such as auk and eider migrate between breeding areas in the High or Low Arctic and wintering areas in the sub-Arctic or boreal zones. The three ice-associated whales, bowhead (Balaena mysticetus), beluga (Delphinapterus leucas), and narwhal (Monodon monoceros), also have migratory populations that move into the High Arctic to feed and nurse their calves in summer and back to wintering areas in the southern extent of drift ice. Walrus and seals (e.g., harp seal Pagophilus groenlandicus and hooded seal Cystophora cristata) undertake similar seasonal migrations. At lower latitudes many species of birds and mammals use sub-Arctic seas for breeding and feeding in summer, while spending the winter at lower latitudes (e.g., northern fur seals Callorhinus ursinus and humpback whales Megaptera novaeangliae).

Although birds and mammals are not likely to be affected by ocean acidification directly, they might be through the food web if their forage base is affected.

Arctic marine food webs include food chains of different types and lengths, characterized by their basis in the water column, ice, and seabed, respectively. Phytoplankton comprise the dominant primary producers in Arctic marine ecosystems, however ice algae also contribute to production – especially in High Arctic areas with multiannual ice. Benthic macroalgae such as kelp do occur in Arctic waters, but ice scour, suitable substrate, and high turbidity from large Arctic rivers limit their growth. Primary production spans two orders of magnitude over the Arctic area, with maxima of $> 500 \text{ g C/m}^2/\text{y}$ in the northern Bering Sea and southern Chukchi Sea, and minima of $< 5 \text{ g C/m}^2/\text{y}$ in the central Arctic Ocean under dense pack-ice. The very high production in the northern Bering Sea reflects the northward flow of nutrient-rich North Pacific water in what has been described as a horizontal upwelling system (Sambrotto et al., 1984; Walsh et al., 1989; Springer and McRoy, 1993; Springer et al., 1996). Nutrients are not fully depleted en route through the Bering Strait, and the Pacific water that forms a layer in the upper halocline in the Canada Basin is still nutrient-rich compared to Atlantic water in the Arctic Ocean.

Ice limits light for photosynthesis and thereby limits the length of the season of primary production, although perhaps not as much as previously thought (e.g., Fortier et al., 2002; Mundy et al., 2009; Arrigo et al., 2012). Melting ice forms a surface layer of low salinity water, which effectively restricts the transport of nutrient-rich water from deeper layers, although upwelling events and wind-induced mixing can break down surface stratification and thus foster blooms (Brugel et al., 2009; Tremblay et al., 2011). This leads to a distinct seasonal and vertical pattern in the distribution and production of phytoplankton in ice-covered Arctic seas (Sakshaug and Skjoldal, 1989; Sakshaug, 2004). The annual primary production on the Arctic shelves with seasonal ice cover is typically rather low, in the range of 20–50 g C/m², while production in the sub-Arctic seas in the North Atlantic sector is typically 100–150 g C/m².

The growth of ice algae is strongly light limited: 2 m of sea ice with some snow cover transmits 1% or less of the incoming sunlight to the surface (Bergmann et al., 1991; Cota et al., 1991; Mundy et al., 2005). Ice algae are adapted to low light and start to grow in spring (April) at these low light levels (Michel et al., 1988; Smith et al., 1988; Bergmann et al., 1991). Successive accumulation of algal biomass absorbs more of the transmitted light, limiting further growth of ice algae (and underlying phytoplankton) through self-shading (Smith et al., 1988; Welch and Bergmann, 1989). The onset of melting improves light conditions, but ice algae are also sloughed from the underside of the ice by the meltwater. Extensive studies in the Canadian Arctic have found annual production of ice algae in the range 1–20 g C/m², with a typical mean value of 5 g C/m² (Smith et al., 1988; Bergmann et al., 1991; Welch et al., 1992; Michel et al., 2006). After the culmination of the icealgae bloom, phytoplankton dominate and usually far exceed the ice algae in terms of annual primary production (Welch et al., 1992; Michel et al., 2006).

While heterotrophic and mixotrophic protists, such as ciliates and dinoflagellates, are important grazers in marine Arctic waters (Levinsen et al., 2000), the predominant herbivores in Arctic marine ecosystems are large calanoid copepods. Three species of *Calanus (C. finmarchicus, C. hyperboreus,* and *C. glacialis)* are important in the Atlantic sector (Conover, 1988; Melle et al., 2004). *Calanus finmarchicus* dominates the sub-Arctic seas (Norwegian and Labrador seas), while *C. hyperboreus* is dominant in Baffin Bay, the Greenland Sea, and the central Arctic Ocean (Østvedt, 1955; Dawson, 1978; Hirche, 1997; Melle and Skjoldal, 1998; Sundby, 2000). *Calanus glacialis* is a shelf species with a circumpolar distribution extending to the Chukchi and northern Bering seas (Mathisen et al., 1996; Melle and Skjoldal, 1998). *Neocalanus* species (*N. cristatus, N. plumchrus*, and *N. flemingeri*) are dominant herbivores in the Bering Sea basin and are advected with the currents up through the Bering Strait region into the Chukchi Sea (Conover, 1988; Springer et al., 1989, 1996; Mathisen et al., 1996). Pteropods also play important roles in Arctic food webs and *Limacina helicina* is a common species in Arctic surface waters after the spring (ice-edge) bloom and is a food source for several species of fish.

The omnivorous euphausids or krill are important components in sub-Arctic seas and form important prey for fish and whales. *Thysanoessa inermis* and *T. raschii* have amphiboreal distributions and are found both in the Bering Sea and the North Atlantic (Mauchline, 1980). These species are advected into the Low Arctic where they can play important roles in food webs (e.g., for bowhead whales, Lowry et al., 2004; Ashjian et al., 2010). Pelagic amphipods also play important roles in Arctic food webs, feeding on small copepods and other zooplankton prey. *Themisto libellula* can grow to 5–6 cm and is perhaps the single most important link between zooplankton and higher trophic-level consumers including many seabirds and seals (Dunbar, 1946; Welch et al., 1992; Melle et al., 2004).

Several small fish species play a similar linking role to higher trophic levels. Arctic cod (*Arctogadus glacialis*) can be an abundant pelagic species in the central Arctic Ocean (Walters, 1961; Aschan et al., 2009). There is little ecological knowledge about this species, which may fall prey to migratory stocks of beluga (e.g., Suydam et al., 2001, 2005). Polar cod (*Boreogadus saida* – termed 'Arctic cod' in North America) has a circumpolar distribution mainly in the Low Arctic. It probably exists with several large migratory (Welch et al., 1992, 1993; Gjøsæter, 1998; Benoit et al., 2008) populations. Polar cod is a pillar of Arctic marine ecosystems, where it constitutes a major prey for many piscivores. Capelin (*Mallotus villosus*) has a similar distribution (Stergiou, 1989; Vilhjalmsson, 1994; Gjøsæter, 1998; Dodson et al., 2007; Praebel et al., 2008; Mecklenburg et al., 2011).

The extensive pelagic and sea-ice associated food webs of Arctic ecosystems are based upon phytoplankton and herbivorous zooplankton. Large copepods, notably Calanus hyperboreus and C. glacialis, are major prey for bowhead whales, which also consume other crustaceans such as amphipods and epibenthic mysids. In sub-Arctic waters, krill are an important food for other baleen whales such as fin (Balaenoptera physalus), humpback, and minke (B. acutorostrata) whales that also eat small fish such as capelin. Beluga and narwhal dive deep and feed mainly on fish and squid. Beluga eats a wide variety of prey (both pelagic and benthic) but probably targets polar cod as its main prey for their seasonal migrations into the Arctic (Bradstreet et al., 1986; Frost and Lowry, 1990; Welch et al., 1992, 1993; O'Corry-Crowe, 2002, 2009). Narwhal also eats polar cod (and Arctic cod) but is considered to be a specialist to feed on deep-water prey including Greenland halibut (Reinhardtius hippoglossoides) and the squid Gonatus fabricii (Laidre and Heide-Jorgensen, 2005). Narwhal is perhaps the only species of Arctic mammal for which winter feeding is more important than summer feeding: satellite-tracking has revealed that they dive 15-20 times a day to depths of around 1000 m or more, for a duration of 5–6 months during winter (Laidre et al., 2003, 2004,a,b; Dietz et al., 2008).

Pelagic crustaceans such as Themisto libellula, and small fish (capelin and polar cod) are the primary prey for large stocks of harp seal, and ringed seal (Pusa hispida) - the most strongly ice-associated of the seals and the primary prey of polar bears. In contrast, hooded seal dives deep to feed on squid (mainly Gonatus fabricii) and various deep-water fish (e.g., Sebastes spp. and Greenland halibut). Bearded seal (Erignathus barbatus), also an ice-associated seal, mainly feeds on benthic invertebrates, as does the walrus (Odobenus rosmarus) that specializes on burrowing bivalves (e.g., Mya truncata, Serripes groenlandicus, and Hiatella arctica). Pacific walrus (subsp. O. r. divergens) is the most numerous subspecies (~ 200 000 individuals) and supported by the extensive productive shallow benthic habitats of the Chukchi Sea. Declining summer ice has deprived the walruses of haul-out platforms, forcing them to change to coastal haul-outs, which has reduced their foraging opportunities. Atlantic walrus (subsp. O. r. rosmarus) number around 20 000-25 000 individuals. Their lower numbers reflect strong overexploitation in the past as well as less productive and less extensive feeding grounds in the Atlantic sector of the Arctic. Historically, the largest population of Atlantic walrus was probably found in the Gulf of St. Lawrence, where they were exterminated by hunting.

Many seabirds feed on ice, in the water, and in the benthos. The High Arctic Ivory gull (*Pagophila eburnea*) and Ross's gull (*Rhodostethia rosea*) feed on pelagic and ice-associated crustaceans, small fish (notably polar cod), and carcasses, feces, and other remnants of marine mammals including polar bear kills. Glaucous gull (*Larus hyperboreus*) is also a generalist predator and scavenger on a wide range of prey including bird eggs and chicks, polar cod, capelin, fish roe, mussels, sea urchins, and other invertebrates in the Arctic. In the Lowand sub-Arctic, black-legged kittiwake (*Rissa tridactyla*) and northern fulmar (*Fulmarus glacialis*) are two abundant feeders on invertebrates and small fish at or close to the sea surface.

Auks - notably dovekie (or little auk, Alle alle) and various species of auklet (least auklet, Aethia pusilla, crested auklet, A. cristatella, and others) - are common plankton feeders and among the most numerous of all seabirds. Their main prey is large calanoid copepods. The larger common and thick-billed murres (Uria aalgae and U. lomvia) are auks that occur in great numbers at large breeding colonies in Low Arctic and sub-Arctic waters, where there is a combination of cliffs and a rich supply of large zooplankton (krill and amphipods) and small fish (e.g., polar cod and capelin). Murres are conspicuous components of Arctic food webs and have several different subspecies (three to four). In sub-Arctic areas the Atlantic puffin (Fratercula arctica) and the Pacific horned puffin (F. corniculata) are abundant; breeding where there is an abundant supply of small fish such as herring and capelin. Black guillemot (Cepphus grylle) is a dispersed generalist species feeding on a variety of small demersal and pelagic fishes and invertebrates, including polar cod, amphipods, sculpins, blennies, and others. In sub-Arctic and Low Arctic coastal environments eiders are abundant and conspicuous. Common eider (Somateria mollissima) and king eider (S. spectabilis) have circumpolar distributions, with the latter being more common in the Arctic whereas several subspecies of common eider occur in boreal and sub-Arctic waters. Spectacled eider (S. fischeri) and Steller's eider (Polysticta stelleri) are found in the Pacific sector (Steller's eider also breeds in western Siberia). Eiders dive in relatively shallow water (to depths of 30-60 m), where they feed on benthic invertebrates, especially mollusks (for common and spectacled eiders), but also crustaceans and echinoderms. In contrast, Steller's eider specializes on feeding in shallow inshore eelgrass (Zostera) or algal beds. Perhaps most remarkably, long-tailed duck (Clangula hyemalis), another abundant circumpolar species, is considered to be the deepest diving waterfowl feeding on various benthic invertebrates, predominantly crustaceans and mollusks at depths over 60 m. Less important species such as scoters (black, white-winged, and surf scoters - Melanitta nigra, M. fusca and M. perspicillata) and mergansers (common and red-breasted mergansers - Mergus merganser and M. serrator), forage mainly in shallow nearshore waters, where scoters eat clams, mussels, periwinkles, and other snails, as well as crabs and shrimps. Mergansers eat mainly small fish in freshwater, estuarine, and coastal habitats.

3.4 Taxon-specific responses to ocean acidification

Information about Arctic species or ecosystem processes is limited. Therefore, the material presented in this section is mainly drawn from non-Arctic studies under the broad assumption that many of these processes will be similar in the Arctic. Each section begins with coverage of what is known from non-Arctic species and is followed by an assessment of what this means for Arctic species.

3.4.1 Effects of ocean acidification on viruses

3.4.1.1 Marine viruses (non-Arctic studies)

There are no studies that definitively show direct effects of ocean acidification on marine viruses. However, since the life cycle and persistence of viruses are directly linked to their hosts, viral processes such as lytic events may be indirectly influenced by ocean acidification. Elevated pCO_2 has been reported to have no effect (Rochelle-Newall et al., 2004), a positive effect, and a negative effect on viral abundance (see review by Danovaro et al., 2011). The effects of ocean acidification on viral infection rates are equally unknown, although several studies indicate that elevated pCO_2 can compromise immune systems of marine invertebrates (e.g., Hernroth et al., 2011; Munari et al., 2012).

3.4.1.2 Polar/Arctic marine viruses: Marine viruses associated with sea ice

Very few data are available on marine viruses associated with sea ice (Maranger et al., 1994; Gowing et al., 2002, 2004; Gowing, 2003; Wells and Deming, 2006). Viral concentrations in sea ice can be 10- to 100-fold greater than concentrations in the underlying water column (Maranger et al., 1994; Gowing et al., 2004), however, the phylogenetic diversity of sea-ice viral communities is unknown at present. Two studies have isolated bacterial viruses from sea ice, both of which would be considered cosmopolitan in their distribution (Borriss et al., 2003; Wells and Deming, 2006), but data on eukaryotic seaice-inhabiting microbes are totally lacking as is their response to ocean acidification.

3.4.2 Effects of ocean acidification on bacteria

3.4.2.1 Bacteria and bacterial processes (non-Arctic studies)

In some studies, bacterial communities responded indirectly to ocean acidification *via* direct effects of acidification on phytoplankton abundance, production, and calcification (McDonald et al., 2009; Liu et al., 2010). Consequently, acidification-induced increases in phytoplankton dissolved organic matter (DOM) production (either by extracellular secretion and/ or grazing by protozoa) may enhance biogeochemical activities in some bacterial communities (although it should be noted that phytoplankton responses to elevated pCO_2 are not consistent between studies – see Section 3.4.3).

Although the bacterial community does not appear to be strongly affected by pCO_2 , elevated pCO_2 generally increases bacterial productivity (Grossart et al., 2006; Allgaier et al., 2008), whereas bacterial abundance does not change with pCO_2 (Allgaier et al., 2008; Paulino et al., 2008). Variable responses to elevated pCO_2 have been reported for extracellular enzyme activity (Grossart et al., 2006; Piontek et al., 2010; Yamada and Suzumura, 2010) and bacterial diversity (reviewed by Liu et al., 2010).

Some special functions of the microbial food web might be affected by ocean acidification, with potential consequences for ocean biogeochemistry. Note also that the activities in the microbial food web have effects on the pH of seawater through processes such as photosynthetic fixation of CO₂ (increasing pH), microbial degradation of organic material to CO, (decreasing pH), and calcite formation (reducing alkalinity). Global-change phenomena occurring in the Arctic that may affect photosynthesis (increased water-column stability, nutrient runoff, improved light conditions due to reduced ice cover, etc.), bacterial activity (increased influx of dissolved organic carbon to the Arctic Ocean from Russian and Canadian rivers, increased denitrification in areas with reduced oxygen content, etc.), or calcification (e.g., changes in the pattern of Emiliania huxleyi blooms due to changes in temperature and currents) will therefore, through the associated microbial processes, modify the geographical and temporal patterns in pH variation created by atmospheric CO₂ (see Chapter 2).

The two microbial processes that have received particular attention in relation to changes in pCO_2 and pH are (1) 'C-overconsumption', whereby more CO_2 is fixed per nitrogen consumed at high pCO_2 , i.e., changing the 'Redfield ratio' of C-fixation, and (2) calcification, where a reduced pH lowers the saturation level for calcite and aragonite and, therefore, makes seawater corrosive to calcium carbonate (CaCO₃). The latter process is covered in detail in Chapter 2, Section 2.3.5.

Increased pCO_2 may stimulate carbon fixation and thus potentially increase C:N (e.g., Leonardos and Geider, 2005). The C:N-ratio of material exported from the photic zone is crucial to the efficiency of the biological C-pump, transferring carbon from the atmosphere to the oceanic interior. Even a small increase in C:N may be important since it could constitute a negative biofeedback where the efficiency of the biological pump increases as atmospheric CO₂ increases. A C:N = 106:16 = 6.6 (molar) stoichiometric coupling in photosynthesis, as implied by the Redfield ratio, is an average value from which large deviations are observed when examining different phytoplankton species and different growth conditions (Geider and La Roche, 2002), implying a rather loose coupling between the acquisitionincorporation processes for carbon and nitrogen. The most convincing data sets indicating C-overconsumption are from the near-natural systems studied in mesocosms. In a study from Raunefjorden in western Norway, Riebesell et al. (2007) quantified the effect to be an increase in C:N from 6 to 8 as manipulated atmospheric CO₂ increased from 350 to 1050 ppm. How this extra organic carbon is partitioned within the system is not well known, but at least parts seem to be directed to the pool of transparent exopolymeric particles (TEP). Models of the C-overconsumption effect have been constructed that combine an increase in particulate organic carbon (POC) with TEP production (Schartau et al., 2007).

Production of TEP has often been associated with diatom growth, but the primary phytoplankton response to an experimental increase in CO₂ seems to be in the picoplanktonic size fraction with very small cells (< 2 μ m) and is consistent with what was observed in a mesocosm experiment in Kongsfjorden, Svalbard, in which the first of a succession of three consecutive blooms exhibited a stimulation of the picoplankton size fraction (Meakin and Wyman, 2011). A phytoplankton-bacteria relationship that has received a lot of attention is the phytoplankton production of the osmolyte DMSP (dimethylsulfoniopropionate) apparently sensitive to acidification, but in a manner variable between species (Spielmeyer and Pohnert, 2012). Mesocosm studies have only revealed small, although statistically significant effects of ocean acidification on DMS (dimethylsulfide) concentration (Vogt et al., 2008). Since bacteria subsequently consume DMSP, probably including the numerically often-dominant SAR11 clade (Reisch et al., 2011), interactions leading to changes in species composition could be possible.

Mesocosm studies in Spitsbergen found few or no substantive effects of elevated pCO_2 on bacterial community composition (Roy et al., 2013).

3.4.2.2 Polar/Arctic bacteria: Bacteria and sea ice

Polar sea ice supports high bacterial productivity and diversity, making this one of the most significant ecosystems in the Arctic and Antarctic (Garrison et al., 1986; Grossmann and Dieckmann, 1994; Amon et al., 2001). Although significant differences in the physics and chemistry of Arctic and Antarctic sea ice cause differences in DOM, particulate organic material (POM), and nutrients, bacterial diversities in Arctic and Antarctic sea ice are similar (Brown and Bowman, 2001; Brinkmeyer et al., 2003; Junge et al., 2004). Bacterial community diversity within sea ice is highly correlated with the diversity in the underlying seawater (Bowman et al., 2012). The response of sea ice bacteria to ocean acidification is unknown.

3.4.3 Effects of ocean acidification on phytoplankton

3.4.3.1 Phytoplankton (non-Arctic studies)

No consistent response to elevated pCO_2 has been observed for marine phytoplankton cell-division rates (Riebesell and Tortell, 2011). For example, ocean acidification has positive effects on division rates in diatoms (e.g., Gervais and Riebesell, 2001), dinoflagellates (Burkhardt et al., 1999), some coccolithophores (Shi et al., 2009; Rickaby et al., 2010), and some cyanobacteria (Barcelos e Ramos et al., 2007; Hutchins et al., 2007); negative (Iglesias-Rodriguez et al., 2008; Langer et al., 2009; Muller et al., 2010) or neutral (Buitenhuis et al., 1999; Zondervan et al., 2002) effects in some coccolithophores; and negative effects in some cyanobacteria (Czerny et al., 2009). Similar variability in response to ocean acidification has been noted in photosynthetic and carbon-fixation rates of marine phytoplankton (Riebesell and Tortell, 2011). Some recent papers concluded that a number of marine phytoplankton are resilient to lowered pH and altered carbonate chemistry (see Hansen et al., 2007; Nielsen et al., 2010, 2011; Riebesell and Tortell, 2011; Renaud et al., 2012). These species would, therefore, be at a competitive advantage over those that are impacted by ocean acidification.

Increasing pCO_2 promotes nitrogen fixation in some cyanobacteria (Barcelos e Ramos et al., 2007; Hutchins et al., 2007; Levitan et al., 2007) and causes a decline in iron availability to phytoplankton (Shi et al., 2010). Ocean acidification can directly influence phytoplankton population community composition (Tortell et al., 2008), and indirectly affect community structure, nutrient utilization, and productivity through nutrient stoichiometry (Blackford, 2010).

Varying responses in calcification in phytoplankton have been a centre of debate in ocean acidification research. Most calcifying phytoplankton show decreased calcification in response to elevated CO_2 / reduced pH. However, some coccolithophores increase their calcification at higher CO_2 whereas others show no response or a non-uniform trend (e.g., Riebesell et al., 2000; Engel et al., 2005; Iglesias-Rodriguez et al., 2008; Barcelos e Ramos et al., 2007; Muller et al., 2010; Fiorini et al., 2011; Raven, 2011).

3.4.3.2 Arctic phytoplankton

Increased pCO_2 has mild negative effects on growth rate in some sea-ice diatoms (Torstensson et al., 2012), however data on other species are lacking. Coccolithophore populations are at risk, although significant gaps exist regarding the contribution of species other than its most abundant and best studied representative, *Emiliania huxleyi*, to polar populations. Non-bloom coccolithophore populations are known to play a major role in carbon export in the subtropics (Sarmiento et al., 2002), but controls on non-bloom populations in polar regions and their contribution to the carbon cycle remain unknown.

More generally, satellite data show that peak phytoplankton blooms now arrive up to 50 days earlier than in 1997 (Kahru et al., 2011). The community composition of these blooms has also shown measurable trends, with recent shifts toward dominance by smaller phytoplankton. Similar trends toward more picophytoplankton and fewer nanophytoplankton have also been associated with ocean acidification (Newbold et al., 2012), and with a freshening of the Arctic Ocean and reduction in nutrient supply (Li et al., 2009). These shifts may not cause substantial changes in total production but are likely to have broader effects on the food web. For example, the smaller size of primary producers may reduce transfers of carbon up the food chain.

3.4.4 Effects of ocean acidification on foraminifera

3.4.4.1 Foraminifera (non-Arctic studies)

Planktonic foraminifera can contribute up to 50% of the total carbonate in the ocean sediment, playing an important role in the carbonate pump (Schiebel, 2002).

Like most actively calcifying taxa, calcareous foraminifera construct their tests internally (Lowenstam and Weiner, 1989). Increasing pCO_2 generally results in less massive tests for calcareous species (Spero et al., 1997; Kuroyanagi et al., 2009; Dias et al., 2010). Survival of foraminifera around seafloor CO_2 -injection cylinders is generally unaffected by moderate (<0.2 pH units) acidification, but extreme reductions (2.0 pH units) cause calcareous forms to disappear entirely while thecate (organic) and agglutinated forms remain unaffected (Dias et al., 2010). These patterns mirror survivorship changes seen at the Triassic-Jurassic boundary, although why that is so is unclear (Hautmann et al., 2008) and at the Paleocene-Eocene Thermal Maximum (Zachos et al., 2005).

3.4.4.2 Arctic foraminifera

Carbonate sedimentation in the Arctic is highly dependent on the presence of planktonic foraminifera, and they are responsible for the majority of calcite production (Hendry et al., 2009). In particular, a large portion of recent carbonate in the Norwegian-Greenland Sea sediments is produced by Neogloboquadrina pachyderma and Globigerina quinqueloba the dominant planktonic species (Carstens and Wefer, 1992; Carstens et al., 1997; Pflaumann et al., 1996). Arctic sedimentation rates vary greatly. On the shelves, sedimentation rates over 30 cm/ky (Polyak et al., 2009; Farmer et al., 2010) have been calculated, whereas rates in basins are less than 1 cm/ky. The planktonic:benthic ratios of foraminifera (Stehli and Creath, 1964) in Arctic shelf seas are generally low, but below 200 m can be ≤ 32 (Murray, 1991; Bergsten, 1994). However, Scott et al. (1989) found that the planktonic to benthic ratio was 1:1 in the deep Arctic where there is no organic influx.

Planktonic and benthic foraminifera typically construct tests of calcium carbonate or of agglutinated sediments, although organic- and siliceous-walled forms are found (Sen Gupta, 2002). Planktonic foraminiferal species diversity is extremely low in the Arctic; assemblages are typically dominated by the polar species *Neogloboquadrina pachyderma*, with the subpolar species *Globigerina quinqueloba*, which is also common (Carstens and Wefer, 1992; Carstens et al., 1997). A few other *Globigerinita* species are found in small (<2%) proportions (Carstens and Wefer, 1992; Arnold and Parker, 1999). Although *N. pachyderma* and *G. quinqueloba* can be found as deep as 200 m, the highest abundances of these species are in the upper 50 m of the water column in northernmost regions (above 83° N), whereas in southern areas they are found primarily below the pycnocline (Carstens and Wefer, 1992).

Benthic foraminiferal species diversity in the Arctic is low, in part because of its oceanographic isolation. Very small forms (mean test size 79 μ m) dominate (Tomanek et al., 2011) in icecovered regions as well as in some other areas (Michaelidis et al., 2005). Agglutinated foraminifera dominate deeper waters below the calcium carbonate compensation depth (Scott and Vilks, 1991), whereas calcareous species are typically found at shallower depths in cold water. Some calcareous forms, predominately hyaline rotalid foraminiferan tests, can also be found below 1000 m (Lagoe, 1976).

There is currently no evidence that calcite undersaturation impacts recent for aminiferal faunas in eastern and central Arctic bottom waters (Bergsten, 1994) or bathyal Holocene sediments in the Sea of Okhotsk (Barash et al., 2008). However, Gregory et al. (2010) attributed dissolution of near-surface for aminifera to changes in water-column saturation. Moy et al. (2009) also suggested that a 30–35% reduction in shell weight of present-day Southern Ocean *Globigerina bulloides* (relative to pre-industrial Holocene samples) was due to increased pCO_2 . Manno et al. (2012a) found no effect of pCO_2 (0.3 pH unit reduction) on survival of *Neogloboquadrina pachyderma*, but shell diameters were significantly smaller under acidification.

Limited evidence indicates that a ~ 0.2 pH unit decline will have few if any effects on survival of foraminifera, but may influence growth rates. Larger declines in pH are likely to reduce the diversity and survival of calcifying species, and hence, decrease sedimentation of inorganic carbon. The broader ecosystem consequences of increased dominance by non-calcifying species are unknown.

3.4.5 Effects of ocean acidification on macroalgae

3.4.5.1 Macroalgae (non-Arctic studies)

Temperate and tropical non-calcifying macroalgae typically show increases in growth and photosynthesis with high pCO₂ (Borowitzka and Larkum, 1976; Hall-Spencer et al., 2008; Connell and Russell, 2010). Calcifying macroalgae, in contrast, show significant reductions in recruitment and growth under elevated pCO₂ (Hall-Spencer et al., 2008; Kuffner et al., 2008; Martin et al., 2008; Gao and Zheng, 2010) and reductions (Martin and Gattuso, 2009; Ries et al., 2009; Robbins et al., 2009) or non-linear responses (Borowitzka and Larkum, 1976; Anderson et al., 2011a; Price et al., 2011) of net calcification. At the same time, photosynthetic rate of calcifiers typically increases as a function of increasing pCO_{2} (Borowitzka and Larkum, 1976; Anderson et al., 2011a; Price et al., 2011). Magnesium content of high-Mg calcite secreted by coralline red algae declined with increasing pCO_2 (Ries, 2010), indicating that coralline red algae may secrete a range

of high-Mg calcites, and that higher-Mg phases preferentially dissolve away under more acidic conditions. The very scant available data indicate growth of calcifying macroalgae may be compromised under future ocean acidification, whereas most non-calcifiers will probably benefit – directly from increased pCO_2 and perhaps indirectly from reduced competition from encrusting calcifying species.

3.4.5.2 Polar/Arctic macroalgae

Studies of polar macroalgae are lacking. Of the few Arctic species that have been studied, non-calcifying red algae (which typically come from habitats with lower natural pH oscillations than green and brown algae) respond positively to higher seawater pCO_2 (Kubler et al., 1999; Gattuso and Hansson, 2011).

Calcified macroalgae may be vulnerable to extreme changes in pH because many polar species are typically weakly calcified. However, data for living Arctic species are lacking. A recent study of temperate and tropical brown macroalgae indicate that in at least in one genus, *Padina*, lower pH conditions may still allow calcification due to increased photosynthesis (Johnson et al., 2012). Studies of post-mortem dissolution rates of the thallus of encrusting Antarctic coralline algae showed that rates were greater under acidification than non-acidification conditions (McClintock et al., 2011).

3.4.6 Effects of ocean acidification on corals

3.4.6.1 Tropical corals

Globally, corals are important calcifiers, and their aragonite skeletons and reefs provide shelter and habitat for a great diversity of other species. Calcification rates of tropical corals are linearly related to $\Omega_{_{\mathrm{Ar}}}$ of the surrounding seawater (Langdon and Atkinson, 2005). Consequently, CO2-induced reductions in Ω_{Ar} are projected to have strong negative effects on corals and coral reefs (Hoegh-Guldberg et al., 2007). Corals calcify internally, and pH within the calcifying polyp varies diurnally by up to 1.8 units (Al-Horani et al., 2003). Nonetheless, acidification to pHs well above daily minima observed within the polyp causes $a \le 50\%$ reduction in calcification and growth rates (Schneider and Erez, 2006; Kroeker et al., 2010, 2013; Hofmann et al., 2011). Effects of elevated pCO₂ on earlier life stages of corals are less well investigated, but available data indicate negative effects on fertilization, larval metamorphosis, and juvenile growth (Albright et al., 2010; Nakamura et al., 2011). The generally negative responses outlined are not universal, and a few studies show adult corals can calcify at very low Ω_{Ar} levels (reviewed by Hofmann et al., 2010; Form and Riebesell, 2012). This ability may be related to high nutritional status.

3.4.6.2 Polar/Arctic corals

Cold-water coral reef-like structures or mounds (formed by scleractinian corals) and coral gardens (mostly gorgonians) provide important habitats for many organisms including commercially important fish (Kutti and Fosså, 2009). Model projections suggest that about 70% of cold-water corals, and especially those at higher latitudes, will be in undersaturated $(\Omega_{Ar} < 1)$ waters by the year 2100 (e.g., Turley et al., 2007; Chapter 2, Section 2.6.3). Although Dullo et al. (2008) suggested that Lophelia may occur in a temperature-salinity niche, very little is known about the distribution of corals in the Arctic, or about possible effects of acidification on cold-water corals. Some data are available for Lophelia pertusa, a common reef builder in the sub-Arctic; acidification of ≤ 0.1 pH units reduced calcification rate of *L. pertusa* by $\leq 29\%$ (Maier et al., 2009; Form and Riebesell, 2012). However, longer-term experiments using gradual declines to lower pHs showed calcification was slightly enhanced over initial values, and net growth was sustained even at $\Omega_{\Lambda r} < 1$ (approx. pH \leq 7.7; Form and Riebesell, 2012). This highlights the need for long-term experiments. Importantly, older skeletons of L. pertusa and the dead reef mounds themselves are exposed to seawater and are, therefore, likely to experience dissolution at $\Omega_{Ar} < 1$. The ecological consequences of this are not yet known.

Available data indicate that ocean acidification may have limited impacts on live cold-water corals (*Lophelia*), especially if adequate food is available. However, there is currently a complete lack of knowledge about the dispersive early life stages and newly settled juveniles. Effects on dead coral structures – and hence reefs – as well as on non-*Lophelia* species, are unknown. The important role of cold-water corals as habitat for other species means this is a critical gap in knowledge.

3.4.7 Effects of ocean acidification on mollusks

3.4.7.1 Mollusks (non-Arctic studies)

Mollusks are highly diverse with over 20 000 marine species (Chapman, 2009). As calcifiers of tremendous ecological and economic importance, they are one of the key groups studied in ocean acidification research. Data are available for three classes of mollusk: bivalves (especially oysters and mussels), gastropods (notably pteropods), and cephalopods. Mollusks have highly diverse body plans, physiology, calcification mechanisms, and ecology. It is therefore not surprising that a wide range of biological responses to ocean acidification has been described, from an apparent positive effect on fitness-related parameters in a few species (e.g., reduced mortality in a bivalve; Range et al., 2011) to no effects (~50% of all tested species) or negative effects (~50% of all tested species). Ocean acidification may impact survival and growth (e.g., Talmage and Gobler, 2009, 2010), calcification (e.g., Ries et al., 2009), physiology (e.g., Melzner et al., 2011), immunology (e.g., Bibby et al., 2008), proteomics (e.g., Tomanek et al., 2011), evolutionary processes (e.g., Sunday et al., 2011), and acclimation (e.g., Parker et al., 2012). In particular, the planktonic pteropods have been relatively well studied due to their ecological importance as a food source, and due to the sensitivity of calcification of a key species, Limacina helicina, to elevated pCO_2 (Comeau et al., 2009, 2010; Lischka et al., 2011).

Acute experiments with bivalves show strong reductions in adult calcification under ocean acidification (Gazeau et al., 2007). However, longer-term incubations demonstrate that adults can calcify at Ω_{Ar} <1 if sufficient food is available

(Michaelidis et al., 2005; Tunnicliffe et al., 2009; Thomsen et al., 2010; Melzner et al., 2011). Clearly, biological control over calcification is possible even where Ω_{Ar} approaches zero, as long as the shell is protected by the organic periostracum and there is sufficient energy to fuel calcification. Early life-history stages tend to be more susceptible (Talmage and Gobler, 2009, 2010; Gazeau et al., 2010; and many others), although successful settlement and dominance of mytilid mussels and other calcifying invertebrates have been observed at seawater pCO_{2} values of 700-1000 µatm (Thomsen et al., 2010). Again, this indicates that either local adaptation and/or adequate energy supplies can also allow early life stages to overcome negative impacts of ocean acidification (see Kelly and Hofmann, 2013). Reductions in growth and calcification have typically been observed at much higher pCO₂, and may be related to elevated metabolic rates and subsequent energy budget re-allocation (Thomsen and Melzner, 2010; Melzner et al., 2011).

The coastal cuttlefish, *Sepia officinalis*, is physiologically robust to 6-week exposure to very high pCO_2 (6000 µatm), although calcification of the cuttlebone increased (Gutowska et al., 2008), which may negatively impact buoyancy regulation. Embryonic *S. officinalis* are more sensitive, showing developmental delay at elevated pCO_2 (>3500 µatm; Hu et al., 2011). Open-water species, such as the squid *Dosidicus gigas*, live in less variable pCO_2 environments and are expected to be more sensitive. *D. gigas* shows mild metabolic depression at $pCO_2 = 1000$ µatm (Rosa and Seibel, 2008). No other data for responses of cephalopods to ocean acidification are available.

3.4.7.2 Polar/Arctic mollusks

Bivalves, pteropod gastropods and cephalopods are important components of the Arctic food web (see Section 3.3.2), and consequently impacts of ocean acidification on these species may have broader emergent effects on Arctic ecosystems. The few available studies of polar/Arctic species have reported negative impacts; however, elevated pCO_2 had no impact on survivorship of the Antarctic bivalve *Laternula elliptica* over a longer exposure period of 120 days. Physiological and gene-expression patterns indicated that individuals may not be able to withstand elevated pCO_2 over longer periods (Cummings et al., 2011).

Pteropods can represent as much as 93% of the total zooplankton biomass in high-latitude regions (Hunt et al., 2008) and play a significant role in contributing to carbonate and organic carbon flux and sequestration (Falk-Petersen et al., 2002; Karnovsky et al., 2008; Manno et al., 2010). They play an important role in the food web, and are food resource for many predators such as herring, salmon, whales, and seabirds (e.g., Doney et al., 2009). Models indicate that a 10% reduction of the pteropod population would cause a 20% reduction in mature body weight of pink salmon (Oncorhynchus gorbuscha; Feely et al., 2010). Ocean acidification decreases calcification in pteropods (Comeau et al., 2009, 2010; Lischka et al., 2011), increases respiration rates (especially in combination with increased temperature; Comeau et al., 2010) and increases mortality of pre-winter juveniles (Lischka et al., 2011) and adults (Bednarsek et al., 2012; but see Comeau et al., 2012a). Although pteropods can calcify in undersaturated waters, it is possible the extra energy cost necessary to counteract shell

dissolution could seriously affect their energy budget (Lischka et al., 2011; Manno et al., 2012b). Recently, model projections of ocean acidification indicated that in much of the Arctic pteropods would be unable to precipitate calcium carbonate by the end of the 21st century (Comeau et al., 2012b).

Available data indicate that calcifying Arctic mollusks are likely to be negatively impacted by ocean acidification, and especially so at sensitive early life-history stages and/or for thin-shelled planktonic species such as the cosomatous pteropods. The effects of ocean acidification on ecologically important non-(or poorly) calcifying species such as squid are unknown, and data on these species are urgently required.

3.4.8 Effects of ocean acidification on echinoderms

3.4.8.1 Echinoderms (non-Arctic studies)

Most echinoderms calcify as adults and larvae, and consequently echinoderms are one of the primary targets for ocean acidification research (Figure 3.2). The many available studies show that echinoderm responses are generally highly species-, populationand individual-specific, ranging from likely species extinction (9% of studied species; such as the brittlestar *Ophiothrix fragilis;* Dupont et al., 2008), through direct or indirect negative effects (such as delay in development) (52% of studies), to positive effects (6% of studies; e.g., Dupont et al., 2010). It is important, however, to note that most published studies are short term (60% less than two weeks) and focus only on one life stage.

3.4.8.2 Polar/Arctic echinoderms

It has been hypothesized that echinoderms - and especially taxa with high Mg-calcite skeletons - will play the role of the 'canary in the coal mine' in polar regions (e.g., McClintock et al., 2011; Sewell and Hofmann, 2011). Available experimental data generally show negative impacts of ocean acidification on a range of processes such as growth. However, these effects do not (as yet) appear to be greater than equivalent responses in temperate and tropical species. For example, some aspects of larval development of the Antarctic sea urchin Sterechinus neumayeri are negatively affected by ocean acidification but to a lesser extent than for temperate and tropical species (Clark et al., 2009). Other reported effects of ocean acidification include: increased metabolic rates in the Arctic brittlestar Ophiocten sericeum (Wood et al., 2011); reduced ingestion and altered metabolism and immune-system responses in the boreo-Arctic urchin Strongylocentrotus dreobachiensis (Dupont and Thorndyke, 2012; Stumpp et al., 2012); developmental delay in the sub-Antarctic urchin Arbacia dufresni (Catarino et al., 2012), and reduced fertilization success in the Antarctic urchin Sterechinus neumayeri (Sewell and Hofmann, 2011).

While some reports from polar echinoderm species show neutral responses to ocean acidification, the majority of responses reported are negative. Echinoderms are key components of benthic carbon cycling in the Arctic (Renaud et al., 2007) and therefore ocean acidification-induced changes in abundances and performance of Arctic echinoderms are likely to have broader impacts on Arctic benthic ecosystems.



Figure 3.2. Brittlestars (e.g. *Amphiura filiformis*, adult (upper) and larvae (middle)) are particularly sensitive to ocean acidification, while some seastars appear to benefit from ocean acidification (e.g. *Crossaster papposus*, larvae (lower)).

3.4.9 Effects of ocean acidification on crustaceans

3.4.9.1 Crustaceans (non-Arctic studies)

Crustaceans are a highly diverse group of organisms, many of which are highly calcified, and many of which also play key roles in Arctic ecosystems. Perhaps the most ecologically important of these are the pelagic copepods, and yet few studies have investigated effects of ocean acidification on copepods. 48



Figure 3.3. The effects of ocean acidification have been investigated in several polar/Arctic crustaceans. For example, barnacles such as *Semibalanus balanoides* (upper), copepods such as *Calanus finmarchicus* (middle) and the shore crab, *Carcinus maenas* (lower). Photos: Fredrik Pleijel.

Most work has focused on effects on processes other than calcification (the exoskeleton of pelagic copepods has little, or no, calcium carbonate), and most studies have found no, or only small, responses to ocean acidification. Even exposure to very high pCO_2 levels (pH 7.4) only reduced survival in one species (of four studied), slightly reduced ($\leq 20\%$) egghatching success in one species (of three studied), and had no statistically significant effect on egg-production rate (four species) (Kurihara et al., 2004a,b; Kurihara and Ishimatsu, 2008; Mayor et al., 2007). Negative effects on larval (nauplius) survival have been reported but only at extreme levels of pCO_2 (Kurihara et al., 2004a,b). These results indicate that planktonic copepods may not be sensitive to ocean acidification at levels expected to occur within the 21st century. However, an Arctic perspective is lacking.

A number of studies have investigated responses of decapod crustaceans to ocean acidification. The balance of evidence indicates that acidification to levels projected for the year 2100 will influence decapod crustaceans substantially (Whiteley, 2011; Flores et al., 2012). Reported impacts include diverse effects on calcification; negative effects on embryonic development (e.g., Egilsdottir et al., 2009); positive, neutral, and negative effects on larvae (e.g., Arnold and Parker, 1999; Walther et al., 2010, 2011; Flores et al., 2012); and negative effects on growth rates and molting frequencies (Whiteley, 2011). Ocean acidification also affects behavioral patterns in hermit crabs (de la Haye et al., 2011) and penaeid prawns (Dissanayake and Ishimatsu, 2011). At an ecosystem level, abundance and diversity of marine benthic communities decline under ocean acidification, but only at pH \leq 7.3 (Hale et al., 2011).

The effects of ocean acidification on barnacles have been investigated in several species. Results vary, with some species showing reduced shell growth (Eliminius modestus) and calcification (Semibalanus balanoides), while others showed increased shell growth (Balanus amphitrite, B. improvisus) and calcification (*B. amphitrite*) (Findlay et al., 2009, 2010a,b; McDonald et al., 2009). Effects on barnacle embryos and larvae also vary: larval condition, cyprid size, larval settlement, and metamorphic success in *B. amphitrite* were all unaffected by very high pCO₂ levels (pH 7.4; McDonald et al., 2009), whereas less extreme levels of ocean acidification had negative effects on S. balanoides (Findlay et al., 2009). Balanus amphitrite shows changes in gene expression, with negative consequences for energy metabolism and respiration (Wong et al., 2011). Work on multiple stressors produced complex effects, for example, the effects of increasing temperature override those of acidification above a critical threshold (13 °C, S. balanoides, Findlay et al., 2010b).

3.4.9.2 Polar/Arctic crustaceans

The boreal/Arctic copepod, *Calanus finmarchicus*, is a key species in the sub-Arctic (see Figure 3.3), and although egg-production rates were unaffected by extreme pCO_2 (8000 ppm = pH 6.95), substantial reductions in egg-hatching success were observed (Mayor et al., 2007). The effects of warming-driven northward-range extension of *C. finmarchicus* (Reygondeau and Beaugrand, 2011) in combination with ocean acidification are unknown. Barnacle species in the Arctic are few and are

limited by the extent of sea ice (which scrapes them from the rocks). Semibalanus balanoides is common in boreal and icefree Arctic waters (the larvae tolerate several weeks of freezing in sea ice; Pineda et al., 2005), and the invasive Balanus improvisus has also been reported from the Arctic. The responses of these species to elevated pCO_2 (typically ~1000 µatm, pH ≈ 7.7) are mixed, but generally negative, showing small reductions in shell calcification, or compensatory calcification, which incurs increased metabolic costs (Findlay et al., 2011). Data for boreo-Arctic decapod species include the shore crab Carcinus maenas, the spider crab Hyas araneus, and the lobster Homarus gammarus. High levels of pCO₂ (pH 7.3) changed gene expression in the gills of the shore crab, although this did not translate into measurable physiological responses (Fehsenfeld et al., 2011). Similar pH drops caused reduced Ca²⁺ uptake by spider crab larvae, especially in populations from Svalbard (the northern margin of the species range; Walther et al., 2011), reduced larval development rate, reduced larval growth and reduced larval condition (Walther et al., 2010), and narrowed the thermal-tolerance window of spider crab adults (Walther et al., 2009). Decreasing pH by 0.3 units had no significant effects on survival, growth (carapace length), and development rates of larvae of the lobster Homarus gammarus, but did cause reductions in Ca2+/Mg2+ content of the larval carapace (Arnold et al., 2009). Interestingly, recent work showed that exposure to pH 7.7 for five months caused reductions in closure strength of the crusher chela (used in feeding and defense against predators) in C. maenas (Landes and Zimmer, 2012).

The available data indicate that although crustaceans in general are relatively robust to ocean acidification, polar and Arctic species show significant reductions in function, especially during the larval stages, which is likely to be translated into lower survivorship and reduced population viability.

3.4.10 Effects of ocean acidification on other invertebrates

3.4.10.1 Other invertebrates (non-Arctic studies)

Beyond the cnidarians (corals), mollusks, echinoderms, and crustaceans (see e.g., Doney et al., 2009), there is only direct evidence on the impact of near-future ocean acidification on five other invertebrate phyla: Acoela (positive effect; Dupont and Thorndyke, 2012), Annelida (negative or no effect; Widdicombe and Needham, 2007), Bryozoa (negative or no effect; e.g., Rodolfo-Metalpa et al., 2010), Nematoda (negative effect; Barry et al., 2004), and Nemertea (no effect; Ericson et al., 2010). This illustrates that little research focus (<5% of literature on invertebrates) is currently placed on those taxa that might be less vulnerable to the anticipated changes in ocean chemistry (Melzner et al., 2009a).

3.4.10.2 Polar/Arctic other invertebrates

Only one published paper considers the impact of ocean acidification on a polar (Antarctic) species. Near-future ocean acidification conditions had no significant effect on fertilization and early embryogenesis of the Nemertean *Parborlasia corrugates* (Ericson et al., 2010).

3.4.11 Effects of ocean acidification on fishes

3.4.11.1 Fishes (non-Arctic studies)

Juvenile and adult fishes have sufficient capacity and flexibility in their acid-base regulation systems to cope with the projected changes in environmental CO₂ levels over the 21st century (Pörtner, 2008; Melzner et al., 2009a,b). However, due to their high surface-to-volume ratios and less developed acid-base regulation systems, eggs and early larval stages might be more sensitive to changes in environmental CO₂ levels (Kikkawa et al., 2003; Ishimatsu et al., 2004). Recent experimental work has demonstrated direct, negative effects of high CO₂ on larval growth and survival of some marine fishes (Baumann et al., 2012; Frommel et al., 2012), but not others (Munday et al., 2009a, 2011a). In addition to growth and survival, laboratory experiments have documented effects of elevated CO₂ on other physiological processes. Rearing at elevated CO₂ concentrations (>1000 µatm CO₂) resulted in hyper-calcification of the otoliths in larval sea bass Atractoscion nobilis (pelagic spawner) (Checkley et al., 2009) and in clownfish (Amphiprion percula; benthic spawning species) reared at ~1700 µatm CO, but not at lower CO₂ concentrations more likely to be observed in the ocean within the next 50-100 years (Munday et al., 2011b).

Little is known about the effects that chronic exposure to high pCO_2 might have on fish reproduction. The few preliminary studies available indicate that impacts are small. Sperm motility of some flatfish species is arrested by mild increases in pCO_2 (Inaba et al., 2003), but similar effects were not observed in the Baltic cod, Gadus morhua (Frommel et al., 2010), or 11 other species from a range of families (Inaba et al., 2003). Sensitivity of fish eggs to elevated CO₂ varies markedly between species, but species tested to date typically have 24h LC50 (lethal concentration resulting in 50% mortality) values well above 10 000 µatm CO₂ (Ishimatsu et al., 2008). Rearing eggs of Atlantic herring in acidified water had no detectable effect on fertilization success, embryonic development, hatch rate, length and weight at hatching, and yolk size (Franke and Clemmesen, 2011). Furthermore, Munday et al. (2009b) did not detect any effect of exposure to ~1000 μ atm CO₂ on the embryonic duration or survival of clownfish eggs. Consequently, the evidence currently indicates that embryos of marine fish are tolerant of increases in oceanic pCO_2 up to the stage of hatching.

Larval stages are predicted to be more sensitive to elevated pCO_2 than adults because they have a larger surface-area-to-volume ratio, and are, therefore, more susceptible to changes in ambient conditions, but also because they might have less developed mechanisms for acid-base balance compensation (Ishimatsu et al., 2008; Melzner et al., 2009a). Although very little is known about the mechanisms and pathways of acid-base regulation in larval fishes, it is clear that they must be capable of acid-base homeostasis, at least in later ontogenetic stages (Brauner, 2008). Larval mortality of the estuarine species *Menidia beryllina* increased when exposed to near-future CO_2 levels (Baumann et al., 2012), and tissue development was disrupted in Atlantic cod reared at high CO_2 (1800 and 4200 µatm CO_2) (Frommel et al., 2012). These studies indicate that larval stages of some marine fishes may be sensitive to rising CO_2 levels in the ocean.

However, Miller et al. (2012) found that reduced growth and survival observed in juvenile anemonefishes *Amphiprion melanopus* reared at high CO_2 levels were reversed when the parents experienced the same CO_2 conditions as the juveniles. This indicates that it is premature to conclude that near-future CO_2 levels will have negative effects on the growth, development, or survival of marine fishes until more studies include exposure to high CO_2 during both the parental and offspring generations, as well as for multiple generations.

Concentrations of CO₂ reported to cause mortality in adult fish are generally an order of magnitude or more higher (>10 000 µatm) than the CO₂ levels projected for the atmosphere and shallow ocean by the end of the 21st century (up to ~1000 ppm under a business-as-usual scenario of CO₂ emissions) (Ishimatsu et al., 2008). Therefore, direct effects of rising CO_2 levels on mortality are unlikely for adults. Very high concentrations of CO₂ (>10 000 µatm) are known to affect respiration, circulation, and metabolism of some fishes (Ishimatsu et al., 2005); however, the effects of CO₂ concentrations projected to occur in the shallow ocean this century are largely unknown. One concern is that increased energetic costs of acid-base balance and/or reduced respiratory capacity may limit the scope for aerobic performance as CO₂ levels rise (Pörtner and Farrell, 2008). In one of the few studies conducted to date, aerobic scope in two tropical cardinal fishes (Ostorhinchus doederleini and O. cyanosoma) declined by >30% when they were exposed to $\sim 1000 \,\mu atm \, CO_2$ at the average summer temperature (29 °C) for the study population and at temperatures up to 3 °C above average (Munday et al., 2009a). In contrast, Atlantic cod maintained their standard and active metabolic rates, critical swimming speeds, and aerobic scope after prolonged exposure (four and 12 months) to 3080 and 5800 µatm CO₂ (Melzner et al., 2009b). These contrasting results indicate that sensitivity to elevated CO₂ varies greatly among species and possibly that aerobic performance of cool-water fishes will be more tolerant of rising pCO_2 than in warm-water fishes. Pörtner and Knust (2007) demonstrated that oxygen limitation due to rising water temperature was likely to be responsible for range contractions and population declines of North Sea eel pout (Zoarces viviparous), and rising CO₂ levels may compound this problem, although this has not yet been tested. In general, the metabolic performance of species and life stages with high-oxygen demand, such as pelagic species and pelagic larvae, are predicted to be most sensitive to elevated oceanic CO₂ levels.

A potentially serious consequence of rising pCO_2 is that it can affect sensory systems and behavior of some marine fishes (Munday et al., 2009c, 2010; Simpson et al., 2011; Domenici et al., 2012). Larval reef fish exposed to elevated CO_2 lose their ability to discriminate between ecologically important chemical cues, such as odors from different habitat types, kin and non-kin, and the smell of predators (Munday et al., 2009c; Dixson et al., 2010). Recent experiments report that the response to auditory cues is also affected (Simpson et al., 2011) and a range of behavioral problems has been detected, including the loss of behavioral lateralization (Domenici et al., 2012). Whether elevated CO_2 causes similar behavioral impairment in larvae of polar fishes is unknown, but should be a priority area for further research.

Recent experiments have demonstrated that adult reef fish also suffer impaired olfactory ability and altered behavior when exposed to elevated pCO₂, with potential effects on predatorprey interactions (Cripps et al., 2011; Ferrari et al., 2011) and homing to resting sites (Devine et al., 2012). Given that adult reef fish experience significant daily fluctuation in pH and pCO_{2} in coral reef habitat (Ohde and van Woesik, 1999; Kuffner et al., 2008), and thus might be expected to have some tolerance to higher CO₂ levels, it should be expected that behavior of adult fish in other habitats, including polar regions, will also be affected. New research shows that elevated pCO_2 alters fish behavior by interfering with brain neurotransmitter function, specifically GABA-A receptors (Nilsson et al., 2012). Given the ubiquity of GABA-A receptors in the vertebrate brain, there is good reason to expect that polar fishes might be susceptible. The ecosystem effects of impaired sensory behavior, altered predator-prey interactions, and changes in behavioral attributes are unknown, but have the potential to be significant.

Striking differences in the responses to elevated CO_2 have been detected, both within and between species. If this variation has a genetic basis, rapid selection of tolerant individuals might be expected (see Munday et al., 2013). Therefore, an understanding of how individuals respond to elevated CO_2 , and the variation in performance within local populations, is needed to make predictions about how species from all habitats and ecosystems will respond to future ocean conditions.

3.4.11.2 Polar/Arctic fishes

The effect of ocean acidification on Arctic fishes is unknown. The most applicable work to date on the direct effects of ocean acidification has focused on two commercially important (and closely related) gadoids, Atlantic cod and walleye pollock (Theragra chalcogramma). Consistent with expectations for larger-bodied fishes, experiments on juveniles of these species found that long-term exposure to elevated CO₂ (>2500 µatm) did not significantly impact growth or swimming capacity (Melzner et al., 2009b; Hurst et al., 2013). Conversely, a study rearing larval cod in mesocosms found increased incidence of developmental anomalies at high-CO₂ levels (1800 and 4200 µatm; Frommel et al., 2012). However, the physiological damage incurred by larval Atlantic cod due to high pCO₂ rapidly disappeared with the development of pH regulatory mechanisms (Frommel et al., 2012). Further, fish reared at those high-CO₂ levels (much higher than would be expected under most scenarios) did not exhibit slower overall growth or behavioral changes (Frommel et al., 2012; Maneja et al., 2013). An important caveat is that these experiments have been conducted with animals from more southerly populations at warmer temperatures, reflecting conditions in the more southerly portions of each species range. It is possible that adaptations of Arctic populations to other climate and habitat features increases sensitivity to ocean acidification or that the physiological ability of these fishes to adapt to elevated CO, may be compromised at lower temperatures when metabolic rates are constrained. This concern extends to other benthic fishes adapted to the persistently low temperatures of the Arctic (Pane and Barry, 2007). Further, whether the sensory and cognitive disruption at elevated CO₂ levels in coral reef fishes occurs in Arctic species is unknown (but see Maneja et al., 2013), as are the ecological consequences of these effects (but see Ferrari et al., 2011).

Polar cod (Boreogadus saida), a circumpolar species, channels up to 75% of the energy from zooplankton to vertebrate predators including seabirds, seals, whales, and polar bears (Bradstreet et al., 1986; Welch et al., 1992). This species typically represents the bulk (≥50%) of ichthyoplankton occurring in Arctic seas, including the Beaufort and Chukchi Seas (Chiperzak et al., 2003a,b; Parker-Stetter et al., 2011; Rand and Logerwell, 2011; Paulic and Papst, 2013; K. Suzuki, University of Laval, unpublished data). Like most marine fishes, the larvae of polar cod feed almost exclusively on copepod nauplii (Drolet et al., 1991; Michaud et al., 1996) while juveniles feed on copepodites (Renaud et al., 2012). The calanoid species Calanus spp. and Pseudocalanus spp. comprise the bulk of prey ingested by polar cod at all life-history stages (Dominique Robert and colleagues, Memorial University of Newfoundland, unpublished data). Potential impacts of increasing pCO_2 on copepods have been examined using the temperate and boreal species Acartia spp. and Calanus finmarchicus (Kurihara, 2008; Whiteley, 2011). While detrimental effects were observed on C. finmarchicus at pCO₂ levels much higher than those predicted in nature in the foreseeable future, no study has reported acidification impact on copepods under CO₂ levels projected for the next two centuries. For instance, even exposure to a high-CO, concentration of 2380 ppm did not affect the survival, growth, and development of Acartia tsuensis at all life stages through two successive generations (Kurihara and Ishimatsu, 2008). Therefore, ocean acidification is unlikely to impact polar cod through decreased prey availability.

3.4.12 Effects of ocean acidification on seabirds and marine mammals

The effect of ocean acidification on seabirds and marine mammals will be through the food chain. For example, those seabirds or mammals that feed upon calcifying forms (e.g., pteropods, bivalves – see Section 3.3) may have to switch to other food sources if these species disappear as a result of ocean acidification.

3.5 Ecosystem-level responses to ocean acidification

The effects of ocean acidification on ecosystems are far more complex than the sum of the individual parts outlined in single-species or single-factor studies. Controlled mesocosm studies (e.g., Riebesell, 2004; Hale et al., 2011; Eklof et al., 2012) provide an approach that has shown that complex emergent results can arise from inter-specific differences in scaling of responses to ocean acidification, often in combination with other stressors. Kordas et al. (2011) gave an excellent example of how climate change effects on a simple predatorprey system (seastars and mussels) can lead to non-intuitive outcomes: increased predation rates by the seastar at higher temperatures outweighed the benefits of increased growth rates of the mussels, so that the overall effect of temperature on the mussel was negative. Equivalent investigations of the effects of ocean acidification on the shore crab Carcinus maenas and its prey, the snail Littorina littorea, showed that negative effects of acidification on both species (decreased chela strength in the crab, and decreased shell strength in the snail), resulted in no net change in predation pressure (Landes and Zimmer, 2012; see also Hoegh-Guldberg and Bruno, 2010; Potera, 2010; Griffith et al., 2011, for effects on food webs). Rossoll et al. (2012) also demonstrated that pCO_2 can impact food quality with negative consequences for the next trophic levels (copepods, fish larvae).

Emergent ecosystem responses to ocean acidification will be a result of biotic and abiotic factors operating independently and in combination. At the biotic level, responses will be determined by the relative scaling of responses to ocean acidification in different species that interact. This principle is exemplified by the predator-prey examples above (Kordas et al., 2011; Landes and Zimmer, 2012), but also includes the influence of nutritional status on tolerance to ocean acidification (e.g., Thomsen et al., 2010), as well as how ocean acidification affects relative competitive ability (for which no relevant examples could be found). Abiotic stressors such as temperature, salinity, and pollutants not only affect species directly, but will also mediate the ecological interactions between them by moderating the relative scalings of their responses to ocean acidification. For example, including the effects of ocean warming in investigations of the effects of ocean acidification on the crab C. maenas preying on the snail L. littorea did not alter the relative scalings. Responses were the same at both temperatures (Landes and Zimmer, 2012), although the capacity for interactions between these variables is clear. To date, there are very few studies that have investigated the combined effects of biotic and abiotic factors on ocean acidification responses in marine waters, including the Arctic (but see Findlay et al., 2010b; Wood et al., 2011, and Eklof et al., 2012). Such studies are urgently needed.

With respect to Arctic food webs, there are as yet no data available. Therefore, there is a pressing need for studies on ecosystem-level responses to ocean acidification, from simple predator-prey interactions to higher-level emergent ecosystem responses.

3.6 Observations in naturally occurring low pH and/or high-CO, regions

Observations in areas that have natural gradients of pH, such as CO₂ vents (Hall-Spencer et al., 2008; Fabricius et al., 2011), estuaries, areas with acid soil runoff (Amaral et al., 2011; Mucci et al., 2011), and upwelling areas, provide opportunities to study the effects of a gradient of low pH on ecosystem-level responses in a variety of coastal ecosystems, as well as the adaptation of individual organisms (Wootton et al., 2008; Thomsen et al., 2010; Barry et al., 2011). The acidified waters near tropical and temperate CO₂ vents affect calcification, recruitment, growth, survival, and species interactions (Hall-Spencer et al., 2008; Fabricius et al., 2011). Many species of microalgae, macroalgae, seagrass, foraminifera, corals, polychaetes, crustaceans, mollusks and bryozoans are remarkably tolerant of long-term exposures to high and variable CO, levels at the vents (Kroeker et al., 2010, 2013; Johnson et al., 2011). However, a fall in mean pH from 8.1 to 7.8 can have detrimental effects on the recruitment of benthic organisms from the plankton (Cigliano et al., 2010) and many adult communities show dramatic reductions in biodiversity along the gradients of

increasing CO₂ in both temperate and tropical coral systems with around 30% fewer species in adult communities at mean pH 7.8 than in adjacent areas at mean pH 8.1 (Hall-Spencer et al., 2008; Fabricius et al., 2011). Important groups, such as coralline algae, calcified foraminifera, and sea urchins are common outside the vent systems but are absent from areas with mean pH \leq 7.8 (Figure 3.4). However, these studies also show many more subtle effects such as changes in the architecture and community composition of corals in response to differential tolerances to ocean acidification (Fabricius et al., 2011) that had not been predicted. These natural analogs have the disadvantage of being constrained to a limited area: mobile species and/or life-stages may choose to move away or be advected to non-acidified environments. Similarly, ocean acidification-sensitive early life stages generated in distant, nonacidified, habitats may be able to recruit into acidified areas once they have grown beyond a size- or tolerance-threshold (e.g., Ferrari et al., 2011). Consequently, although natural ocean acidification analogs provide a valuable picture of some ecosystem-level effects this picture is inevitably incomplete.

0.5

pH reduction

A key question is how results from acidified areas in temperate and tropical systems might translate to polar waters. For example, the effects of naturally high CO₂ on mussels vary dramatically depending on food availability; populations in the Baltic Sea are able to settle and grow well but are unable to do so in the oligotrophic conditions of the Tyrrhenian Sea (Thomsen et al., 2010; Melzner et al., 2011; Rodolfo-Metalpa et al., 2011). To better understand the effects of low pH / saturation state on Arctic ecosystems, ecological shifts that occur along natural pH / carbonate-saturation gradients such as in upwelling areas or in fjords and in areas affected by tundra soil runoff are important to investigate. Areas of naturally lower pH in Arctic waters may be an important focus of future studies to reveal which types of organisms are resistant to ocean acidification and so provide a glimpse of the future diversity and functioning of marine ecosystems as anthropogenic CO₂ emissions rise.

3.7 Caveats

1.5

3.7.1 Time-scales of experiments in the context of acclimation and adaptation

The literature on the impact of ocean acidification is mostly based on short-term perturbation experiments. For example, 66% of all studies exploring the impact of ocean acidification on echinoderms consider exposure times of less than two weeks (data updated from Dupont et al., 2010). However, longer acclimation times (more than ten weeks) to high pCO_{2} trigger negative effects on survival on such long-lived adult invertebrates (Pörtner et al., 2004; Shirayama and Thornton, 2005; Kurihara et al., 2008). Moreover, short exposure times are not sufficient to allow (or take into account) potential acclimation to a new environment. For example, the sea urchin Strongylocentrotus purpuratus is able to reshape its skeleton and change its behavior in a matter of eight to 20 weeks when exposed to a different habitat structure (Hernández and Russell, 2010). As a consequence, short-term exposure may both over-(no acclimation, no selection, no positive carry-over effects) and underestimate (not enough time to induce lethal effects, no negative carry-over effects) the real impacts of a chronic exposure to high pCO₂. The roles of natural variability and rate of change are other neglected parameters that may lead to overand underestimation of the real impact of ocean acidification on marine species. For practical reasons, most perturbation experiments published to date used abrupt changes in pH. However, abrupt changes can lead to an overestimate of the real impact on fitness (e.g., pCO, changes on terrestrial ecosystems; Klironomos et al., 2005).

One of the critical gaps in knowledge is the lack of information on long-term exposure taking into account phenotypic plasticity and natural selection during multigeneration exposures. In other words: is there enough genetic variability and plasticity for populations to cope with near-

Figure 3 biodivers occurs in and non-Atmosph required equivalen a pH gra For calcifiers taxa, and correspon by climate based on

1.0

Figure 3.4. Change in Mediterranean coastal biodiversity as a function of declining pH. The biodiversity remaining (percentage of taxa that occurs in areas with no pH reduction [open square]) is shown for calcifying taxa (51 taxa total, white circles) and non-calcifying taxa (71 taxa total, black circles). Atmospheric pCO, levels (ppm CO,) that would be required to cause pH changes in ocean surface waters equivalent to those observed at three locations along a pH gradient are indicated by dotted vertical lines. For calcifiers, non-calcifiers, and all taxa combined, exponential regressions explained 99%, 90%, and 88% of the variance. Fitted regressions indicate a loss of biodiversity of about 40% for non-calcifiers and all taxa, and about 70% for calcifiers, for a pH reduction corresponding to the atmospheric pCO₂ level projected by climate models for 2100. Source: Barry et al. (2011), based on Hall-Spencer et al. (2008).



100

80

60

40

20

0

0.0

Taxa remaining, %

future ocean acidification? Phenotypic plasticity occurs when individual genotypes produce different phenotypes in different environmental conditions. This is an adaptive strategy, but is complicated by the fact that plasticity is itself genetically determined and can be under strong selective pressure. Thus, species and populations experiencing high environmental variability may have higher acclimation potential to additional stressors. Natural selection occurs when some genotypes have higher fitness and, therefore, are selected within a population during environmental changes. These parameters are critical and yet have only been considered in relatively few studies (Dupont and Thorndyke, 2008; Kurihara et al., 2008; Parker et al., 2012; but see Garrard et al., 2013). Evolutionary perspectives are only now starting to be considered. Using different clones of bryozoans, Pistevos et al. (2011) demonstrated the existence of genotype variation, which may sustain populations by natural selection during falling pH. Sunday et al. (2011) showed that the sea urchin S. franciscanus has vastly greater levels of phenotypic and genetic variation for larval size in future CO₂ conditions compared to the mussel Mytilus trossulus. Further, selectively bred fast-growing oysters were more resilient to ocean acidification (Parker et al., 2012).

In a recent paper, Parker et al. (2012) pre-exposed adult oysters to high pCO_2 (856 µatm) for five weeks during reproductive conditioning and showed a positive carry-over effect on larval growth. On the other hand, exposure to an environmental stressor also can induce negative carry-over effects that persist into later stages. For example, exposure to an environmental stressor during the pelagic phase can reduce juvenile performance and be exacerbated if stressful conditions persist (Emlet and Sadro, 2006). Adult sea urchin pre-exposure to elevated pCO, had a direct negative impact on subsequent larval settlement success. Five to nine times fewer offspring reached the juvenile stage in cultures using gametes collected from adults previously acclimated to high pCO₂ for four months (Dupont and Thorndyke, 2012). In conclusion, considering impacts on a single life-history stage can also lead to misinterpretation of the impact of high pCO_2 on a given species.

3.7.2 Multiple stressors

In the Arctic Ocean, the main drivers of change in the nearfuture will be warming and multi-year ice melt, which will drive shifts in geographic range - and hence species composition of ecosystems (e.g., Beaugrand et al., 2008) - as well as changes in physiology. Some cold-water adapted high polar species may be marginalized and lost as their physical niche contracts (as in montane streams, e.g., Sauer et al., 2011), or suffer reduced productivity (e.g., Torstensson et al., 2012). At the same time temperature may favor an increase in general productivity that could benefit other species (e.g., Miller et al., 2012). The combination of temperature changes with ocean acidification has the potential to further constrain benefits (Pörtner and Knust, 2007), although available evidence is as yet limited. In addition to increased pCO_2 (i.e., ocean acidification), the additional influence of ice melt to freshen the water column may also have an effect on organisms, although again, data are lacking. A large meta-analysis of multiple stressors in climate change from nonpolar regions showed that effects of a wide range of stressors were additive (26% of cases), synergistic (36% of cases) or antagonistic (38% of cases; Crain et al., 2008). The distributions of additive, synergistic and antagonistic effects were also reflected in studies involving ocean acidification, although the number of studies involved was very low (Crain et al., 2008).

3.8 Research priorities

There is an almost complete lack of information on the effects of ocean acidification (in isolation or in combination with other environmental stressors) on keystone species and processes in the Arctic. Accordingly, there is an urgent need for focused research on the likely impacts of ocean acidification on a range of taxa and processes. Some of these are listed below. Note that these points are not mutually exclusive and should, whenever possible, be combined. Further, notwithstanding logistical difficulties, experiments should be conducted *in situ* whenever possible and/ or using pelagic and benthic mesocosms (Figure 3.5).

• In field studies, locate pH and CO₂ sensors in the same places that assessments of biological and ecological responses are being conducted.



Figure 3.5. Two experimental approaches to investigate the impact of ocean acidification on marine species: laboratory-based perturbation experiments (left, Maj Arnberg) and large-scale field-based mesocosms (right, Maike Nicolai).

- · Assess responses of key components of Arctic and sub-Arctic food webs to ocean acidification and determine whether their responses are in any way different from those observed in other geographic regions, for example: viral and microbial communities; heterotrophic and mixotrophic protists; planktonic and under-ice microalgae and how seasonal changes in sea-ice coverage influences primary production and carbon transfer through the food web; clarify the relative effects of the acidification-induced changes in proportion of carbon species used in calcification $(CO_2, HCO_3^{-}, and CO_3^{-2})$; for a miniferans (e.g., determine the respective functional roles of calcifying and non-calcifying foraminifera in order to assess the ecosystem consequences of diversity shifts under ocean acidification); corals (e.g., determine the effects of ocean acidification on Lophelia and non-Lophelia Arctic coral species and the effects of acidification-mediated loss of unprotected coral structure on coral-associated biodiversity); crustaceans, especially the copepods Calanus finmarchicus, C. hyperboreus and C. glacialis, amphipods (Themisto libelulla), pteropods (mostly Limacina helicina) and commercially important crabs; fish, especially Arctic cod (Arctogadus glacialis), polar cod (Boreogadus saida), and capelin; seabirds and marine mammals (e.g., assess the dietary flexibility of seabirds and marine mammals to, for example, switch to food sources not affected by ocean acidification).
- Adaptive capacity and environmental plasticity of all stages of the life cycle, including possible local adaptations. This could include (but need not be limited to): long-term transgenerational and selective-breeding experiments; long-term studies investigating adaptations and acclimation capacity; long-term studies investigating the potential for effects to 'carry-over' to subsequent life stages and/or generations; address the issue of whether the increased generation time of many Arctic species confers reduced adaptive capacity to near-future climate change (in comparison to temperate and tropical species); multipopulation studies to quantify genetic and phenotypic intra-specific variation of key traits influencing fitness; effects of multiple stressors (ocean acidification, carbonate saturation state, temperature, salinity, oxygen, toxins, etc.) on species-level and ecosystem-level processes (including trophic interactions); assess the effects of food limitation / nutritional status on species-level and ecosystemlevel processes (e.g., do Arctic species show the same amelioration of ocean acidification impacts under high food / nutrient levels?).

3.9 Conclusions and recommendations

- Based on studies from other oceans, it is highly likely that a significant change will occur in Arctic marine ecosystems due to ocean acidification.
- Presently, too few data are available to determine how vulnerable Arctic ecosystems are to ocean acidification.
- It is likely that some organisms will benefit (e.g., seagrasses, some phytoplankton) from a high-CO₂ world, while others will be disadvantaged (e.g., calcifiers), possibly to the point of local extinction.
- Early life stages and life-stage transitions (e.g., molting) are, in general, more susceptible to ocean acidification.
- There are major knowledge gaps on ecological interactions and on individual species response through acclimation and adaptation.
- Ocean acidification impacts must be assessed in the context of other ecosystem drivers associated with change in the Arctic (e.g., warming, loss of ice cover, freshening of surface waters).

Ocean acidification may cause changes to Arctic marine systems at the organism and ecosystem levels. At the organism level, there will be both direct effects on physiology and behavior and indirect effects via, for example, trophic interactions (e.g., food availability and quality). Ocean acidification will affect energy flux through food webs and material flux through changes in the vertical movement of organic and inorganic matter. Pelagic and benthic calcifiers are at greatest risk to ocean acidification. The early life stages of both invertebrates and vertebrates, particularly larval forms with limited autoregulatory capacity, will generally be more susceptible. Sessile benthic organisms will in general be more susceptible to ocean acidification and are likely to be excluded from some regions in which they were previously present. On the other hand, productivity of autotrophs (e.g., dinoflagellates) are likely to increase as may primary productivity (as a result of higher CO₂ and temperature). Organisms inhabiting regions that have always exhibited marked fluctuations in pH and CO, may prove highly resilient to ocean acidification. Northward movement of some organisms - driven by warming of the oceans - may be limited by direct or indirect effects of ocean acidification. These potential impacts of ocean acidification on the Arctic are considered in the modeling scenarios presented in Chapter 4.

4. Potential economic and social impacts of ocean acidification on Arctic fisheries

Lead authors: U. Rashid Sumaila, William W.L. Cheung, Sarah Cooley, Ola Flaaten, Vicky W.Y. Lam, Nathalie Hilmi, Alain Safa, Helene Amundsen, Arild Gjertsen, Grete K. Hovelsrud

4.1 Introduction

The objective of this chapter is to review existing knowledge on the impact of ocean acidification on social and economic aspects of Arctic fisheries. A secondary objective is to discuss how change in ocean pH under future scenarios of anthropogenic carbon dioxide emission may present a risk to marine fisheriesbased economies, the food security and culture of indigenous communities, and recreational fisheries values in the Arctic. The chapter builds on information presented earlier in the report on the chemistry (Chapter 2) and biological effects (Chapter 3) of ocean acidification in the Arctic. The information presented here shows that ocean acidification is likely to affect the abundance, productivity and distribution of marine species although the magnitude and direction of the changes are uncertain. This, in turn, is likely to affect the price of fish, the cost of fishing, and the fisheries benefits to the indigenous and local populations of the Arctic. The chapter concludes by providing suggestions about ways to manage and counteract the effects of ocean acidification on Arctic fisheries. However, given that the effects of ocean acidification are difficult to isolate from the effects of other anthropogenic impacts, this chapter concludes by stressing the need to strengthen marine management in general.

4.2 People in the Arctic

Estimates of population numbers for the Arctic depend on the geographical boundaries used: varying from between 2 and 4 million people (ACIA, 2005) to 9.9 million people (Duhaime and Caron, 2009). Of these, approximately 2.3 million people live within about 50 km of the coasts of the Arctic Ocean and its marginal seas (see Table 4.1). Estimates of the population of indigenous peoples vary between 400 000 and 1.3 million (AMAP, 2011a), and the proportion of indigenous peoples out of the total population in the various Arctic regions varies greatly (Table 4.1). These include many different ethnic groups. For instance, in the Russian Arctic, there are an estimated 2 million people with native ethnic status (Aslaksen et al., 2009).

In many Arctic regions, non-local and non-indigenous populations are the majority populations. Only in Nunavik and Nunavut in Canada, and in Greenland is more than 80% of the population indigenous or local; in other regions the share is much lower (Table 4.1). The number of local indigenous people in the Arctic increased by approximately 1.5% annually between 1990 and 2010, but their proportion of the total population declined over this period due to immigration (Rasmussen, 2011).

The general demographic trends in the Arctic region show a diminishing number of people in the active workforce, an ageing population with higher education levels, and decreasing household size. The Nordic countries, including the Faroe Islands and Greenland, are experiencing a decline in birth rates, combined with emigration of young people (in some areas particularly women), which adds to a skewed population distribution. In parts of the Canadian Arctic, birth rates remain high enough to enable populations to grow despite emigration, while in parts of the Russian Arctic low birth rates are leading to declining populations (Rasmussen, 2011).

Many of the indigenous peoples can be characterized by a mixed-economy lifestyle that combines subsistence activities with cash income (ACIA, 2005; Poppel, 2006; Larsen et al., 2010; AMAP, 2011a; Poppel et al., 2011). The Arctic is experiencing major social and environmental changes, while retaining indigenous languages, and engaging in traditional activities (ensuring continuity and a connection to nature) (Aslaksen et al., 2009; Poppel, 2011). Hunting, fishing, herding livestock and gathering other food are important activities for securing daily needs as well as a source of income, and they help maintain cultural identity and social roles (AHDR, 2004; Hovelsrud et al., 2011). Such activities require investment in appropriate equipment, and thus some cash income.

The nature and level of subsistence activities, including fishing and berry picking, and hunting marine mammals, birds or big game, varies among different regions in the Arctic. With the exception of Alaska, these activities are not included in official statistics (Aslaksen et al., 2009). It is therefore difficult to assess their contribution to household economies. However, large structural changes are occurring and agricultural, hunting and fishing communities are experiencing emigration and negative population growth. While many studies uphold the nutritional value of traditional food, the economic and cultural importance of harvesting activities is decreasing (Rasmussen, 2011). Changes in the availability of marine resources dues to stressors such as ocean acidification could exacerbate these trends.

4.3 Socio-economic factors

The economy of the circumpolar north (see Chapter 1, Figure 1.3 for a definition of 'circumpolar north') is large in relation to the population size. Gross Domestic Product (GDP) for the region was estimated at USD 225 billion in 2003. This estimate does not include production in the subsistence sector and is therefore an underestimate. Fisheries are relatively important, with the catch in 2002 accounting for over 10% of the world's total catch of wild fish and over 5% of the crustacean catch (Statistics Norway, 2007). The region also has some aquaculture production. However, most of the aquaculture and much of the wild catch is produced in the sub-Arctic regions (see Chapter 1, Figure 1.4 for a definition of 'sub-Arctic regions').

Differences in disposable income per capita across Arctic countries are smaller than differences in gross regional product (GRP) per capita, and the gap between Arctic and non-Arctic

Table 4.1. Basic economic and social conditions, Arctic regions, 2006. Source: modified from Duhaime and Caron (2009).

Region	Total population	Approx. % of population living within ca. 50km of coast of Arctic Ocean and its marginal seas	Share of aboriginal peoples in the total population	Share of women in the total population	Share of children aged 0-14 years in the total population	Life expectancy in years	Infant mortality per thousand live births	Share of tertiary education graduates in the total population	Personal disposable income, USD-PPP	Dependency ratio	Composite index ¹
Alaska	670 053	72	13.1	48.5	21.5	76.7	6.7	24.7	32 811	0.6	9
Labrador	26 364	0	37.8	49.3	20.6	76.1	4.4	9.4	19 044	1.3	6
NWT	41 465	13	49.8	48.8	23.9	79.1	4.2	19.4	30 339	0.7	8
Nunavik	10 815	100	89.2	49.1	36.3	63.5	17.3	9.6	19 532	1.9	4
Nunavut	29 475	100	84.5	48.7	33.9	70.4	10.0	11.9	24 495	1.6	5
Yukon	30 375	<1	25.0	49.7	18.8	76.4	11.0	23.4	29 761	1.0	8
Faroe Islands	48 183	100	0.0	48.1	22.8	78.9	4.4	23.0	15 275	0.7	7
Lapland	184 935	<1	0.8	49.9	16.3	78.6	5.9	20.7	14 000	1.5	7
Oulu	465 018	0		49.7	19.8	79.0	4.2	22.7	13 847	1.4	7
Greenland	56 901	100	88.6	47.0	24.8	68.3	15.4	n.d.	15 237	0.9	5
Iceland	299 891	98	0.0	49.6	21.8	81.2	1.4	23.5	17 957	0.8	8
Finnmark	72 937	93	9.2	49.2	20.5	77.6	4.3	21.4	18 687	1.1	7
Nordland	236 257	100		50.0	19.3	79.4	3.3	19.8	18 700	1.2	7
Troms	153 585	99		49.6	19.7	79.0	3.7	25.1	18 550	1.0	8
Norrbotten	251 886	0	3.6	49.3	15.6	79.5	5.1	13.6	14 721	1.3	6
Västerbotten	257 581	0		50.0	16.1	80.4	3.1	19.4	14 139	1.2	7
Arkhangelsk	1 280 200	29	0.5	53.3	16.3	64.8	10.2	12.1	7 465	1.1	3
Chukchi	50 500	46	20.9	47.9	21.7	58.9	23.2	14.6	19 267	0.3	5
Evenk	17 000	0	19.3	50.0	24.2	59.1	21.3	11.5	9 765	0.5	4
Karelia	693 100	3	0.0	54.2	15.5	63.8	7.6	13.7	6 734	1.0	3
Khanty-Mansii	1 488 300	0	1.4	50.8	19.9	68.8	7.5	15.9	16 851	0.7	6
Komi	974 600	0	1.0	52.5	17.5	64.2	7.0	12.2	10 710	1.1	4
Koryak	22 600	23	34.2	50.0	22.0	56.0	33.0	9.9	12 389	0.6	3
Magadan	168 500	0	8.7	51.6	17.0	63.4	14.2	15.4	10 682	0.8	4
Murmansk	857 000	41	0.2	51.6	15.7	65.2	10.3	15.5	9 853	0.9	5
Nenets	42 000	13	14.3	51.2	22.3	62.2	15.2	9.9	•••	0.5	4
Sakha	950 000	2	2.4	51.5	23.6	65.6	10.6	14.6	10 733	1.0	5
Taimyr	38 400	4	19.0	51.8	22.9	63.8	7.4	13.3	11 641	0.7	5
Yamal-Nenets	532 600	2	5.9	50.7	21.3	68.9	13.0	16.8	20 447	0.5	6

¹ This index is based on the six indicators: female proportion, life expectancy, infant mortality, tertiary education rate, personal disposable income, and dependency rate.



Figure 4.1. Gross regional product per capita and disposal household income per capita. Note: PPP denotes purchasing power parity, which relates the value of the U.S. dollar to what it can actually buy in each country relative to the USA. Source: Duhaime and Caron (2009).

regions is considerably smaller than the difference in GRP per capita (Figure 4.1). This is largely explained by mechanisms of national income redistribution. As shown, disposable income per capita is highest in the United States. A comparison of disposable income per capita between countries only roughly indicates differences in welfare, however¹. Hence, the above figures may overestimate the actual welfare level of Arctic regions when compared with non-Arctic regions. On the other hand, subsistence activities are more widespread in the Arctic regions than elsewhere in these countries and contribute to overall welfare.

For indigenous peoples, marine resources contribute to their economy, livelihoods, nutrition, health and well-being. The contributions from these subsistence activities are not considered in official Arctic GRP assessments, yet their importance in the mixed cash-subsistence economies of the Arctic is well documented (e.g., Hovelsrud, 1999; Rasmussen, 2005; Poppel, 2006; Kruse et al., 2008; Aslaksen et al., 2009; Poppel and Kruse, 2009). For example, locally produced food accounts for more than half of the diet in Inuit households (Poppel et al., 2007). An example illustrating the difficulties with attaching values in GRP accounts is that of whale meat: this can either be distributed among kin in non-monetary exchanges or sold locally (Nuttall, 2002), complicating rigorous assessments of the value gained. This example highlights the potential conflict between traditional indigenous perspectives and modern commercial interests that commonly influences discussions on Arctic natural resources. Arctic regions where traditional subsistence activities play a more dominant role (e.g., Greenland and northern Canada) have a much lower GRP compared to the largest economies in the Arctic (AMAP, 2011a).

4.4 Marine fisheries in the Arctic region

Catch statistics supplied by the UN Food and Agriculture Organization (FAO) represent a primary source for describing the marine fisheries in the Arctic region in more detail. The Arctic marine area extends over five FAO fisheries statistical areas (Figure 4.2), of which only one (FAO Fishing Area 18: Arctic Sea) lies completely within the Arctic region. The other four (FAO Fishing Areas 21: Northwest Atlantic; 27: Northeast Atlantic; 61: Northwest Pacific; 67: Northeast Pacific) extend well beyond the Arctic. However, the Arctic areas of these four regions contain important fisheries.

4.4.1 FAO Fishing Area 18 (Arctic Sea)

Fishing area 18 covers the central Arctic Ocean Basin and the following Arctic shelf seas: the Kara, Laptev, East Siberian and Chukchi Seas, the Beaufort Sea, the Canadian Arctic Archipelago, and Hudson Bay/Hudson Strait (see Figure 4.2). The scope for undertaking commercial fisheries in this area is limited by the presence of sea ice during large parts of the year. This is reflected in catch data reported to FAO; for example, the FAO database only records catches of some 460 and 590 t of cod/haddock/hake by Russia in 2008 and 2010, respectively. However, subsistence fishing by people living along the coasts of these Arctic seas is likely to be extensive. Zeller et al. (2011) attempted 'educated guesses' of the historic catches in the region and suggested catches of around 10 000 t in the mid-2000s (declining from around 24 000 t in 1950). The species involved were mostly salmonids (coregonids, Arctic char, trout and salmon species). About 80% of the reconstructed catch was attributed to Russia, while Alaska and Canada shared the remaining 20% equally.

¹ For better comparability, public expenditures to household services should be added to the disposable incomes, and differences in price levels between Arctic and non-Arctic regions (generally higher in Arctic regions) should be taken into account.



With the observed and projected decline in sea ice in response to global warming, the scope for exploiting the fish and shellfish resources in this area will increase, with the inherent danger that unregulated catches may lead to overfishing of stocks.

In 2009, the US North Pacific Fishery Management Council (NPFMC) approved, and the National Marine Fisheries Service (NMFS) implemented a management plan based on a precautionary perspective that covers the Arctic waters of the United States in the Chukchi and Beaufort Seas. This Fishery Management Plan for Fish Resources of the Arctic Management Area (Arctic FMP) prohibits commercial fishing in the Arctic waters of the region that are under US jurisdiction until more information is available to support sustainable fisheries management. Subsistence fishing is not affected.

4.4.2 FAO Fishing Area 21 (Northwest Atlantic)

The International Commission for the Northwest Atlantic Fisheries (ICNAF) maintains catch statistics based on a comprehensive division of FAO Fishing Area 21 into several sub-areas (see Figure 4.2).

For the Arctic areas, catches from the Baffin Bay/Davis Strait areas west of Greenland (sub-areas 0A and 0B, 1A to 1F, and 2G to 2J) are considered to represent fisheries mainly in the Low Arctic. The average total catch (1992–2011) in these areas amounts to 209 000 t. Of the two countries bordering the region, Canada is responsible for 31% of the catch, while Greenland takes 63% (Table 4.2). Table 4.3 lists the main species (groups) landed over the past 20 years and for two ten-year periods separately.

A further 273 000 t is caught in sub-areas 3K-3P, fisheries on the Grand Banks that are located just outside of the AMAP Arctic area. About 65% of this catch is taken by Canada. The remainder is taken by other countries fishing mainly in international waters off Flemish Cap, and thus not contributing to the economy of the Arctic region.

The catches of species fished in Fishing Area 21 (sub-areas 0, 1, and 2) since 1992 are dominated by pandalid shrimp (on average 71% of the total catch) (Figure 4.3), followed by Greenland halibut (16%) and redfish (over 5% in the latter half of the period); other species groups represent smaller fractions. Remarkably, invertebrates (crustaceans, mollusks, echinoderms) represent more than 45% of the total catch also in the 21.3 sub-area.

Table 4.2. Average total annual catch by country in FAO Fishing Area 21 (ICNAF sub-areas 0, 1 and 2) over the past two decades (1992-2011). Source: UN Food and Agricultural Organization (www.fao.org/fishery/topic/3457/en).

FAO Fishing Area 21	Average annual catch, Kt		
Total Arctic region (sub-areas 0, 1 and 2)	209		
By country			
Canada	65 (31%)		
Greenland	131 (63%)		
Russia	4 (2%)		
Faroe Islands	1 (<1%)		
Iceland	1 (<1%)		
Other (non-Arctic)	7 (3%)		

Table 4.3. Average annual catches by species in the Arctic part of FAO Fishing Area 21 (ICNAF sub-area 0, 1, and 2) over the past two decades (1992-2011). Source: UN Food and Agricultural Organization (www.fao. org/fishery/topic/3457/en).

Fishing Area 21	Average annual catch, t					
	1992-2011	1992-2001	2002-2011			
Capelin	88	118	59			
Cod	4934	1579	8289			
Other gadoids	1201	1747	656			
Grenadiers	114	174	54			
Wolffish	251	87	414			
Lumpfish	3003	1069	4938			
Redfish	8256	2758	13755			
Greenland halibut	34550	29061	40040			
Witch flounder	7	3	12			
Halibut	60	95	24			
Other flatfish	3	2	3			
Salmonids	111	191	30			
Other fish	431	663	200			
Sharks	19	37	0			
Rays and skates	7	7	6			
Shrimp	148608	117430	179787			
Other crustaceans	6038	7327	4748			
Echinoderms	16	21	11			
Mollusks	1656	1539	1772			
Total	209352	163909	254796			

4.4.3 FAO Fishing Area 27 (Northeast Atlantic)

Within FAO Fishing Area 27 (Northeast Atlantic), sub-areas 27.1 and 27.14 can be considered to represent the High/Low Arctic (see Figure 4.2 and Chapter 1, Figure 1.4 for a definition of this area), while sub-areas 27.2 and 27.5 essentially represent the Low/sub-Arctic region (see Figure 4.2). The average total catch (1992-2011) by sub-area (Table 4.4) indicates that the vast majority is taken in the sub-Arctic regions (84%). Iceland and Norway take about equal shares, followed by Russia and the Faroe Islands. Non-Arctic countries fishing in the area together take 4%, while Greenland lands only 1%. Table 4.5 lists the average catches of the most important species (groups) landed over the past 20 years and for two ten-year periods separately. The top ten species (herring, capelin, cod, blue whiting, saithe, haddock, redfish, mackerel, pandalid shrimp, Greenland halibut) account for 91% of the total average catch. Among these, blue whiting and mackerel do not represent separate stocks, but form part of much larger oceanic stocks that have their centre of distribution (and spawn) in more temperate waters. The other species represent (often multiple) stocks that reside and spawn within the Arctic region. Herring and capelin are pelagic species that spawn near the coast (capelin even inshore), but migrate offshore to feed. Cod, saithe and haddock are typical demersal species that remain on the shelf to spawn and feed, although they migrate over large distances. Redfish, Greenland halibut and pandalid shrimp represent deep-water species and their stock structure is less well known. Although there are differences between the two periods, these data do not suggest that there have been large Table 4.4. Average total annual catch by sub-area and country in FAO Fishing Area 27 over the past two decades (1992-2011). Source: UN Food and Agricultural Organization (www.fao.org/fishery/topic/3457/en).

Fishing Area 27	Average annual catch, Kt		
Total Arctic region (sub-areas 1, 2, 5 and 14)	4457		
By sub-area			
27.1	507 (11%)		
27.2	1990 (45%)		
27.5	1739 (39%)		
27.14	221 (5%)		
By country			
Iceland	1638 (37%)		
Norway	1557 (35%)		
Russia	752 (17%)		
Faroe Islands	292 (7%)		
Greenland	25 (1%)		
Other (non-Arctic)	192 (4%)		

Table 4.5. Average annual catches by species in the Arctic part of FAO Fishing Area 27 (ICES sub-areas 1, 2, 5, and 14) over the past two decades (1992-2011). Source: UN Food and Agricultural Organization (www.fao. org/fishery/topic/3457/en).

Fishing Area 27	Average annual catch, Kt					
	1992-2011	1992-2001	2002-2011			
Herring	1092	972	1211			
Other clupeoids	1	1	0			
Capelin	1004	1334	673			
Cod	770	853	687			
Blue whiting	460	309	611			
Saithe	255	224	287			
Haddock	205	167	242			
Ling species	23	20	26			
Tusk	22	24	20			
Polar cod	22	23	20			
Other gadoids	19	22	15			
Argentines	23	17	29			
Wolffish species	36	38	35			
Redfish species	166	192	141			
Lumpfish	7	8	5			
Mackerel	125	129	121			
Greenland halibut	42	43	41			
Plaice	11	13	8			
Other flatfish	15	18	13			
Salmonids	1	1	0			
Other fish	21	17	25			
Sharks	3	4	1			
Rays and skates	2	2	2			
Nephrops	2	2	2			
Shrimp	81	112	49			
Other crustaceans	7	2	12			
Echinoderms	1	0	1			
Bivalves and gastropods	21	30	13			
Plants	21	21	21			
Total	4456	4600	4313			





FAO Area 27 (sub-areas 27.1, 27.2, 27.5, 27.14)



long-term trends in any of the species. The rank order has mostly been maintained. However, there are major differences in the rank order of the dominant species in the different sub-areas and also large annual fluctuations (Figure 4.4).

In the Barents Sea (sub-area 27.1), the three species dominating the catch are cod, capelin and haddock, with marginal contributions from the others. Cod yields a consistently high catch, while capelin periodically exceeds the cod catch but is virtually zero between these periods. This is a consequence of the management system that only opens the fishery when the stock is predicted to be above some threshold in order to set aside enough food for predators. The cod and haddock in this area migrate to sub-area 27.2 to spawn.

In the Norwegian Sea (sub-area 27.2), the herring stock of spring spawners started to recover in the 1970s, grew through the 1980s due to strict harvest regulations, and since the early

1990s has contributed almost consistently more than 50% of the catch with cod coming second. The third-ranked species is blue whiting, but large catches were restricted to the early 2000s, when these fish apparently penetrated waters farther north than is usually the case.

In Icelandic and Faroese waters (sub-area 27.5), the bulk of the catch consisted of capelin and blue whiting, but their contribution has declined strongly over the past ten years. Recently, the top ten species are more equally distributed in terms of their contribution to the total catch. Also, the rest of the group represents a major component.

In Greenlandic waters (sub-area 27.14), deep-water species (redfish, Greenland halibut, pandalid shrimp) are consistently well represented in the total catch, although capelin is by far the dominant species on average. However, as is the case in sub-area 27.1 capelin yields have fluctuated widely.

Figure 4.4. Trend in the top-ten species reported for the Arctic part of FAO Fishing Area 27 over the past two decades. Source: UN Food and Agricultural Organization (www.fao.org/fishery/topic/3457/en). For key to species see Figure 4.3.



4.4.4 FAO Fishing Areas 61 (Northwest Pacific) and 67 (Northeast Pacific)

The statistical information for FAO Major Fishing Areas 61 (Northwest Pacific) and 67 (Northeast Pacific) does not include detailed information by sub-area that can be used to separate Arctic fisheries from non-Arctic fisheries. Wild fish catches in the eastern and western Bering Sea were 1.84 and 0.52 million t, respectively in 2002 (Statistics Norway, 2007) with an additional 0.01 million t of crustaceans (snow crabs) caught in the eastern Bering Sea. Pollock was the main species caught comprising approximately 80% of the wild fish catch in the eastern Bering Sea and 77% of the catch in the western Bering Sea. Pacific salmon, flatfish and other ground fish species made up most of the remainder of the catches.

4.5 Impacts of ocean acidification on Arctic fisheries

On the basis that ocean acidification is likely to affect the abundance, productivity and distribution of marine species in different ways (see Chapter 3 for further details), the ecosystem services associated with these species will also be affected. However, the magnitude and direction of the changes expected at the ecosystem level are uncertain, because the effects of ocean acidification cannot be seen in isolation from the other factors (such as temperature rise and ice melt) that evoke responses in individual species with respect to distribution and population dynamics. These factors may strengthen or counteract each other and until more insight is gained regarding such interactions, predicting the effects on ecosystem services must involve much speculation. To date, there has been little research on the economic impacts of ocean acidification on fisheries. Those studies that have been undertaken have been species-specific and have focused on shell-forming species (e.g., Cooley and Doney, 2009; Narita et al., 2012). A comprehensive, broad-based approach for understanding the impact of ocean acidification on marine ecosystems and their services, such as fisheries (including the socio-economic dimensions), is almost non-existent (but see Armstrong et al., 2012; Hilmi et al., 2013).

Global simulation modeling (e.g., Cheung et al., 2010) suggested that climate change may lead to increases in potential fisheries yield in the Arctic, while follow-up studies with a model that accounts for hypothesized physiological effects of ocean acidification suggested that there may be a substantial reduction in potential yield in the North Atlantic if waters become more acidic (Cheung et al., 2011). Predicting the effects of ocean acidification on potential yield is complicated by species interactions and multiple stressors on ecosystems. Using both ecosystem models with explicit trophic linkages (Ecopath with Ecosim) and a Dynamic Bioclimate Envelope model that incorporates ecophysiology and spatial population dynamics of marine species, ocean acidification in combination with hypoxia is projected to reduce maximum catch potential by 2050 relative to 2000 in the North Atlantic under different assumptions regarding future greenhouse gas emissions scenarios (IPCC SRES A1B or A2) (Ainsworth et al., 2011; Cheung et al., 2011). However, these projections are sensitive to the assumed vulnerability to ocean acidification of the modeled

organisms, which is poorly understood (see Chapter 3). Responses of marine species may interact with other human stressors such as overfishing, which might exacerbate the impacts of ocean acidification on yield (e.g., Wootton et al., 2008; Russell et al., 2009; Ainsworth et al., 2011). The estimated changes in quantity, quality and predictability of catches have direct implications for fisheries (Sumaila et al., 2011).

Decreasing pH in ocean surface waters is synergistic with other effects of climate change, such as increases in ocean temperature, changes in salinity, and possible expansion of anoxic zones (Stramma et al., 2010), but the combined effects of these factors in the Arctic are still unknown. Despite the fact that the impact of ocean acidification is still uncertain, further increases in ocean acidity can be expected to add more stress to marine ecosystems and the human communities that depend on them.

Increasing amounts of dissolved carbon dioxide in the ocean is expected to reduce, in particular, the growth of calcifying marine species such as mollusks (see Chapter 3 for further details). A reduction in the productivity of commercially targeted mollusks could have substantial impacts on their fisheries, and hence, on the communities that depend on these for food and income (Cooley and Doney, 2009; Cooley et al., 2009; Narita et al., 2012).

The description of fisheries in the Arctic region (see Section 4.4) clearly indicates large differences in species composition by area, and because ecosystems differ by region, a universal response to ocean acidification would seem unlikely. Most probably, shelf systems cannot be treated in the same manner as oceanic or deep-water systems. Moreover, any effects of ocean acidification on potential yield will be embedded in responses to other factors affecting these systems, specifically global warming. Unless the quantitative effects of each of these factors separately can be evaluated, any prediction of their integrated effects remains speculative.

Nevertheless, a few speculations may be inferred from the previous chapters:

- If ocean acidification is expected to be stronger in areas of melting sea ice (Chapter 3), sub-Arctic regions may suffer less than Low Arctic regions.
- If ocean acidification is expected to be stronger in surface waters than in deep waters, shelf species may be affected more than deep-water species.
- If ocean acidification is expected to be stronger in colder regions, global warming might to some extent compensate for the effects.
- If food webs are less complex in Low Arctic regions than in sub-Arctic regions (as supported by fewer species dominating the total yield), and therefore less resilient to ocean acidification, the negative effects on fisheries in terms of catch composition and total output may be greater.
- If mollusks are more susceptible to effects of ocean acidification, fisheries in the Low Arctic region of the northwestern Atlantic would suffer more than in the northeastern Atlantic, because bivalves and gastropods score higher among the species groups dominating the catches there.

4.5.1 Economic impacts of ocean acidification on Arctic fisheries

Environmental impacts of economic activity are not included in standard GDP, as opposed to 'green GDP', and it is therefore a challenge to develop environmental statistics and environmental indicators that can be applied in conjunction with economic indicators (Statistics Norway, 2009).

Kite-Powell (2009) has provided a global perspective on the economics of ocean acidification. Although there is still no detailed study on the economic impact of ocean acidification on global fisheries, Armstrong et al. (2012) suggested that ocean acidification may have negative as well as positive effects on fisheries and aquaculture.

Clearly, more work is needed to help reveal the potential impact of ocean acidification on the fisheries of the Arctic, as well as on marine ecosystems in general. Despite valuable attempts (Hilmi et al., 2013), a comprehensive, broad-based approach for understanding the impact of ocean acidification on the marine ecosystem, fisheries, and eventually the associated socio-economic dimensions, is still lacking. One approach to estimating these impacts is to use 'Economic Valuation' based on changes in exvessel prices, fishing costs and projected catch under different ocean acidification scenarios. Other than the change in landed value or total revenue, ocean acidification may also affect the cost of fishing by changing the fishing effort (the number of fishing days, travel distance, gear to be employed, etc.) required to catch the same amount of fish, when the distribution and abundance of target species are affected by ocean acidification. The wages earned may also be affected. Because fisheries represent a primary industry (i.e., an industry that relies on basic raw materials and whose product - fish - is used by other sectors of the economy, such as retail), change in catch potential may also indirectly affect other dependent sectors, from boat building to international transport (Dyck and Sumaila, 2010). Thus, it is crucial to take indirect economic activities into account when assessing the full economic impact of ocean acidification on fisheries. It should be noted, however, that this type of analysis is yet to be done. To have a more reliable economic model for estimating impacts of ocean acidification, a better understanding of the biophysical responses of marine species and the ecosystem to ocean acidification in the Arctic is necessary.

4.6 Impacts of ocean acidification on Arctic indigenous peoples and local communities

The indigenous peoples in the Arctic continue, to a large extent, to rely on food harvested locally. This makes them vulnerable to the effects of environmental contaminants that concentrate in Arctic food chains (Huntington et al., 1998), and to reduced harvests due to the consequences of other environmental stressors. These include the consequences of climate change (loss of sea ice) as well as other factors related to human development (pollution from atmospheric transport). Marine ecosystems in the Arctic region involve a small number of key species at each trophic level (Murray et al. 1998) and food webs tend not to be especially complex because
species diversity is relatively low. Most of the species present, however, are important food sources for indigenous peoples (Huntington et al., 1998; Figure 4.5). Adequate management of these resources is crucial, and countries exploiting the resources in the Low Arctic and sub-Arctic regions have developed management procedures that are at least aimed at preventing overfishing. In the High Arctic region (mainly FAO Fishing Area 18), where only subsistence fishing occurs, fish stocks are not under management plans. The exception is Alaska, where a precautionary management plan prohibits commercial fishing until more information has been put in place to support sustainable fisheries management.

Table 4.6 lists the marine fish and shellfish species, marine mammals, and seabirds commonly harvested in the Arctic region (between May and October) for consumption and trade, with an estimate of how likely they are to be affected by ocean acidification either directly (via its effects on life functions of the organism) or indirectly (via its effects on a major prey of the organism). Table 4.6 shows that some indigenous groups depend on marine species that are at the highest risk of impact from ocean acidification impacting on their food security (see Turley and Boot, 2011: their figure 13.2; Gattuso et al., 2011).

Table 4.7 summarizes the marine organisms that some Arctic indigenous peoples depend on, and shows the variety between geographical locations (Huntington et al., 1998). Most indigenous

groups harvest a range of organisms and can adapt by shifting their harvest to species not affected by ocean acidification, but there may be distinct seasonal losses or specific cultural losses resulting from changing harvests of some of these organisms. These impacts, combined with rising ocean temperatures and the resulting sea-ice reduction, suggest that the livelihoods of indigenous peoples will be increasingly compromised (Hovelsrud et al., 2011).

Developing an economic assessment of the potential ocean acidification driven losses for indigenous peoples and other local communities could underestimate the multiple values of marine harvests in Arctic cultures. Some peoples view natural resources as economic assets, whereas others view them as sources of cultural identity (Nuttall, 2002). Economically-based forms of accounting often insufficiently acknowledge political, cultural, or social factors and therefore provide limited insight into risk to Arctic peoples. For many Arctic indigenous peoples, the impact of a combination of changes, including climate change and ocean acidification, poses challenges in economic, social, cultural and environmental terms. The availability of and access to resources is fundamental to Arctic cultures and identities and for livelihood activities such as fishing (e.g., Nuttall, 2005; Ford and Furgal, 2009; Hovelsrud et al., 2011; Hoover et al., 2013). As such, ocean acidification represents a potential risk to Arctic food systems, as well as to cultures and livelihoods.



Table 4.6. Arctic organisms harvested by indigenous peoples and their risk of effects of ocean acidification. Major prey items which are very likely or extremely likely to experience direct effects of ocean acidification are indicated. Risk of indirect effects from trophic relationships are indicated in the final column of the table: harvested species that prey on many directly impacted species have high risk; species that prey on a mixture of directly impacted species and non-impacted species have medium risk; and species that prey mostly on non-impacted species have low risk. Harvested organisms that are at greatest risk are at the top of each subgroup. Source: adapted from Huntington et al. (1998), Nuttall (2005), and Kroeker et al. (2013).

	Harvested organism	Major	r prey of l	Risk of prey-related (i.e. indirect) ocean acidification effects								
		Mollusks other than squid ^a	$Echinoderms^{a}$	Crustaceans (shrimp, crab) ^b	Phyto/Zooplankton/ Krill ^b	Squid ^b	Smaller fish (incl. juvenile forms)	Invertebrates (benthic, worms, etc.)	Marine mammals	Plants	Insects	
Fish	Atlantic wolffish/ocean catfish	×	×	×								High
	Rough dab	×		×			×	×				Medium
	Redfish	×		×				×				Medium
	Arctic char	×					×				×	Medium
	Haddock		×	×			×					Medium
	Atlantic cod, Greenland halibut, Mackerel			×	×		×					Medium
	Salmon			×	×		×				×	Medium
	Blue whiting, herring			×	×							Medium
	Blue ling			×			×	×				Medium
	Greenland cod			×			×					Low
	Muksun, Siberian sturgeon, tusk			×				×				Medium
	Nelma/inconnu, ling				×		×					Low
	Capelin				×							Medium
	Coalfish, saithe						×					Low
	Greenland shark						×		×			Low
	Plaice							×				Low
Shellfish	Crab ^b	×	×	×				×				Medium
	Clams, scallops ^a				×							Low
	Shrimp ^b				×							Low
	Norway lobster/langoustine ^b						×	×				Low
Marine mammals	Bearded, harbor and hooded seals	×		×			×					Medium
	Walrus	×		×				×				Medium
	Narwhal			×		×	×					Medium
	Beluga whales			×			×	×				Low
	Harp and ringed seals			×			×					Medium
	Bowhead whales				×	×						Medium
	Fin and minke whales				×		×					Low
	Fur seals, pilot whales, sealions					×	×					Medium
	Common seals						×					Low
	Polar bear								×			Low
Seabirds	Arctic tern	×		×			×				×	Medium
	Ducks	×					×			×	×	Medium
	Sea gulls	×					×	×				Medium
	Eider	×						×				Medium
	Dovekie			×	×							Medium
	Thick-billed murre			×		×	×					Medium
	Black guillemot			×			×					Medium
	Kittiwakes				×		×					Low
	Cranes, geese						×			×	×	Low
	Swans									×		Low

^a Evidence exists that these organisms are *extremely likely* to be directly affected by ocean acidification; ^b evidence exists that these organisms are *very likely* to be directly affected by ocean acidification.

Table 4.7. Marine and coastal organisms harvested by Arctic peoples. Omission of a particular relationship in this table indicates lack of data rather than a lack of dependence. Source: adapted from Huntington et al. (1998).

	Harvested animal	Aleut, Alutiiq	Athabascan	Siberian Yupik	Inuit	Kalaallit	Icelanders	Faeroese	Northern Norwegians	Saami	Murmansk Oblast	Nenets and Yamalo- Nenets Autonomous Okrug	Taimyr Autonomous Okrug	Arctic Districts of Sakha Republic
Fish	Arctic char				×	×				×			×	
	Atlantic cod/cod		0 * * * * * * * * * * * * * * * *			×	×	×	×	×	×			
	Atlantic wolffishª/ ocean catfishª, capelin					×	×		×					
	Blue ling, blue whiting, coalfish, tusk, ling, mackerel							×	×					
	Greenland cod, saithe, Greenland shark					×								
	Greenland halibut/ halibut					×	×	×	×		×			
	Haddock						×	×	×		×			
	Herring						×	×	×		×		×	
	Muksun, nelma/inconnu												×	
	Plaice, rockfish										×			
	Redfish						×	×	×					
	Rough dab					×								
	Salmon	×	×			×			×	×	×			
	Siberian sturgeon												×	×
Shellfish	Norway lobster/langoustine, scallops						×		×		0 * * * * * * * * * * * * *			
	Shrimp					×	×		×					
Marine mammals	Bearded seals												×	
	Beluga whales		×		×	×						×	×	
	Bowhead whales			×										
	Common seals												×	×
	Fin whales, harbour, harp, hooded, seals; minke whales, Narwhal; polar bear					×			×					
	Pilot whales							×						
	Ringed seals					×							×	×
	Sealions													
	Walrus			×		×								
Seabirds	Arctic tern, dovekie, eider, sea gulls					×								
	Black guillemot, thick-billed murre					×					×			
	Cranes													
	Ducks				×							×	×	×
	Geese				×	×				×		×	×	×
	Kittiwakes										×			

^a Most likely to be affected by ocean acidification because of direct effects or a predominance of prey species that will suffer direct effects (see Table 4.6).

4.7 Impact of ocean acidification on the recreational value of Arctic ecosystems

Local people and tourists utilize many parts of the Arctic nature for recreational purposes, and have done so for decades, and even centuries. The tourism industry is one of the fastest growing global industries (Cisneros-Montemayor and Sumaila, 2010), also in the Arctic. In this region, the industry is based both on use values (e.g., fishing and hunting) and non-use values (e.g., whale and landscape watching) that people ascribe to nature. Some consumers are willing to pay a premium to travel from their lower-latitude countries to remote Arctic areas by plane or cruise ship (which contributes to the emission of greenhouse gases) to fish for their preferred trophy fish, though this does not represent the majority of recreational fishermen (Aas, 2008; Borch et al., 2011; NJFF, 2011). For many local people, their appreciation of nature is supplemented by its value as part of their cultural heritage and the fact that it provides a foundation for their commercial livelihoods. Recreational attractions are found on land, in freshwater and at sea.

In 2002, the Earth Summit in Johannesburg focused on sustainable tourism development. The World Tourism Organization defines sustainable tourism as *tourism which leads to management of all resources in such a way that economic, social and aesthetic needs can be fulfilled while maintaining cultural integrity, essential ecological processes, biological diversity and life support systems.* In addition, there is an increasing focus on eco-tourism and responsible tourism. In 2007, the Davos Declaration aimed at increasing knowledge of the links between tourism and climate change. In 2008, the Green Economy Report considered tourism as a main component of the Green Economy Initiative and created the term 'green tourism'. Finally, Rio+20 reinforced the global community's commitment to sustainable development in 2012.

With regard to tourism in the Arctic, there are three categories potentially at risk:

- Land-based tourism in the coastal zone.
- Recreational activities in inshore waters by local people and long-distance foreign tourists.
- Ocean and coastal cruising.

The Arctic marine ecosystem is changing owing to several stressors, including ocean acidification (see Chapters 1, 2 and 3). This will affect lower-trophic (e.g., krill and fish) and higher-trophic (e.g., seabirds and marine mammals) species. Whale watching is a rapidly expanding form of ecotourism that contributes to coastal economies, also in countries with whaling activities, such as Iceland, Norway and the USA (Cisneros-Montemayor and Sumaila, 2010; Cisneros-Montemayor et al., 2010). With the rapidly changing environmental conditions in the Arctic area, whale migration routes may change, thus enabling mixing between populations that was previously inhibited (Heide-Jørgensen et al., 2010). Walrus (*Odobenus rosmarus*) may be affected by ocean acidification (Jay et al., 2011), as well as most cetaceans and pinnipeds, but to what extent is still not known.

Acidification of the ocean may, or may not, have direct and indirect economic and employment effects on tourism activities

in the Arctic region. A major issue is which species are affected, and to what extent, and how these are valued by tourists. To the extent that people ascribe different values to particular species and ecosystems as a whole, the economic effects on the local and national economy may differ. This is due to both natural and social causes, of which the latter includes demographic shifts, income changes, economic structural changes and trade. Recreational fishing has been defined by the FAO as Fishing of aquatic animals that do not constitute the individual's primary resource to meet nutritional needs and are not generally sold or otherwise traded on export, domestic or black markets (quoted from Borch et al., 2011). Marine fishing tourism in Norway is an example of a growing industry based on the recreational value of fish resources and other natural amenities. The long and mainly sheltered Norwegian coastline makes recreational salt-water fishing attractive for increasing numbers of people from other European countries. Historically, domestic people, both rural and urban, have used the fish resources not only for commercial purposes, but also for nutritional needs and recreation. Long distance recreational fishers are still quite few.

The mix of fish species caught in recreational fisheries varies across countries and regions (e.g., Aas, 2008). Recreational fishers are attracted by the fish itself and by the natural amenity services at the fishing spot. Well-managed stocks with fish that taste good, look nice, are big, and are strong fighters (trophy fish) are in greatest demand and fishers are willing to pay for the recreational provision of nature (Toivonen et al., 2004; Flaaten, 2011). Vølstad et al. (2011) provided data on catches in Norway's coastal tourist fishery and from this concluded that tourist fishers behave opportunistically by simply catching what is available. Cod and saithe are the dominant species in the north of Norway, while saithe, mackerel and ling are the major species in the south. But the key question that remains is which of these species will be most affected by global warming and ocean acidification?

If ocean acidification mainly affects crustaceans and mollusks at lower trophic levels, the fishing tourism industry and recreational fishing may hardly be affected unless there is some indirect impact via the food web. Recreational fishing of marine crustaceans is not common in the countries bordering the Arctic Ocean, but there are some exceptions, such as lobster (*Homarus gamarus*) and red king crab (*Paralithodes camtschaticus*) in Norway. Note that red king crab in this country is an alien species that many people think should be eradicated.

Tourists and local recreational fishers in the AMAP area appear adaptive and target those species available in their areas. To the extent that ocean acidification will affect some species more than others this may in the future alter the relative composition of recreational catches by tourists and local residents, but it is not likely that people's overall pleasure and welfare will be affected by such a change. Over this long period, commercial harvests of some species have risen and declined (e.g., North Sea cod, *Gadus morhua*), others have fluctuated, but still sustain great fisheries (e.g., Norwegian spring-spawning herring *Clupea harengus*). As a result of human carbon dioxide emissions, the average acidity of surface ocean waters worldwide is now about 30% higher than at the start of the Industrial Revolution (Chapter 1, Section 1.1). It is not known if and to what extent this 30% increase in ocean acidification has contributed to the decline in some stocks and fluctuations in others or what will be the impacts of future changes in Arctic Ocean acidification.

Fishing is important for recreational activities of tourists and local people in the Arctic. It may therefore be of grave concern if important species are impacted by acidification, directly or indirectly through effects on important prey species as well as on the phytoplankton and zooplankton communities. Increased research into ocean acidification and its implications for specific ecosystems and stocks now takes place in many countries, also in the Arctic (e.g., Cooley et al., 2009; IMR, 2011). A study of economic impacts of global warming on the Barents Sea fisheries indicated that the management system in operation seems to be of greater importance for future fishery performance than possible environmental changes caused by global warming (Eide, 2007).

4.8 Marine management in the face of ocean acidification

Identifying specific management measures to stem the impacts of ocean acidification is virtually impossible because ocean acidification is occurring in combination with many other anthropogenic stressors affecting marine ecosystems. Hence, in addition to mitigating the emissions of greenhouse gases, it is crucial that marine ecosystems are strengthened, for example by not overfishing, to make them sufficiently resilient to withstand the effects of multiple stressors, including ocean acidification. To this end, the following management measures need to be reinforced where they exist or introduced where they do not.

- In 1996, Arctic nations put in place a cooperative framework, embodied in the Arctic Council, where a priority given to scientific research and with less emphasis placed on remedial action. The existing arrangement is voluntary, with few resources for cooperative action. Changes in the Arctic are due to global factors (ACIA, 2005), but the consequences (ecological aspects and development) will occur at the regional level. Arctic nations would have to fight global developments through cooperation and mutual efforts to protect their marine environment and their traditional way of life.
- There is a need to establish and maintain, with appropriate management, marine protected areas to increase resilience of marine habitats to ocean acidification and support fish and shellfish populations for sustainable marine harvests. Marine protected areas need to be designed carefully in order to be successful (Polacheck, 1990; Sumaila et al., 2000; Fernandes et al., 2009; Flaaten and Mjølhus, 2010; Osmond et al., 2010).
- Adaptive management of aquaculture species and methods (e.g., selection of resistant species, and selective breeding and improvement of aquaculture practices) might provide solutions to impacts of ocean acidification on specific seafood sources to enhance future economic and social opportunities.
- Arctic nations need to coordinate their efforts to promote sustainable development of the natural and living resources and of the region's people. New development opportunities and threats will also appear from expanding economic

activities (new shipping routes, oil and gas exploration, new commercial fishing). These create new management challenges that require cooperation between the different Arctic nations and provide a transboundary approach that includes a monitoring system to evaluate the health of ecosystems.

- The initiative for an ecosystem-based management system for the Arctic environment adopted by the Arctic Council needs to be further strengthened.
- The Arctic region needs a sustainable tourism strategy. This is because tourism is predicted to increase at the same time as the impacts of ocean acidification on tourism are predicted to increase.
- It would be beneficial to apply traditional knowledge in marine management systems to tackle ocean acidification and other global change multi-stressors. See Binder and Hanbidge (1993), Joks (2003), Manseau et al. (2005) and Berkes et al. (2007), for discussions of the role of traditional knowledge in the management of marine ecosystems.

4.9 Concluding remarks

Fish stocks would be more resilient to ocean acidification if the combined stresses from overfishing, habitat degradation, pollution runoff, land-use transformation, competing aquatic resource uses and other anthropogenic factors are minimized. People have had to adapt to changes in fish stocks and catches over time owing to other factors, such as seasonal, inter-annual and multi-decadal climate variability. Experience gathered over the years could inform the adaptation strategies needed even though the scale and speed of the expected changes in relation to ocean acidification are much greater than have been observed to date. Possible adaptation strategies in the face of dwindling fish stocks include restricting the use of particular gear types and the implementation of livelihood diversification measures.

Adaptation and mitigation measures can either be carried out in anticipation of future effects, or in response to effects once they have occurred. Some can be implemented via public institutions, others by private individuals. Responses to the direct effects of adverse events on fisheries infrastructure and communities may be more effective if they are anticipatory, as part of a long-term participatory, broad-based approach to fisheries management (Sumaila et al., 2011). Such an integrated approach has the potential to increase ecosystem and community resilience and to provide a valuable framework for dealing with ocean acidification.

Ocean acidification is expected to affect fish stocks, marine ecosystems and the commercial, subsistence, and recreational fisheries of the Arctic. However, many aspects of this issue are still unknown. Interdisciplinary studies that investigate fisheries responses to ocean acidification are needed. This contribution is therefore only a start in the long journey to determine the bio-economic effects of ocean acidification in the Arctic. Ultimately, reducing carbon dioxide emissions is the only way to control increases in ocean acidification. Thus, it is important that while adaptation measures may help to some degree, all with interests in the fisheries of the Arctic sector should strengthen the case for lower carbon dioxide emissions. Lead Authors: Richard Bellerby, Howard I. Browman, U. Rashid Sumaila

5.1 The changing Arctic

In concert with the rising atmospheric concentration of carbon dioxide (CO₂), the ocean inorganic carbon inventory is increasing due to a net air-sea flux of CO₂ at the surface. Atmospheric CO₂ levels are the highest for at least 800 000 years and, generally, global ocean surface CO₂ increases are following the atmospheric change. Consequently, the pH of the ocean is going down and the relative speciation of inorganic carbon is changing at a rate likely to have been unsurpassed over the past 55 million years (Kump et al., 2009). This process has been termed ocean acidification; following the increase in the concentration of hydrogen ions in seawater. The Arctic is inherently susceptible to ocean acidification as it has a low buffer capacity and as such will exhibit greater changes in ocean acidification per unit CO₂ increase. As in most of the world oceans, the major driver of ocean acidification in the Arctic Ocean is directly attributable to the increasing atmospheric carbon load following the combustion of fossil fuels and land use changes. However, rapid ocean warming and ice melt are accelerating ocean acidification over most of the Arctic. Carbonate chemistry responds not only to the trends in atmospheric CO, but also to shorter-term ocean climate variability. The solubility and biological pumps exert a huge influence on seasonal changes in carbonate chemistry, where the greatest surface annual changes are associated with the seasonal-ice zone over the Arctic shelves. Ocean acidification will affect some nutrient and micronutrient availability for biological production and may change the bioavailability of some trace metals. The Arctic Ocean has already crossed significant geochemical thresholds following ocean acidification and, without significant reductions in atmospheric CO₂ concentrations, ocean-wide aragonite (one of the two most common biogenic calcium carbonate minerals in the oceans) undersaturation will occur. These changes will be long-lasting with high levels of ocean acidification persisting for many thousands of years unless CO, emissions are curtailed. Even if emissions peak today, the system will not return to pre-industrial levels for tens of thousands of years. Ocean acidification is only one of the stressors on the Arctic Ocean system. In concert with ongoing ocean warming and freshening are changes to ocean productivity and heterotrophy which, along with the increasing organic carbon supply are rapidly changing the oxygen distributions in the Arctic. Changing sea-ice cover is changing the light and turbulence regimes and therefore impacting on air-sea fluxes and ocean productivity. All these processes interact with the carbon cycle and thus may either compound or alleviate ocean acidification.

5.2 Experimental direction

There is an almost complete lack of information on the effects of ocean acidification – in isolation or in combination with other environmental stressors such as temperature, salinity, oxygen concentration – on keystone species and processes in the Arctic. Accordingly, there is an urgent need for focused research on

the likely impacts of ocean acidification on a range of taxa and processes. Notwithstanding logistic difficulties, experiments should be conducted in situ whenever possible and/or using pelagic and benthic mesocosms. It will also be important to conduct experiments over timescales that are more relevant ecologically than the very short-term experiments typical of this field to date. To increase confidence in the conclusions and interpretations of experiments, multiple biological and ecological end-points should be assessed simultaneously. The responses of key components of Arctic and sub-Arctic food webs to ocean acidification must be assessed and compared with those observed for organisms in other geographic regions. Such studies would need, for example, to consider the following: viral and microbial communities; heterotrophic and mixotrophic protists; planktonic and under-ice microalgae, including how seasonal changes in sea-ice coverage influence primary production and carbon transfer through the food web; the effects of ocean acidification on Arctic coral species and the effects of acidification-mediated loss of unprotected coral structure on coral-associated biodiversity; crustaceans, especially the copepods Calanus finmarchicus, C. hyperboreus and C. glacialis, amphipods (Themisto libelulla), pteropods (mostly Limacina helicina) and commercially important crabs; fish, especially Arctic cod (Arctogadus glacialis), polar cod (Boreogadus saida - termed 'Arctic cod' in North America), and capelin (Mallotus villosus); seabirds; and marine mammals (e.g., to assess the dietary flexibility of seabirds and marine mammals in switching to food sources not affected by ocean acidification). Adaptive capacity (over multiple generations) and environmental plasticity of all stages of an organism's life-cycle must be assessed. This could include (but need not be limited to): longterm transgenerational and selective-breeding experiments; long-term studies investigating adaptations and acclimation capacity; long-term studies investigating the potential for effects to 'carry-over' to subsequent life stages and/or generations; the issue of whether the increased generation time of many Arctic species confers reduced adaptive capacity to near-future climate change (in comparison to temperate and tropical species); multi-population studies to quantify genetic and phenotypic intra-specific variation of key traits influencing fitness; effects of multiple stressors (ocean acidification, carbonate saturation state, temperature, salinity, oxygen, toxins...) on species-level and ecosystem-level processes (including trophic interactions); and the effects of food limitation / nutritional status on specieslevel and ecosystem-level processes (e.g., do Arctic species show the same amelioration of ocean acidification impacts under high food / nutrient levels?).

5.3 Socio-economic advances

A comprehensive assessment of the socio-economic consequences of ocean acidification needs to build upon a solid scientific understanding of the chemical changes and biophysical effects of these changes. As stated at the start of the previous section, "There is [at present] an almost complete lack of information on the effects of ocean acidification" on the biophysics of the ocean. Given this limitation, the only avenue left to economists and other social scientists is to rely on theoretical and simulation modeling together with scenario building to keep advancing our socio-economic understanding of the potential impacts of ocean acidification.

Based on these approaches, it is concluded that fish stocks will be more resilient to ocean acidification if the combined stresses from overfishing, habitat degradation, pollution runoff, landuse transformation, competing aquatic resource uses and other anthropogenic factors are minimized. People have had to adapt to changes in fish stocks and catches over time resulting from fishing pressure combined with other factors, such as seasonal, inter-annual and multi-decadal climate variability. Experience gathered over the years could inform the adaptation strategies needed; however, the scale and speed of the expected changes in relation to ocean acidification and ongoing climate change are generally much greater than have been encountered to date. Possible adaptation strategies in the face of dwindling fish stocks include restricting the use of particular gear types and the implementation of livelihood diversification measures. Adaptation and mitigation measures can either be carried out in anticipation of future effects, or in response to impacts once they have occurred. Some can be implemented via public institutions, others by private individuals. Responses to the direct effects of adverse events on fisheries infrastructure and communities may be more effective if they are anticipatory, as part of a long-term participatory, broad-based approach to fisheries management. Such an integrated approach has the potential to increase ecosystem and community resilience and to provide a valuable framework for dealing with ocean acidification.

Ocean acidification is expected to affect fish stocks, marine ecosystems and the commercial, subsistence, and recreational fisheries of the Arctic. However, many aspects of this issue are still unknown. Interdisciplinary studies that investigate fisheries responses to ocean acidification are needed. This contribution is therefore only a start in the long journey to determining the bio-economic effects of ocean acidification in the Arctic. Ultimately, reducing CO_2 emissions is the only way to control increases in ocean acidification. While adaptation measures may help to some degree, all with interests in the Arctic fisheries should strengthen the case for lower CO_2 emissions.

5.4 An interdisciplinary approach

The important advances that have come from recent cooperation between many scientific research disciplines and marine resource end-users (including fishers and indigenous peoples groups) illustrate an awareness of the need to collaborate at the science and policy levels in order to understand the extent and consequences of Arctic Ocean acidification. It is apparent that the changes in Arctic Ocean acidification are happening faster than in any other ocean basin. This requires a unique research perspective that challenges the paradigm that ocean carbonate chemistry follows atmospheric CO_2 increases directly. Despite the lack of *in situ* data, current understanding of the marine carbonate system is sufficient to state that ocean CO_2 will increase faster in the Arctic than elsewhere and that critical thresholds for particular geochemical and biological processes will be reached first in the Arctic. As a consequence, experimentation on organisms and ecosystems and the interpretation of experimental results of biotic response should not be restricted to only following the 'average state' or timescales implied by the scenarios of atmospheric CO_2 evolution developed by the Intergovernmental Panel on Climate Change (IPCC). Doing so could result in incomplete or inadequate information to inform the policy debate. Instead, each unique habitat, ecosystem or biome should be studied to determine the potential impacts of future ocean acidification on it. The information on future ocean acidification should dictate the range of conditions that need to be addressed in biological experimentation. This will in-turn furnish relevant information for climate mitigation and socioeconomic policy measures.

5.5 Monitoring networks

The Arctic is one of the least understood regions on the planet and current understanding of the marine components of the Arctic carbon cycle is based largely on very few measurements combined with extrapolation of what is known about the functioning of other ocean systems. It is of paramount importance that long-term, dedicated marine carbonate system observation programs are developed. As the Arctic is undergoing changes in other biogeochemical and ecological stressors, these programs should be integrated within a framework that monitors the changes in the key variables including oxygen, temperature, salinity and nutrients. These programs should be in the form of repeated surveys, in situ observational platforms such as moorings, drifting buoys and ocean gliders. New technologies are required to enable measurements under Arctic conditions where the exceptional challenges, including low temperatures and ice cover, require advances in instrument design. It is imperative that the physicochemical monitoring is coordinated closely with biological observations. Management of platform design, observational logistics and data management should, preferably, be coordinated at the international level.

References

Aagaard, K. and E.C. Carmack, 1989. The role of sea ice and other fresh water in the Arctic circulation. Journal of Geophysical Research, 94:14,485-14,498.

Aagaard, K., L.K. Coachman and E.C. Carmack, 1981. On the halocline of the Arctic Ocean. Deep-Sea Research, 28:529-545.

Aas, Ø. 2008. Global Challenges in Recreational Fisheries. Øystein, A.A.S. (Ed.). Blackwell.

Abrahamsen, E.P., M.P. Meredith, K.K. Falkner, S. Torres-Valdes, M.J. Lemg, M.B. Alkire, S. bacon, S.W. Laxon, I. Polyakov and V. Ivanov, 2009. Tracer-derived freshwater composition of the Siberian continental shelf and slope following the extreme Arctic summer of 2007. Geophysical Research Letters, 36: L07602.

ACIA, 2005. Arctic Climate Impact Assessment. Cambridge University Press, 1042p.

Aguilar-Islas, A.M., M.P. Hurst, K.N. Buck, B. Sohst, G.J. Smith, M.C. Lohan and K.W. Bruland, 2007. Micro- and macronutrients in the southeastern Bering Sea: Insight into iron-replete and iron-depleted regimes. Progress in Oceanography, 73:99-126.

AHDR, 2004, Arctic Human Development Report. Stefansson Arctic Institute. Cambridge University Press.

Ahner, B.A., N.M. Price and F.M.M. Morel, 1994. Phytochelatin production by marine phytoplankton at low free metal ion concentrations: Laboratory studies and filed data from Massachusetts Bay. Proceedings of the National Academy of Sciences, 91:8433-8436.

Ainsworth, C.H., J.F. Samhouri, D.S. Busch, W.W.L. Cheung, J. Dunne and T.A. Okey, 2011. Potential impacts of climate change on Northeast Pacific marine foodwebs and fisheries. ICES Journal of Marine Science, 68:1217-1229.

Al-Horani, F.A., S.M. Al-Moghrabi and D. de Beer, 2003. The mechanism of calcification and its relation to photosynthesis and respiration in the scleractinian coral *Galaxea fascicularis*. Marine Biology, 142:419-426.

Albright, R., B. Mason, M. Miller and C. Langdon, 2010. Ocean acidification compromises recruitment success of the threatened Caribbean coral *Acropora palmata*. Proceedings of the National Academy of Sciences of the United States of America, 107:20400-20404.

Alkire, M.B., K.K. Falkner, I. Rigor, M. Steele and J. Morison, 2007. The return of Pacific waters to the upper layers of the central Arctic Ocean. Deep Sea Research I, 54:1509-1529.

Allgaier, M., U. Riebesell, M. Vogt, R. Thyrhaug and H.P. Grossart, 2008. Coupling of heterotrophic bacteria to phytoplankton bloom development at different pCO_2 levels: a mesocosm study. Biogeosciences, 5:1007-1022.

Alling, V., D. Porcelli, C.M. Morth, L.G. Anderson, L. Sanchez-Garcia, O. Gustafsson, P.S. Andersson and C. Humborg, 2012. Degradation of terrestrial organic carbon, primary production and out-gassing of CO_2 in the Laptev and East Siberian Seas as inferred from delta C-13 values of DIC. Geochimica et Cosmochimica Acta, 95:143-159.

Alonso-Saez, L, A.S. Waller, D.R. Mende, K.Bakker, H. Farnelid, P.L. yager, C. Lovejoy, J-E. Tremblay, M. Potvin, F. Heinrich, M. Estrada, L. Riemann, P. Bork, C. Pedros-Alio and S. Bertilisson, 2012. Role for urea in nitrification by polar marine Archaea. Proceedings of the National Academy of Sciences, 109:17732-17733.

AMAP, 1997. Arctic Pollution Issues: A State of the Arctic Environment Report. Arctic Monitoring and Assessment Programme, Oslo, Norway, xii+188 pp.

AMAP, 1998. AMAP Assessment Report: Arctic Pollution Issues. Arctic Monitoring and Assessment Programme, Oslo, Norway, 859 pp.

AMAP, 2002. Arctic Pollution 2002: Persistent Organic Pollutants, Heavy Metals, Radioactivity, Human Health, Changing Pathways. Arctic Monitoring and Assessment Programme, Oslo, Norway, xii+112 pp.

AMAP, 2009. Arctic Pollution 2009. Arctic Monitoring and Assessment Programme, Oslo, Norway, xi+83pp.

AMAP, 2011a. Snow, Water, Ice and Permafrost in the Arctic (SWIPA): Climate Change and the Cryosphere. Arctic Monitoring and Assessment Programme, Oslo, Norway, xii+538 pp.

AMAP, 2011b. AMAP Assessment 2011: Mercury in the Arctic. Arctic Monitoring and Assessment Programme, Oslo, Norway, xiv+193 pp.

AMAP, 2012. Arctic Climate Issues 2011: Changes in Arctic Snow, Water, Ice and Permafrost. SWIPA 2011 Overview Report. Arctic Monitoring and Assessment Programme (AMAP), Oslo. 112 pp.

Amaral, V., H.N. Cabral and M.J. Bishop, 2011. Resistance among wild invertebrate populations to recurrent estuarine acidification. Estuarine Coastal and Shelf Science, 93:460-467.

Amels, P., H. Elias and K-J. Wannowius, 1997. Kinetics and mechanism of the oxidation of dimethyl sulfide by hydroperoxides in aqueous medium: A study on the potential contribution of liquid-phase oxidation of dimethyl sulfide in the atmosphere. Journal of the Chemical Society, Faraday Transactions, 93:2537-2544.

Amiel, D. and J.K. Cochran, 2008. Terrestrial and marine POC fluxes derived from ²³⁴Th distributions and δ^{13} C measurements on the Mackenzie Shelf. Journal of Geophysical Research, 113:C03S06.

Amiel, D., J.K. Cochran and D.J. Hirschberg, 2002. ²³⁴Th/²³⁸U disequilibrium as an indicator of the seasonal export flux of particulate organic carbon in the North Water. Deep-Sea Research II, 49:5191-5209.

Amon, R.M.W., H.P. Fitznar and R. Benner, 2001. Linkages among the bioreactivity, chemical composition, and diagenetic state of marine dissolved organic matter. Limnology and Oceanography, 46:287-297.

Andersen, M.B., C.H. Stirling, D. Porcelli, A.N. Halliday, P.S. Andersson and M. Baskaran, 2007. The tracing of riverine U

in Arctic seawater with very precise ²³⁴U/²³⁸U measurements. Earth and Planetary Science Letters, 259:171-185.

Anderson, L.G. and A. Olsen, 2002. Air-sea flux of anthropogenic carbon dioxide in the North Atlantic. Geophysical Research Letters, 29:16-1–16-4.

Anderson, L.G., K. Olsson and M. Chierici, 1998. A carbon budget for the Arctic Ocean. Global Biogeochemical Cycles, 12:455-465.

Anderson, L.G., E. Falck, E.P. Jones, S. Jutterström and J.H. Swift, 2004. Enhanced uptake of atmospheric CO_2 during freezing of seawater: a field study in Storfjorden, Svalbard. Journal of Geophysical Research, 109:C06004.

Anderson, L.G., T. Tanhua, G. Björk, S. Hjalmarsson, E.P. Jones, S. Jutterström, B. Rudels, J.H. Swift and I. Wåhlstöm, 2010. Arctic ocean shelf-basin interaction: An active continental shelf CO_2 pump and its impact on the degree of calcium carbonate solubility. Deep-Sea Research, 57:869-879.

Anderson, L.G., G. Bjork, S. Jutterstrom, I. Pipko, N. Shakhova, I. Semiletov and I. Wahlstrom, 2011a. East Siberian Sea, an Arctic region of very high biogeochemical activity. Biogeosciences, 8:1745-1754.

Anderson, L.G., G. Björk, S. Jutterström, I. Pipko, N. Shakhova, I.P. Semiletov and I. Wåhlström, 2011b. East Siberian Sea, an Arctic region of very high biogeochemical activity. Biogeosciences Discussions, 8:1137-1167.

Andersson, P.S., D. Porcelli, M. Frank, G. Björk, R. Dahlqvist and O. Gustafsson, 2008. Neodymium isotopes in seawater from the Barents Sea and Fram Strait Arctic-Atlantic gateways. Geochimica et Cosmochimica Acta, 72:2854-2867.

Anthony, K.R.N., D.I. Kline, G. Diaz-Pulido, S. Dove and O. Hoegh-Guldberg, 2008. Ocean acidification causes bleaching and productivity loss in coral reef builders. Proceedings of the National Academy of Sciences of the United States of America, 105:17442-17446.

Archer, D.A., 2005. Fate of fossil fuel CO_2 in geologic time. Journal of Geophysical Research, 110:C09S05.

Archer, D., M. Eby, V. Brovkin, A. Ridgwell, L. Cao, U. Mikolajewicz, K. Caldeira, K. Matsumoto, G. Munhoven, A. Montenegro and K. Tokos, 2009a. Atmospheric lifetime of fossil fuel carbon dioxide. Annual Reviews of Earth and Planetary Sciences, 37:117-134.

Archer, D., B. Buffett and V. Brovkin, 2009b. Ocean methane hydrates as a slow tipping point in the global carbon cycle. Proceedings of the National Academy of Sciences, 106:20596-20601.

Archer, S.D., S.A. Kimmance, J.A. Stephens, F.E. Hopkins, R.G.J. Bellerby, K.G. Schulz, J. Piontek and A. Engel, 2012. Contrasting responses of DMS and DMSP to ocean acidification in Arctic waters. Biogeosciences Discussions, 9:12803-12843.

Ardelan, M.V. and E. Steinnes, 2010. Changes in mobility and solubility of the redox sensitive metals Fe, Mn and Co at the seawater-sediment interface following CO_2 seepage. Biogeosciences, 7:569-583. Armstrong, C., S. Holen, S. Navrud and I. Seifert, 2012. The Economics of Ocean Acidification – a scoping study. Fram Centre. 57 pp.

Arnold, A.J. and W.C. Parker, 1999. Biogeography of Planktonic Foraminifera. In: Sen Gupta, B.K. (Ed.). Modern Foraminifera, pp. 103-122. Kluwer Academic Publishers.

Arnold, K.E., C. Wells and J.I. Spicer, 2009. Effect of an insect juvenile hormone analogue, Fenoxycarb (R) on development and oxygen uptake by larval lobsters *Homarus gammarus* (L.). Comparative Biochemistry and Physiology C-Toxicology and Pharmacology, 149:393-396.

Arora, V., J. Scinocca, G. Boer, J. Christian, K. Denman, G. Flato, V. Kharin, W. Lee and W. Merryfield, 2011. Carbon emission limits required to satisfy future representative concentration pathways of greenhouse gases. Geophysical Research Letters, 38:L05, 805.

Arrigo, K.R. and G.L. van Dijken, 2004. Annual cycles of sea ice and phytoplankton in Cape Bathurst polynya, southeastern Beaufort Sea, Canadian Arctic. Geophysical Research Letters, 31: L08304.

Arrigo, K.R., T. Mock and M.P. Lizotte, 2010. Primary producers in sea ice. In: Thomas, D.N. and G.S. Dieckmann (Eds.). Sea Ice, pp. 283-326. Second Edition, Wiley.

Arrigo, K.R., D.K. Perovich, R.S. Pickart and 28 others, 2012. Massive phytoplankton blooms under Arctic sea ice. Science, 336:1408.

Årthun, M., R.G.J. Bellerby and C. Schrum, 2012. Spatiotemporal variability of air–sea CO₂ fluxes in the Barents Sea, as determined from empirical relationships and modeled hydrography. Journal of Marine Systems, 98-99:40-50.

Aschan, M., O.V. Karamushko, I. Byrkjedal, R. Wienerroither, I.V. Borkin and J.S. Christiansen, 2009. Records of the gadoid fish *Arctogadus glacialis* (Peters, 1874) in the European Arctic. Polar Biology, 32:963-970.

Ashjian, C.J., S.R. Braund, R.G. Campbell, J.C. George, J. Kruse, W. Maslowski, S.E. Moore, C.R. Nicolson, S.R. Okkonen, B.F. Sherr, E.B. Sherr and Y.H. Spitz, 2010. Climate variability, oceanography, bowhead whale distribution, and inupiat subsistence whaling near Barrow, Alaska. Arctic, 63:179-194.

Aslaksen, I., W. Dallmann, D.L. Holen, E. Høydahl, J. Kruse, B. Poppel, M. Stapleton and E.I. Turi, 2009. Interdependency of subsistence and market economies in the Arctic. In: Glomsrød, S. and I. Aslaksen (Eds.). The Economy of the North 2008, pp.75-98. Statistics Norway, Oslo.

Avgoustidi, V., P.D. Nightingale, I. Joint, M. Steinke, S.M. Turner, F.E. Hopkins and P.S. Liss, 2012. Decreased marine dimethyl sulfide production under elevated CO_2 levels in mesocosm and in vitro studies. Environmental Chemistry, 9:399-404.

Azetsu-Scott, K., M. Starr, M. Harvey, F. Saucier, C. Hillaire-Marcel and D. Slauenwhite, 2008. Freshwater and carbon dynamics in Hudson Bay: Results from MERICA 2003-2006. In: Arctic Change 2008, Québec, QC, 9-12 December, 2008.

Azetsu-Scott, K., A. Clarke, K. Falkner, J. Hamilton, E.P. Jones, C. Lee, B. Petrie, S. Prinsenberg, M. Starr and P. Yeats, 2010.

Calcium carbonate saturation states in the waters of the Canadian Arctic Archipelago and the Labrador Sea. Journal of Geophysical Research, 115:C11021.

Bacon, M.P., C-A. Huh and R.M. Moore, 1989. Vertical profiles of some natural radionuclides over the Alpha Ridge, Arctic Ocean. Earth and Planetary Science Letters, 95:15-22.

Barash, M.S., T.A. Khusid, A.G. Matul, M.P. Chekhovskaya, N. Biebow, D. Nuernberg and R. Tiedemann, 2008. Distribution of benthic foraminifera in Upper Quaternary sediments of the Deryugin Basin (Sea of Okhotsk). Oceanology, 48:105-113.

Barcelos e Ramos, J., H. Biswas, K.G. Schulz, J. LaRoche and U. Riebesell, 2007. Effect of rising atmospheric carbon dioxide on the marine nitrogen fixer *Trichodesmium*. Global Biogeochemical Cycles, 21:GB2028.

Barry, J.P., K.R. Buck, C.F. Lovera, L. Kuhnz, P.J. Whaling, E.T. Peltzer, P. Walz and P.G. Brewer, 2004. Effects of direct ocean CO_2 injection on deep-sea meiofauna. Journal of Oceanography, 60:759-766.

Barry, J.P., S. Widdicombe and J.M. Hall-Spencer, 2011. Effects of ocean acidification on marine biodiversity and ecosystem function. In: Gattuso, J.P. and L. Hansson (Eds.). Ocean Acidification, pp. 192-209. Oxford University Press.

Bates, N.R., 2006. Air-sea CO 2 fluxes and the continental shelf pump of carbon in the Chukchi Sea adjacent to the Arctic Ocean. Journal of Geophysical Research, 111(C10): 1-21.

Bates, N. and J. Mathis, 2009. The Arctic Ocean marine carbon cycle: Evaluation of air-sea CO₂ exchanges, ocean acidification impacts and potential feedbacks. Biogeosciences, 6:2433-2459.

Bates, N.R., M.H.P. Best and D.A. Hansell, 2005. Spatiotemporal distribution of dissolved inorganic carbon and net community production in the Chukchi and Beaufort Seas. Deep-Sea Research Part II, 52: 3303-3323.

Bates, N.R., S.B. Moran, D.A. Hansell and J.M. Mathis, 2006. An increasing CO_2 sink in the Arctic Ocean due to sea-ice loss? Geophysical Research Letters, 33:L23609.

Bates, N.R., J.T. Mathis and L.W. Cooper, 2009. Ocean acidification and biologically induced seasonality of carbonate mineral saturation states in the western Arctic Ocean. Journal of Geophysical Research, 114:C11007.

Baumann, H., S.C. Talmage and C.J. Gobler, 2012. Reduced early life growth and survival in a fish in direct response to increased carbon dioxide. Nature Climate Change, 2:38-41.

Beaugrand, G., M. Edwards, K. Brander, C. Luczak and F. Ibanez, 2008. Causes and projections of abrupt climate-driven ecosystem shifts in the North Atlantic. Ecology Letters, 11:1157-1168.

Bednarsek, N., G.A. Tarling, D.C.E. Bakker, S. Fielding, E.M. Jones, H.J. Venables, P. Ward, A. Kuzirian, B. Leze, R.A. Feely and E.J. Murphy, 2012. Extensive dissolution of live pteropods in the Southern Ocean. Nature Geoscience, 5:881-885.

Bellerby, R.G.J., A. Olsen, T. Furevik and L.A. Anderson, 2005. Response of the surface ocean CO_2 system in the Nordic Seas and North Atlantic to climate change. In: Drange, H., T.M. Dokken, T. Furevik, R. Gerdes and W. Berger (Eds.). Climate Variability in the Nordic Seas. Geophysical Monograph Series, 158:189-198.

Bellerby, R.G.J., K.G. Schulz, U. Riebesell, C. Neill, G. Nondal, E. Heegaard, T. Johannessen and K.R. Brown, 2008. Marine ecosystem community carbon and nutrient uptake stoichiometry under varying ocean acidification during the PeECE III experiment. Biogeosciences, 5:1517-1527.

Bellerby, R.G.J., A. Silyakova, G. Nondal, D. Slagstad, J. Czerny, T. Lange and A. Ludwig, 2012. Marine carbonate system evolution during the EPOCA Arctic pelagic ecosystem experiment in the context of simulated Arctic ocean acidification. Biogeosciences Discussions, 9:15541-15565.

Beman, J.M., C-E. Chow, A.L. King, Y. Feng, J.A. Fuhrman, A. Andersson, N.R. bates, B.N. Popp abnd D.A. Hutchins, 2011. Global declines in oceanic nitrification rates as a consequence of ocean acidification. Proceedings of the National Academy of Sciences, 108:208-213.

Benoit, D., Y. Simard and L. Fortier, 2008. Hydroacoustic detection of large winter aggregations of Arctic cod (*Boreogadus saida*) at depth in ice-covered Franklin Bay (Beaufort Sea). Journal of Geophysical Research, 113:C06S90.

Bergmann, M.A., H.E. Welch, J.E. Butler-Walker and T.D. Siferd, 1991. Ice algal photosynthesis at Resolute and Saqvaqjuac in the Canadian Arctic. Journal of Marine Systems, 2:43-52.

Bergsten, H., 1994. Recent benthic foraminifera of a transect from the North Pole to the Yermak Plateau, eastern central Arctic Ocean. Marine Geology, 119:251-267.

Berkes, F., M.K. Berkes and H. Fast, 2007. Collaborative integrated management in Canada's North: The role of local and traditional knowledge and community-based monitoring. Coastal Management, 35:143-162.

Biastoch, A., T. Treude, L.H. Rüpke, U. Riebesell, C. Roth, E.B. Burwicz, W. Park, M. Latif, C.W. Bönning, G. Madec and K. Wallmann, 2011. Rising Arctic Ocean temperatures cause gas hydrate destabilization and ocean acidification. Geophysical Research Letters, 38:L08602.

Bibby, R., S. Widdicombe, H. Parry, J. Spicer and R. Pipe, 2008. Effects of ocean acidification on the immune response of the blue mussel *Mytilus edulis*. Aquatic Biology, 2:67-74.

Binder, L.N. and B. Hanbidge, 1993. Aboriginal people and resource co-management. The Inuvialuit of the western Arctic and resource co-management under a land claims settlement. In: Inglis, J.T. (Ed.). Traditional Ecological Knowledge. Concepts and Cases. Canadian Museum of Nature, Ottawa.

Bishop, J.K.B., 1989. Regional extremes in particulate matter composition and flux: effects on the chemistry of the ocean interior. In: Berger, W.H., V.S. Smetacek and G. Wefer (Eds.). Productivity of the Ocean: Present and Past, pp. 117-137. Wiley & Sons.

Blackford, J.C., 2010. Predicting the impacts of ocean acidification: Challenges from an ecosystem perspective. Journal of Marine Systems, 81:12-18.

Blais, M., J.-É. Tremblay, A.D. Jungblut, J. Gagnon, J. Martin, M. Thaler and C. Lovejoy, 2012. Nitrogen fixation and

identification of potential diazotrophs in the Canadian Arctic. Global Biogeochemical Cycles, 26:GB3022.

Borch, T., M. Moilanen and F. Olsen, 2011. Marine fishing tourism in Norway: structure and economic effects. Økonomisk Fiskeriforskning, 1:1-17.

Borowitzka, M.A. and A.W.D. Larkum, 1976. Calcification in the green alga *Halimeda*. III. The sources of inorganic carbon for photosynthesis and calcification and a model of the mechanism of calcification. Journal of Experimental Botany, 27:879-893.

Borriss, M., E. Helmke, R. Hanschke and T. Schweder, 2003. Isolation and characterization of marine psychrophilic phagehost systems from Arctic sea ice. Extremophiles, 7:377-384.

Bowman, J.S., S. Rasmussen, N. Blom, J.W. Deming, S. Rysgaard and T. Sicheritz-Ponten, 2012. Microbial community structure of Arctic multiyear sea ice and surface seawater by 454 sequencing of the 16S RNA gene. Isme Journal, 6:11-20.

Bradstreet, S.W., K.J. Finley, A.D. Sekerak, W.B. Griffith, M.F. Fabijan, C.R. Evans and H.E. Stallard, 1986. Aspects of the biology of Arctic cod (*Boreogadus saida*) and its importance in arctic marine food chains. Canadian Technical Reports of Fisheries and Aquatic Sciences, No. 1491.

Brauner, C. J., 2008. Acid-base balance. In: Finn, R.N. and B. G. Kapoor (Eds.). Fish Larval Physiology, pp. 185–198. Science Publishers.

Breitbarth, E., R.L. Bellerby, C.C. Neill, M.V. Ardelan, M. Meyerhöfer, E. Zöllner, P.L. Croot and U. Riebessell, 2010. Ocean acidification affects iron speciation during a coastal seawater mesocosm experiment. Biogeosciences, 7:1065-1073.

Brinkmeyer, R., K. Knittel, J. Jurgens, H. Weyland, R. Amann and E. Helmke, 2003. Diversity and structure of bacterial communities in Arctic versus Antarctic pack ice. Applied and Environmental Microbiology, 69:6610-6619.

Broecker, W.S. and T-H. Peng, 1982. Tracers in the Sea. Lamont-Doherty Geological Observatory, Columbia University, Palisades, New York. 690 pp.

Brown, M.V. and J.P. Bowman, 2001. A molecular phylogenetic survey of sea-ice microbial communities (SIMCO). FEMS Microbiology Ecology, 35:267-275.

Brown, C.W. and J.A. Yoder, 1994. Coccolithophorid blooms in the global ocean. Journal of Geophysical Research, 99:7467-7482.

Brugel, S., C. Nozais, M. Poulin, J-É Tremblay, L.A. Miller, K.G. Simpson, Y. Gratton and S. Demers, 2009. Phytoplankton biomass and production in the southeastern Beaufort Sea in autumn 2002 and 2003. Marine Ecology Progress Series, 377:63-77.

Bruland, K.W., 1989. Complexation of zinc by natural organic ligands in the central North Pacific. Limnology and Oceanography, 34:269-285.

Bruland, K.W., 1992. Complexation of cadmium by natural organic ligands in the central North Pacific. Limnology and Oceanography, 37:1008-1017.

Buitenhuis, E.T., H.J.W. de Baar and M.J.W. Veldhuis, 1999. Photosynthesis and calcification by *Emiliania huxleyi* (Prymnesiophyceae) as a function of inorganic carbon species. Journal of Phycology, 35:949-959.

Burkhardt, S., U. Riebesell and I. Zondervan, 1999. Effects of growth rate, CO_2 concentration, and cell size on the stable carbon isotope fractionation in marine phytoplankton. Geochimica et Cosmochimica Acta, 63:3729-3741.

Byrne, R.H., 2010. Comparative carbonate and hydroxide complexation of cations in seawater. Geochimica et Cosmochimica Acta, 74:4312-4321.

Byrne, R.H., L.R. Kump and K.J. Cantrell, 1988. The influence of temperature and pH on trace metal speciation in seawater. Marine Chemistry, 25:163-181.

Cai,W.-J., L. Chen, B. Chen, Z. Gao, S.H. Lee, J. Chen, D. Pierrot, K. Sullivan, Y.Wang, X. Hu, W.-J. Huang, Y. Zhang, S. Xu, A. Murata, J.M. Grebmeier, E.P. Jones and H. Zhang, 2010. Decrease in the CO_2 uptake capacity in an ice-free Arctic ocean basin. Science, 329:5991.

Caldeira, K. and M.E. Wickett, 2003. Anthropogenic carbon and ocean pH. Nature, 425:365.

Cameron-Smith, P., S. Elliott, M. Maltrud, D. Erickson and O. Wingenter, 2011. Changes in dimethyl sulfide oceanic distribution due to climate change. Geophysical Research Letters, 38:L07704.

Capodaglio, G., G. Scarponi, G. Toscano and P. Cescon, 1991. Cadmium complexation in surface seawater of Terra-Nova Bay (Antarctica). Annali Di Chimica, 81:279-296.

Carmack, E.C. and D.C. Chapman, 2003. Wind-driven shelf/ basin exchange on an Arctic shelf: The joint roles of ice cover extent and shelf-break bathymetry. Geophysical Research Letters, 30:1778.

Carmack, E.C., D. Barber, J. Christensen, R.W. Macdonald, B. Rudels and E. Sakshaug, 2006. Climate variability and physical forcing of the food webs and the carbon budget on panarctic shelves. Progress in Oceanography, 71:145-181.

Carpenter, L.J., S.D. Archer and R. Beale, 2012. Oceanatmosphere trace gas exchange. Chemical Society Reviews, 41: 6473-6506.

Carstens, J. and G. Wefer, 1992. Recent distribution of planktonic foraminifera in the Nansen Basin, Arctic Ocean. Deep-Sea Research A, 39:S507-S524.

Carstens, J., D. Hebbeln and G. Wefer, 1997. Distribution of planktic foraminifera at the ice margin in the Arctic (Fram Strait). Marine Micropaleontology, 29:257-269.

Catarino, A.I., C.De Ridder, M. Gonzalez, P. Gallardo and P. Dubois, 2012. Sea urchin *Arbacia dufresnei* (Blainville 1825) larvae response to ocean acidification. Polar Biology, 35:455-461.

Cavalieri, D., C. Parkinson, P. Gloersen and H.J. Zwally, 1996 and updated yearly. Sea ice concentrations from Nimbus-7 SMMR and DMSP SSM/I-SSMIS Passive Microwave Data, [1996-01-01; 2010-12-31]. National Snow and Ice Data Center, Boulder, Colorado USA. Digital media. Chang, B.X. and Devol, A.H., 2009. Seasonal and spatial patterns of sedimentary denitrification rates in the Chukchi sea. Deep Sea Research Part II: Topical Studies in Oceanography, 56(17): 1339-1350.

Chapman, A.D., 2009. Numbers of Living Species in Australia and the World. 2nd Edition. Report for the Australian Biological Resources Study. Canberra, Australia.

Charlson, R.J., Lovelock, J.E., Andreae, M.O., Warren, S.G., 1987. Oceanic phytoplankton, atmospheric sulfur, cloud albedo and climate. Nature 326, 655-661.

Checkley, D.M., A.G. Dickson, M. Takahashi, J.A. Radich, N. Eisenkolb and R. Asch, 2009. Elevated CO_2 enhances otolith growth in young fish. Science, 324:1683.

Cheung, W., V. Lam, J. Sarmiento, K. Kearney, R. Watson, D. Zeller and D. Pauly, 2010. Large-scale redistribution of maximum fisheries catch potential in the global ocean under climate change. Global Change Biology, 16:2435.

Cheung, W.W.L., J. Dunne, J. Sarmiento and D. Pauly, 2011. Integrating ecophysiology and plankton dynamics into projected changes in maximum fisheries catch potential under climate change in the Northeast Atlantic. ICES Journal of Marine Science, 68:1008-1018.

Chierici, M. and A. Fransson, 2009. Calcium carbonate saturation in the surface water of the Arctic Ocean: Undersaturation in freshwater influenced shelves. Biogeosciences, 6:2421-2432.

Chierici, M., A. Fransson, B. Lansard, L.A. Miller, A. Mucci, E. Shadwick, H. Thomas, J.-E. Tremblay and T.N. Papakyriakou, 2011. Impact of biogeochemical processes and environmental factors on the calcium carbonate saturation state in the Circumpolar Flaw Lead in the Amundsen Gulf, Arctic Ocean. Journal of Geophysical Research, 116:C00G09. doi: 10.1029/2011JC007184.

Chiperzak, D.B., G.E. Hopky, M.J. Lawrence, D.F. Schmid and J.D. Reist, 2003a. Larval and Post-larval Fish Data from the Canadian Beaufort Sea Shelf, July to September, 1985. Canadian Data Report of Fisheries and Aquatic Sciences, 1119, iv+116 pp.

Chiperzak, D.B., G.E. Hopky, M.J. Lawrence, D.F. Schmid and J.D. Reist, 2003b. Larval and Post-larval Fish Data from the Canadian Beaufort Sea Shelf, July to September, 1986. Canadian Data Report of Fisheries and Aquatic Sciences, 1120, iv+153 pp.

Christensen, J., J. Murray, A. Devol and L. Codispoti, 1987. Denitrification in continental shelf sediments has major impact on the oceanic nitrogen budget. Global Biogeochemical Cycles, 1:97-116.

Christian, J., V. Arora, G. Boer, C. Curry, K. Zahariev, K. Denman, G. Flato, W. Lee, W. Merryfield, N. Roulet and J. Scinocca, 2010. The global carbon cycle in the Canadian Earth System Model (CanESM1): Pre-industrial control simulation. Journal of Geophysical Research, 115:G03014.

Christman, G.D., M.T. Cottrell, B.N. Popp, E. Gier and D.L. Kirchman, 2011. Abundance, diversity, and activity of ammonia-oxidizing prokaryotes in the coastal Arctic Ocean in summer and winter. Applied and Environmental Microbiology, 77:2026-2034. Cigliano, M., M.C. Gambi, R. Rodolfo-Metalpa, F.P. Patti and J.M. Hall-Spencer, 2010. Effects of ocean acidification on invertebrate settlement at volcanic CO_2 vents. Marine Biology, 157:2489-2502.

Cisneros-Montemayor, A.M. and U.R. Sumaila, 2010. A global valuation of ecosystem-based marine recreation. Journal of Bioeconomics, 12:245-268.

Cisneros-Montemayor, A.M., U.R. Sumaila, K. Kaschner and D. Pauly, 2010. The global potential for whale watching. Marine Policy, 34:1273-1278.

Clark, D., M. Lamare and M. Barker, 2009. Response of sea urchin pluteus larvae (Echinodermata: Echinoidea) to reduced seawater pH: a comparison among a tropical, temperate, and a polar species. Marine Biology, 156:1125-1137.

Clegg, S.L. and M. Whitfield, 1995. A chemical model of seawater including dissolved ammonia and the stoichiometric dissociation constant of ammonia in estuarine water and seawater from -2 to 40°C. Geochimica et Cosmochimica Acta, 59:2403-2421.

Coale, K.H. and K.W. Bruland, 1988. Copper complexation in the northeast Pacific. Limnology and Oceanography, 33:1084-1101.

Cochran, J.K., D.J. Hirschberg, H.D. Livingston, K.O. Buesseler and R.M. Key, 1995. Natural and anthropogenic radionuclide distributions in the Nansen Basin, Arctic Ocean: Scavenging rates and circulation timescales. Deep-Sea Research II, 42:1495-1517.

Codispoti, L.A., G.E. Friederich and D.W. Hood, 1986. Variability in the inorganic carbon system over the southeastern Bering Sea shelf during spring 1980 and spring-summer 1981. Continental Shelf Research, 5:133-160.

Cohen, A.L. and M. Holcomb, 2009. Why corals care about ocean acidification: Uncovering the mechanism. Oceanography, 22:118-127.

Collins, W., N. Bellouin, M. Doutriaux-Boucher, N. Gedney, P. Halloran, T. Hinton, J. Hughes, C. Jones, M. Joshi, S. Liddicoat, G. Martin, F. OConnor, J. Rae, C. Senior, S. Sitch, I. Totterdell, A. Wiltshire and S. Woodward, 2011. Development and evaluation of an earth-system model - HadGEM2. Geoscientific Model Development, 4:1051-1075.

Comeau, S., G. Gorsky, R. Jeffree, J.-L. Teyssié and J.-P. Gattuso, 2009. Impact of ocean acidification on a key Arctic pelagic mollusc (*Limacina helicina*). Biogeosciences, 6:1877-1882.

Comeau, S., R. Jeffree, J.-L. Teyssié and J.-P. Gattuso, 2010. Response of the Arctic pteropod *Limacina helicina* to projected future environmental conditions. PloS One 5:e11362.

Comeau, S., J.-P. Gattuso, A.-M. Nisumaa and J. Orr, 2012a. Impact of aragonite saturation state changes on migratory pteropods. Proceedings of the Royal Society B, 279:732-738.

Comeau, S., S. Alliouane and J.P. Gattuso, 2012b. Effects of ocean acidification on overwintering juvenile Arctic pteropods *Limacina helicina*. Marine Ecology Progress Series, 456:279-284.

Connell, S.D. and B.D. Russell, 2010. The direct effects of increasing CO_2 and temperature on non-calcifying organisms: increasing the potential for phase shifts in kelp forests. Proceedings of the Royal Society B, 277:1409-1415.

Conover, R.J., 1988. Comparative life histories in the genera Calanus and neocalanus in high latitudes of the northern hemisphere. Hydrobiologia, 167:127-142.

Cooley, S.R. and S.C. Doney, 2009. Anticipating ocean acidification's economic consequences for commercial fisheries. Environmental Research Letters, 4:024007.

Cooley, S.R., H.L. Kite-Powell and S.C. Doney, 2009. Ocean acidification's potential to alter global marine ecosystem services. Oceanography, 22:172-181.

Cota, G.F., L. Legendre, M. Gosselin and R.G. Ingram, 1991. Ecology of bottom ice algae: I. Environmental controls and variability. Journal of Marine Systems, 2:257-277.

Crain, C.M., K. Kroeker and B.S. Halpern, 2008. Interactive and cumulative effects of multiple human stressors in marine systems. Ecology Letters, 11:1304-1315.

Cripps, I.L., P.L. Munday and M.I. McCormick, 2011. Ocean acidification affects prey detection by a predatory reef fish. PloS One, 6:e22736.

Croot, P.L., J.W. Moffett and L. Brand, 2000. Production of extracellular Cu complexing ligands by eucaryotic phytoplankton in response to Cu stress. Limnology and Oceanography, 45:619-627.

Crumbliss, A.L. and J.M. Harrington, 2009. Iron sequestration by small molecules: Thermodynamic and kinetic studies of natural siderophores and synthetic model compounds. In: Rudi van, E. and D.H. Colin (Eds.). Advances in Inorganic Chemistry, pp. 179-250. Academic Press.

Cullen, J.T. and R.M. Sherrell, 2005. Effects of dissolved carbon dioxide, zinc, and manganese on the cadmium to phosphorus ratio in natural phytoplankton assemblages. Limnology and Oceanography, 50:1193-1204.

Cullen, J.T., T.W. Lane, F.M.M. Morel and R.M. Sherrell, 1999. Modulation of cadmium uptake in phytoplankton by seawater CO, concentration. Nature, 402:165-167.

Cummings, V., J. Hewitt, A. Van Rooyen, K. Currie, S. Beard, S. Thrush, J. Norkko, N. Barr, P. Heath, N. J. Halliday, R. Sedcole, A. Gomez, C. McGraw and V. Metcalf, 2011. Ocean acidification at high latitudes: potential effects on functioning of the Antarctic bivalve *Laternula elliptica*. PloS One, 6:e16069.

Czerny, J., J. Barcelos e Ramos and U. Riebesell, 2009. Influence of elevated CO₂ concentrations on cell division and nitrogen fixation rates in the bloom-forming cyanobacterium *Nodularia spumigena*. Biogeosciences, 6:1865-1875.

Damm, E., U. Schauer, B. Rudels and C. Haas, 2007. Excess of bottom-released methane in an Arctic shelf sea polynya in winter. Continental Shelf Research, 27:1692-1701.

Danielsson, L.G. and S. Westerlund, 1983. Trace metals in the Arctic Ocean. In: Wong, C.S., E. Boyle, K.W. Bruland, D. Burton and E.D. Goldberg (Eds.). Trace Metals in the Sea, pp. 85-95. Plenum Press.

Danovaro, R., C. Corinaldesi, A. Dell'Anno, J.A. Fuhrman, J.J. Middelburg, R.T. Noble and C.A. Suttle, 2011. Marine viruses and global climate change. FEMS Microbiology Reviews, 35:993-1034.

Dawson, J.K., 1978. Vertical distribution of *Calanus hyperboreus* in a central Arctic location. Limnology and Oceanography, 23: 950-957.

de Jesus Mendes, P.A. and L. Thomsen, 2012. Effects of ocean acidification on the ballast of surface aggregates sinking through the twilight zone. PLoS ONE, 7:e50865.

de la Haye, K.L., J.I. Spicer, S. Widdicombe and M. Briffa, 2011. Reduced sea water pH disrupts resource assessment and decision making in the hermit crab *Pagurus bernhardus*. Animal Behaviour, 82:495-501.

De la Rocha, C.L., 2006. The biological pump. In: Holland, H.D. and K.K. Turekian (Eds.). The Oceans and Marine Geochemistry, pp. 83-111. Elsevier-Pergamon.

de Souza, M.P. and D.C. Yoch, 1995. Comparative physiology of dimethyl sulfide production by dimethylsulfoniopropionate lyase in *Pseudomonas doudoroffii* and *Alcaligenes* sp. Strain M3a. Applied and Environmental Microbiology, 61:3986-3991.

Deal, C.J., N. Steiner, J. Christian, J.C. Kinney, K. Denman, S. Elliott, G. Gibson, M. Jin, D. Lavoie, S. Lee, W. Lee, W. Maslowski, J. Wang and E. Watanabe, 2013. Progress and challenges in biogeochemical modeling of the Pacific Arctic Region. In: Grebmeier, J.M. and W. Maslowski (Eds.). The Pacific Arctic Region: Ecosystem Status and Trends in a Rapidly Changing Environment. Springer Publishing, accepted.

deConto, R.M., S. Galeotti, M. Pagani, D. Tracy, K. Schaefer, T. Zhang, D. Pollard and D.J. Beerling, 2012. Past extreme warming events linked to massive carbon release from thawing permafrost. Nature, 484:87-91.

Denman, K., G. Brasseur, A. Chidthaisong, P. Ciais, P. Cox, R. Dickinson, D. Hauglustaine, C. Heinze, E. Holland, D. Jacob, U. Lohmann, S. Ramachandran, P. da Silva Dias, S. Wofsy and X. Zhang, 2007. Couplings between changes in the climate system and biogeochemistry. In: Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press.

Denman, K., J.R. Christian, N. Steiner, H.O. Pörtner and Y. Nojiri, 2011. Potential impacts of future ocean acidification on marine ecosystems and fisheries: current knowledge and recommendations for future research. ICES Journal of Marine Science, 68:1019-1029.

Devine, B.M., P.L. Munday and G.P. Jones, 2012. Homing ability of adult cardinal fish is affected by elevated carbon dioxide. Oecologia, 168:269-276.

Devol, A.H., L.A. Codispoti and J.P. Christensen, 1997. Summer and winter denitrification rates in western Arctic shelf sediments. Continental Shelf Research, 17:1029-1033.

Dias, B.B., B. Hart, C.W. Smart and J.M. Hall-Spencer, 2010. Modern seawater acidification: the response of foraminifera to high- CO_2 conditions in the Mediterranean Sea. Journal of the Geological Society, 167:843-846.

Dickson, A.G. and F.J. Millero, 1987. A comparison of the equilibrium constants for the dissociation of carbonic acid in seawater media, Deep-Sea Research, 34:1733-1743.

Dickson, A.G. and J.P. Riley, 1978. The effect of analytical error on the evaluation of the components of the aquatic carbondioxide system. Marine Chemistry, 6:77-85.

Dickson, A.G., C.L. Sabine and J.R. Christian (Eds.), 2007. Guide to best practices for ocean CO_2 measurements. PICES Special Publication 3, 191 pp.

Dietz, R., M.P. Heide-Jorgensen, P. Richard, J. Orr, K. Laidre and H.C. Schmidt, 2008. Movements of narwhals (*Monodon monoceros*) from Admiralty Inlet monitored by satellite telemetry. Polar Biology, 31:1295-1306.

Dissanayake, A. and A. Ishimatsu, 2011. Synergistic effects of elevated CO_2 and temperature on the metabolic scope and activity in a shallow-water coastal decapod (*Metapenaeus joyneri*; Crustacea: Penaeidae). ICES Journal of Marine Science, 68:1147-1154.

Dissard, D., G. Nehrke, G.J. Reichart and J. Bijma, 2010. The impact of salinity on the Mg/Ca and Sr/Ca ratio in the benthic foraminifera *Ammonia tepida*: Results from culture experiments. Geochimica et Cosmochimica Acta, 74:928-940.

Dodson, J.J., S. Tremblay, F. Colombani, J.E. Carscadden and F. Lecomte, 2007. Trans-Arctic dispersals and the evolution of a circumpolar marine fish species complex, the capelin (*Mallotus villosus*). Molecular Ecology, 16:5030-5043.

Domenici, P., B. Allan, M.I. McCormick and P.L. Munday, 2012. Elevated carbon dioxide affects behavioural lateralization in a coral reef fish. Biology Letters, 8:78-81.

Doney, S.C., N. Mahowald, I. Lima, R.A. Feely, F.T. Mackenzie, J.-F. Lamarque and P.J. Rasch, 2007. Impact of anthropogenic atmospheric nitrogen and sulfur deposition on ocean acidification and the inorganic carbon system. Proceedings of the National Academy of Sciences, USA, 104:14,580-14,585.

Doney, S.C., V.J. Fabry, R.A. Feely and J.A. Kleypas, 2009. Ocean acidification: The other CO₂ problem. Annual Reviews of Marine Science, 1:169-192.

Dore, J.E., R. Lukas, D.W. Sadler, M.J. Church and D.M. Karl, 2009. Physical and biogeochemical modulation of ocean acidification in the central North Pacific. Proceedings of the National Academy of Sciences, 106:12235-12240.

Drolet, R., L. Fortier, D. Ponton and M. Gilbert, 1991. Production of fish larvae and their prey in sub-Arctic southeastern Hudson Bay. Marine Ecology Progress Series, 77:105-118.

Dufresne, J.-L., M.-A. Foujols, S. Denvil and 57 others, 2013. Climate change projections using the IPSL-CM5 Earth System Model: from CMIP3 to CMIP5. Climate Dynamics, 40:2123-2165.

Duhaime, G. and A. Caron, 2009. Economic and social conditions in Arctic regions. In: Glomsrød, S. and I. Aslaksen, I. (Eds), 2009. The Economy of the North 2008, pp. 11-26. Statistics Norway, Oslo.

Dullo, W.-C., S. Floegel and A. Rueggeberg, 2008. Cold-water coral growth in relation to the hydrography of the Celtic and Nordic European continental margin. Marine Ecology Progress Series, 371:165-176. Dunbar, M.J., 1946. On *Themisto libellula* in Baffin Island Coastal Waters. Journal of the Fisheries Research Board of Canada, 6e:419-434.

Dunbar, M.J., 1968. Ecological Development in Polar Regions. Prentice Hall.

Dunne, J.P., J.G. John, A.J. Adcroft, S.M. Griffies, R.W. Hallberg, E.S. Shevliakova, R.J. Stouffer, W. Cooke, K.A. Dunne, M.J. Harrison, J.P. Krasting, S.L. Malyshev, P.C.D. Milly, P.J. Phillipps, L.T. Sentman, B.L. Samuels, M.J. Spelman, M. Winton, A.T. Wittenberg and N. Zadeh, 2012. GFDL's ESM2 global coupled climate-carbon Earth System Models. Part I: Physical formulation and baseline simulation characteristics. Journal of Climate, 25:6646-6665.

Dunne, J.P., J.G. John, E.N. Shevliakova, R.J. Stouffer, J.P. Krasting, S.L. Malyshev, P.C.D. Milly, L.T. Sentman, A.J. Adcroft, W. Cooke, K.A. Dunne, S.M. Griffies, R.W. Hallberg, M.J. Harrison, H. Levy, A.T. Wittenberg, P.J. Phillipps and N. Zadeh, 2013. GFDL's ESM2 global coupled climate-carbon Earth System Models. Part II: Carbon system formulation and baseline simulation characteristics. Journal of Climate, 26:2247-2267.

Dupont, S. and M. Thorndyke, 2008. Ocean acidification and its impact on the early life-history stages of marine animals. CIESM Monograph 36.

Dupont, S. and M. Thorndyke, 2012. Relationship between CO₂driven changes in extracellular acid-base balance and cellular immune response in two polar echinoderm species. Journal of Experimental Marine Biology and Ecology, 424:32-37.

Dupont, S., J.N. Havenhand, W. Thorndyke, L.S. Peck and M.C. Thorndyke, 2008. CO₂-driven ocean acidification radically affects larval survival and development in the brittlestar *Ophiothrix fragilis*. Marine Ecology Progress Series, 373:285-294.

Dupont, S., O. Ortega-Martinez and M. Thorndyke, 2010. Impact of near-future ocean acidification on echinoderms. Ecotoxicology, 19:449-462.

Dyck, A.J. and U.R. Sumaila, 2010. Economic impact of ocean fish populations in the global fishery. Journal of Bioeconomics, 12:227-243.

Edmonds, H.N., S.B. Moran, J.A. Hoff, J.N. Smith and R.L. Edwards, 1998. Protactinium-231 and Thorium-230 abundances and high scavenging rates in the western Arctic Ocean. Science, 280:405-407.

Edmonds, H.N., S.B. Moran, H. Cheng and R.L. Edwards, 2004. ²³⁰Th and ²³¹Pa in the Arctic Ocean: implications for particle fluxes and basin-scale Th/Pa fractionation. Earth and Planetary Science Letters, 227:155-167.

Egilsdottir, H., J.I. Spicer and S.D. Rundle, 2009. The effect of CO_2 acidified sea water and reduced salinity on aspects of the embryonic development of the amphipod *Echinogammarus marinus* (Leach). Marine Pollution Bulletin, 58:1187-1191.

Eide, A., 2007. Economic impacts of global warming – the case of the Barents Sea fisheries. Natural Resource Modeling, 20:61-83.

Eklof, J.S., C. Alsterberg, J.N. Havenhand, K. Sundback, H.L. Wood and L. Gamfeldt, 2012. Experimental climate change weakens the insurance effect of biodiversity. Ecology Letters, 15:864-872.

Ekwurzel, B., P. Schlosser, R.A. Mortlock, R.G. Fairbanks and J.H. Swift, 2001. River runoff, sea ice meltwater, and Pacific water distribution and mean residence times in the Arctic Ocean. Journal of Geophysical Research, 106:9075-9092.

Elliott, S., M. Reagan, G. Moridis and P.C. Smith, 2010. Geochemistry of clathrate derived methane in Arctic ocean waters. Geophysical Research Letters, 37:L12607.

Ellwood, M.J. and C.M.G. van den Berg, 2001. Determination of organic complexation of cobalt in seawater by cathodic stripping voltammetry. Marine Chemistry, 75:33-47.

Else, B.G.T., T.N. Papakyriakou, R.J. Galley, W.M. Drennan, L.A. Miller and H. Thomas, 2011. Wintertime CO_2 fluxes in an Arctic polynya using eddy covariance: Evidence for enhanced air-sea gas transfer during ice formation. Journal of Geophysical Research, 116:C00G03. doi: 10.1029/2010JC006760.

Else, B.G.T., T.N. Papakyriakou, R.J. Galley, A. Mucci, M. Gosselin, L.A. Miller, E.H. Shadwick and H. Thomas, 2012. Annual cycles of pCO_2 sw in the southeastern Beaufort Sea: New understandings of air-sea CO_2 exchange in arctic polynya regions. Journal of Geophysical Research, 117:C00G13. doi: 10.1029/2011JC007346.

Emlet, R.B. and S.S. Sadro, 2006. Linking stages of life history: How larval quality translates into juvenile performance for an intertidal barnacle (*Balanus glandula*). Integrative and Comparative Biology, 46:334-346.

Engel, A., I. Zondervan, K. Aerts, L. Beaufort, A. Benthien, L. Chou, B. Delille, J.P. Gattuso, J. Harlay, C. Heemann, L. Hoffmann, S. Jacquet, J. Nejstgaard, M.D. Pizay, E. Rochelle-Newall, U. Schneider, A. Terbrueggen and U. Riebesell, 2005. Testing the direct effect of CO_2 concentration on a bloom of the coccolithophorid *Emiliania huxleyi* in mesocosm experiments. Limnology and Oceanography, 50:493-507.

Ericson, J.A., M.D. Lamare, S.A. Morley and M.F. Barker, 2010. The response of two ecologically important Antarctic invertebrates (*Sterechinus neumayeri* and *Parborlasia corrugatus*) to reduced seawater pH: effects on fertilisation and embryonic development. Marine Biology, 157:2689-2702.

Fabricius, K.E., C. Langdon, S. Uthicke, C. Humphrey, S. Noonan, G. De'ath, R. Okazaki, N. Muehllehner, M. S. Glas and J.M. Lough, 2011. Losers and winners in coral reefs acclimatized to elevated carbon dioxide concentrations. Nature Climate Change, 1:165-169.

Fabry, V.J., B.A. Seibel, R.A. Feely and J.C. Orr, 2008. Impacts of ocean acidification on marine fauna and ecosystem processes. ICES Journal of Marine Sciences, 65:414-432.

Fabry, V.J., J.B. McClintock, J.T. Mathis and J.M. Grebmeier, 2009. Ocean acidification at high latitudes: the bellwether. Oceanography, 22:160-171.

Falk-Petersen, S., T.M. Dahl, C.L. Scott, J.R. Sargent, B. Gulliksen, S. Kwasniewski, H. Hop and R.M. Millar, 2002.

Lipid biomarkers and trophic linkages between ctenophores and copepods in Svalbard waters. Marine Ecology Progress Series, 227:187-194.

Farmer, J.R., T.M. Cronin, R. Thunell, L.D. Keigwin and D.A. Willard, 2010. Holocene climate variability in the Beaufort Sea, Arctic Ocean from benthic foraminifers, stable isotopes and pollen. American Geophysical Union, Fall Meeting 2010, abstract #PP21B-1689.

Feely, R.A., C.L. Sabine, K. Lee, W. Berelson, J.A. Kleypas, V.J. Fabry and F.J. Millero, 2004. Impact of anthropogenic CO_2 on the CaCO₃ system in the oceans. Science, 305:362-366.

Feely, R.A., C.L. Sabine, J.M. Hernandez-Ayon, D. Ianson and B. Hales, 2008. Evidence for upwelling of corrosive "acidified" water onto the continental shelf. Science, 320:1490-1492.

Feely, R.A., R. Wanninkhof, J.E. Stein, M.F. Sigler, E.B. Jewett, L.F. Arzayus, D.K. Gledhill, A.J. Sutton and N.O.A.S. Committee, 2010. NOAA Ocean and Great Lakes Acidification Research Plan, U.S. Department of Commerce, National Oceanic and Atmospheric Administration.

Fehsenfeld, S., R. Kiko, Y. Appelhans, D.W. Towle, M. Zimmer and F. Melzner, 2011. Effects of elevated seawater pCO_2 on gene expression patterns in the gills of the green crab, *Carcinus maenas*. BMC Genomics, 12:488.

Fernandes, L., J. Day, B. Kerrigan, D. Breen, G. De'ath, B. Mapstone, R. Coles, T. Done, H. Marsh, I. Poiner, T. Ward, D. Williams and R, Kenchington, 2009. A process to design a network of marine no-take areas: lessons from the Great Barrier Reef. Ocean and Coastal Management, 52:439-447.

Ferrari, M.C.O., M.I. McCormick, P.L. Munday, M.G. Meekan, D.L. Dixson, O. Lonnstedt and D.P. Chivers, 2011. Putting prey and predator into the CO_2 equation – qualitative and quantitative effects of ocean acidification on predator-prey interactions. Ecology Letters, 14:1143-1148.

Findlay, H.S., T. Tyrrell, R.G.J. Bellerby, A. Merico and I. Skjelvan, 2008. Carbon and nutrient mixed layer dynamics in the Norwegian Sea. Biogeosciences, 5:1395-1410.

Findlay, H.S., M.A. Kendall, J.I. Spicer and S. Widdicombe, 2009. Future high CO₂ in the intertidal may compromise adult barnacle *Semibalanus balanoides* survival and embryonic development rate. Marine Ecology Progress Series, 389:193-202.

Findlay, H.S., M.T. Burrows, M.A. Kendall, J.I. Spicer and S. Widdicombe, 2010a. Can ocean acidification affect population dynamics of the barnacle *Semibalanus balanoides* at its southern range edge? Ecology, 91:2931-2940.

Findlay, H.S., M.A. Kendall, J.I. Spicer and S. Widdicombe, 2010b. Relative influences of ocean acidification and temperature on intertidal barnacle post-larvae at the northern edge of their geographic distribution. Estuarine Coastal and Shelf Science, 86:675-682.

Findlay, H.S., H.L. Wood, M.A. Kendall, J.I. Spicer, R.J. Twitchett and S. Widdicombe, 2011. Comparing the impact of high CO_2 on calcium carbonate structures in different marine organisms. Marine Biology Research, 7:565-575. Fiorini, S., J.J. Middelburg and J.-P. Gattuso, 2011. Testing the effects of elevated pCO_2 on coccolithophores (Prymnesiophyceae): comparison between haploid and diploid life stages. Journal of Phycology, 47:1281-1291.

Flaaten, O., 2011. Recreational fishing. In: Fisheries Economics and Management, pp. 152-164. Norwegian College of Fishery Science, University of Tromsø.

Flaaten, O. and E. Mjølhus 2010. Nature reserves as a bioeconomic management tool – a simplified modeling approach. Environmental and Resource Economics, 47:125-148.

Flores, H., A. Atkinson, S. Kawaguchi, B.A. Krafft, G. Milinevsky, S. Nicol, C. Reiss, G.A. Tarling, R. Werner, E.B. Rebolledo, V. Cirelli, J. Cuzin-Roudy, S. Fielding, J.J. Groeneveld, M. Haraldsson, A. Lombana, E. Marschoff, B. Meyer, E.A. Pakhomov, E. Rombola, K. Schmidt, V. Siegel, M. Teschke, H. Tonkes, J.Y. Toullec, P.N. Trathan, N. Tremblay, A.P. Van de Putte, J.A. van Franeker and T. Werner, 2012. Impact of climate change on Antarctic krill. Marine Ecology Progress Series, 458:1-19.

Ford, J.D. and C. Furgal 2009. Foreword to the special issue: Climate Change Impacts, Adaptation and Vulnerability in the Arctic. Polar Research, 28:1-9.

Forest, A., M. Sampei, H. Hattori, R. Makabe, H. Sasaki, M. Fukuchi, P. Wassmann and L. Fortier, 2007. Particulate organic carbon fluxes on the slope of the Mackenzie Shelf (Beaufort Sea): Physical and biological forcing on shelf-basin exchanges. Journal of Marine Systems, 68:39-54.

Form, A.U. and U. Riebesell, 2012. Acclimation to ocean acidification during long-term CO_2 exposure in the cold-water coral *Lophelia pertusa*. Global Change Biology, 18:843-853.

Forster, P., V. Ramaswamy, P. Artaxo, T. Berntsen, R. Betts, D.W. Fahey, J. Haywood, J. Lean, D.C. Lowe, G. Myhre, R. Priinn, G. Raga, M. Schulz and R. Van Dorland, 2007. Changes in atmospheric constituents and in radiative forcing. In: S. Solomon et al. (Eds.). Climate Change 2007: The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press.

Fortier, M., L. Fortier, C. Michel and L. Legendre, 2002. Climatic and biological forcing of the vertical flux of biogenic particles under seasonal Arctic sea ice. Marine Ecology Progress Series, 225:1-16.

Frank, M., D. Porcelli, P. Andersson, M. Baskaran, G. Björk, P.W. Kubik, B. Hattendorf and D. Guenther, 2009. The dissolved beryllium isotope composition of the Arctic Ocean. Geochimica et Cosmochimica Acta, 73:6114-6133.

Franke, A. and C. Clemmesen, 2011. Effect of ocean acidification on early life stages of Atlantic herring (*Clupea harengus* L.). Biogeosciences, 8:3697-3707.

Fransson, A., M. Chierici, L.G. Anderson, I. Bussman, G. Kattner, E.P. Jones and J.H. Swift, 2001. The importance of shelf processes for the modification of chemical constituents in the waters of the eastern Arctic Ocean. Continental Shelf Research, 21:225-242.

Fransson, A., M. Chierici and Y. Nojiri, 2006. Increased net CO_2 outgassing in the upwelling region of the southern Bering Sea in a period of variable marine climate between 1995 and 2001. Journal of Geophysical Research, 111:C08008.

Fransson, A., M. Chierici and Y. Nojiri, 2009. New insights into the spatial variability of the surface water carbon dioxide in varying sea ice conditions in the Arctic Ocean. Continental Shelf Research, 29:1317-1328.

Friedlingstein, P., P. Cox, R. Betts, L. Bopp, W. von Bloh, V. Brovkin, P. Cadule, S. Doney, M. Eby, I. Fung, G. Bala, J. John, C. Jones, F. Joos, T. Kato, M. Kawamiya, W. Knorr, K. Lindsay, H.D. Matthews, T. Raddatz, P. Rayner, C. Reick, E. Roeckner, K-G. Schnitzler, K. Strassmann, A.J. Weaver, C. Yoshikawa and N. Zeng, 2006. Climate carbon cycle feedback analysis: Results from the c4mip model intercomparison. Journal of Climate, 19:3337-3353.

Frommel, A.Y., V. Stiebens, C. Clemmesen and J. Havenhand, 2010. Effect of ocean acidification on marine fish sperm (Baltic cod: *Gadus morhua*). Biogeosciences, 7:3915-3919.

Frommel, A.Y., R. Maneja, D. Lowe, A.M. Malzahn, A.J. Geffen, A. Folkvord, U. Piatkowski, T.B.H. Reusch and C. Clemmesen, 2012. Severe tissue damage in Atlantic cod larvae under increasing ocean acidification. Nature Climate Change, 2:42-46.

Frost, K.J. and L.F. Lowry, 1990. Advances in research on the beluga whale, *Delphinapterus leucas*. University of California, Dept. of Fisheries and Oceans.

Gangstø, R., M. Gehlen, B. Schneider, L. Bopp, O. Aumont and F. Joos, 2008. Modeling the marine aragonite cycle: changes under rising carbon dioxide and its role in shallow water CaCO₃ dissolution. Biogeosciences, 5:1057-1072.

Gao, K. and Y. Zheng, 2010. Combined effects of ocean acidification and solar UV radiation on photosynthesis, growth, pigmentation and calcification of the coralline alga *Corallina sessilis* (Rhodophyta). Global Change Biology, 16:2388-2398.

Garneau, M.-È., W.F. Vincent, L. Alonso-Sáez, Y. Gratton and C. Lovejoy, 2006. Prokaryotic community structure and heterotrophic production in a river-influenced coastal arctic ecosystem. Aquatic Microbial Ecology, 42:27-40.

Garnier, J.-M., J.-M. Martin, J.-M. Mouchel and K. Sioud, 1996. Partitioning of trace metals between the dissolved and particulate phases and particulate surface reactivity in the Lena River estuary and the Laptev Sea (Russia). Marine Chemistry, 53:269-283.

Garrard, S., R.C. Hunter, A.Y. Frommel, A.C. Lane, J.C. Phillips, R. Cooper, R. Dineshram, U. Cardini, S.J. McCoy, M. Arnberg, B.G. Rodrigues Alves, S. Annane, M.R. de Orte, A. Kumar, G.V. Aguirre-Martínez, R.H. Maneja, M.D. Basallote, F. Ape, A. Torstensson and M.M. Bjoerk, 2013. Biological impacts of ocean acidification: a postgraduate perspective on research priorities. Marine Biology, 160:1789-1805.

Garrison, D.L., C.W. Sullivan and S.F. Ackley, 1986. Sea ice microbial communities in Antarctica. Bioscience, 36:243-250.

Gattuso, J.P. and L. Hansson, 2009. European Project on Ocean Acidification (EPOCA): Objectives, products and scientific highlights. Oceanography, 22:190-201.

Gattuso, J.P. and L. Hansson, 2011. Ocean Acidification. Oxford University Press.

Gattuso, J.-P., J. Bijma, M. Gehlen, U. Riebesell and C. Turley, 2011. Ocean acidification: knowns, unknowns, and perspectives. In: Gattuso, J.-P. and L. Hansson (Eds.). Ocean Acidification, pp. 291-312. Oxford University Press.

Gazeau, F., C. Quiblier, J.M. Jansen, J.P. Gattuso, J.J. Middelburg and C.H.R. Heip, 2007. Impact of elevated CO_2 on shellfish calcification. Geophysical Research Letters, 34:L07603.

Gazeau, F., J.P. Gattuso, C. Dawber, A.E. Pronker, F. Peene, J. Peene, C.H.R. Heip and J.J. Middelburg, 2010. Effect of ocean acidification on the early life stages of the blue mussel *Mytilus edulis*. Biogeosciences, 7:2051-2060.

Geider, R.J. and J. La Roche, 2002. Redfield revisited: variability of C : N : P in marine microalgae and its biochemical basis. European Journal of Phycology, 37:1-17.

Gervais, F. and U. Riebesell, 2001. Effect of phosphorus limitation on elemental composition and stable carbon isotope fractionation in a marine diatom growing under different CO_2 concentrations. Limnology and Oceanography, 46:497-504.

Giesbrecht, K.E., L.A. Miller, S. Zimmermann, E. Carmack, W.K. Johnson, R.W. Macdonald, F. McLaughlin, A. Mucci, W.J. Williams, C.S. Wong and M. Yamamoto-Kawai, 2013. Measurements of the dissolved inorganic carbon system and associated biogeochemical parameters in the Canadian Arctic, 1974-2009. Earth System Science Data Discussions, 6:223-254.

Gihring, T.M., G. Lavik, M.M.M. Kuypers and J.E. Kostka, 2010. Direct determination of nitrogen cycling rates and pathways in Arctic fjord sediments (Svalbard, Norway). Limnology and Oceanography, 55:740-752.

Giles, K.A., S.W. Laxon and A.L. Ridout, 2008. Circumpolar thinning of Arctic sea ice following the 2007 record ice extent minimum. Geophysical Research Letters, 35:L22502. doi: 10.1029/2008GL035710.

Giles, K.A., S.W. Laxon, A.L. Ridout, D.J. Wingham and S. Bacon, 2013. Western Arctic Ocean freshwater storage increased by wind-driven spin-up of the Beaufort Gyre. Nature Geoscience, 5:194-197 doi:10.1038/ngeo1379.

Giraudeau, J., M. Grelaud, S. Solignac, J.T. Andrews, M. Moros and E. Jansen, 2010. Millennial-scale variability in Atlantic water advection to the Nordic Seas derived from Holocene coccolith concentration records. Quaternary Science Reviews, 29:1276-1287.

Gjøsæter, H., 1998. The population biology and exploitation of capelin (*Mallotus villosus*) in the Barents Sea. Sarsia, 83:453-496.

Gledhill, M. and C.M.G. van den Berg, 1994. Determination of complexation of iron(III) with natural organic complexing ligands in seawater using cathodic stripping voltammetry. Marine Chemistry, 47:41-54.

Gonzalez-Davila, M., J.M. Santana-Casiano and F.J. Millero, 2005. Oxidation of iron (II) nanomolar with H_2O_2 in seawater. Geochimica et Cosmochimica Acta, 69:83-93.

González-Dávila, M., J.M. Santana-Casiano, A.G. González, N. Pérez and F.J. Millero, 2009. Oxidation of copper(I) in seawater at nanomolar levels. Marine Chemistry, 115:118-124.

Gosink, T.A., J.G. Pearson and J.J. Kelley, 1976. Gas movement through sea ice. Nature, 263:41-42.

Gowing, M.M., 2003. Large viruses and infected microeukaryotes in Ross Sea summer pack ice habitats. Marine Biology, 142:1029-1040.

Gowing, M.M., B.E. Riggs, D.L. Garrison, A.H. Gibson and M.O. Jeffries, 2002. Large viruses in Ross Sea late autumn pack ice habitats. Marine Ecology Progress Series, 241:1-11.

Gowing, M.M., D.L. Garrison, A.H. Gibson, J.M. Krupp, M.O. Jeffries and C.H. Fritsen, 2004. Bacterial and viral abundance in Ross Sea summer pack ice communities. Marine Ecology Progress Series, 279:3-12.

Granéli, E. and C. Haraldsson, 1993. Can increased leaching of trace metals from acidified areas influence phyto-plankton growth in coastal waters? Ambio, 22:308-311.

Granéli, E. and M.O. Moreira, 1990. Effects of river water of different origin on the growth of marine dinoflagellates and diatoms in laboratory culture. Journal of Experimental Marine Biology and Ecology, 136: 89-106.

Grebmeier, J.M., L.W. Cooper, H.M. Feder and B.I. Sirenko, 2006. Ecosystem dynamics of the Pacific-influenced Northern Bering and Chukchi Seas in the Amerasian Arctic. Progress in Oceanography, 71:331-361.

Gregory, T.R., C.W. Smart, M.B. Hart, G. Masse, L.L. Vare and S.T. Belt, 2010. Holocene palaeoceanographic changes in Barrow Strait, Canadian Arctic: foraminiferal evidence. Journal of Quaternary Science, 25:903-910.

Griffith, G.P., E.A. Fulton and A.J. Richardson, 2011. Effects of fishing and acidification-related benthic mortality on the southeast Australian marine ecosystem. Global Change Biology, 17:3058-3074.

Griffith, D.R., A.P. McNichol, L. Xu, F.A. McLaughlin, R.W, Macdonald, K.A. Brown and T.I. Eglinton, 2012. Carbon dynamics in the western Arctic Ocean: insights from full-depth carbon isotope profiles of DIC, DOC, and POC. Biogeosciences, 9:1217-1224.

Groom, S.B. and P.M. Holligan, 1987. Remote sensing of coccolithophore blooms. Advances in Space Research, 7:73-78.

Grossart, H.-P., M. Allgaier, U. Passow and U. Riebesell, 2006. Testing the effect of CO_2 concentration on the dynamics of marine heterotrophic bacterioplankton. Limnology and Oceanography, 51:1-11.

Grossmann, S. and G.S. Dieckmann, 1994. Bacterial standing stock, activity, and carbon production during formation and growth of sea ice in the Weddell Sea, Antarctica. Applied and Environmental Microbiology, 60:2746-2753.

Guay, C.K. and K.K. Falkner, 1997. Barium as a tracer of Arctic halocline and river waters. Deep Sea Research II, 44:1543-1569.

Guay, C.K. and K.K. Falkner, 1998. A survey of dissolved barium in the estuaries of major Arctic rivers and adjacent seas. Continental Shelf Research, 18:859-882.

Guay, C.K.H., F.A. McLaughlin and M. Yamamoto-Kawai, 2009. Differentiating fluvial components of upper Canada Basin waters on the basis of measurements of dissolved barium combined with other physical and chemical tracers. Journal of Geophysical Research, 114:C00A09.

Guieu, C., W.W. Huang, J.-M. Martin and Y.Y. Yong, 1996. Outflow of trace metals into the Laptev Sea by the Lena River. Marine Chemistry, 53: 255-267.

Guinotte, J.M. and V.J. Fabry, 2008. Ocean acidification and its potential effects on marine ecosystems. Annals of the New York Academy of Science, 1134:320-342.

Guinotte, J.M., J. Orr, S. Cairns, A. Freiwald, L. Morgan and R. George, 2006. Will human-induced changes in seawater chemistry alter the distribution of deep-sea scleractinian corals? Frontiers in Ecology and the Environment, 4:141-146.

Guo, L., I. Semiletov, Ö. Gustafsson, J. Ingri, P. Andersson, O. Dudarev and D. White, 2004. Characterization of Siberian Arctic coastal sediments: Implications for terrestrial organic carbon export. Global Biogeochemical Cycles, 18:GB1036 1-10.

Gutowska, M.A., H.O. Pörtner and F. Melzner, 2008. Growth and calcification in the cephalopod *Sepia officinalis* under elevated seawater *p*CO₂. Marine Ecology Progress Series, 373:303-309.

Hale, R., P. Calosi, L. McNeill, N. Mieszkowska and S. Widdicombe, 2011. Predicted levels of future ocean acidification and temperature rise could alter community structure and biodiversity in marine benthic communities. Oikos, 120:661-674.

Hall, I.R. and C.I. Measures, 1998. The distribution of Al in the IOC stations of the North Atlantic and Norwegian Sea between 52° and 65° North. Marine Chemistry, 61:69-85.

Hall-Spencer, J.M., R. Rodolfo-Metalpa, S. Martin, E. Ransome, M. Fine, S.M. Turner, S.J. Rowley, D. Tedesco and M.-C. Buia, 2008. Volcanic carbon dioxide vents show ecosystem effects of ocean acidification. Nature, 454:96-99.

Hansen, P.J., 2002. Effect of high pH on the growth and survival of marine phytoplankton: implications for species succession. Aquatic Microbial Ecology, 28:279-288.

Hansen, B., and S. Østerhus, 2000. North Atlantic – Nordic Seas exchanges. Progress in Oceanography, 45:109-208.

Hansen, P.J., N. Lundholm and B. Rost, 2007. Growth limitation in marine red-tide dinoflagellates: effects of pH versus inorganic carbon availability. Marine Ecology Progress Series, 334:63-71.

Hansen, B., S. Østerhus, W.R. Turrell, S. Jonsson, H. Valdimarasson, H. hatun and S.M. Olsen, 2008. The inflow of Atlantic Water, heat and salt to the Nordic Seas across the Greenland-Scotland Ridge. In: Dickson, R.R., J. Meincke and P. Rhines (Eds.). Arctic-Subarctic Ocean Fluxes, pp. 15-43. Springer.

Hansson, I., 1973. The determination of dissociation constants of H_2CO_3 in synthetic sea water in the salinity range of 20-40 o/oo and temperature range of 5-30 °C. Acta Chemica Scandinavica, 27:931-944.

Harada, N., M. Sato, K. Oguri and 12 others, 2012. Enhancement of coccolithophorid blooms in the Bering Sea by recent environmental changes. Global Biogeochemical Cycles, 26:GB2036.

Hargrave, B.T., B. von Bodungen, P. Stoffyn-Egli and P.J. Mudie, 1994. Seasonal variability in particle sedimentation under permanent ice cover in the Arctic Ocean. Continental Shelf Research, 14:279-293.

Harrington, J. and A. Crumbliss, 2009. The redox hypothesis in siderophore-mediated iron uptake. BioMetals, 22:679-689.

Harrison, W., E.J.H. Head, R.J. Conover, A.R. Longhurst and D.D. Sameot, 1985. The distribution and metabolism of urea in the eastern Canadian Arctic. Deep Sea Research I, 32:23-42.

Hautmann, M., M.J. Benton and A. Tomasovych, 2008. Catastrophic ocean acidification at the Triassic-Jurassic boundary. Neues Jahrbuch Fur Geologie Und Palaontologie-Abhandlungen, 249:119-127.

Heggie, D.T., 1982. Copper in surface waters of the Bering Sea. Geochimica et Cosmochimica Acta, 46:1301-1306.

Heide-Jørgensen, M.P., K.L. Laidre, D. Borchers, T.A. Marques, H. Stern and M.J. Simon, 2010. The effect of sea ice loss on beluga whales (*Delphinapterus leucas*) in West Greenland. Polar Research, 29:198-208.

Hendry, K.R., R.E.M. Rickaby, M.P. Meredith and H. Elderfield, 2009. Controls on stable isotope and trace metal uptake in *Neogloboquadrina pachyderma* (sinistral) from an Antarctic seaice environment. Earth and Planetary Science Letters, 278:67-77.

Hernández, J.C. and M.P. Russell, 2010. Substratum cavities affect growth-plasticity, allometry, movement and feeding rates in the sea urchin *Strongylocentrotus purpuratus*. Journal of Experimental Biology, 213:520-525.

Hernroth, B., S. Baden, M. Thorndyke and S. Dupont, 2011. Immune suppression of the echinoderm *Asterias rubens* (L.) following long-term ocean acidification. Aquatic Toxicology, 103:222-224.

Hester, K.C., E.T. Pelzer, W.J. Kirkwood and P.G. Brewer, 2008. Unanticipated consequences of ocean acidification: A noisier ocean at lower pH. Geophysical Research Letters, 35:L19601.

Hilmi, N., D. Allemand, S. Dupont, A. Safa, G. Haraldsson, P.L.D. Nunes, C. Moore, C. Hattam, S. Reynaud, J.M. Hall-Spencer, M. Fine, C. Turley, R. Jeffree, J. Orr, P.L. Munday and S.R. Cooley, 2013. Towards improved socio-economic assessments of ocean acidification's impacts. Marine Biology, doi:10.1007/s00227-012-2031-5.

Hirche, H.J., 1997. Life cycle of the copepod *Calanus hyperboreus* in the Greenland Sea. Marine Biology, 128:607-618.

Hoegh-Guldberg, O. and J.F. Bruno, 2010. The impact of climate change on the world's marine ecosystems. Science, 328:1523-1528.

Hoegh-Guldberg, O., P.J. Mumby, A.J. Hooten, R.S. Steneck, P. Greenfield, E. Gomez, C.D. Harvell, P.F. Sale, A.J. Edwards, K. Caldeira, N. Knowlton, C.M. Eakin, R. Iglesias-Prieto, N. Muthiga, R.H. Bradbury, A. Dubi and M.E. Hatziolos, 2007. Coral reefs under rapid climate change and ocean acidification. Science, 318:1737-1742.

Hofmann, G.E., J.P. Barry, P.J. Edmunds, R.D. Gates, D.A. Hutchins, T. Klinger and M.A. Sewell, 2010. Ocean acidification

impacts on calcifying marine organisms in marine ecosystems. Annual Review of Ecology Evolution and Systematics, 41:127-147.

Hofmann, G.E., J.E. Smith, K.S. Johnson, U. Send, L.A. Levin, F. Micheli, A. Paytan, N.N. Price, B. Peterson, Y. Takeshita, P.G. Matson, E.D. Crook, K.J. Kroeker, M.C. Gambi, E.B. Rivest, C.A. Frieder, P.C. Yu and T.R. Martz, 2011. High-frequency dynamics of ocean pH: A multi-ecosystem comparison. PLoS ONE, 6:e28983.

Hölemann, J.A., M. Schirmacher and A. Prange, 2005. Seasonal variability of trace metals in the Lena River and the southeastern Laptev Sea: Impact of the spring freshet. Global and Planetary Change, 48:112-125.

Holland, M., C. Bitz and B. Tremblay, 2006. Future abrupt reductions in the summer Arctic sea ice. Geophysical Research Letters, 33:L23503.

Holligan, P.M., E. Fernandez, J. Aiken, W.M. balch, P. Boyd, P.H. Burkhill, M. Finch, S.B. Groom, G. Malin, K. Muller, C.C. Trees and S.M. Turner, 1993. A biogeochemical study of the coccolithophore, *Emiliania huxleyi*, in the North Atlantic. Global Biogeochemical Cycles, 7:879-900.

Honjo, S., S.J. Manganini, R. Krishfield and R. Francois, 2008. Particulate organic carbon fluxes to the ocean interior and factors controlling the biological pump: A synthesis of global sediment trap programs since 1983. Progress in Oceanography, 76:217-285.

Honjo, S., R.A. Krishfield, T.I. Eglinton, S.J. Manganini, J.N. Kemp, K. Doherty, J. Hwang, T.K. McKee and T. Takizawa, 2010. Biological pump processes in the cryopelagic and hemipelagic Arctic Ocean: Canada Basin and Chukchi Rise. Progress in Oceanography, 85:137-170.

Hoover, C., M. Bailey, J. Higdon, S. Ferguson and U.R. Sumaila, 2013. Estimating the value of narwhal and beluga hunts in Hudson Bay, Nunavut. Arctic, 66:1-16.

Hopkins, F.E., S.M. Turner, P.D. Nightingale, M. Steinke, D. Bakker and P.S. Liss, 2010. Ocean acidification and marine trace gas emissions. Proceedings of the National Academy of Sciences, 107:760-765.

Hopkins, F.E., S.A. Kimmance, J.A. Stephens, R.G.J. Bellerby, C.P.D. Brussard, J. Czerny, K.G. Schulz and S.D. Archer, 2012. Response of halocarbons to ocean acidification in the Arctic. Biogeosciences Discussions, 9:8199-8239.

Hovelsrud, G.K., 1999. The integrative role of seals in an East Greenlandic hunting village. Arctic Anthropology, 36:37-50.

Hovelsrud, G.K., B. Poppel, B. van Oort and J.D. Reist, 2011. Arctic societies, cultures, and peoples in a changing cryosphere. Ambio, 40:100-110.

Hu, M.Y., Y.-C. Tseng, M. Stumpp, M.A. Gutowska, R. Kiko, M. Lucassen and F. Melzner, 2011. Elevated seawater pCO_2 differentially affects branchial acid-base transporters over the course of development in the cephalopod *Sepia officinalis*. American Journal of Physiology, 300:R1100-R1114.

Hunt, B.P.V., E.A. Pakhomov, G.W. Hosie, V. Siegel, P. Ward and K. Bernard, 2008. Pteropods in Southern Ocean ecosystems. Progress in Oceanography, 78:193-221.

Hunter, K.A., P.S. Liss, V. Surapipith, F. Detener, R. Duce, M. Kanakidou, N. Kubilay, N. Mahowald, G. Okin, M. Sarin, M. Uematu and T. Zhu, 2011. Impacts of anthropogenic SOx, NOx and NH₃ on acidification of coastal waters and shipping lanes. Geophysical Research Letters, 38:L13602.

Huntington, H.P., J.H. Mosli and V. Shustov, 1998. Peoples of the Arctic: Characteristics of human populations relevant to pollution issues. In: AMAP Assessment Report: Arctic Pollution Issues, pp. 141-182. Arctic Monitoring and Assessment Programme, Oslo, Norway.

Hurst, M.P. and K.W. Bruland, 2007. An investigation into the exchange of iron and zinc between soluble, colloidal, and particulate size-fractions in shelf waters using low-abundance isotopes as tracers in shipboard incubation experiments. Marine Chemistry, 103:211-226.

Hurst, T.P., E.R. Fernandez and J.T. Mathis, 2013. Effects of ocean acidification on hatch size and larval growth of walleye pollock (*Theragra chalcogramma*). ICES Journal of Marine Science, 70:812-822.

Hutchins, D.A., F.X. Fu, Y. Zhang, M.E. Warner, Y. Feng, K. Portune, P.W. Bernhardt and M.R. Mulholland, 2007. CO₂ control of *Trichodesmium* N-2 fixation, photosynthesis, growth rates, and elemental ratios: Implications for past, present, and future ocean biogeochemistry. Limnology and Oceanography, 52:1293-1304.

Iglesias-Rodriguez, M.D., P.R. Halloran, R.E.M. Rickaby, I.R. Hall, E. Colmenero-Hidalgo, J.R. Gittins, D.R. H. Green, T. Tyrrell, S.J. Gibbs, P. von Dassow, E. Rehm, E.V. Armbrust and K.P. Boessenkool, 2008. Phytoplankton calcification in a high-CO, world. Science, 320:336-340.

Ilyina, T., K.D. Six, J. Segschneider, E. Maier-Reimer, H. Li and I. Nunez-Riboni, 2013. Global ocean biogeochemistry model HAMOCC: Model architecture and performance as component of the MPI-Earth System Model in different CMIP5 experimental realizations. Journal of Advances in Modeling Earth Systems, 5:287-315.

IMR, 2011. Ocean Acidification. Marine Research News, no. 15-2011. Institute of Marine Research, Bergen.

Inaba, K., C. Dreanno and J. Cosson, 2003. Control of flatfish sperm motility by CO_2 and carbonic anhydrase. Cell Motility and the Cytoskeleton, 55:174-187.

IPCC, 2007. Climate Change 2007: The Physical Science Basis. Solomon, S., D. Qin, M. Manning, Z. Chen, M. Marquis, K.B. Averyt, M. Tignor and H.L. Miller (Eds.). Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, 996 pp.

IPCC, 2011. Workshop Report of the Intergovernmental Panel on Climate Change Workshop on

Impacts of Ocean Acidification on Marine Biology and Ecosystems. Field, C.B., V. Barros, T.F.

Stocker, D. Qin, K.J. Mach, G.-K. Plattner, M.D. Mastrandrea, M. Tignor and K.L. Ebi (Eds.). Intergovernmental Panel on Climate Change, Working Group II Technical Support Unit, 164 pp.

Ishimatsu, A., T. Kikkawa, M. Hayashi, K.S. Lee and J. Kita, 2004. Effects of CO_2 on marine fish: Larvae and adults. Journal of Oceanography, 60:731-741.

Ishimatsu, A., M. Hayashi, K.S. Lee, T. Kikkawa and J. Kita, 2005. Physiological effects on fishes in a high- CO_2 world. Journal of Geophysical Research, 110:C09S09.

Ishimatsu, A., M. Hayashi and T. Kikkawa, 2008. Fishes in high-CO₂, acidified oceans. Marine Ecology Progress Series, 373:295-302.

Jay, C.V., B.G. Marcot and D.C. Douglas, 2011. Projected status of the Pacific walrus (*Odobenus rosmarus divergens*) in the twenty-first century. Polar Biology, 34:1065-1084.

Jeansson, E., A. Olsen, T. Eldevik, I. Skjelvan, A.M. Omar, S.K. Lauvset, J.E.Ø. Nilsen, R.G.J. Bellerby, T. Johannessen and E. Falck, 2011. The Nordic Seas carbon budget: Sources, sinks, and uncertainties. Global Biogeochemical Cycles, 25:GB4010.

Joks, S., 2003. Reports No. 2: Co-management of marine resources in Arctic areas with respect to indigenous peoples and traditional ecological knowledge. Initiative Arctic Council, Saami Council.

Johnson, V.R., C. Brownlee, R.E.M. Rickaby, M. Graziano, M. Milazzo and J.M. Hall-Spencer, 2011. Responses of marine benthic microalgae to elevated CO₂. Marine Biology, 1-12.

Johnson, V.R., B.D. Russell, K.E. Fabricius, C. Brownlee and J.M. Hall-Spencer, 2012. Temperate and tropical brown macroalgae thrive, despite decalcification, along natural CO_2 gradients. Global Change Biology, 18:2792-2803.

Jones, E.P. and L.G. Anderson, 1986. On the origin of the chemical properties of the Arctic Ocean halocline. Journal of Geophysical Research, 91:10,759-10,767.

Jones, E.P. and L.G. Anderson, 1990. On the origin of the properties of the Arctic Ocean halocline north of Ellesmere Island: Results from the Canadian Ice Island. Continental Shelf Research, 10:485-498.

Jones, E.P., B. Rudels and L.G. Anderson, 1995. Deep waters of the Arctic Ocean: Origin and circulation. Deep-Sea Research I, 42:737-760.

Jones, E.P., J.H. Swift, L.G. Anderson, M. Lipizer, G. Civitarese, K.K. Falkner, G. Kattner and F. McLaughlin, 2003. Tracing Pacific water in the North Atlantic Ocean. Journal of Geophysical Research, 108:13-1 - 13-10.

Joos, F., T. L. Frölicher, M. Steinacher and G.-K. Plattner, 2011. Impact of climate change mitigation on ocean acidification projections. In: J.-P. Gattuso and L. Hansson (Eds.). Ocean Acidification, pp. 272-290. Oxford University Press.

Junge, K., H. Eicken and J.W. Deming, 2004. Bacterial activity at 2 to 20 °C in Arctic wintertime sea ice. Applied and Environmental Microbiology, 70:550-557.

Jutterström, S. and L.G. Anderson, 2005. The saturation of calcite and aragonite in the Arctic Ocean. Marine Chemistry, 94:101-110.

Jutterström, S. and L.G. Anderson, 2010. Uptake of CO₂ by the Arctic Ocean in a changing climate. Marine Chemistry, 122:96-104.

Jutterström, S., L.G. Anderson, N.R. Bates, R. Bellerby, T. Johannessen, E.P. Jones, R.M. Key, X. Lin, A. Olsen and A.M. Omar, 2010. Arctic Ocean data in CARINA. Earth System Science Data, 2:71-78.

Kadko, D. and K. Aagaard, 2009. Glimpses of Arctic Ocean shelf–basin interaction from submarine-borne radium sampling. Deep-Sea Research I, 56:32-40.

Kadko, D. and R. Muench, 2005. Evaluation of shelf-basin interaction in the western Arctic by use of short-lived radium isotopes: The importance of mesoscale processes. Deep Sea Research II, 52:3227-3244.

Kahru, M., V. Brotas, M. Manzano-Sarabia and B.G. Mitchell, 2011. Are phytoplankton blooms occurring earlier in the Arctic? Global Change Biology, 17:1733-1739.

Kaltin, S. and L.G. Anderson, 2005. Uptake of atmospheric carbon dioxide in Arctic shelf seas: evaluation of the relative importance of processes that influence pCO_2 in water transported over the Bering-Chukchi Sea shelf. Marine Chemistry, 94:67-79.

Kaltin, S., L.G. Anderson, K. Olsson, A. Fransson and M. Chierici, 2002. Uptake of atmospheric carbon dioxide in the Barents Sea. Journal of Marine Systems, 38:31-45.

Karlsson, R., A. Karlsson, O. Bäckman, B.R. Johansson and S. Hulth, 2009. Identification of key proteins involved in the anammox reaction. FEMS Microbiology Letters, 297:87-94.

Karnovsky, N.J., K.A. Hobson, S. Iverson and G.L. Hunt, Jr., 2008. Seasonal changes in diets of seabirds in the North Water Polynya: a multiple-indicator approach. Marine Ecology Progress Series, 357:291-299.

Kasamatsu, N., T. Hirano, S. Kudoh, T. Odate and M. Fukuchi, 2004. Dimethylsulfoniopropionate production by psychrophilic diatom isolates. Journal of Phycology, 40:874-878.

Kelley, J.J., 1968. Carbon dioxide in the seawater under the arctic ice. Nature, 218:862-864.

Kelley, J.J., L.L. Longerich and D.W. Hood, 1971. Effect of upwelling, mixing, and high primary productivity on CO_2 concentrations in surface waters of the Bering Sea. Journal of Geophysical Research, 76:8687-8693.

Kelly, M.W. and G.E. Hofmann, 2013. Adaptation and the physiology of ocean acidification. Functional Ecology, 27:980-990.

Key, R.M., T. Tanhua, A. Olsen, M. Hoppema, S. Jutterström, C. Schirnick, S. van Heuven, A. Kozyr, X. Lin, A. Velo, D.W.R. Wallace and L. Mintrop, 2009. The CARINA data synthesis project: introduction and overview. Earth System Science Data Discussions, 2:579-624.

Kikkawa, T., A. Ishimatsu and J. Kita, 2003. Acute CO₂ tolerance during the early developmental stages of four marine teleosts. Environmental Toxicology, 18:375-382.

Kim, J.-M., K. Lee, E.J. Yang and 11 others, 2010. Enhanced production of oceanic dimethylsulfide resulting from CO_2 -induced grazing activity in a high CO_2 world. Environmental Science and Technology, 44:8140-8143.

King, D.W., 1998. Role of carbonate speciation on the oxidation rate of Fe(II) in aquatic systems. Environmental Science and Technology, 32:2997-3003.

King, D.W. and R. Farlow, 2000. Role of carbonate speciation on the oxidation of Fe(II) by H₂O₂. Marine Chemistry, 70:201-209.

Kite-Powell, H.L., 2009. A global perspective on the economics of ocean acidification. Journal of Marine Education, 25:25-29.

Kivimäe, C., R.G.J. Bellerby, A. Fransson, M. Reigstad and T. Johannessen, 2010. A carbon budget for the Barents Sea. Deep-Sea Research I, 57:1532-1542.

Kleypas, J.A., R.W. Buddemeier, D. Archer, J.P. Gattuso, C. Langdon and B.N. Opdyke, 1999. Geochemical consequences of increased atmospheric carbon dioxide on coral reefs. Science, 284:118-120.

Kleypas, J.A., R.A. Feely, V.J. Fabry, C. Langodon, C.L. Sabine and L.L. Robbins, 2006. Impacts of Ocean Acidification on Coral Reefs and Other Marine Calcifiers: A Guide for Future Research. Report of a workshop held 18–20 April 2005, St. Petersburg, Florida. Sponsored by the National Science Foundation, the National Oceanic and Atmospheric Administration and the U.S. Geological Survey, 88 pp.

Klironomos, J.N., M.F. Allen, M.C. Rillig, J. Piotrowski, S. Makvandi-Nejad, B.E. Wolfe and J.R. Powell, 2005. Abrupt rise in atmospheric CO_2 overestimates community response in a model plant-soil system. Nature, 433:621-624.

Klunder, M.B., D. Bauch, P. Laan, H.J.W. de Baar, S. van Heuven and S. Ober, 2012a. Dissolved iron in the Arctic shelf seas and surface waters of the central Arctic Ocean: Impact of Arctic river water and ice-melt. Journal of Geophysical Research, 117:C01027.

Klunder, M.B., P. Laan, R. Middag, H.J.W. de Baar and K. Bakker, 2012b. Dissolved iron in the Arctic Ocean: Important role of hydrothermal sources, shelf input and scavenging removal. Journal of Geophysical Research, 117:C04014.

Koike, I. and A. Hattori, 1979. Estimates of denitrification in sediments of the Bering Sea shelf. Deep Sea Research A, 26:409-415.

Kordas, R.L., C.D.G. Harley and M.I. O'Connor, 2011. Community ecology in a warming world: The influence of temperature on interspecific interactions in marine systems. Journal of Experimental Marine Biology and Ecology, 400:218-226.

Kroeker, K.J., R.L. Kordas, R.N. Crim and G.G. Singh, 2010. Metaanalysis reveals negative yet variable effects of ocean acidification on marine organisms. Ecology Letters, 13:1419-1434.

Kroeker, K.J., R.L. Kordas, R. Crim, I.E. Hendriks, L. Ramajo, G.S. Sihgh, C.M. Duarte and J.-P. Gattuso, 2013. Impacts of ocean acidification on marine organisms: quantifying sensitivities and interaction with warming. Global Change Biology, 19:1884-1896.

Kruse, J., B. Poppel, L. Abryutina, G. Duhaime, S. Martin, M. Poppel, M., Kruse, E. Ward, P. Cochran and V. Hanna, 2008. Survey of Living Conditions in the Arctic (SLiCA). In: Møller, V., D. Huschka and A.C. Michalos (Eds.). Barometers of Quality of Life Around the Globe: How Are We Doing? pp. 107-134. Springer Verlag. Kubler, J.E., A.M. Johnston and J.A. Raven, 1999. The effects of reduced and elevated CO_2 and O_2 on the seaweed *Lomentaria articulata*. Plant Cell and Environment, 22:1303-1310.

Kuenen, J.G., 2008. Anammox bacteria: from discovery to application. Nature Reviews Microbiology, 6:320-326.

Kuenzler, E.J. and J.P. Perras, 1965. Phosphatases of marine algae. Biological Bulletin, 128:271-284.

Kuffner, I.B., A.J. Andersson, P.L. Jokiel, K.S. Rodgers and F.T. Mackenzie, 2008. Decreased abundance of crustose coralline algae due to ocean acidification. Nature Geoscience, 1:114-117.

Kump, L.R., T.J. Bralower and A. Ridgwell, 2009. Ocean acidification in deep time. Oceanography, 22:94-107.

Kurihara, H., 2008. Effects of CO_2 -driven ocean acidification on the early developmental stages of invertebrates. Marine Ecology Progress Series, 373:275-284.

Kurihara, H. and A. Ishimatsu, 2008. Effects of high CO₂ seawater on the copepod (*Acartia tsuensis*) through all life stages and subsequent generations. Marine Pollution Bulletin, 56:1086-1090.

Kurihara, H., S. Shimode and Y. Shirayama, 2004a. Sub-lethal effects of elevated concentration of CO_2 on planktonic copepods and sea urchins. Journal of Oceanography, 60:743-750.

Kurihara, H., S. Shimode and Y. Shirayama, 2004b. Effects of raised CO₂ concentration on the egg production rate and early development of two marine copepods (*Acartia steueri* and *Acartia erythraea*). Marine Pollution Bulletin, 49:721-727.

Kurihara, H., M. Matsui, H. Furukawa, M. Hayashi and A. Ishimatsu, 2008. Long-term effects of predicted future seawater CO_2 conditions on the survival and growth of the marine shrimp *Palaemon pacificus*. Journal of Experimental Marine Biology and Ecology, 367:41-46.

Kuroyanagi, A., H. Kawahata, A. Suzuki, K. Fujita and T. Irie, 2009. Impacts of ocean acidification on large benthic foraminifers: Results from laboratory experiments. Marine Micropaleontology, 73:190-195.

Kutti, T. and J.H. Fosså, 2009. Coral survey off Northern Norway, Benthic Habitats and Shellfish Research Group. Institute of Marine Research, Cruise Report *R/V H. Mosby* – cruise No. 2009615; *M/S Øyfisk* cruise No. 2009813.

Kwok, R., G. Cunningham, M. Wensnahan, I. Rigor and H. Zwally, 2009. Thinning and volume loss of the Arctic ocean sea ice cover: 2003-2008. Journal of Geophysical Research, 114:C07005.

Lagoe, M.B., 1976. Species diversity of deep-sea benthic Foraminifera from the central Arctic Ocean. Geological Society of America Bulletin, 87:1678-1683.

Laidre, K.L. and M.P. Heide-Jorgensen, 2005. Winter feeding intensity of narwhals (*Monodon monoceros*). Marine Mammal Science, 21:45-57.

Laidre, K.L., M.P. Heide-Jorgensen, R. Dietz, R.C. Hobbs and O.A. Jorgensen, 2003. Deep-diving by narwhals *Monodon monoceros*: differences in foraging behavior between wintering areas? Marine Ecology Progress Series, 261:269-281.

Laidre, K.L., M.P. Heide-Jorgensen, O.A. Jorgensen and M.A. Treble, 2004a. Deep-ocean predation by a high Arctic cetacean. ICES Journal of Marine Science, 61:430-440.

Laidre, K.L., M.P. Heide-Jorgensen, M.L. Logdson, R.C. Hobbs, P. Heagerty, R. Dietz, O.A. Jorgensen and M.A. Treble, 2004b. Seasonal narwhal habitat associations in the high Arctic. Marine Biology, 145:821-831.

Landes, A. and M. Zimmer, 2012. Acidification and warming affect both a calcifying predator and prey, but not their interaction. Marine Ecology Progress Series, 450:1-10.

Langdon, C. and M.J. Atkinson, 2005. Effect of elevated pCO_2 on photosynthesis and calcification of corals and interactions with seasonal change in temperature/irradiance and nutrient enrichment. Journal of Geophysical Research, 110:C09S07.

Langer, G., M. Geisen, K.H. Baumann, J. Klas, U. Riebesell, S. Thoms and J.R. Young, 2006. Species-specific responses of calcifying algae to changing seawater carbonate chemistry. Geochemistry Geophysics Geosystems, 7:12.

Langer, G., G. Nehrke, I. Probert, J. Ly and P. Ziveri, 2009. Strainspecific responses of *Emiliania huxleyi* to changing seawater carbonate chemistry. Biogeosciences, 6:2637-2646.

Lannig, G., A.S. Cherkasov, H.O. Pörtner, C. Bock and I.M. Sokolova, 2008. Cadmium-dependent oxygen limitation affects temperature tolerance in eastern oysters (*Crassostrea virginica* Gmelin). American Journal of Physiology, 294:R1338-R1346.

Lansard, B., A. Mucci, L.A. Miller, R.W. Macdonald and Y. Gratton, 2012. Seasonal vaiability of water mass distribution in the southeastern Beaufort Sea determined by total alkalinity and δ^{18} O. Journal of Geophysical Research, 117:C03003. doi: 10.1029/2011JC007346.

Larsen, N.J., P. Schweitzer and G. Fondahl, 2010. Arctic Social Indicators – a follow-up of the Arctic Human Development Report. TemaNord 2010:519. Nordic Council of Ministers.

Lavoie, D., K.L. Denman and R.W. Macdonald, 2010. Effects of future climate change on primary productivity and export fluxes in the Beaufort Sea. Journal of Geophysical Research, 115:C04018.

Lehmann, M.F., D.M. Sigman, D.C. McCorkle, B.G. Brunelle, S. Hoffman, M. Kienast, G. Cane and J. Clement, 2005. Origin of the deep Bering Sea nitrate deficit: Constraints from the nitrogen and oxygen isotopic composition of water column nitrate and benthic nitrate fluxes. Global Biogeochemical Cycles, 19:GB4005.

Leonardos, N. and R.J. Geider, 2005. Elevated atmospheric carbon dioxide increases organic carbon fixation by *Emiliania huxleyi* (Haptophyta) under nutrient-limited high-light conditions. Journal of Phycology, 41:1196-1203.

Lepore, K., and S.B. Moran, 2007. Seasonal changes in thorium scavenging and particle aggregation in the western Arctic Ocean. Deep-Sea Research I, 54:919-938.

Lepore, K., S.B. Moran, J.M. Grebmeier, L.W. Cooper, C. Lalande, W. Maslowski, V. Hill, N.R. Bates, D.A. Hansell, J.T. Mathis and R.P. Kelly, 2007. Seasonal and interannual changes

in particulate organic carbon export and deposition in the Chukchi Sea. Journal of Geophysical Research, 112:C10024.

Levasseur, M., 2011. Ocean science: If Gaia could talk. Nature Geoscience, 4:351-352.

Levinsen, H., J.T. Turner, T.G. Nielsen and B.W. Hansen, 2000. On the trophic coupling between protists and copepods in Arctic marine ecosystems. Marine Ecology Progress Series, 204:65-77.

Levitan, O., G. Rosenberg, I. Setlik, E. Setlikova, J. Grigel, J. Klepetar, O. Prasil and I. Berman-Frank, 2007. Elevated CO₂ enhances nitrogen fixation and growth in the marine cyanobacterium *Trichodesmium*. Global Change Biology, 13:531-538.

Li, W.K.W., F.A. McLaughlin, C. Lovejoy and E.C. Carmack, 2009. Smallest algae thrive as the Arctic Ocean freshens. Science, 326:539.

Lischka, S., J. Buedenbender, T. Boxhammer and U. Riebesell, 2011. Impact of ocean acidification and elevated temperatures on early juveniles of the polar shelled pteropod *Limacina helicina*: mortality, shell degradation, and shell growth. Biogeosciences, 8:919-932.

Liu, K.-K., L. Atkinson, C.T.A. Chen, S. Gao, J. Hall, R.W. Macdonald, L. Talaue McManus and R. Quiñones, 2000. Exploring continental margin carbon fluxes on a global scale. Eos, Transactions, American Geophysical Union, 81:641, 642, 644.

Liu, J., M.G. Weinbauer, C. Maier, M. Dai and J.-P. Gattuso, 2010. Effect of ocean acidification on microbial diversity and on microbe-driven biogeochemistry and ecosystem functioning. Aquatic Microbial Ecology, 61:291-305.

Long, Z. and W. Perrie, 2013. Impacts of climate change on fresh water content and sea surface height in the Beaufort Sea. Ocean Modelling, 71:127-139.

Loose, B. and P. Schlosser, 2011. Sea ice and its effect on CO_2 flux between the atmosphere and the Southern Ocean interior. Journal of Geophysical Research, 116:C11019.

Loose, B., L.A. Miller, S. Elliott and T. Papakyriakou, 2011. Sea ice biogeochemistry and material transport across the frozen interface. Oceanography, 24:202-218.

Louis, Y., C. Garnier, V. Lenoble, D. Omanovic, S. Mounier and I. Pizeta, 2009. Characterisation and modelling of marine dissolved organic matter interactions with major and trace cations. Marine Environmental Research, 67:100-107.

Lowenstam, H.A. and S. Weiner, 1989. On Biomineralization. Oxford University Press.

Lowry, L.F., G. Sheffield and J.C. George, 2004. Bowhead whale feeding in the Alaskan Beaufort Sea, based on stomach contents analyses. Journal of Cetacean Research and Management, 6:215-223.

Lueker, T.J., A.G. Dickson and C.D. Keeling, 2000. Ocean pCO_2 calculated from dissolved inorganic carbon, alkalinity, and equations for K1 and K2: validation based on laboratory measurements of CO_2 in gas and seawater at equilibrium. Marine Chemistry, 70:105-119.

Macdonald, R. and C. Gobeil, 2012. Manganese sources and sinks in the Arctic Ocean with reference to periodic enrichments in basin sediments. Aquatic Geochemistry, 18:565-591.

Maier, C., J. Hegeman, M.G. Weinbauer and J.P. Gattuso, 2009. Calcification of the cold-water coral *Lophelia pertusa* under ambient and reduced pH. Biogeosciences, 6:1671-1680.

Maldonado, M.T. and N.M. Price, 2001. Reduction and transport of organically bound iron by *Thalassiosira oceanica* (Bacillariophyceae). Journal of Phycology, 37:298-309.

Maldonado, M.T., R.F. Strzepek, S. Sander and P.W. Boyd, 2005. Acquistion of iron bound to strong organic complexes, with different Fe binding groups and photochemical reactivities, by plankton communities in Fe-limited subantarctic waters. Global Biogeochemical Cycles, 19:GB4S23. doi: 10.1029/2005GB002481.

Maneja, R.H., A.Y. Frommel, H.I. Browman, C. Clemmesen, A.J. Geffen, A. Folkvord, U. Piatkowski, C.M.F. Durif, R. Bjelland and A.B. Skiftesvik, 2013. The swimming kinematics of larval Atlantic cod, *Gadus morhua* L., are resilient to elevated seawater *p*CO₂. Marine Biology, 160:1963-1972.

Manno, C., V. Tirelli, A. Accornero and S.F. Umani, 2010. Importance of the contribution of *Limacina helicina* faecal pellets to the carbon pump in Terra Nova Bay (Antarctica). Journal of Plankton Research, 32:145-152.

Manno, C., N. Morata and R. Bellerby, 2012a. Effect of ocean acidification and temperature increase on the planktonic foraminifer *Neogloboquadrina pachyderma* (sinistral). Polar Biology, 35:1311-1319.

Manno, C., N. Morata and R. Primicerio, 2012b. *Limacina retroversa*'s response to combined effects of ocean acidification and sea water freshening. Estuarine, Coastal and Shelf Science, 113:163-171.

Manseau, M., B. Parlee and G.B. Ayles, 2005. A place for traditional ecological knowledge in resource management. In: Berkes, F., R. Huebert, H. Fast, M. Manseau and A. Diduck (Eds.). Breaking Ice: Renewable Resource and Ocean Management in the Canadian North, pp. 141-164. University of Calgary Press.

Maranger, R., D.F. Bird and S.K. Juniper, 1994. Viral and bacterial dynamics in Arctic sea ice during the spring algal bloom near Resolute, NWT, Canada. Marine Ecology Progress Series, 111:121-127.

Mart, L., H.W. Nürnberg and D. Dyrssen, 1984. Trace metal levels in the Eastern Arctic Ocean. Science of the Total Environment, 39:1-14.

Martens-Habbena, W., P.M. Berube, H. Urakawa, J.R. de la Torre and D.A. Stahl, 2009. Ammonia oxidation kinetics determine niche separation of nitrifying Archaea and Bacteria. Nature, 461:976-979.

Martin, S. and J.-P. Gattuso, 2009. Response of Mediterranean coralline algae to ocean acidification and elevated temperature. Global Change Biology, 15:2089-2100.

Martin, J.M., D.M. Guan, F. Elbaz-Poulichet, A.J. Thomas and V.V. Gordeev, 1993. Preliminary assessment of the distributions

of some trace elements (As, Cd, Cu, Fe, Ni, Pb and Zn) in a pristine aquatic environment: The Lena River estuary (Russia). Marine Chemistry, 43:185-199.

Martin, S., R. Rodolfo-Metalpa, E. Ransome, S. Rowley, M.C. Buia, J.P. Gattuso and J. Hall-Spencer, 2008. Effects of naturally acidified seawater on seagrass calcareous epibionts. Biology Letters, 4:689-692.

Martinez, J.S., M.G. Haygood and A. Butler, 2001. Identification of a natural desferrioxamine siderophore produced by a marine bacterium. Limnology and Oceanography, 46:420-424.

Maslanik, J., C. Fowler, J. Stroeve, S. Drobot, H. Zwally, D. Yi and W. Emery, 2007. A younger, thinner ice cover: increased potential for rapid, extensive ice loss. Geophysical Research Letters, 34:L24501.

Maslowski, W., J.C. Kinney, M. Higgins and A. Roberts, 2012. The future of Arctic sea ice. Annual Review of Earth and Planetary Sciences, 40:625-654.

Mathis, J.T., R.S. Pickart, D.A. Hansell, D. Kadko and N.R. Bates, 2007. Eddy transport of organic carbon and nutrients from the Chukchi Shelf: Impact on the upper halocline of the western Arctic Ocean. Journal of Geophysical Research, 112:C05011. doi: 10.1029/2006JC003899.

Mathis, J.T., J.N. Cross and N.R. Bates, 2011a. Coupling primary production and terrestrial runoff to ocean acidification and carbonate mineral suppression in the eastern Bering Sea. Journal of Geophysical Research, 116:C02030. doi: 10.1029/2010JC006453.

Mathis J.T., J.N. Cross J.N. and N.R. Bates, 2011b. The role of ocean acidification in systematic carbonate mineral suppression in the Bering Sea. Geophysical Research Letters, 38:L19602. doi: 10.1029/2011GL048884.

Mathis, J.T., R.S. Pickart, R.H. Byrne, C.L. McNeil, G.W.K. Moore, L.W. Juranek, X. Liu, J. Ma, R.A. Easley, M.M. Elliot, J.N. Cross, S.C. Reisdorph, F. Bahr, J. Morison, T. Lichendorf and R.A. Feely, 2012. Storm-induced upwelling of high pCO_2 waters onto the continental shelf of the western Arctic Ocean and implications for carbonate mineral saturation states. Geophysical Research Letters, 39:L07606. doi: 10.1029/2012GL051574.

Mathisen, O.A., K.O. Coyle and A.S.G.C. Program, 1996. Ecology of the Bering Sea: a review of Russian literature. Alaska Sea Grant College Program, University of Alaska Fairbanks.

Mauchline, J., 1980. The biology of mysids and euphausiids. Advances in Marine Biology, 18:1-681.

Mayor, D.J., C. Matthews, K. Cook, A.F. Zuur and S. Hay, 2007. CO₂-induced acidification affects hatching success in *Calanus finmarchicus*. Marine Ecology Progress Series, 350:91-97.

McClelland, J.W., S. Déry, B.J. Peterson, R.M. Holmes and E. Wood, 2006. A pan-Arctic evaluation of changes in river discharge during the latter half of the 20th century. Geophysical Research Letters, 33:L06715.

McClelland, J.W., R.M. Holmes, K.H. Dunton and R.W. Macdonald, 2012. The Arctic Ocean estuary. Estuaries and Coasts, 35:353-368.

McClintock, J.B., M.O. Amsler, R.A. Angus, R.C. Challener, J.B. Schram, C.D. Amsler, C.L. Mah, J. Cuce and B.J. Baker, 2011. The Mg-calcite composition of Antarctic echinoderms: Important implications for predicting the impacts of ocean acidification. Journal of Geology, 119:457-466.

McDonald, M.R., J.B. McClintock, C.D. Amsler, D. Rittschof, R.A. Angus, B. Orihuela and K. Lutostanski, 2009. Effects of ocean acidification over the life history of the barnacle *Amphibalanus amphitrite*. Marine Ecology Progress Series, 385:179-187.

McElhany, P. and S.D. Busch, 2012. Appropriate pCO_2 treatments in ocean acidification experiments. Marine Biology, October:1-6.

McGuire, A.D., L.G. Anderson, T.R. Christensen, S. Dallimore, L. Guo, D.J. Hayes, M. Heimann, T.D. Lorenson, R.W. Macdonald and N. Roulet, 2009. Sensitivity of the carbon cycle in the Arctic to climate change. Ecological Monographs, 79:523-555.

McGuire, A.D., R.W. Macdonald, E.A.G. Schuur, J.W. Harden, P. Kuhry, D.J. Hayes, T.R. Christensen and M. Heimann, 2010. The carbon budget of the northern cryosphere region. Current Opinion in Environmental Sustainability, 2:231-236.

McLaughlin, F.A., E.C. Carmack, R.W. Macdonald and J.K.B. Bishop, 1996. Physical and geochemical properties across the Atlantic/Pacific water mass front in the southern Canadian Basin. Journal of Geophysical Research, 101:1183-1197.

McLaughlin, F.A., E.C. Carmack, R.G. Ingram, W.J. Williams and C. Michel, 2006. Oceanography of the Northwest Passage. The Sea, 14:1213-1244.

McNeil, B.I. and R.J. Matear, 2007. Climate change feedbacks on future oceanic acidification. Tellus, 59B:191-198.

McPhee, M.G. and T.P. Stanton, 1996. Turbulence in the statically unstable oceanic boundary layer under Arctic leads. Journal of Geophysical Research, 101:6409-6428.

McPhee, M.G., A. Proshutinsky, J.H. Morison, M. Steele and M.B. Alkire, 2009. Rapid change in freshwater content of the Arctic Ocean. Geophysical Research Letters, 36:L10602.

Meakin, N.G. and M. Wyman, 2011. Rapid shifts in picoeukaryote community structure in response to ocean acidification. ISME Journal, 5:1397-1405.

Measures, C.I., 1999. The role of entrained sediments in sea ice in the distribution of aluminium and iron in the surface waters of the Arctic Ocean. Marine Chemistry, 68:59-70.

Mecklenburg, C., P. Møller and D. Steinke, 2011. Biodiversity of arctic marine fishes: taxonomy and zoogeography. Marine Biodiversity, 41:109-140.

Mehrbach, C., C.H. Culberson, J.E. Hawley and R.M. Pytkowicz, 1973. Measurement of the apparent dissociation constants of carbonic acid in seawater at atmospheric pressure. Limnology and Oceanography, 18:897-907.

Melle, W. and H.R. Skjoldal, 1998. Reproduction and development of *Calanus finmarchicus*, *C. glacialis* and *C. hyperboreus* in the Barents Sea. Marine Ecology Progress Series, 169:211-228.

87

Melle, W., B. Ellertsen and H.R. Skjoldal, 2004. The Norwegian Sea Ecosystem. Tapir Academic Press.

Melling, H. and R.M. Moore, 1995. Modification of halocline source waters during freezing on the Beaufort Sea shelf: evidence from oxygen isotopes and dissolved nutrients. Continental Shelf Research, 15:89-113.

Melzner, F., M.A. Gutowska, M. Langenbuch, S. Dupont, M. Lucassen, M.C. Thorndyke, M. Bleich and H.-O. Pörtner, 2009a. Physiological basis for high CO₂ tolerance in marine ectothermic animals: pre-adaptation through lifestyle and ontogeny? Biogeosciences, 6:2313-2331.

Melzner, F., S. Göbel, M. Langenbuch, M.A. Gutowska, H.O. Pörtner and M. Lucassen, 2009b. Swimming performance in Atlantic Cod (*Gadus morhua*) following long-term (4-12 months) acclimation to elevated seawater pCO_2 . Aquatic Toxicology, 92:30-37.

Melzner, F., P. Stange, K. Trübenbach, J. Thomsen, I. Casties, U. Panknin, S.N. Gorb and M.A. Gutowska, 2011. Food supply and seawater pCO_2 impact calcification and internal shell dissolution in the blue mussel *Mytilus edulis*. PLoS One, 6:e24223.

Melzner, F., J. Thomsen, W. Koeve, A. Oschlies, M.A. Gutowska, H.W. Bange, H.P. Hansen and A. Körtzinger, 2013. Future ocean acidification will be amplified by hypoxia in coastal habitats. Marine Biology, 160:1875-1888.

Metzger, R., F.J. Sartoris, M. Langenbuch and H.O. Pörtner, 2007. Influence of elevated CO_2 concentrations on thermal tolerance of the edible crab *Cancer pagurus*. Journal of Thermal Biology, 32:144-151.

Merico, A., T. Tyrrell, C.W. Brown, S.B. Groom and P.I. Miller, 2003. Analysis of satellite imagery for *Emiliania huxleyi* blooms in the Bering Sea before 1997. Geophysical Research Letters, 30:1337.

Michaelidis, B., C. Ouzounis, A. Paleras and H.O. Pörtner, 2005. Effects of long-term moderate hypercapnia on acidbase balance and growth rate in marine mussels *Mytilus* galloprovincialis. Marine Ecology Progress Series, 293:109-118.

Michaud, J., L. Fortier, P. Rowe and R. Ramseier, 1996. Feeding success and survivorship of Arctic cod larvae, *Boreogadus saida*, in the northeast water polynya (Greenland Sea). Fisheries Oceanography, 5:120-135.

Michel, C., L. Legendre, S. Demers and J.C. Therriault, 1988. Photoadaptation of sea ice microalgae in springtime – Photosynthesis and carboxylating enzymes. Marine Ecology Progress Series, 50:177-185.

Michel, C., R.G. Ingram and L.R. Harris, 2006. Variability in oceanographic and ecological processes in the Canadian Arctic Archipelago. Progress in Oceanography, 71:379-401.

Middag, R., H.J.W. de Baar, P. Laan and K. Bakker, 2009. Dissolved aluminium and the silicon cycle in the Arctic Ocean. Marine Chemistry, 115:176-195.

Middag, R., H.J.W. de Baar, P. Laan and M.B. Klunder, 2011. Fluvial and hydrothermal input of manganese into the Arctic Ocean. Geochimica et Cosmochimica Acta, 75:2393-2408. Middelboe, A.L. and P.J. Hansen, 2007. Direct effects of pH and inorganic carbon on macroalgal photosynthesis and growth. Marine Biology Research, 3:134-144.

Miller, L.A., P.L. Yager, K.A. Erickson, D. Amiel, J. Bâcle, J.K. Cochran, M.-È. Garneau, M. Gosselin, D.J. Hirschberg, B. Klein, B. LeBlanc and W.L. Miller, 2002. Carbon distributions and fluxes in the North Water, 1998 and 1999. Deep-Sea Research II, 49:5151-5170.

Miller, L.A., T.N. Papakyriakou, R.E. Collins, J.W. Deming, J.K. Ehn, R.W. Macdonald, A. Mucci, O. Owens, M. Raudsepp and N. Sutherland, 2011. Carbon dynamics in sea ice: A winter flux time series. Journal of Geophysical Research, 116:C02028. doi: 10.1029/2009JC006058.

Miller, G.M., S.-A. Watson, J.M. Donelson, M.I. McCormick and P.L. Munday, 2012. Parental environment mediates impacts of increased carbon dioxide on a coral reef fish. Nature Climate Change, 2:858-861.

Millero, F.J., 1989. Effect of ionic interactions on the oxidation of Fe(II) and Cu(I) in natural waters. Marine Chemistry, 28:1-18.

Millero, F.J. and S. Sotolongo, 1989. The oxidation of Fe(II) with H_2O_2 in seawater. Geochimica et Cosmochimica Acta, 53:1867-1873.

Millero, F.J., S. Sotolongo and M. Izaguirre, 1987. The oxidation kinetics of Fe(II) in seawater. Geochimica et Cosmochimica Acta, 51:793-801.

Millero, F.J., R. Woosley, B. Ditrolio and J. Waters, 2009. Effect of ocean acidification on the speciation of metals in seawater. Oceanography, 22:72-85.

Milligan, A.J., D.E. Varela, M.A. Brzezinski and F. Morel, 2004. Dynamics of silicon metabolism and silicon isotopic discrimination in a marine diatom as a function of pCO_2 . Limnology and Oceanography, 49:322-329.

Milligan, A.J., C.E. Mioni and F.M.M. Morel, 2009. Response of cell surface pH to *p*CO₂ and iron limitation in the marine diatom *Thalassiosira weissflogii*. Marine Chemistry, 114:31-36.

Moffett, J.W. and L.E. Brand, 1996. Production of strong, extracellular Cu chelators by marine cyanobacteria in response to Cu stress. Limnology and Oceanography, 41:388-395.

Moffett, J.W. and C. Dupont, 2007. Cu complexation by organic ligands in the sub-arctic NW Pacific and Bering Sea. Deep-Sea Research I, 54:586-595.

Moffett, J.W. and R.G. Zika, 1983. Oxidation kinetics of Cu(I) in seawater: Implications for its existence in the marine environment. Marine Chemistry, 13:239-251.

Moffett, J.W., R.G. Zika and L.E. Brand, 1990. Distribution and potential sources and sinks of copper chelators in the Sargasso Sea. Deep-Sea Research, 37:27-36.

Moore, R.M., 1981. Oceanographic distributions of zinc, cadmium, copper and aluminium in waters of the central Arctic. Geochimica et Cosmochimica Acta, 45:2475-2482.

Moore, R.M. and J.N. Smith, 1986. Disequilibria between ²²⁶Ra, ²¹⁰Pb and ²¹⁰Po in the Arctic Ocean and the implications for

chemical modification of the Pacific water inflow. Earth and Planetary Science Letters, 77:285-292.

Moore, T.S., M.D. Dowell and B.A. Franz, 2012. Detection of coccolithophore blooms in ocean color satellite imagery: A generalized approach for use with multiple sensors. Remote Sensing of Environment, 117:249-263.

Moran, S.B., J.A. Hoff, K.O. Buesseler and R.L. Edwards, 1995. High precision ²³⁰Th and ²³²Th in the Norwegian Sea and Denmark by thermal ionization mass spectrometry. Geophysical Research Letters, 22:2589-2592.

Moran, S.B., K.M. Ellis and J.N. Smith, 1997. ²³⁴Th/²³⁸U disequilibrium in the central Arctic Ocean: implications for particulate organc carbon export. Deep-Sea Research II, 44:1593-1606.

Morel, F.M.M., A.B. Kustka and Y. Shaked, 2008. The role of unchelated Fe in the iron nutrition of phytoplankton. Limnology and Oceanography, 53:400-404.

Moss, R.H., J.A. Edmonds, K.A. Hibbard, M.R. Manning, S.K. Rose, D.P. van Vuuren, T.R. Carter, S. Emori, M. Kainuma, T. Kram, G.A. Meehl, J.F.B. Mitchell, N. Nakicenovic, K. Riahi, S.J. Smith, R.J. Stouffer, A.M. Thomson, J.P. Weyant and T.J. Wilbanks, 2010. The next generation of scenarios for climate change research and assessment. Nature, 463:747-756.

Moulin, L., A.I. Catarino, T. Claessens and P. Dubois, 2011. Effects of seawater acidification on early development of the intertidal sea urchin *Paracentrotus lividus* (Lamarck 1816). Marine Pollution Bulletin, 62:48-54.

Moy, A.D., W.R. Howard, S.G. Bray and T.W. Trull, 2009. Reduced calcification in modern Southern Ocean planktonic foraminifera. Nature Geoscience, 2:276-280.

Mucci, A., 1983. The solubility of calcite and aragonite in seawater at various salinities, temperatures, and one atmosphere total pressure. American Journal of Science, 283:780-799.

Mucci, A., B. Lansard, L.A. Miller and T.N. Papakyriakou, 2010. CO_2 fluxes across the air-sea interface in the southeastern Beaufort Sea: Ice-free period. Journal of Geophysical Research, 115:C04003. doi: 10.1029/2009JC005330.

Mucci, A., M. Starr, D. Gilbert and B. Sundby, 2011. Acidification of Lower St. Lawrence Estuary bottom waters. Atmosphere-Ocean, 49:206-218.

Muller, M.N., K.G. Schulz and U. Riebesell, 2010. Effects of longterm high CO_2 exposure on two species of coccolithophores. Biogeosciences, 7:1109-1116.

Munari, M., V. Matozzo, G. Chemello and M.G. Marin, 2012. Combined effects of seawater acidification and diclofenac on immune parameters of the clam *Ruditapes philippinarum*. Comparative Biochemistry and Physiology A, 163:S5-S5.

Munday, P.L., N.E. Crawley and G.E. Nilsson, 2009a. Interacting effects of elevated temperature and ocean acidification on the aerobic performance of coral reef fishes. Marine Ecology Progress Series, 388:235-242.

Munday, P.L., J.M. Donelson, D.L. Dixson and G.G.K. Endo, 2009b. Effects of ocean acidification on the early life history

of a tropical marine fish. Proceedings of the Royal Society B, 276:3275-3283.

Munday, P.L., D.L. Dixson, J.M. Donelson, G.P. Jones, M.S. Pratchett, G.V. Devitsina and K.B. Doving, 2009c. Ocean acidification impairs olfactory discrimination and homing ability of a marine fish. Proceedings of the National Academy of Sciences of the United States of America, 106:1848-1852.

Munday, P.L., D.L. Dixson, M.I. McCormick, M. Meekan, M.C.O. Ferrari and D.P. Chivers, 2010. Replenishment of fish populations is threatened by ocean acidification. Proceedings of the National Academy of Sciences of the United States of America, 107:12930-12934.

Munday, P.L., M. Gagliano, J.M. Donelson, D.L. Dixson and S.R. Thorrold, 2011a. Ocean acidification does not affect the early life history development of a tropical marine fish. Marine Ecology Progress Series, 423:211-221.

Munday, P.L., V. Hernaman, D.L. Dixson and S.R. Thorrold, 2011b. Effect of ocean acidification on otolith development in larvae of a tropical marine fish. Biogeosciences, 8:1631-1641.

Munday, P.L., M.I. McCormick, M. Meekan, D.L. Dixson, S.-A. Watson, D.P. Chivers and M.C.O. Ferrari, 2013. Selective mortality associated with variation in CO₂ tolerance in a marine fish. Ocean Acidification, 1:1-5.

Mundy, C.J., D.G. Barber and C. Michel, 2005. Variability of snow and ice thermal, physical and optical properties pertinent to sea ice algae biomass during spring. Journal of Marine Systems, 58:107-120.

Mundy, C.J., M. Gosselin, J. Ehn, Y. Gratton, A. Rossnagel, D.G. Barber, J. Martin, J-E. Tremblay, M. Palmer, K.R. Arrigo, G. Darnis, L. Fortier, B. Else and T. Papakyriakou, 2009. Contribution of under-ice primary production to an ice-edge upwelling phytoplankton bloom in the Canadian Beaufort Sea. Geophysical Research Letters, 36:L17601.

Murata, A. and T. Takizawa, 2003. Summertime CO_2 sinks in shelf and slope waters of the western Arctic Ocean. Continental Shelf Research, 23:753-776.

Murray, J.W., 1991. Ecology and Palaeoecology of Benthic Foraminifera. Adison Wesley Longman, 398 pp.

Murray, J.L., B.G.E. de March and B.T. Hargrave, 1998. Ecological characteristics of the Arctic. In: AMAP Assessment Report: Arctic Pollution Issues, pp. 117-139. Arctic Monitoring and Assessment Programme, Oslo, Norway.

Nakamura, M., S. Ohki, A. Suzuki and K. Sakai, 2011. Coral larvae under ocean acidification: Survival, metabolism and metamorphosis. Plos One, 6:e14521.

Nakaoka, S., A. Aiki, T. Nakazawa, G. Hashida, S. Morimoto, T. Yamanouchi and H. Yoshikawa-Inoue, 2006. Temporal and spatial variations of oceanic pCO_2 and air-sea CO_2 flux in the Greenland Sea and the Barents Sea. Tellus, 58:148-161.

Nakayama, Y., S. Fujita, K. Kuma and K. Shimada, 2011. Iron and humic-type fluorescent dissolved organic matter in the Chukchi Sea and Canada Basin of the western Arctic Ocean. Journal of Geophysical Research, 116:C07031. Nakicenovic, N., J. Alcamo, G. Davis and 25 others, 2000. IPCC Special Report on Emissions Scenarios. Intergovernmental Panel on Climate Change. Technical Report. Cambridge University Press.

Narita, D., K. Rehdanz and R.S.J. Tol, 2012. Economic costs of ocean acidification: a look into the impacts of global shellfish production. Climate Change, 113:1049-1063.

Nedashkovsky, A.P. and A.P. Makshtas, 2010. Emission of CO_2 to atmosphere during Arctic sea ice formation. Problems of the Arctic and Antarctic, No.3 (86), p. 35-44 (in Russian).

Nedashkovsky, A.P. and Shvetsova, M.G., 2010. Total inorganic carbon in sea ice. Oceanology, 50:861-868.

Newbold, L.K., A.E. Oliver, T. Booth, B. Tiwari, T. DeSantis, M. Maguire, G. Andersen, C.J. van der Gast and A.S. Whiteley, 2012. The response of marine picoplankton to ocean acidification. Environmental Microbiology, 14:2293-2307.

Nielsen, L.T., H.H. Jakobsen and P.J. Hansen, 2010. High resilience of two coastal plankton communities to twentyfirst century seawater acidification: Evidence from microcosm studies. Marine Biology Research, 6:542-555.

Nielsen, L.T., G.M. Hallegraeff, S.W. Wright and P.J. Hansen, 2011. Effects of experimental seawater acidification on an estuarine plankton community. Aquatic Microbial Ecology, 65:271-285.

Nilsson, G.E., D.L. Dixson, P. Domenici, M.I. McCormick, C. Sorensen, S.A. Watson and P.L. Munday, 2012. Near-future carbon dioxide levels alter fish behaviour by interfering with neurotransmitter function. Nature Climate Change, 2:201-204.

Nimmo, M., C.M.G. van den Berg and J. Brown, 1989. The chemical speciation of dissolved nickel, copper, vanadium and iron in Liverpool Bay, Irish Sea. Estuarine, Coastal and Shelf Science, 29:57-74.

Nishino, S., K. Shimada, M. Itoh and S. Chiba, 2009. Vertical double silicate maxima in the sea-ice reduction region of the western Arctic Ocean: Implications for an enhanced biological pump due to sea-ice reduction. Journal of Oceanography, 65:871-883.

Nishino, S., T. Kikuchi, M. Yamamoto-Kawai, Y. Kwwaguchi, T. Hirawake and M. Itoh, 2011. Enhancement/reduction of biological pump depends on ocean circulation in the sea-ice reduction regions of the Arctic Ocean. Journal of Oceanography, 67:305-314.

Nitishinsky, M., L.G. Anderson and J.A. Hölemann, 2007. Inorganic carbon and nutrient fluxes on the Arctic Shelf. Continental Shelf Research, 27:1584-1599.

NJFF, 2011. Norges Jeger- og Fiskerforbund (Norwegian Association of Hunters and Fishers) www.njff.no/portal/page/ portal/njff/artikkel?displaypage=TRUE&element_id=67207 accessed 20 November 2011.

Nomura, D., H. Eicken, R. Gradinger and K. Shirasawa, 2010a. Rapid physically driven inversion of the air-sea ice CO_2 flux in the seasonal landfast ice off Barrow, Alaska after onset of surface melt. Continental Shelf Research, 30:1998-2004. Nomura, D., H. Yoshikawa-Inoue, T. Toyota and K. Shirasawa, 2010b. Effects of snow, snowmelting and refreezing processes on air-sea-ice CO₂ flux. Journal of Glaciology, 56:262-270.

Not, C., K. Brown, B. Ghaleb and C. Hillaire-Marcel, 2012. Conservative behavior of uranium vs. salinity in Arctic sea ice and brine. Marine Chemistry, 130-1:33-39.

Nuttall, M., 2002. Protecting the Arctic: Indigenous Peoples and Cultural Survival. Routledge, Abingdon, UK.

Nuttall, M., 2005. Hunting, herding, fishing and gathering: Indigenous peoples and renewable resource use in the Arctic. In: Arctic Climate Impact Assessment, pp. 649-690. Cambridge University Press.

O'Brien, M.C., R.W. Macdonald, H. Melling and K. Iseki, 2006. Geochemistry and physical forcing of sediment transport and deposition in the Canadian Beaufort Sea. Continental Shelf Research, 26:41-81.

O'Brien, M.C., H. Melling, T.F. Pederson and R.W. Macdonald, 2011. The role of eddies and energetic ocean phenomena in the transport of sediment from shelf to basin in the Arctic. Journal of Geophysical Research, 116:C08001.

O'Brien, M.C., H. Melling, T.F. Pederson and R.W. Macdonald, 2013. The oceanographic context of particle flux in the Canada Basin of the Arctic Ocean. Deep-Sea Research I, 71:1-20.

O'Corry-Crowe, G.M., 2002. Encyclopedia of Marine Mammals. Academic Press.

O'Corry-Crowe, G.M., 2009. Encyclopedia of Marine Mammals. Elsevier Science.

O'Dowd, C.D., M.C. Facchini, F. Cavalli, D. Ceburnis, M. Mircea, S. Decesari, S. Fuzzi, Y.J. Yoon and J.-P. Putaud, 2004. Biogenically driven organic contribution to marine aerosol. Nature, 431:676-680.

Ohde, S. and R. van Woesik, 1999. Carbon dioxide flux and metabolic processes of a coral reef, Okinawa. Bulletin of Marine Science, 65:559-576.

Olafsson, J., S.R. Olafsdottir, A. Benoit-Cattin, M. Danielsen, T.S. Arnarson and T. Takahashi, 2009. Rate of Iceland Sea acidification from time series measurements. Biogeosciences, 6:2661-2668.

Olsen, A., A.M. Omar, R.G.J. Bellerby, T. Johanessen, U. Ninnemann, K.R. brown, K.A. Olsson, J. Olafsson, G. Nondal, C. Kivimäe, S. Kringstad and S. Olafsdottir, 2006. Magnitude and origin of the anthropogenic CO₂ increase and ¹³C Suess effect in the Nordic seas since 1981. Global Biogeochemical Cycles, 20:GB3027.

Olsen, A., R.M. Key, E. Jeansson, E. Falck, J. Olafsson, S. van Heuven, I. Skjelvan, A.M. Omar, K.A. Olsson, L.G. Anderson, S. Jutterström, F. Rey, T. Johannessen, R.G.J. Bellerby, J. Blindheim, J.L. Bullister, B. Pfeil, X. Lin, A. Kozyr, C. Schirnick, T. Tanhua and D.W.R. Wallace, 2009. Overview of the Nordic Seas CARINA data and salinity measurements. Earth System Science Data, 1:25-34.

Olsen, A., A.M. Omar, E. Jeansson, L.G. Anderson and R.G.J. Bellerby, 2010. Nordic seas transit time distributions and anthropogenic CO₂. Journal of Geophysical Research, 115:C05005.

Olsen, M.S., Callaghan, T.V., Reist, J.D., Reiersen, L.O., Dahl-Jensen, D., Granskog, M.A., Goodison, B., Hovelsrud, G.K., Johansson, M., Kallenborn, R., Key, J., Klepikov, A., Meier, W., Overland, J.E., Prowse, T.D., Sharp, M., Vincent, W.F. and Walsh, J.E., 2011. The changing Arctic cryosphere and likely consequences: An overview. Ambio, 40: 111-118.

Omar, A., T. Johannessen, S. Kaltin and A. Olsen, 2003. Anthropogenic increase of oceanic pCO_2 in the Barents Sea surface water. Journal of Geophysical Research, 108:3388.

Omar, A., T. Johannessen, R.G.J. Bellerby, A. Olsen, L.G. Anderson and C. Kivimäe, 2005. Sea-ice and brine formation in Storfjorden: Implications for the Arctic wintertime air-sea CO_2 flux. In: Drange, H., T. Dokken, T. Furevik, R. Gedes and W. Berger (Eds.). The Nordic Seas: An Integrated Perspective. Geophysical Monograph Series, 158:177-187.

Omar, A.M., T. Johannessen, A. Olsen, S. Kaltin and F. Rey, 2007. Seasonal and interannual variability of the air-sea CO₂ flux in the Atlantic sector of the Barents Sea. Marine Chemistry, 104:203-213.

Orr, J.C., V.J. Fabry, O. Aumont, L. Bopp, S.C. Doney, R.A. Feely, A. Gnanadesikan, N. Gruber, A. Ishida, F. Joos, R.M. Key, K. Lindsay, E. Maier-Reimer, R. Matear, P. Monfray, A. Mouchet, R.G. Najjar, G.K. Plattner, K.B. Rodgers, C.L. Sabine, J.L. Sarmiento, R. Schlitzer, R.D. Slater, I.J. Totterdell, M.F. Weirig, Y. Yamanaka and A. Yool, 2005. Anthropogenic ocean acidification over the twenty-first century and its impact on calcifying organisms. Nature, 437:681-686.

Osmond, M., S. Airame, M. Caldwell and J. Day, 2010. Lessons for marine conservation planning: A comparison of three protected area planning processes. Ocean and Coastal Management, 53:41-51.

Østvedt, O.J., 1955. Hvalrådets Skrifter, I kommisjon hos J. Dybwad.

Pane, E.F. and J.P. Barry, 2007. Extracellular acid-base regulation during short-term hypercapnia is effective in a shallow-water crab, but ineffective in a deep-sea crab. Marine Ecology Progress Series, 334:1-9.

Papakyriakou, T. and L. Miller, 2011. Springtime CO₂ exchange over seasonal sea ice in the Canadian Arctic Archipelago. Annals of Glaciology, 52:215-224.

Parker, L.M., P.M. Ross, W.A. O'Connor, L. Borysko, D.A. Raftos and H.-O. Pörtner, 2012. Adult exposure influences offspring response to ocean acidification in oysters. Global Change Biology, 18:82-92.

Parker-Stetter, S.L., J.K. Horne and T.J. Weingartner, 2011. Distribution of polar cod and age-0 fish in the US Beaufort Sea. Polar Biology, 34:1543-1557.

Passow, U. and C. Carlson, 2012. The biological pump in a high CO, world. Marine Ecology Progress Series, 470:249-271.

Paulic, J.E. and M.H. Papst, 2013. Larval and early juvenile fish distribution and assemblage structure in the Canadian Beaufort Sea during July–August, 2005. Journal of Marine Systems, in press.

Paulino, A.I., J.K. Egge and A. Larsen, 2008. Effects of increased atmospheric CO_2 on small and intermediate sized osmotrophs during a nutrient induced phytoplankton bloom. Biogeosciences, 5:739-748.

Paull, C.K., W. Ussler III, S.R. Dallimore, S.M. Blasco, T.D. Lorenson, H. Melling, B.E. Medioli, F.M. Nixon and F.A. McLaughlin, 2007. Origin of pingo-like features on the Beaufort Sea shelf and their possible relationship to decomposing methane gas hydrates. Geophysical Research Letters, 34:L01603.

Pflaumann, U., J. Duprat, C. Pujol and L.D. Labeyrie, 1996. SIMMAX: A modern analog technique to deduce Atlantic sea surface temperatures from planktonic foraminifera in deep-sea sediments. Paleoceanography, 11:15-35.

Pineda, J., C. DiBacco and V. Starczak, 2005. Barnacle larvae in ice: Survival, reproduction, and time to post-settlement metamorphosis. Limnology and Oceanography, 50:1520-1528.

Piontek, J., M. Lunau, N. Haendel, C. Borchard, M. Wurst and A. Engel, 2010. Acidification increases microbial polysaccharide degradation in the ocean. Biogeosciences, 7:1615-1624.

Pipko, I.I., I.P. Semiletov, P.Y. Tischenko, S.P. Pugach and J.P. Christensen, 2002. Carbonate chemistry dynamics in Bering Strait and the Chukchi Sea. Progress in Oceanography, 55:77-94.

Pipko, I.I., I.P. Semiletov, P. Ya. Tishchenko, S.P. Pugach and N.I. Savel'eva, 2008. Variability of the carbonate system parameters in the coast–shelf zone of the East Siberian Sea during the autumn season. Oceanology, 48:54-67.

Pipko, I., S.P. Pugach and I.P. Semiletov, 2009. The autumn distribution of the CO_2 partial pressure in bottom waters of the East Siberian Sea. Doklady Earth Sciences, 425:345-349.

Pipko, I.I., I.P. Semiletov, S.P. Pugach, I. Wåhlström and L.G. Anderson, 2011. Interannual variability of air-sea CO_2 fluxes and carbonate system parameters in the East Siberian Sea. Biogeosciences, 8:1987-2007.

Pistevos, J.C.A., P. Calosi, S. Widdicombe and J.D.D. Bishop, 2011. Will variation among genetic individuals influence species responses to global climate change? Oikos, 120:675-689.

Pitzer, K.S., 1973. Thermodynamics of electrolytes. I. Theoretical basis and general equations. Journal of Physical Chemistry, 77:268-277.

Pohl, C., G. Kattner and M. Schulzbaldes, 1993. Cadmium, copper, lead and zinc on transects through Arctic and eastern Atlantic surface and deep waters. Journal of Marine Systems, 4:17-29.

Polacheck, T., 1990. Year round closed areas as a management tool. Natural Resource Modeling, 4:327-354.

Polyak, L., J. Bischof, J.D. Ortiz, D.A. Darby, J.E.T. Channell, C. Xuan, D.S. Kaufman, R. Lovlie, D.A. Schneider, D.D. Eberl, R.E. Adler and E.A. Council, 2009. Late Quaternary stratigraphy and sedimentation patterns in the western Arctic Ocean. Global and Planetary Change, 68:5-17.

Poppel, B., 2006. Interdependency of subsistence and market economies in the Arctic. In: Glomsrød, S. and I. Aslaksen (Eds.). The Economy of the North, pp.65-80. Statistics Norway, Oslo.

Poppel, B. and J. Kruse, 2009. The importance of a mixed cash and harvest herding based economy to living in the Arctic: an analysis based on Survey of Living Conditions in the Arctic (SLiCA). In: Muller, V. and D. Huschka (Eds.). Quality of Life in the New Millennium: Advances in the Quality-of-Life Studies, Theory and Research, pp. 27-42. Springer Verlag.

Poppel, B., J. Kruse, G. Duhaime and L. Abryutina, 2007. SLiCA Results. Anchorage: Institute of Social and Economic Research, University of Alaska Anchorage.

Poppel, B., J. Kruse, A.R. Broderstad, B.M. Eliassen, S. Martin, M. Melhus, S. Olsvig and C. Westin, 2011. SLiCA Survey of living conditions in the Arctic. Lessons learned.

Porcelli, D., P.S. Andersson, M. Baskaran, M. Frank, G. Björk and I. Semiletov, 2009. The distribution of neodymium isotopes in Arctic Ocean basins. Geochimica et Cosmochimica Acta, 73:2645-2659.

Pörtner, H.O., 2001. Climate change and temperaturedependent biogeography: oxygen limitation of thermal tolerance in animals. Naturwissenschaften, 88:137-146.

Pörtner, H.O., 2002. Climate variations and the physiological basis of temperature dependent biogeography: systemic to molecular hierarchy of thermal tolerance in animals. Comparative Biochemistry and Physiology A, 132:739-761.

Pörtner, H.O., 2006. Climate-dependent evolution of Antarctic ectotherms: An integrative analysis. Deep-Sea Research II, 53:1071-1104.

Pörtner, H.O., 2008. Ecosystem effects of ocean acidification in times of ocean warming: a physiologist's view. Marine Ecology Progress Series, 373:203-217.

Pörtner, H.O., 2010. Oxygen- and capacity-limitation of thermal tolerance: a matrix for integrating climate-related stressor effects in marine ecosystems. Journal of Experimental Biology, 213:881-893.

Pörtner, H.O. and A.P. Farrell, 2008. Ecology, physiology and climate change. Science, 322:690-692.

Pörtner, H.O. and R. Knust, 2007. Climate change affects marine fishes through the oxygen limitation of thermal tolerance. Science, 315:95-97.

Pörtner, H.O., M. Langenbuch and A. Reipschlager, 2004. Biological impact of elevated ocean CO_2 concentrations: Lessons from animal physiology and earth history. Journal of Oceanography, 60:705-718.

Potera, C., 2010. Will ocean acidification erode the base of the food web? Environmental Health Perspectives, 118:A157-A157.

Praebel, K., J.I. Westgaard, S.E. Fevolden and J.S. Christiansen, 2008. Circumpolar genetic population structure of capelin *Mallotus villosus*. Marine Ecology Progress Series, 360:189-199.

Price, N.M. and F.M.M. Morel, 1990. Role of extracellular enzymatic reactions in natural waters. In: W. Stumm (Ed.). Aquatic Chemical Kinetics: Reaction Rates of Processes in Natural Waters, pp. 235-257. Wiley.

Price, N.N., S.L. Hamilton, J.S. Tootell and J.E. Smith, 2011. Species-specific consequences of ocean acidification for the calcareous tropical green algae *Halimeda*. Marine Ecology Progress Series, 440:67-78.

Quinn, P.K. and T.S. Bates, 2011. The case against climate regulation via oceanic phytoplankton sulfur emissions. Nature, 480:51-56.

Rampal, P., J. Weiss and D. Marsan, 2009. Positive trend in the mean speed and deformation rate of Arctic sea ice, 1979-2007. Journal of Geophysical Research Oceans, 114:1-14.

Rand, K.M. and E.A. Logerwell, 2011. The first demersal trawl survey of benthic fish and invertebrates in the Beaufort Sea since the late 1970s. Polar Biology, 34:475-488.

Range, P., M.A. Chicharo, R. Ben-Hamadou, D. Pilo, D. Matias, S. Joaquim, A.P. Oliveira and L. Chicharo, 2011. Calcification, growth and mortality of juvenile clams *Ruditapes decussatus* under increased pCO_2 and reduced pH: Variable responses to ocean acidification at local scales? Journal of Experimental Marine Biology and Ecology, 396:177-184.

Rasmussen, R.O., 2005. Analyse av fangererhvervet I Grønland. [In Danish] 158 pp. Grønnland Hjemmestyre, Direktoratet for fangst og fiskeri, Nuuk, Greenland.

Rasmussen, R.O. (Ed.), 2011, Megatrends, TemaNord 2011:527. Nordic Council of Ministers, Copenhagen.

Raven, J.A., 2011. Effects on marine algae of changed seawater chemistry with increasing atmospheric CO_2 . Biology and Environment, Proceedings of the Royal Irish Academy, 111B:1-17.

Reid, R.T. and A. Butler, 1991. Investigation of the mechanism of iron acquisition by the marine bacterium *Alteromonas luteoviolaceus*: Characterization of siderophore production. Limnology and Oceanography, 36:1783-1792.

Reigstad, M., P. Wassmann, C.W. Riser, S. Oygarden and F. Rey, 2002. Variations in hydrography, nutrients and chlorophyll a in the marginal ice-zone and the central Barents Sea. Journal of Marine Systems, 38:9-29.

Reisch, C.R., M.J. Stoudemayer, V.A. Varaljay, I.J. Amster, M.A. Moran and W.B. Whitman, 2011. Novel pathway for assimilation of dimethylsulphoniopropionate widespread in marine bacteria. Nature, 473:208-+.

Renaud, P.E., N. Morata, W.G. Ambrose, Jr., J.J. Bowie and A. Chiuchiolo, 2007. Carbon cycling by seafloor communities on the eastern Beaufort Sea shelf. Journal of Experimental Marine Biology and Ecology, 349:248-260.

Renaud, P.E., J. Berge, O. Varpe, O.J. Lonne, J. Nahrgang, C. Ottesen and I. Hallanger, 2012. Is the poleward expansion by Atlantic cod and haddock threatening native polar cod, *Boreogadus saida*? Polar Biology, 35:401-412.

Reygondeau, G. and G. Beaugrand, 2011. Future climate-driven shifts in distribution of *Calanus finmarchicus*. Global Change Biology, 17:756-766.

Rickaby, R.E.M., J. Henderiks and J.N. Young, 2010. Perturbing phytoplankton: response and isotopic fractionation with changing carbonate chemistry in two coccolithophore species. Climate of the Past, 6:771-785.

Riebesell, U., 2004. Effects of CO_2 enrichment on marine phytoplankton. Journal of Oceanography, 60:719-729.

Riebesell, U. and P.D. Tortell, 2011. Effects of ocean acidification on pelagic organisms and ecosystems. In: Gattuso, J.P. and L. Hansson (Eds.). Ocean Acidification, pp. 99-121. Oxford University Press. Riebesell, U., I. Zondervan, B. Rost, P.D. Tortell, R.E. Zeebe and F.M.M. Morel, 2000. Reduced calcification of marine plankton in response to increased atmospheric CO₂. Nature, 407:364-367.

Riebesell, U., K.G. Schulz, R.G.J. Bellerby, M. Botros, P. Fritsche, M. Meyerhöfer, C. Neill, G. Nondal, A. Oschlies, J. Wohlers and E. Zöllner, 2007. Enhanced biological carbon consumption in a high CO₂ ocean. Nature, 450:545-548.

Riebesell, U., V.J. Fabry, L. Hansson and J.-P. Gattuso (Eds.), 2010. Guide to best practices for ocean acidification research and data reporting. Publications Office of the European Union, 260 pp.

Ridgwell, A. and J.C. Hargreaves, 2007. Regulation of atmospheric CO_2 by deep-sea sediments in an Earth system model. Global Biogeochemical Cycles, 21:GB2008.

Ries, J.B., 2010. Review: geological and experimental evidence for secular variation in seawater Mg/Ca (calcite-aragonite seas) and its effects on marine biological calcification. Biogeosciences, 7:2795-2849.

Ries, J.B., A.L. Cohen and D.C. McCorkle, 2009. Marine calcifiers exhibit mixed responses to CO_2 -induced ocean acidification. Geology, 37:1131-1134.

Ripperger, S., M. Rehkämper, D. Porcelli and A.N. Halliday, 2007. Cadmium isotope fractionation in seawater – A signature of biological activity. Earth and Planetary Science Letters, 261:670-684.

Robbins, L.L., P.O. Knorr and P. Hallock, 2009. Response of *Halimeda* to ocean acidification: field and laboratory evidence. Biogeosciences Discussions, 6:4895-4918.

Rochelle-Newall, E., B. Delille, M. Frankignoulle, J.P. Gattuso, S. Jacquet, U. Riebesell, A. Terbruggen and I. Zondervan, 2004. Chromophoric dissolved organic matter in experimental mesocosms maintained under different pCO_2 levels. Marine Ecology Progress Series, 272:25-31.

Rodolfo-Metalpa, R., C. Lombardi, S. Cocito, J.M. Hall-Spencer and M.C. Gambi, 2010. Effects of ocean acidification and high temperatures on the bryozoan *Myriapora truncata* at natural CO₂ vents. Marine Ecology – an Evolutionary Perspective, 31:447-456.

Rodolfo-Metalpa, R., F. Houlbreque, E. Tambutte, F. Boisson, C. Baggini, F.P. Patti, R. Jeffree, M. Fine, A. Foggo, J.P. Gattuso and J.M. Hall-Spencer, 2011. Coral and mollusc resistance to ocean acidification adversely affected by warming. Nature Climate Change, 1:308-312.

Roleda, M.Y., P.W. Boyd and C.L. Hurd, 2012. Before ocean acidification: calcifier chemistry lessons. Journal of Phycology, 48:840-843.

Rosa, R. and B.A. Seibel, 2008. Synergistic effects of climaterelated variables suggest future physiological impairment in a top oceanic predator. Proceedings of the National Academy of Sciences, 105:20776-20780.

Rossoll, D., R. Bermudez, H. Hauss, K.G. Schulz, U. Riebesell, U. Sommer and M. Winder, 2012. Ocean acidification-induced food quality deterioration constrains trophic transfer. Plos One, 7:e34737.

Roy, R.N., L.N. Roy, K.M. Vogel, C. Porter-Moore, T. Pearson, C.E. Good, F.J. Millero and D.M. Campbell, 1993. The dissociation constants of carbonic acid in seawater at salinities 5 to 45 and temperatures 0 to 45°C. Marine Chemistry, 44:249-267.

Roy, A.-S., S.M. Gibbons, H. Schunck, S. Owens, J.G. Caporaso, M. Sperling, J.I. Nissimov, S. Romac, L. Bittner, M. Mühling, U. Riebesell, J. LaRoche and J. A. Gilbert, 2013. Ocean acidification shows negligible impacts on high-latitude bacterial community structure in coastal pelagic mesocosms, Biogeosciences, 10:555-566.

Rudels, B., E.P. Jones, U. Schauer and P. Eriksson, 2004. Atlantic sources of the Arctic Ocean surface and halocline waters. Polar Research, 23:181-208.

Rue, E.L. and K.W. Bruland, 1995. Complexation of iron(III) by natural organic ligands in the central North Pacific as determined by a new competitive ligand equilibration/ adsorptive cathodic stripping voltammetric method. Marine Chemistry, 50:117-138.

Russell, B.D., J.-A.I. Thompson, L.J. Falkenberg and S.D. Connell, 2009. Synergistic effects of climate change and local stressors: CO_2 and nutrient-driven change in subtidal rocky habitats. Global Change Biology, 15:2153-2162.

Rutgers van der Loeff, M.M., R.M. Key, J. Scholten, D. Bauch and A. Michel, 1995.²²⁸Ra as a tracer for shelf water in the Arctic Ocean. Deep-Sea Research II, 42:1533-1553.

Rutgers van der Loeff, M., S. Kuhne, M. Washner, H. Höltzen, M. Frank, B. Ekwurzel, M. mensch and V. Rachols, 2003. ²²⁸Ra and ²²⁶Ra in the Kara and Laptev seas. Continental Shelf Research, 23:113-124.

Rysgaard, S., R.N. Glud, M.K. Sejr, J. Bendtsen and P.B. Christensen, 2007. Inorganic carbon transport during sea ice growth and decay: A carbon pump in polar seas. Journal of Geophysical Research, 112:C03016.

Rysgaard, S., R. Glud, M. Sejr, M. Blicher and H. Stahl, 2008. Denitrification activity and oxygen dynamics in Arctic sea ice. Polar Biology, 31:527-537.

Rysgaard, S., J. Bendtsen, L.T. Pedersen, H. Ramløv and R.N. Glud, 2009. Increased CO_2 uptake due to sea ice growth and decay in the Nordic Seas. Journal of Geophysical Research, 114:C09011.

Rysgaard, S., J. Bendtsen, B. Delille, G.S. Dieckmann, R.N. Glud, H. Kennedy, J. Mortensen, S. Papadimitriou, D.N. Thomas and J.-L. Tison, 2011. Sea ice contribution to the air-sea CO_2 exchange in the Arctic and Southern Oceans. Tellus, 63B:823-830.

Sabine, C.L., R.A. Feely, N. Gruber, R.M. Key, K. Lee, J.L. Bullister, R. Wanninkhof, C.S. Wong, D.W.R. Wallace, B. Tilbrook, F.J. Millero, T.-H. Pent, A. Kozyr, T. Ono and A.F. Rios, 2004. The oceanic sink for anthropogenic CO_2 . Science, 305:367-371.

Saito, M.A. and J.W. Moffett, 2001. Complexation of cobalt by natural organic ligands in the Sargasso Sea as determined by a new high-sensitivity electrochemical cobalt speciaiton method suitable for open ocean work. Marine Chemistry, 75:49-68.

Saito, M.A., J.W. Moffett and G.R. DiTullio, 2004. Cobalt and nickel in the Peru upwelling region: A major flux of labile cobalt

utilized as a micronutrient. Global Biogeochemical Cycles, 18:GB4030. doi: 10.1029/2003GB002216.

Sakshaug, E., 2004. Primary and secondary production in the Arctic Seas. In: Stein, R. and R.W. Macdonald (Eds.). The Organic Carbon Cycle in the Arctic Ocean, pp. 57-81. Springer.

Sakshaug, E. and H.R. Skjoldal, 1989. Life at the ice edge. Ambio, 18:60-67.

Sambrotto, R.N., J.J. Goering and C.P. McRoy, 1984. Large yearly production of phytoplankton in the western Bering Strait. Science, 225:1147-1150.

Sambrotto, R.N., C. Mordy, S.I. Zeeman, P.J. Stabeno and S.A. Macklin, 2008. Physical forcing and nutrient conditions associated with patterns of Chl a and phytoplankton productivity in the southeastern Bering Sea during summer. Deep-Sea Research II, 55:1745-1760.

Santana-Casiano, J.M., M. Gonzalez-Davila and F.J. Millero, 2005. Oxidation of nanomolar levels of Fe(II) with oxygen in natural waters. Environmental Science and Technology, 39:2073-2079.

Sarmiento, J.L., J. Dunne, A. Gnanadesikan, R.M. Key, K. Matsumoto and R. Slater, 2002. A new estimate of the $CaCO_3$ to organic carbon export ratio. Global Biogeochemical Cycles, 16(4):1107.

Sauer, J., S. Domisch, C. Nowak and P. Haase, 2011. Low mountain ranges: summit traps for montane freshwater species under climate change. Biodiversity and Conservation, 20:3133-3146.

Schartau, M., A. Engel, J. Schroeter, S. Thoms, C. Voelker and D. Wolf-Gladrow, 2007. Modelling carbon overconsumption and the formation of extracellular particulate organic carbon. Biogeosciences, 4:433-454.

Schauer, U., B. Rudels, E.P. Jones, L.G. Anderson, R.D. Muench, G. Björk, J.H. Swift, V. Ivanov and A.-M. Larsson, 2002. Confluence and redistribution of Atlantic water in the Nansen, Amundsen and Makarov basins. Annales Geophysicae, 20:257-273.

Schiebel, R., 2002. Planktic foraminiferal sedimentation and the marine calcite budget. Global Biogeochemical Cycles, 16(4):1065.

Schiermeier, Q., 2012. Ice loss shifts Arctic cycles. Nature, 489:185-186.

Schmittner, A., A. Oschlies, H.D. Matthews and E.D. Galbraith, 2008. Future changes in climate, ocean circulation, ecosystems, and biogeochemical cycling simulated for a business-as-usual CO_2 emission scenario until year 4000 AD. Global Biogeochemical Cycles, 22:GB1013.

Schmitz, W.J., 1995. On the interbasin-scale thermohaline circulation. Reviews of Geophysics, 33:151-173.

Schneider, K. and J. Erez, 2006. The effect of carbonate chemistry on calcification and photosynthesis in the hermatypic coral *Acropora eurystoma*. Limnology and Oceanography, 51:1284-1293.

Schneider, B., L. Bopp, M. Gehlen, J. Segschneider, T.L. Frölicher, P. Cadule, P. Friedlingstein, S.C. Doney, M.J. Behrenfeld and F. Joos, 2008. Climate-induced interannual variability of marine primary and export production in three global coupled climate carbon cycle models. Biogeosciences, 5:597-614.

Scholten, J.C., M.M. Rutgers van der Loeff and A. Michel, 1995. Distribution of ²³⁰Th and ²³¹Pa in the water column in relation to the ventilation of the deep Arctic basins. Deep Sea Research II, 42:1519-1531.

Scott, D.B. and G. Vilks, 1991. Benthonic foraminifera in the surface sediments of the deep-sea Arctic Ocean. Journal of Foraminiferal Research, 21:20-38.

Scott, D.B., P.J. Mudie, V. Baki, K.D. Mackinnon and F.E. Cole, 1989. Biostratigraphy and Late Cenozoic paleoceanography of the Arctic Ocean – Foraminiferal, lithostratigraphy and isotopic evidence. Geological Society of America Bulletin, 101:260-277.

Seibel, B.A. and P.J. Walsh, 2003. Biological impacts of deep-sea carbon dioxide injection inferred from indices of physiological performance. Journal of Experimental Biology, 206:641-650.

Seitz, K., J. Buxmann, D. Pöhler, T. Sommer, J. Tschritter, C. O'Dowd and U. Platt, 2009. The spatial distribution of the reactive iodine species IO from simultaneous active and passive DOAS observations. Atmospheric Chemistry and Physics Discussions, 9:21371-21398.

Semiletov, I., A. Makshtas, S.-I. Akasofu and E.L. Andreas, 2004. Atmospheric CO₂ balance: The role of Arctic sea ice. Geophysical Research Letters, 31:L05121.

Semiletov, I., O. Dudarev, V. Luchin, A. Charkin, K.-H. Shin and N. Tanaka, 2005. The East Siberian Sea as a transition zone between Pacific-derived waters and Arctic shelf waters. Geophysical Research Letters, 32:L10614.

Semiletov, I.P., I.I. Pipko, I. Repina and N.E. Shakhova, 2007. Carbonate chemistry dynamics and carbon dioxide fluxes across the atmosphere-ice-water interface in the Arctic Ocean: Pacific sector of the Arctic. Journal of Marine Systems, 66:204-226.

Semiletov, I.P., I.I. Pipko, N.E. Shakhova, O.V. Dudarev, S.P. Pugach, A.N. Charkin, C.P. McRoy, D. Kosmach and Ö Gustafsson, 2011. Carbon transport by the Lena River from its headwaters to the Arctic Ocean, with emphasis on fluvial input of terrestrial particulate organic carbon vs. carbon transport by coastal erosion. Biogeosciences, 8:2093-2143.

Semiletov, I.P, N.E. Shakhova, V.I. Sergienko, I.I. Pipko and O.V. Dudarev, 2012. On carbon transport and fate in the East Siberian Arctic land-shelf-atmosphere system. Environmental Research Letters, 7:015101.

Sen Gupta, B.K., 2002. Modern Foraminifera. Kluwer Academic Publishers.

Serreze, M.C., 2011. Re thinking the sea-ice tipping point. Nature, 471:47-48.

Serreze, M.C., A.P. Barrett, A.G. Slater, R.A. Woodgate, K. Aagaard, R.B. Lammers, M. Steele, R. Moritz, M. Meredith and C.M. Lee, 2006. The large-scale freshwater cycle of the Arctic. Journal of Geophysical Research, 111:C11010.

Sewell, M.A. and G.E. Hofmann, 2011. Antarctic echinoids and climate change: a major impact on the brooding forms. Global Change Biology, 17:734-744.

Shadwick, E.H., H. Thomas, M. Chierici, B. Else, A. Fransson, C. Michel, L.A. Miller, A. Mucci, A. Niemi, T.N. Papakyriakou and J.-É. Tremblay, 2011. Seasonal variability of the inorganic carbon system in the Amundsen Gulf region of the southeastern Beaufort Sea. Limnology and Oceanography, 56:303-322.

Shaked, Y., A.B. Kustka, F.M.M. Morel and Y. Erel, 2004. Simultaneous determination of iron reduction and uptake by phytoplankton. Limnology and Oceanography: Methods, 2:137-145.

Shaked, Y., A.B. Kustka and F.M.M. Morel, 2005. A general kinetic model for iron acquisition by eukaryotic phytoplankton. Limnology and Oceanography, 50:872-882.

Shakhova, N.E., I. Semiletov, A. Salyuk, V. Yusupov, D. Kosmach and Ö. Gustafsson, 2010. Extensive methane venting to the atmosphere from sediments of the East Siberian Arctic shelf. Science, 327:1246-1250.

Sharma, V.K. and F.J. Millero, 1989. The oxidation of Cu(I) with H_2O_2 in natural waters. Geochimica et Cosmochimica Acta, 53:2269-2276.

Sharp, M., Burgess, D., Cogley, J.G., Ecclestone, M., Labine, C. and Wolken, G.J., 2011. Extreme melt on Canada's Arctic ice caps in the 21st century. Geophysical Research Letters, 38: L11501.

Shi, D., Y. Xu and F.M.M. Morel, 2009. Effects of the pH/pCO_2 control method on medium chemistry and phytoplankton growth. Biogeosciences, 6:1199-1207.

Shi, D.L., Y. Xu, B.M. Hopkinson and F.M.M. Morel, 2010. Effect of ocean acidification on iron availability to marine phytoplankton. Science, 327:676-679.

Shimada, K., T. Kamoshida, M. Itoh, S. Nishino, E. Carmack, F. McLaughlin, S. Zimmermann and A. Proshutinsky, 2006. Pacific Ocean inflow: Influence on catastrophic reduction of sea ice cover in the Arctic Ocean. Geophysical Research Letters, 33:L08605.

Shirayama, Y. and H. Thornton, 2005. Effect of increased atmospheric CO_2 on shallow water marine benthos. Journal of Geophysical Research, 110:C09S08.

Simpson, K.G., J.-É. Tremblay, Y. Gratton and N.M. Price, 2008. An annual study of inorganic and organic nitrogen and phosphorus and silicic acid in the southeastern Beaufort Sea. Journal of Geophysical Research, 113:C07016.

Simpson, S.D., P.L. Munday, M.L. Wittenrich, R. Manassa, D.L. Dixson, M. Gagliano and H.Y. Yan, 2011. Ocean acidification erodes crucial auditory behaviour in a marine fish. Biology Letters, 7:917-920.

Skjelvan, I., T. Johannessen and L.A. Miller, 1999. Interannual variability of fCO_2 in the Greenland and Norwegian Seas. Tellus, 51B:477-489.

Skjelvan, I., E. Falck, F. Rey, F. and S.B. Kringstadt, 2008. Inorganic carbon time series at Ocean Weather Station M in the Norwegian Sea. Biogeosciences, 5:549-560.

Slagstad, D., I.H. Ellingsen and P. Wassmann, 2011. Evaluating primary and secondary production in an Arctic Ocean void of summer sea ice: an experimental simulation approach. Progress in Oceanography, 90:117-131. Smith, R.E.H., J. Anning, P. Clement and G. Cota, 1988. Abundance and production of ice algae in Resolute Passage, Canadian Arctic. Marine Ecology Progress Series, 48:251-263.

Smith, J.N., S.B. Moran and R.W. Macdonald, 2003. Shelf-basin interactions in the Arctic Ocean based on ²¹⁰Pb and Ra isotope tracer distributions. Deep Sea Research I, 50:397-416.

Smyth, T.J., T. Tyrrell and B. Tarrant, 2004. Time series of coccolithophore activity in the Barents Sea, from twenty years of satellite imagery. Geophysical Research Letters, 31:L11302.

Solomon, S., D. Qin, M. Manning, Z. Chen, M. Marquis, K.B. Averyt, M. Tignor and H.L. Miller (Eds), 2007. Climate Change 2007: The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press.

Spasojevic, I., S.K. Armstrong, T.J. Brickman and A.L. Crumbliss, 1999. Electrochemical behavior of the Fe(III) complexes of the cyclic hydroxamate siderophores alcaligin and desferrioxamine E. Inorganic Chemistry, 38:449-454.

Spero, H.J., J. Bijma, D.W. Lea and B.E. Bemis, 1997. Effect of seawater carbonate concentration on foraminiferal carbon and oxygen isotopes. Nature, 390:497-500.

Spielmeyer, A. and G. Pohnert, 2012. Influence of temperature and elevated carbon dioxide on the production of dimethylsulfoniopropionate and glycine betaine by marine phytoplankton. Marine Environmental Research, 73:62-69.

Springer, A.M. and C.P. McRoy, 1993. The paradox of pelagic food webs in the northern Bering Sea. III. Patterns of primary production. Continental Shelf Research, 13:575-599.

Springer, A.M., C.P. McRoy and K.R. Turco, 1989. The paradox of pelagic food webs in the northern Bering Sea. II. Zooplankton communities. Continental Shelf Research, 9:359-386.

Springer, A.M., C.P. McRoy and M.V. Flint, 1996. The Bering Sea Green Belt: Shelf-edge processes and ecosystem production. Fisheries Oceanography, 5:205-223.

Statistics Norway, 2006. The Economy of the North. Glomsrød, S. and I. Aslaksen (Eds.). Statistics Norway, Oslo.

Statistics Norway, 2007. Annual Report 2006. Plans and Reports, No. 2007/11. www.ssb.no/a/english/about_ssb/annual_report/2006/annual_report_2006.pdf. Accessed 25 April 2013.

Statistics Norway, 2009. The Economy of the North 2008. Glomsrød, S. and I. Aslaksen (Eds.). Statistics Norway, Oslo.

Stefels, J. and L. Dijkhuizen, 1996. Characteristics of DMSPlyase in *Phaeocystis* sp. (Prymnesiophyceae). Marine Ecology Progress Series, 131:307-313.

Steffen, W., Å. Persson, L. Deutsch, J. Zalasiewicz, M. Williams, K. Richardson, C. Crumley, P. Crutzen, C. Folke, L. Gordon, M. Molina, R. Veerabhadran, J. Rockström, M. Scheffer, H.J. Schellnhuber and U. Svedin, 2011. The anthropocene: From global change to planetary stewardship. Ambio, 40:739-761.

Stehli, F.G. and W.B. Creath, 1964. Foraminiferal ratios and regional environments. American Association of Petroleum Geologists Bulletin, 48:1810-1827.

Stein, R. and R.W. Macdonald, 2004a. Organic carbon budget: Arctic Ocean versus global ocean. In: Stein, R. and R.W. Macdonald (Eds.). The Arctic Ocean Organic Carbon Cycle: Present and Past, pp. 315-322. Springer.

Stein, R. and R.W. Macdonald (Eds.), 2004b. The Organic Carbon Cycle in the Arctic Ocean. Springer.

Steinacher, M., F. Joos, T.I. Frolicher, G.-K. Plattner and S.C. Doney, 2009. Imminent ocean acidification in the Arctic projected with the NCAR global climate carbon cycle-climate model. Biogeosciences, 6:515-533.

Steinacher, M., F. Joos, T. Frölicher, L. Bopp, P. Cadule, V. Cocco, S. Doney, M. Gehlen, K. Lindsay, J. Moore, B. Schneider and J. Segschneider, 2010. Projected 21st century decrease in marine productivity: a multi-model analysis. Biogeosciences, 7:979-1005.

Steiner, N.S., W.G. Lee and J.R. Christian, 2013. Enhanced gas fluxes in small sea ice leads and cracks – effects on CO_2 exchange and ocean acidification. Journal of Geophysical Research: Oceans, 118:1195-1205.

Stergiou, K.I., 1989. Capelin *Mallotus villosus* (Pisces, Osmeridae), glaciations and speciation – a nomothetic approach to fisheries ecology and reproductive biology. Marine Ecology Progress Series, 56:211-224.

Stockton, W.L. and T.E. DeLaca, 1982. Food falls in the deepsea – occurrence, quality, and significance. Deep-Sea Research A, 29:157-169.

Stramma, L., S. Schmidtko, L.A. Levin and G.G. Johnson, 2010. Ocean oxygen minima expansions and their biological impacts. Deep-Sea Research I, 57:587-595.

Stroeve, J., M. Serreze, F. Fetterer, T. Arbetter, W. Meier, J. Maslanik and K. Knowles, 2005. Tracking the Arctic's shrinking ice cover: Another extreme September minimum in 2004. Geophysical Research Letters, 32:L04501.

Stroeve, J., M.M. Holland, W. Meier, T. Scambos and M. Serreze, 2007. Arctic sea ice decline: Faster than forecast. Geophysical Research Letters, 34:L09501.

Stroeve, J.C., V. Kattsov, A. Barrett, M. Serreze, T. Pavlova, M. Holland and W.N. Meier, 2012a. Trends in Arctic sea ice extent from CMIP5, CMIP3 and observations. Geophysical Research Letters, 39:L16502.

Stroeve, J.C., M.C. Serreze, M.M. Holland, J.E. Kay, J. Malanik and A.P. Barrett, 2012b. The Arctic's rapidly shrinking sea ice cover: a research synthesis. Climatic Change, 110:1005-1027.

Stumpp, M., K. Truebenbach, D. Brennecke, M.Y. Hu and F. Melzner, 2012. Resource allocation and extracellular acid-base status in the sea urchin *Strongylocentrotus droebachiensis* in response to CO₂ induced seawater acidification. Aquatic Toxicology, 110:194-207.

Sumaila, U.R., S. Guénette, J. Alder and R. Chuenpagdee, 2000. Addressing the ecosystem effects of fishing using marine protected areas. ICES Journal of Marine Science, 57:752-760.

Sumaila, U.R., W.W.L. Cheung, V.W.Y. Lam, D. Pauly and S. Herrick, 2011. Climate change impacts on the biophysics and economics of world fisheries. Nature Climate Change, 1:449-456.

Sun, J., D.A. Hutchins, Y. Feng, E.L. Seubert, D.A. Caron and F.-X. Fu, 2011. Effects of changing pCO_2 and phosphate availability on domoic acid production and physiology of the marine harmful bloom diatom *Pseudo-nitzschia multiseries*. Limnology and Oceanography, 56:829-840.

Sunda, W.G. and S.A. Huntsman, 2005. Effect of CO_2 supply and demand on zinc uptake and growth limitation in a coastal diatom. Limnology and Oceanography, 50:1181-1192.

Sunday, J.M., R.N. Crim, C.D.G. Harley and M.W. Hart, 2011. Quantifying rates of evolutionary adaptation in response to ocean acidification. PLoS ONE, 6:e22881.

Sundby, S., 2000. Recruitment of Atlantic cod stocks in relation to temperature and advection of copepod populations. Sarsia, 85:277-298.

Suydam, R.S., L.F. Lowry, K.J. Frost, G.M. Corry-Crowe and D. Pikok, 2001. Satellite tracking of eastern Chukchi Sea beluga whales into the Arctic Ocean. Arctic, 54:237-243.

Suydam, R.S., L.F. Lowry and K.J. Frost, 2005. Distribution and movements of beluga whales from the eastern Chukchi Sea stock during summer and early autumn. OCS Study Report MMS 2005-035: i-iii, 1-37.

Takahashi, T., S.C. Sutherland, C. Sweeney, A. Poisson, N. Metzl, T. Tilbrook, N. Bates, R. Wanninkhof, R.A. Feely, C. Sabine, J. Olafsson and Y. Nojiri, 2002. Global sea-air CO_2 flux based on climatological suface ocean pCO_2 , and seasonal biological and temperature effects. Deep-Sea Research II, 49:1601-1622.

Talmage, S.C. and C.J. Gobler, 2009. The effects of elevated carbon dioxide concentrations on the metamorphosis, size, and survival of larval hard clams (*Mercenaria mercenaria*), bay scallops (*Argopecten irradians*), and Eastern oysters (*Crassostrea virginica*). Limnology and Oceanography, 54:2072-2080.

Talmage, S.C. and C.J. Gobler, 2010. Effects of past, present, and future ocean carbon dioxide concentrations on the growth and survival of larval shellfish. Proceedings of the National Academy of Sciences of the United States of America, 107:17246-17251.

Taylor, J., M. Tranter and G. Munhoven, 2002. Carbon cycling and burial in the glacially influenced polar North Atlantic. Paleoceanography, 17:1-1–1-13.

Taylor, J.R., K.K. Falkner, U. Schauer and M. Meredith, 2003. Quantitative considerations of dissolved barium as a tracer in the Arctic Ocean. Journal of Geophysical Research, 108:3374.

The Royal Society, 2005. Ocean Acidification due to Increasing Atmospheric Carbon Dioxide. Policy Document 12/05. Available at: www.royalsoc.ac.uk

Thomas, D.N., S. Papadimitriou and C. Michel, 2010. Biogeochemistry of sea ice. In: Thomas, D.N. and G.S. Dieckmann (Eds.). Sea Ice, pp. 425-467. Chichester.

Thomas, H., E. Shadwick, F. Dehairs, B. Lansard, A. Mucci, J. Navez, Y. Gratton, F. Prowe, M. Chierici, A. Fransson, T.N. Papakyriakou, E. Sternberg, L.A. Miller, J.-É. Tremblay and C. Monnin, 2011. Barium and carbon fluxes in the Canadian Arctic Archipelago. Journal of Geophysical Research, 116:C00G08. doi: 10.1029/2011JC007120. Thomsen, J. and F. Melzner, 2010. Moderate seawater acidification does not elicit long-term metabolic depression in the blue mussel *Mytilus edulis*. Marine Biology, 157:2667-2676.

Thomsen, J., M.A. Gutowska, J. Saphorster, A. Heinemann, K. Trubenbach, J. Fietzke, C. Hiebenthal, A. Eisenhauer, A. Kortzinger, M. Wahl and F. Melzner, 2010. Calcifying invertebrates succeed in a naturally CO₂-rich coastal habitat but are threatened by high levels of future acidification. Biogeosciences, 7:3879-3891.

Thuróczy, C.-E., L.J.A. Gerringa, M. Klinder, P. Laan, M. Le Guitton and H.J.W. de Baar, 2011. Distinct trends in the speciation of iron between the shallow shelf seas and the deep basins of the Arctic Ocean. Journal of Geophysical Research, 116:C10009.

Toivonen, A.L., E. Roth, S. Navrud, G. Gudbergsson, H. Appelblad, B. Bengtsson and P. Tuunainen, 2004. The economic value of recreational fisheries in Nordic countries. Fisheries Management and Ecology, 11:1-14.

Tomanek, L., M.J. Zuzow, A.V. Ivanina, E. Beniash and I.M. Sokolova, 2011. Proteomic response to elevated pCO_2 level in eastern oysters, *Crassostrea virginica*: evidence for oxidative stress. Journal of Experimental Biology, 214:1836-1844.

Torstensson, A., M. Chierici and A. Wulff, 2012. The influence of increased temperature and carbon dioxide levels on the benthic/sea ice diatom *Navicula directa*. Polar Biology, 35:205-214.

Tortell, P.D., C.D. Payne, Y.Y. Li, S. Trimborn, B. Rost, W.O. Smith, C. Riesselman, R.B. Dunbar, P. Sedwick and G.R. DiTullio, 2008. CO_2 sensitivity of Southern Ocean phytoplankton. Geophysical Research Letters, 35:L04605.

Trees, C., J. Aiken, H.-J. Hirche and S. Groom, 1992. Bio-optical variability across the Arctic Front. Polar Biology, 12:455-461.

Tremblay, J J.-É., H. Hattori, C. Michel, M. Ringuette, Z.P. Mei, C. Lovejoy, L. Fortier, K.A. Hobson, D. Amiel and K. Cochran, 2006. Trophic structure and pathways of biogenic carbon flow in the eastern North Water Polynya. Progress in Oceanography, 71:402-425.

Tremblay, J.-É., S. Bélanger, D.G. Barber, M. Asplin, J. Martin, G. Darnis, L. Fortier, Y. Gratton, H. Link, P. Archambault, A. Sallon, C. Michel, W.J. Williams, B. Phillipe and M. Gosselin, 2011. Climate forcing multiplies biological productivity in the coastal Arctic Ocean. Geophysical Research Letters, 38:L18604.

Treshnikov, A.F., 1959. Surface waters in the Arctic basin. Problemy Artiki, 7:5-14.

Trimble, S.M., M. Baskaran and D. Porcelli, 2004. Scavenging of thorium isotopes in the Canada Basin of the Arctic Ocean. Earth and Planetary Science Letters, 222:915-932.

Tsurushima, N., S. Watanabe and S. Tsunogai, 1996. Methane in the East China Sea water. Journal of Oceanography, 52:221-233.

Tunnicliffe, V., K.T.A. Davies, D.A. Butterfield, R.W. Embley, J.M. Rose and W.W. Chadwick Jr, 2009. Survival of mussels in extremely acidic waters on a submarine volcano. Nature Geoscience, 2:344-348.

Turley, C.M. and J. Blackford, 2005. The other CO₂ problem. Marine Conservation, 6:11 and in a film animation by Ridgeway School. www.opendemocracy.net/globalizationclimate_change_debate/article_2480.jsp

Turley, C. and K. Boot, 2011. The ocean acidification challenges facing science and society. In: Gattuso J.-P. and L. Hansson (Eds.). Ocean Acidification, pp. 249-271. Oxford University Press.

Turley, C.M., J.M. Roberts and J.M. Guinotte, 2007. Corals in deep-water: will the unseen hand of ocean acidification destroy cold-water ecosystems? Coral Reefs, 26:445-448.

Turley, C., M. Eby, A.J. Ridgwell, D.N Schmidt, H.S. Findlay, C. Brownlee, U. Riebesell, V.J. Fabry, R.A. Feely and J.-P. Gattuso, 2010. The societal challenge of ocean acidification. Marine Pollution Bulletin, 60:787-792.

Turner, J.T., 2002. Zooplankton fecal pellets, marine snow and sinking phytoplankton blooms. Aquatic Microbial Ecology, 27:57-102.

Turner, D.R., M. Whitfield and A.G. Dickson, 1981. The equilibrium speciation of dissolved components in freshwater and seawater at 25C and 1 atm pressure. Geochimica et Cosmochimca Acta, 45:855-881.

van den Berg, C.M.G., 1985. Determination of the zinc complexing capacity in seawater by cathodic stripping voltammetry of zinc-APDC complex ions. Marine Chemistry, 16:121-130.

Vancoppenolle, M., L. Bopp, G. Madec, J. Dunne, T. Ilyina, P.R. Halloran and N. Steiner, 2013. Future Arctic Ocean primary productivity from cmip5 simulations: Uncertain outcome, but consistent mechanisms. Global Biogeochemical Cycles, 27:605-619.

Vermeij, G.J. and P.D. Roopnarine, 2008. The coming Arctic invasion. Science, 321:780-781.

Vilhjalmsson, H., 1994. The Icelandic capelin stock. Capelin, *Mallotus villosus* (Muller) in the Iceland–Greenland–Jan Mayen area. Rit Fiskideildar, 13:1-281.

Vogt, M., M. Steinke, S. Turner, A. Paulino, M. Meyerhoefer, U. Riebesell, C. LeQuere and P. Liss, 2008. Dynamics of dimethylsulphoniopropionate and dimethylsulphide under different CO_2 concentrations during a mesocosm experiment. Biogeosciences, 5:407-419.

Volk T. and M.I. Hoffert, 1985. Ocean carbon pumps: Analysis of relative strengths and efficiencies in ocean-driven atmospheric CO, changes. Geophysical Monograph Series, 32:99-110.

Vølstad, J.H., K. Korsbrekke, K.H. Nedreaas, M. Nilsen, G.N. Nilsson, M. Pennington, S. Subbey and R. Wienerroither, 2011. Probability-based surveying using self-sampling to estimate catch and effort in Norway's coastal tourist fishery. ICES Journal of Marine Science, 68:1785-1791.

Vonk, J.E., L. Sanchez-Garcia, I. Semiletov, O. Dudarev, T. Eglinton, A. Andersson and O. Gustafsson, 2010. Molecular and radiocarbon constraints on sources and degradation of terrestrial organic carbon along the Kolyma paleoriver transect, East Siberian Sea. Biogeosciences, 7:3153-3166.

Vonk, J.E., L. Sanchez-Garcia, B.E. van Dongen, D. Kosmach, A. Charkin, I.P. Semiletov, O.V. Dudarev, N. Shakhova, P. Roos, T.I. Eglinton, A. Andersson and O. Gustafsson, 2012. Activation of old carbon by erosion of coastal and subsea permafrost in Arctic Siberia. Nature, 489:137-140.

Vraspir, J.M. and A. Butler, 2009. Chemistry of marine ligands and siderophores. Annual Review of Marine Science, 1:43-63.

Wadhams, P., 2012. Arctic ice cover, ice thickness and tipping points. Ambio, 41:23-33.

Wåhlström, I., A. Omstedt, G. Björk and L.G. Anderson, 2012. Modelling the CO_2 dynamics in the Laptev Sea, Arctic Ocean: Part I. Journal of Marine Systems, 102-104: 29-38.

Wåhlström, I., A. Omstedt, G. Björk and L.G. Anderson, 2013. Modelling the CO_2 dynamics in the Laptev Sea, Arctic Ocean: Part II. Sensitivity of fluxes to changes in the forcing. Journal of Marine Systems, 111-112:1-10.

Walsh, J.E., 1991. The Arctic as a bellwether. Nature, 352:19-20.

Walsh, J.J., C.P. McRoy, L.K. Coachman, J.J. Goering, J.J. Nihoul, T.E. Whitledge, T.H. Blackburn, P.L. Parker, C.D. Wirick, P.G. Shuert, J.M. Grebmeier, A.M. Springer, R.D. Tripp, D.A. Hansell, S. Djenidi, E. Deleersnijder, K. Henriksen, B.A. Lund, P. Andersen, F.E. Mullerkarger and K. Dean, 1989. Carbon and nitrogen cycling within the Bering/Chukchi Seas: Source regions for organic matter effecting AOU demands of the Arctic Ocean. Progress in Oceanography, 22:277-359.

Walters, V., 1961. Winter abundance of Arctogadus glacialis in the polar basin. Copeia, 1961:236-237.

Walther, K., F.J. Sartoris, C. Bock and H.O. Pörtner, 2009. Impact of anthropogenic ocean acidification on thermal tolerance of the spider crab *Hyas araneus*. Biogeosciences, 6:2207-2215.

Walther, K., K. Anger and H.O. Pörtner, 2010. Effects of ocean acidification and warming on the larval development of the spider crab *Hyas araneus* from different latitudes (54° vs. 79° N). Marine Ecology Progress Series, 417:159-170.

Walther, K., F.J. Sartoris and H.O. Pörtner, 2011. Impacts of temperature and acidification on larval calcium incorporation of the spider crab *Hyas araneus* from different latitudes (54° vs. 79° N). Marine Biology, 158:2043-2053.

Watanabe, S., T. Hajima, K. Sudo, T. Nagashima, T. Takemura, H. Okajima, T. Nozawa, H. Kawase, M. Abe, T. Yokohata, T. Ise, H. Sato, E. Kato, K. Takata, S. Emori and M. Kawamiya, 2011. MIROC-ESM 2010: model description and basic results of CMIP5-20c3m experiments. Geoscientific Model Development, 4:845-872.

Weiss, R.F., 1974. Carbon dioxide in water and seawater: The solubility of a non-ideal gas. Marine Chemistry, 2:203-215.

Welch, H.E. and M.A. Bergmann, 1989. Seasonal development of ice algae and its prediction from environmental factors near Resolute, NWT, Canada. Canadian Journal of Fisheries and Aquatic Sciences, 46:1793-1804.

Welch, H.E., M.A. Bergmann, T.D. Siferd, K.A. Martin, M.F. Curtis, R.E. Crawford, R.J. Conover and H. Hop, 1992. Energy flow through the marine ecosystem of the Lancaster Sound region, Arctic Canada. Arctic, 45:343-357.

Welch, H.E., R.E. Crawford and H. Hop, 1993. Occurrence of Arctic cod (*Boreogadus saida*) schools and their vulnerability to predation in the Canadian High Arctic. Arctic, 46:331-339.

Wells, L.E. and J.W. Deming, 2006. Modelled and measured dynamics of viruses in Arctic winter sea-ice brines. Environmental Microbiology, 8:1115-1121.

Westbrook, G.K., K.E. Thatcher, E.J. Rohling, A.M. Piotrowski, H. Pälike, A.H. Osborne, E.G. Nisbet, T.A. Minshull, M. Lanoisellé, R.H. James, V. Hühnerbach, D. Green, R.E. Fisher, A.J. Crocker, A. Chabert, C. Bolton, Beszczynska-Möller, C. Berndt, and A. Aquilina, 2009. Escape of methane gas from the seabed along the West Spitsbergen continental margin. Geophysical Research Letters, 36, L15608, doi:10.1029/2009GL039191.

White, D., L.D. Hinzman, L. Alessa, J. Cassano, M. Chambers, K. Falkner, J. Francis, W.J. Gutowski Jr., M. Holland, R.M. Holmes, H. Huntington, D. Kane, A. Kliskey, C. Lee, J. McClelland, B. Peterson, T.S. Rupp, F. Straneo, M. Steele, R. Woodgate, D. Yang, K. Yoshikawa and T. Zhang, 2007. The arctic freshwater system: Changes and impacts. Journal of Geophysical Research, 112:G04S54.

Whiteley, N.M., 2011. Physiological and ecological responses of crustaceans to ocean acidification. Marine Ecology Progress Series, 430:257-271.

Wicks, L.C. and J.M. Roberts, 2012. Benthic invertebrates in a high CO_2 world. Oceanography and Marine Biology: An Annual Review, 50:127-188.

Widdicombe, S. and H.R. Needham, 2007. Impact of CO₂induced seawater acidification on the burrowing activity of *Nereis virens* and sediment nutrient flux. Marine Ecology Progress Series, 341:111-122.

Wilhelm, S.W. and C.G. Trick, 1994. Iron-limited growth of cyanobacteria: Multiple siderophore production is a common response. Limnology and Oceanography, 39:1979-1984.

Williamson, P. and C. Turley, 2012. Ocean acidification in a geoengineering context. Philosophical Transactions of the Royal Society A, 370:4317-4342.

Wingenter, O.W., K.B. Haase, M. Zeigler, D.R. Blake, F. Sc. Rowland, B.C. Sive, A.I. Paulino, R. Rhyrhaug, A. Larsen, K. Schulz, M. Meyerhöfer and U. Riebesell, 2007. Unexpected consequences of increasing $\rm CO_2$ and ocean acidity on marine production of DMS and CH2CII: Potential climate impacts. Geophysical Research Letters, 34:L05710.

Winn, C. and F. Millero, 1993. Hydrographic, chemical and carbon data obtained during the R/V Thomas Thompson cruise in the Pacific Ocean during WOCE Section P14N (EXPOCODE 325023_1), 05 July - 02 September, 1993. http://cdiac.ornl.gov/ oceans/woce_p14n.html. Carbon Dioxide Information Analysis Center, Oak Ridge National Laboratory, Tennessee.

Wong, K.K.W., A.C. Lane, P.T.Y. Leung and V. Thiyagarajan, 2011. Response of larval barnacle proteome to CO_2 -driven seawater acidification. Comparative Biochemistry and Physiology D, 6:310-321.

Wood, H.L., J.I. Spicer, M.A. Kendall, D.M. Lowe and S. Widdicombe, 2011. Ocean warming and acidification;

implications for the Arctic brittlestar *Ophiocten sericeum*. Polar Biology, 34:1033-1044.

Wootton, J.T., C.A. Pfister and J.D. Forester, 2008. Dynamic patterns and ecological impacts of declining ocean pH in a high-resolution multi-year dataset. Proceedings of the National Academy of Sciences of the United States of America, 105:18848-18853.

Xu, Y., D. Shi, L. Aristilde and F.M.M. Morel., 2012. The effect of pH on the uptake of zinc and cadmium in marine phytoplankton: Possible role of weak complexes. Limnology and Oceanography, 57:293-304.

Yager, P.L., D.W.R. Wallace, K.M. Johnson, J. Smith, P.J. Minnett and J.W. Deming, 1995. The Northeast Water Polynya as an atmospheric CO_2 sink: a seasonal rectification hypothesis. Journal of Geophysical Research, 100:4389-4398.

Yamada, N. and M. Suzumura, 2010. Effects of seawater acidification on hydrolytic enzyme activities. Journal of Oceanography, 66:233-241.

Yamamoto, A., M. Kawamiya, A. Ishida, Y. Yamanaka and S. Watanabe, 2012. Impact of rapid sea-ice reduction in the Arctic Ocean on the rate of ocean acidication. Biogeosciences, 9:2365-2375.

Yamamoto-Kawai, M., N. Tanaka and S. Pivovarov, 2005. Freshwater and brine behaviors in the Arctic Ocean deduced from historical data of δ^{18} O and alkalinity (1929-2002 AD). Journal of Geophysical Research, 110:C10003.

Yamamoto-Kawai, M., E. Carmack and F.A. McLaughlin, 2006. Nitrogen balance and Arctic throughflow. Nature, 443:43.

Yamamoto-Kawai, M., F.A. McLaughlin, E.C. Carmack, S. Nishino and K. Shimada, 2009. Aragonite undersaturation in the Arctic Ocean: Effects of ocean acidification and sea ice melt. Science, 326:1098-1100.

Yamamoto-Kawai, M., F.A. McLaughlin and E.C. Carmack, 2011. Effects of ocean acidification, warming and melting of sea ice on aragonite saturation of the Canada Basin surface water. Geophysical Research Letters, 38:L03601.

Yeats, P.A. and S. Westerlund, 1991. Trace metal distributions at an Arctic Ocean ice island. Marine Chemistry, 33:261-277.

Yool, A. and M.J.R. Fasham, 2001. An examination of the 'continental shelf pump' in an open ocean general circulation model. Global Biogeochemical Cycles, 15:831-844.

Yu, W., L.Q. Chen, J.P. Cheng, J.H. He, M.D. Yin and Z. Zeng, 2010. Th-234-derived particulate organic carbon export flux in the western Arctic Ocean. Chinese Journal of Oceanology and Limnology, 28:1146-1151.

Zachos, J.C., U. Rohl, S.A. Schellenberg, A. Sluijs, D.A. Hodell, D.C. Kelly, E. Thomas, M. Nicolo, I. Raffi, L. J. Lourens, H. McCarren and D. Kroon, 2005. Rapid acidification of the ocean during the Paleocene–Eocene thermal maximum. Science, 308:1611-1615.

Zeller, D., S. Booth, E. Pakhomov, W. Swartz and D. Pauly, 2011. Arctic fisheries catches in Russia, USA and Canada: Baselines for neglected ecosystems. Polar Biology, 34:955-973.
Zemmelink, H.J., B. Delille, J.L. Tison, E.J. Hintsa, L. Houghton and J.W.H. Dacey, 2006. CO₂ deposition over the multi-year ice of the western Weddell Sea. Geophysical Research Letters, 33:L13606.

Zhang, J., R. Lindsay, A. Schweiger and I. Rigor, 2012. Recent changes in the dynamic properties of declining Arctic sea ice: A model study. Geophysical Research Letters, 39:L20503.

Zimmermann, B., D. Porcelli, M. Frank, P.S. Andersson, M. Baskaran, D.-C Lee and A.N. Halliday, 2009. Hafnium isotopes in Arctic Ocean water. Geochimica et Cosmochimica Acta, 73:3218-3233.

Zondervan, I., B. Rost and U. Riebesell, 2002. Effect of CO₂ concentration on the PIC/POC ratio in the coccolithophore *Emiliania huxleyi* grown under light-limiting conditions and different daylengths. Journal of Experimental Marine Biology and Ecology, 272:55-70.

Zou, H.Y., J.L. Huang, F. Fang and J.S. Guo, 2011. Characterization of amoA and hao genes responsible for ammonia oxidation reaction in CANON system, In: Environmental Biotechnology and Materials Engineering, Pts 1-3, pp. 1014-1019. Advanced Materials Research. Trans Tech Publications Ltd.

Glossary

$\Omega_{_{ m Ar}}$	Saturation state of aragonite	O ₂	Oxygen
$\Omega_{_{\mathrm{Ca}}}$	Saturation state of calcite	ОМ	Organic matter
A _T	Total carbonate alkalinity	pCO ₂	Partial pressure of carbon dioxide
С	Carbon	рН	Acidity or concentration of hydrogen ions
$Ca^{2+} / [Ca^{2+}]$	Calcium ion / Concentration of calcium ions	POC	Particulate organic carbon
CaCO ₃	Calcium carbonate	POM	Particulate organic matter
C _{ant}	Anthropogenic carbon dioxide	TED	
Cd	Cadmium	TEP	iransparent exopolymeric particles
CH_4	Methane	Zn	Zinc
CO ₂	Carbon dioxide		
C _T	Total inorganic (dissolved) carbon		
Cu	Copper		
DMS	Dimethylsulfide		
DMSP	Dimethylsulfoniopropionate		
DOC	Dissolved organic carbon		
DOM	Dissolved organic matter		
ESM	Earth system model		
FAO	United Nations Food and Agriculture Organization		
Fe	Iron		
GDP	Gross domestic product		
GRP	Gross regional product		
${ m H}^{_{+}}$ / $[{ m H}^{_{+}}]$	Hydrogen ion / Concentration of hydrogen ions		
IPCC AR5	Fifth Assessment Report of the Intergovernmental Panel on Climate Change		
Mg^{2+}	Magnesium ion		
Ν	Nitrogen		
NH ₃	Ammonia		
NH_4^{+}	Ammonium		

Arctic Monitoring and Assessment Programme

The Arctic Monitoring and Assessment Programme (AMAP) was established in June 1991 by the eight Arctic countries (Canada, Denmark, Finland, Iceland, Norway, Russia, Sweden and the United States) to implement parts of the Arctic Environmental Protection Strategy (AEPS). AMAP is now one of six working groups of the Arctic Council, members of which include the eight Arctic countries, the six Arctic Council Permanent Participants (indigenous peoples' organizations), together with observing countries and organizations.

AMAP's objective is to provide 'reliable and sufficient information on the status of, and threats to, the Arctic environment, and to provide scientific advice on actions to be taken in order to support Arctic governments in their efforts to take remedial and preventive actions to reduce adverse effects of contaminants and climate change'.

AMAP produces, at regular intervals, assessment reports that address a range of Arctic pollution and climate change issues, including effects on health of Arctic human populations. These are presented to Arctic Council Ministers in 'State of the Arctic Environment' reports that form a basis for necessary steps to be taken to protect the Arctic and its inhabitants.

This report has been subject to a formal and comprehensive peer review process. The results and any views expressed in this series are the responsibility of those scientists and experts engaged in the preparation of the reports.

The AMAP Secretariat is located at Gaustadalléen 21, N-0349 Oslo, Norway. For further information regarding AMAP or ordering of reports, please contact the AMAP Secretariat or visit the AMAP website (www.amap.no).

AMAP Secretariat

Gaustadalléen 21 N-0349 Oslo, Norway

T +47 21 08 04 80 F +47 21 08 04 85

www.amap.no ISBN - 978-82-7971-082-0



ARCTIC COUNCIL

Arctic Monitoring and Assessment Programme

